Spectral estimates for singular systems

Dissertation an der Fakultät für Mathematik, Informatik und Statistik der Ludwig-Maximilians-Universität München



eingereicht von Charlotte Isabel Dietze 04.02.2025

- 1. Gutachter: Prof. Dr. Phan Thành Nam
- 2. Gutachter: Prof. Dr. Stéphane Nonnenmacher
- 3. Gutachter: Prof. Dr. Daniel Grieser

Tag der mündlichen Prüfung: 17.4.2025

Eidesstattliche Versicherung

(Siehe Promotionsordnung vom 12.07.2011, §8, Abs. 2, Pkt. 5.)

Hiermit erkläre ich an Eides statt, dass die Dissertation von mir selbstständig, ohne unerlaubte Beihilfe angefertigt ist.

Charlotte Isabel Dietze Fürstenfeldbruck, 04.02.2025

Acknowledgements

I am deeply indebted to Phan Thành Nam for his continued support and for taking a lot of time to discuss with me. His optimism and vast array of ideas for tackling different mathematical problems have always been inspiring. I also thank him for introducing me to several of his collaborators and friends at various conferences. His very generous, almost infinite travel budget allowed me to make many unforgettable experiences travelling to numerous conferences, workshops, and it also allowed me to meet many collaborators. This has had a great impact on my PhD experience, and it has also laid the foundation for future collaborations. In particular, I am grateful to him for allowing me to spend in total twelve months at Institut des Hautes Études Scientifiques.

I would like to express my deepest gratitude to Laure Saint-Raymond for her support throughout my PhD. As my mentor at Institut des Hautes Études Scientifiques, she made my visits possible. I particularly enjoyed the conference in Rome she co-organised, which gave me the opportunity to meet mathematicians and physicists working on internal waves. I am looking forward to continuing the work with her during my postdoc. She introduced me to Yves Colin de Verdière, with whom we have been working together on a project. In the discussions with Laure and Yves, which I have always enjoyed a lot, I did not only learn about several mathematical techniques and ideas, but I also learned how to discuss mathematics in French.

I would like to especially thank Yves, who has helped and supported me a lot. In addition to our collaboration, I would like to thank him for the invitation to Grenoble and for the amazing days we spent in Bretagne, where I particularly enjoyed our walks and climbing in the rocks. Yves suggested me to work on a joint project with him, Maarten de Hoop and Emmanuel Trélat. This project has been a great source of joy and motivation, and I am looking forward to continuing this work.

I would like to thank Emmanuel Trélat for his enormous support, for taking a lot of time to discuss and for patiently answering my questions. I particularly appreciate his very clear and understandable explanations, which have helped me to learn a lot from our discussions. Furthermore, I would like to thank him for the many invitations to lunch.

I was lucky to share my offices with Sami Fersi, Larry Read and Jian Wang. In particular, I would like to thank Sami for always keeping me updated about the latest news in football, for many walks, dinners and discussions at lunch and for listening. I would like to thank Larry for the many mathematical discussions we had, the many tea and lunch breaks, and for his encouragement. I am also very happy that we could work together on a joint project, which, ironically, we discussed mostly over Zoom. I would like to thank Jian Wang for several helpful discussions and for his detailed explanations and I am looking forward to continuing working with him during my postdoc.

I have learnt a lot from working with different collaborators, for instance on how they think about mathematics or effective working strategies, and I would like to thank all of them for their time they spent working with me.

It was a pleasure working with Ayman Kachmar and Vladimir Lotoreichik. I have always been impressed by their mathematical carefulness as well as their very friendly and respectful way of communication. I would like to thank Konstantin Pankrashkin for introducing me to the local tea and kale during a visit to Oldenburg. About a year later, he suggested and discussed an open problem with me at a conference, which led to the fastest paper during my PhD. I appreciated the collaboration with Jinyeop Lee and I would like to thank him for providing a lot of useful advice and support. I will keep the many journeys to different workshops both of us attended in good memory. I thank Lea Boßmann for some helpful advice and taking care of the vast majority of typesetting of a joint paper with Phan Thành Nam. I also had in total three weeks of very intense and efficient discussions with Simon Larson, both in Gothenburg and in Munich, where I learnt a lot.

I would like to thank Florent Noisette and Jonas Peteranderl for organising a workshop for young researchers together with me. While the organisation was certainly more stressful and time-consuming than we expected, I am glad that we shared this experience. I would also like to thank Florent for some very nice discussions and walks during a summer school in Grenoble and the dinners at the student residence we organised with other participants.

There are several PhD students and postdocs from LMU (in addition to those I have already mentioned), whom I would particularly like to thank: I would like to thank Jakob Stern for numerous lunch breaks we have passed together, walks near the institute as well as many mathematical discussions. I would also like to thank

Leonard Wetzel for the lunch and coffee breaks we have spent together. I thank Simone Rademacher for her help in Oberwolfach, for her support and advice. I was also happy to spend several lunch breaks and time at workshops with Emanuela Giacomelli. I would like to thank Dinh Thi Nguyen for his advice, as well as for an amazing hike in the Calanques.

I am very grateful to Heinz Siedentop who has mentored me since my bachelor's thesis and has helped me in some key moments during my PhD. I also thank Thomas Sørensen for the long discussions on pseudodifferential operators and some useful advice.

I would like to thank Daniel Grieser and Stéphane Nonnenmacher for taking the time to read the thesis and being the referees. Furthermore I thank Peter Müller and Thomas Sørensen for accepting to be the examiners.

I am also thankful to Edith Höchst for the administrative support at LMU and taking the time to carefully fill in many travel permit and reimbursement forms with me. At this point, I would also like to thank the staff at IHES who are all doing an exceptionally good job. In particular, I would like to thank Mélanie Hajdinjak, Ingrid Peeters, Aurélie Gayraud, Elisabeth Jasserand and Clément Nivoy.

Special thanks goes to Hayette Bendjema for preparing the tea breaks in the afternoon at IHES and providing emotional support. I particularly appreciate her social skills and I found her memory of all the names of the visitors and their dietary restrictions very impressive.

I would also like to thank a number of PhD students and postdocs with whom I have spent time at IHES. I would like to thank Veronica Fantini who has helped me with many things and provided a lot of helpful advice. I thank Arthur Touati for the great and detailed advice he gave to me on many things. I would like to thank Angeliki Menegaki for many discussions and letting me use her office for a week. Furthermore, I would like to thank Kihyun Kim for several very helpful discussions and his advice. I thank Prasuna Bandi and Shuddhodan Kadattur Vasudevan for many homecooked Indian dinners, discussions and trips to Paris. I would also like to thank Yilin Wang, who warmly welcomed me when I arrived at IHES and who has always contributed to a positive atmosphere. I would also like to mention Mirek Olšak, who organised amazing board game evenings at Ormaille. I would particularly thank Aleix Gimenez who let me use his apartment for a week and for nice discussions at tea time. I thank Jiaming Xia, Hongbin Chen, Sami Douba, Dmitry Kubrak, François Jacopin, Marius Kjærsgaard, Irene Ren, Edmund Heng and many others for nice discussions at tea time and lunch.

Moreover, I would like to thank Nicolas Camps for his support and advice. I thank Zhuolin Li for her advice and the nice time we spent together at two summer schools, where I particularly remember the wonderful time we spent together in Cetraro. I would like to thank Sascha Lill for the time we spent together at different summer schools and conferences. Furthermore, I would like to thank Ngoc Nhi Nguyen for her support.

I would also like to mention the students who attended my tutorials. I enjoyed working with them a lot. I thank Peter Philip for the very good organisation of the numerics course and providing a lot of material, which has helped me a lot with the teaching.

I acknowledge the financial support from Deutsche Forschungsgemeinschaft (DFG project Nr. 426365943), the European Research Council via the ERC CoG RAM-BAS (Project-Nr. 10104424), the visiting programme of IHES through the Jean-Paul Gimon Fund and the Engie Foundation, the Erasmus+ programme, and from the LMU Mentoring programme.

Moreover, I would like to thank other friends who I have not mentioned yet. They have been an important source of emotional support and encouragement. In particular, I would like to mention Tobi for visiting me numerous times and the very long walks we did together. I thank Ute for many lunch breaks we spent together and for spending time together during walks or visits to the museum. I am grateful for the nice evenings with Anne and Kajetan, the discussions with them and their encouragement. I would also like to thank Chokri for his support and our discussions. I was very happy that I could visit Kevin every time I visited IST and I also appreciated our discussions a lot. Furthermore, I would like to thank Nicco, in particular for organising the Christmas cookie meeting for the last two years, which brought together many friends. I thank Katie for her support and for discussing some math together over Zoom during covid times. I would like to thank Anna for our meetings with amazing cake and wonderful discussions. I am very grateful to Vro for her support, and for the evenings and afternoons we have spent together.

I particularly thank my brother for his IT support which helped me numerous times. I would like to thank my entire family for their support, for listening a lot and always being there for me.

Zusammenfassung

Diese Doktorarbeit besteht aus vier Artikeln, die sich mit Eigenwertabschätzungen und Eigenwertasymptotiken, isoperimetrischen Ungleichungen und Funktionalungleichungen befassen.

Wir beweisen eine universelle Abschätzung für die Anzahl der negativen Eigenwerte von Schrödingeroperatoren auf Hölder-Gebieten mit Neumann-Randbedingungen. Anschließend benutzen wir diese Abschätzung, um eine semiklassische Asymptotik für die jeweiligen Schrödingeroperatoren zu zeigen. Wir diskutieren außerdem Fälle, in denen diese Abschätzung und Asymptotik nicht gelten.

In einem anderen Artikel zeigen wir verschiedene Weyl-Asymptotiken für den Laplace-Beltrami-Operator auf singulären Riemannschen Mannigfaltigkeiten. Die Motivation für das betrachtete Modell stammt von der Untersuchung der Ausbreitung von Schallwellen in Gasplaneten.

Zudem zeigen wir eine isoperimetrische Ungleichung für das Massenträgheitsmoment von der Menge aller Punkte in einem Gebiet mit festem Abstand zum Rand. Mithilfe dieser Ungleichung kann man eine isoperimetrische Ungleichung für den ersten Eigenwert von magnetischen Robin-Laplace-Operatoren zeigen.

Außerdem beweisen wir eine stärkere Version der Hardy-Ungleichung. Alle Terme in unserer Ungleichung haben dasselbe Skalierungsverhalten.

Abstract

This thesis consists of four works that deal with estimates and asymptotics for eigenvalues, isoperimetric and functional inequalities.

We prove a universal bound for the number of negative eigenvalues of Schrödinger operators on Hölder domains with Neumann boundary conditions. This bound is used to deduce semiclassical asymptotics for those operators. We also discuss a class of examples where those bounds and asymptotics fail.

In another work, we prove Weyl asymptotics for the Laplace-Beltrami operator on a class of singular Riemannian manifolds. This model is motivated by the study of the propagation of sound waves in gas planets.

We also show an isoperimetric inequality for moments of inertia of inner parallel curves. This inequality can be used to prove an isoperimetric inequality for the first eigenvalue of the magnetic Robin Laplacian.

Furthermore, we prove an improved version of Hardy's inequality. Our inequality only contains terms that scale in the same way.

Declaration of own contribution to the publications

This thesis consists of the following four research papers:

- [14] Semiclassical estimates for Schrödinger operators with Neumann boundary conditions on Hölder domains. Preprint 2023, arXiv:2304.01587
- [9] (with Yves Colin de Verdière, Maarten de Hoop and Emmanuel Trélat) Weyl formulae for some singular metrics with application to acoustic modes in gas giants. Preprint 2024, arXiv:2406.19734
- [16] (with Ayman Kachmar and Vladimir Lotoreichik) Isoperimetric inequalities for inner parallel curves. J. Spectr. Theory 14.4 (2024), 1537–1562, DOI: 10.4171/JST/534
- [17] (with Phan Thành Nam) Hardy-Sobolev interpolation inequalities.
 Calc. Var. 63, 184 (2024), DOI: https://doi.org/10.1007/s00526-024-02800-x

I will explain my own contribution to those papers in the following.

• [14] Semiclassical estimates for Schrödinger operators with Neumann boundary conditions on Hölder domains. Preprint 2023, arXiv:2304.01587

Phan Thành Nam suggested this project to me and we discussed the project in great detail together.

• [9] (with Yves Colin de Verdière, Maarten de Hoop and Emmanuel Trélat) Weyl formulae for some singular metrics with application to acoustic modes in gas giants. Preprint 2024, arXiv:2406.19734

For the subcritical case, I initially had a proof idea that used an argument similar to Rozenblum's proof of the CLR inequality by covering the domain by boxes, each of which corresponds to at most one eigenvalue. One of my collaborators found the change of functions, which was the key ingredient to make the analysis more convenient so we could avoid weighted measures. This was later particularly helpful for the proofs in the critical and supercritical case.

The proofs presented in this paper rely on heat kernel estimates, which mostly uses ideas from my collaborators. However, in parallel to the development of the proofs using heat kernels, I found alternative proofs involving Dirichlet-Neumann bracketing, which we mentioned in a remark in the paper. I explain these proofs in Section 3 below. My collaborators took care of large parts of the editing, and I carefully checked everything.

 [16] (with Ayman Kachmar and Vladimir Lotoreichik) Isoperimetric inequalities for inner parallel curves. J. Spectr. Theory 14.4 (2024), 1537–1562, DOI: 10.4171/JST/534

The idea for this project came from a talk from Ayman Kachmar, where he mentioned our result as an open problem. After many helpful discussions, I had the idea for the construction of the closed curve covering the inner parallel curve, which is the key ingredient of the paper. Vladimir Lotoreichik found a very important reference, which allowed us to use a certain result on the structure of inner parallel curves, which is intuitive but has a very lengthy proof. I wrote very detailed handwritten drafts, and my collaborators did large parts of the editing, which I then checked.

[17] (with Phan Thành Nam) Hardy-Sobolev interpolation inequalities.
 Calc. Var. 63, 184 (2024), DOI: https://doi.org/10.1007/s00526-024-02800-x

Many ideas of this work were found during discussions with Phan Thành Nam. He had the idea of using monotonicity for expressions of the form A(B - A), which was a key idea of the proof. I found two important references that greatly helped us with the proof and I also found the counterexamples detailed in the paper. Furthermore, I had the idea of how to extend the first result to Lorentz spaces. While I took care of writing detailed handwritten drafts for our results, Phan Thành Nam took care of most of the editing of the paper and afterwards, I checked everything.

Contents

Ι	Int	roduction	1		
1	Gen	General introduction			
2	Semiclassical estimates for Schrödinger operators on Hölder do-				
	mai	ns	6		
	2.1	Overview of [14]	6		
	2.2	Context for [14]	7		
	2.3	The min-max principle	8		
	2.4	Dirichlet-Neumann bracketing	10		
3	Weyl formulae for some singular metrics with application to acous-				
	tic r	modes in gas giants	14		
	3.1	Overview of [9]	14		
	3.2	Context for [9]	15		
	3.3	Normal form and change of variables	17		
	3.4	Statement of [9, Theorem 1] in the subcritical, critical and supercrit-			
		ical case	17		
	3.5	Hausdorff dimension	18		
	3.6	Discreteness of the spectrum	18		
	3.7	Quasi-isometric metric	19		
	3.8	Change of function and change of measure	19		
	3.9	Separation of variables and the one-dimensional operator P_{μ}	20		
	3.10	Application of the Weyl asymptotics for M	21		
	3.11	Unitary equivalence of P_{μ} and scaling $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	22		
	3.12	Removing zero and subleading contributions	22		
	3.13	End of the proof in the critical case	24		
	3.14	End of the proof in the supercritical case	26		
	3.15	End of the proof in the subcritical case	27		

4	Isop	perimetric inequalities for inner parallel curves	29
	4.1	Overview of [16]	29
	4.2	Context for [16]	30
	4.3	The dumbbell example	31
	4.4	General proof strategy	32
	4.5	A first naive try for the construction	33
	4.6	General ideas on the modification of the closed curve $\ldots \ldots \ldots$	35
	4.7	Construction of the closed curve Σ_t	36
5	Har	dy-Sobolev interpolation inequalities	38
	5.1	Overview of [17]	38
	5.2	Context of [17]	39
	5.3	Radial example	40
	5.4	Application of the radial example to (5.3)	41
	5.5	Bubbling example	42
	5.6	Outlook: Existence of optimisers?	43
II	A	rticles	50
6	Semiclassical estimates for Schrödinger operators on Hölder do- mains 5		
7	Weyl formulae for some singular metrics with application to acous- tic modes in gas giants 9		
8	Isoperimetric inequalities for inner parallel curves		
9	Hardy-Sobolev interpolation inequalities 14		

Part I

Introduction

Chapter 1

General introduction

One of the great early successes of quantum mechanics was its rigorous proof of atomic stability, a phenomenon that classical physics could not explain. In the simplest case of the hydrogen atom, the question of stability asks whether there exists a constant C > 0 such that for all $u \in H^1(\mathbb{R}^3)$ with $||u||_{L^2} = 1$, we have

$$\int_{\mathbb{R}^3} |\nabla u|^2 - \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|} dx \ge -C.$$
 (1.1)

Here the left-hand side of (1.1) represents the energy of the wave function describing an electron in a hydrogen atom. Put differently, we ask if the energy for L^2 normalised wave functions u is bounded from below.

In the case of the hydrogen atom, the optimal constant in (1.1) is known to be C = 1/4 using explicit computations. In the following, I would like to explain an alternative approach of proving (1.1) for some (non-optimal) C > 0 that relies on Hardy's inequality, which can be seen as an uncertainty principle. Hardy's inequality states that for every function $u \in \dot{H}^1(\mathbb{R}^3)$

$$\int_{\mathbb{R}^3} |\nabla u|^2 \ge \frac{1}{4} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|^2} dx.$$
(1.2)

Now note that

$$0 \leqslant \left(1 - \frac{1}{2|x|}\right)^2 = 1 - \frac{1}{|x|} + \frac{1}{4|x|^2}, \quad \text{so } \quad \frac{1}{|x|} \leqslant \frac{1}{4|x|^2} + 1.$$
(1.3)

We obtain using (1.3) and (1.2) that for all $u \in H^{(\mathbb{R}^3)}$ with $||u||_{L^2} = 1$

$$\int_{\mathbb{R}^3} |\nabla u|^2 - \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|} dx \ge \int_{\mathbb{R}^3} |\nabla u|^2 - \frac{1}{4} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|^2} - \int_{\mathbb{R}^3} |u|^2 \ge -1.$$
(1.4)

This shows that the hydrogen atom is stable in the sense of (1.1).

The quantity

$$\inf_{u \in H^1(\mathbb{R}^3), \|u\|_{L^2} = 1} \left(\int_{\mathbb{R}^3} |\nabla u|^2 - \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|} dx \right)$$
(1.5)

is equal to the lowest eigenvalue (which is usually called the ground state) of the Schrödinger operator¹ $-\Delta - 1/|x|$. The eigenfunctions of $-\Delta - 1/|x|$ correspond to the bound states of the hydrogen atom. The eigenvalues of $-\Delta - 1/|x|$ can be computed using the min-max principle, see Section 2.3 below.

More generally, one may consider Schrödinger operators $-\Delta + V$ on domains $\Omega \subset \mathbb{R}^d, d \in \mathbb{N}$, where $V : \Omega \to \mathbb{R}$ is usually called a potential. The corresponding quadratic form describing the energy of L^2 -normalised wave functions u is

$$Q(u) := \int_{\Omega} |\nabla u|^2 + \int_{\Omega} V(x) |u(x)|^2 \, dx \,. \tag{1.6}$$

If $\Omega \not\subseteq \mathbb{R}^d$ is a domain and we consider Q(u) for all $u \in H^1(\Omega)$, we say that we take Neumann boundary conditions². If we only consider all $u \in H^1_0(\Omega)$ (where $H^1_0(\Omega)$ denotes the closure of all smooth compactly supported functions in Ω with respect to the $H^1(\Omega)$ norm), we refer to this as Dirichlet boundary conditions.

Similarly to (1.5), the lowest eigenvalue³ of $-\Delta + V$ is characterised by the infimum of Q(u) taken over all u in the quadratic form domain (here $H^1(\Omega)$ or $H^1_0(\Omega)$) with $||u||_{L^2} = 1$. Thus, the lowest eigenvalue of Schrödinger operators with Neumann boundary conditions is less than or equal to the lowest eigenvalue of the corresponding Schrödinger operator with Dirichlet boundary conditions. This is also true for higher eigenvalues. Thus, the number of eigenvalues below a certain threshold, say λ , of a Schrödinger operator with Neumann boundary conditions is larger than or equal to this number for the corresponding Schrödinger operator with Dirichlet boundary conditions.

In a seminal work, Weyl proved in 1911 [50] that for any bounded domain $\Omega \subset \mathbb{R}^d$, the number of eigenvalues below λ of the Dirichlet Laplacian $-\Delta$ on Ω , which we

¹For simplicity, we ignore domain issues and the question of self-adjointness here.

²Strictly speaking, we would like to assume that V is such that $\int_{\Omega} V|u|^2 < \infty$ for all $u \in H^1(\Omega)$ here in order to ensure that $Q(u) < \infty$ for all $u \in H^1(\Omega)$.

³Since we do not know in general if the operator has discrete spectrum, we should rather speak of the lowest min-max value.

denote by $N(\lambda)$, satisfies the following asymptotics

$$N(\lambda) = C_d |\Omega| \lambda^{d/2} + o\left(\lambda^{d/2}\right) \text{ as } \lambda \to \infty, \qquad (1.7)$$

where $C_d > 0$ is a constant that only depends on the dimension d.

Since then, deducing similar asymptotics has remained an active field of research. For instance, one can ask if similar asymptotics hold in the case of the Neumann Laplacian. There the situation is more complicated, and in fact, there are bounded domains such that the corresponding Neumann Laplacian has zero in its essential spectrum [29]. Furthermore, instead of considering a Laplace operator $-\Delta$, one can ask for asymptotics for the number of negative eigenvalues of Schrödinger operators $-\Delta + \lambda V$ as $\lambda \to \infty$. The result is well-known for Lipschitz domains, see for example [22]. My work [14] described in Section 2 below studies this question for the Neumann Laplacian on Hölder domains.

Another direction is to prove Weyl asymptotics for the Laplace-Beltrami operator on Riemannian manifolds. In [9] with Yves Colin de Verdière, Maarten de Hoop and Emmanuel Trélat, see Section 3 below, we prove Weyl asymptotics for the Laplace-Beltrami operator on a singular Riemannian manifold. Our model comes from the study of the propagation of soundwaves in gas planets and has links to sub-Riemannian geometry.

In certain cases, geometric inequalities can be useful to deduce properties of the spectrum. Together with Ayman Kachmar and Vladimir Lotoreichik [16], we proved weighted isoperimetric inequalities for smooth, bounded, and simply connected domains. More precisely, we show that the moment of inertia of inner parallel curves for domains with fixed perimeter attains its maximum for a disk. This inequality, which was previously only known for convex domains, allows us to extend an isoperimetric inequality for the first eigenvalue of the magnetic Robin Laplacian.

One can also investigate other functional inequalities and their properties. In a joint work with Phan Thành Nam [17], we derive a family of interpolation estimates which improve Hardy's inequality (1.2) and are critical in some sense. We also determine all optimisers among radial functions in the corresponding radial problem and discuss open questions.

This thesis contains the works [14, 9, 16, 17]. In this introduction, we give a brief overview of each of the papers. We also explain the context, mention the most relevant known results and references, and describe how they are linked to the paper.

Furthermore, for each paper, we choose to explain or emphasise certain aspects, which we find useful to simplify the reading or to give additional background information. In particular, we will explain the concept of min-max values and Dirichlet-Neumann bracketing in Section 2. In Section 3, we give an alternative proof sketch of the main result of [9] using Dirichlet-Neumann bracketing. In Section 4, we explain the intuition and key steps of the elementary geometric construction we used in [16]. In Section 5, we focus on different counterexamples and give an outlook on the remaining open question of existence of optimisers for our functional inequality.

Chapter 2

Semiclassical estimates for Schrödinger operators on Hölder domains

2.1 Overview of [14]

In a work of mine [14], I considered Schrödinger operators $-\Delta + V$ on bounded Hölder domains $\Omega \subset \mathbb{R}^d$ (that is, the boundary of Ω is locally the graph of a Höldercontinuous function) with Neumann boundary conditions, where the potential V is a non-positive function on Ω . My work [14] was motivated by the detailed study [41] of Weyl asymptotics for the Laplacian $-\Delta$ on bounded Hölder domains with Neumann boundary conditions.

I proved the validity of the same leading order eigenvalue asymptotics as for Schrödinger operators on Lipschitz domains, under certain assumptions on the potential V in the optimal range of Hölder exponents. More precisely, if $N(-\Delta + \lambda V)$ denotes the number of negative eigenvalues of the Schrödinger operator $-\Delta + \lambda V$, then

$$N(-\Delta + \lambda V) = C_d \lambda^{d/2} \int_{\Omega} |V|^{d/2} + o\left(\lambda^{d/2}\right) \text{ as } \lambda \to \infty, \qquad (2.1)$$

where $C_d > 0$ is a universal constant only depending on the dimension d. My result is valid in all dimensions $d \ge 2$, but for simplicity, let us assume $d \ge 3$ in the following explanations.

Furthermore, I provided an example, which shows that the assumptions on the potential V in the setting of Hölder domains indeed need to be stronger than in the case of Lipschitz domains (where $V \in L^{d/2}(\Omega)$ is sufficient). This is a surprising

result since the condition $V \in L^{d/2}(\Omega)$ seems very natural in view of (2.1).

The proof of (2.1) relies on the new universal bound for the number $N(-\Delta+V)$ of negative eigenvalues of Schrödinger operators $-\Delta+V$

$$N(-\Delta+V) \leqslant C ||V|||^{d/2}, \tag{2.2}$$

where $\|\|\cdot\|\|$ is a weighted L^p -norm for p > d/2 with a weight that grows near the boundary $\partial\Omega$. The proof of (2.2) is inspired by the proof of Rozenblum of the "classical" Cwikel-Lieb-Rozenblum inequality [45], but I made substantial modifications and adaptations.

I covered the domain Ω by smaller domains, which are carefully chosen rectangles intersected with Ω such that each of the smaller domains supports at most one negative eigenvalue of the Schrödinger operator $-\Delta + V$. To this end, I proved a new Poincaré-Sobolev inequality and a new Besicovitch-type covering theorem for those smaller domains. The proof of the Poincaré-Sobolev inequality uses a Sobolev embedding for Hölder domains [34]. The key ingredient for the Besicovitch-type covering theorem is that the smaller domains are chosen such that the ratio of the side-lengths of each rectangle are comparable for all chosen rectangles intersecting each other. The number of negative eigenvalues $N(-\Delta + V)$ can then be estimated by the number of smaller domains chosen using the min-max principle, see Section 2.3 below.

One of the main challenges in this problem was to get an exponent of d/2 on the right-hand side of (2.2). This is essential for deducing (2.1). Previous results, see for example [23], could only get an exponent that is strictly larger than d/2. On the technical side, the careful choice of the smaller domains, an application of Hölder's inequality for sums of real numbers and the use of the weighted L^p -norm $||| \cdot |||$ made it possible to obtain (2.2) with the exponent d/2.

2.2 Context for [14]

For the Dirichlet Laplacian on domains $\Omega \subset \mathbb{R}^d$ with finite measure, we always have the leading-order Weyl asymptotics (1.7) [50, 44]. For the Neumann Laplacian, the situation is much more delicate. Intuitively speaking, the eigenfunctions of the Neumann Laplacian can grow near the boundary and if the boundary is very rough, in particular with many outward pointing cusps, then many eigenfunctions can accumulate near the boundary leading to a larger number of eigenvalues than what one would expect from (1.7). There are bounded domains that have zero in the essential spectrum of the Neumann Laplacian, and one can even construct for any closed $S \subset [0, \infty)$ bounded domains such that the essential spectrum of the Neumann Laplacian is given by S [29].

If the domain Ω is an H^1 -extension domain, that is, we can extend functions in $H^1(\Omega)$ to H^1 -functions on the entire space \mathbb{R}^d in such a way that their $H^1(\mathbb{R}^d)$ norm is comparable with their $H^1(\Omega)$ -norm, and similarly for the L^2 -norms, then a classical argument shows that the Neumann Laplacian has discrete spectrum and the leading-order Weyl asymptotics (1.7) hold, see for example [22, Theorem 3.20]. Lipschitz domains are extension domains, but Hölder domains for a Hölder exponent $\gamma < 1$ are in general not extension domains. Thus, it is a priori unclear if we have (1.7) for Hölder domains.

In [41], the authors considered the Neumann Laplacian on bounded Hölder domains $\Omega \subset \mathbb{R}^d$, $d \ge 2$. Denoting the Hölder exponent of the function locally describing the boundary by γ , they showed that if $\gamma \in ((d-1)/d, 1)$, then the usual Weyl asymptotics (1.7) as in the case of Lipschitz domains hold. Note that a Hölder exponent $\gamma = 1$ means that Ω is a Lipschitz domain. Furthermore, if $\gamma \in (0, (d-1)/d]$, then they provided an explicit counterexample for a domain $\Omega \subset \mathbb{R}^d$ with Hölder exponent γ and such that (1.7) fails. It is remarkable that they could identify a critical value (d-1)/d, where the asymptotic behaviour changes drastically.

My work [14] treats similar questions in the case of Schrödinger operators $-\Delta + V$ on bounded Hölder domains $\Omega \subset \mathbb{R}^d$ with Neumann boundary conditions.

In the following, I will explain the min-max principle and Dirichlet-Neumann bracketing, which have been important tools for my work [14].

2.3 The min-max principle

This section is devoted to an explanation of the min-max principle, which goes back to [43, 20, 13]. It is a very powerful technique in spectral theory, see [37, Theorem 12.1] for an overview.

Let us first introduce the min-max principle in a general setting [37, Theorem 12.1, Version 3]. If \mathcal{H} is a Hilbert space and $A : \mathcal{D}(A) \to \mathcal{H}$ is a self-adjoint operator on \mathcal{H} with domain $\mathcal{D}(A)$ and with $A \ge -C$ for some C > 0, then we define by

$$\mu_n(A) := \inf_{M \subset \mathcal{D}(A), \dim(M) = n} \sup_{u \in M, \|u\| = 1} \langle u, Au \rangle$$
(2.3)

the *n*-th min-max value for all $n \in \mathbb{N}$. The *M* in the infimum in (2.3) is a subspace of $\mathcal{D}(A)$ of dimension *n*. If *A* has at least *n* discrete eigenvalues below its essential spectrum, then $\mu_n(A)$ agrees with the *n*-th lowest eigenvalue of *A*, counted with multiplicity.

In particular, if A has compact resolvent, then (2.3) gives a variational characterisation of the eigenvalues of A. More generally, instead of considering M as an *n*-dimensional subspace of $\mathcal{D}(A)$ in (2.3), we can also take M as an *n*-dimensional subspace of a form core of the corresponding quadratic form $Q(\cdot)$, and replace $\langle u, Au \rangle$ by Q(u). This is very convenient as we need not know the domain of A explicitly, but it suffices to work on the level of quadratic forms.

While we can refer to the $\mu_n(A)$ from (2.3) as min-max values (think of the infimum as a minimum and the supremum as a maximum), there is also a version of max-min values [37, Theorem 12.1, Version 2]

$$\nu_n(A) := \sup_{L \subset \mathcal{D}(A), \dim(L) = n-1} \inf_{u \in L^\perp, \|u\| = 1} \langle u, Au \rangle.$$
(2.4)

Again, one can alternatively work with the corresponding quadratic form and a quadratic form core, and furthermore, by the proof of [37, Theorem 12.1, Version 2], it suffices to assume that L is an n-1-dimensional subspace of our Hilbert space \mathcal{H} , which is sometimes referred to as Glazman's lemma [22, Theorem 1.26]. One can show that for all $n \in \mathbb{N}$

$$\mu_n(A) = \nu_n(A), \qquad (2.5)$$

see [37, Theorem 12.1, Version 2, 3]. Thus, we have two very different variational characterisations of the n-th eigenvalue below the essential spectrum of an operator. Both of them are very helpful in applications (see also Section 2.4).

For instance, the second characterisation can be useful when estimating the value of the second eigenvalue for a Neumann Laplacian: Suppose that we consider a box $\Omega := [0, 1]^d$ in dimension $d \ge 3$ and we consider the Schrödinger operator $-\Delta + V$ with Neumann boundary conditions and $V \in L^{d/2}(\Omega)$. Assume that we have shown that

$$\int_{\Omega} |\nabla u|^2 + \int_{\Omega} V|u|^2 \ge 0 \quad \text{for all } u \in H^1(\Omega) \text{ with } \int_{\Omega} u = 0.$$
 (2.6)

Note that the corresponding quadratic form Q(u) is the left-hand side of (2.6) with quadratic form domain $H^1(\Omega)$ (since we consider the Neumann Laplacian). Then defining L as the subspace of $H^1(\Omega)$ that is spanned by the constant function on Ω , the condition $\int_{\Omega} u = 0$ is equivalent to saying $u \in L^{\perp}$. Thus, (2.6) states that $Q(u) \ge 0$ for all $u \in L^{\perp}$. We deduce that $\nu_2(-\Delta + V) \ge 0$, or put differently, the number of negative eigenvalues of $-\Delta + V$ is at most one. The argument we presented here is classical [45] and we also use it in different variations in [14], for instance in the proof of [14, Lemma 1.8]. In practice, if Ω is a box as in our example, a sufficient condition for (2.6) is a sufficiently small negative part of V in the $L^{d/2}(\Omega)$ -norm.

2.4 Dirichlet-Neumann bracketing

In this section, we explain Dirichlet-Neumann bracketing. A historically particularly relevant work is due to Weyl [50]. We would like to refer to [47, Theorem 7.5.28] for a more recent source that also includes historical remarks.

I will explain Dirichlet-Neumann bracketing for a domain Ω in euclidean space that is decomposed into two smaller domains Ω_1 and Ω_2 . I will allow for mixed boundary conditions for Ω_1 and Ω_2 , which appear naturally when imposing Dirichlet or Neumann boundary conditions on Ω and performing the Dirichlet-Neumann bracketing. More generally, this technique can also be used for Schrödinger operators or for the Laplace-Beltrami operator on a Riemannian manifold. We used Dirichlet-Neumann bracketing in both [14] and [9].

Let us consider an open set $\Omega \subset \mathbb{R}^d$, $d \in \mathbb{N}$. Recall that

$$H^{1}(\Omega) = \left\{ u \in L^{2}(\Omega) \mid u \text{ weakly differentiable and } \nabla u \in L^{2}(\Omega) \right\}.$$
 (2.7)

Then the Neumann Laplacian on Ω is the unique self-adjoint operator that corresponds to the quadratic form

$$\int_{\Omega} |\nabla u|^2 \quad \text{for } u \in H^1(\Omega).$$
(2.8)

Let us now define for every $E \subset \mathbb{R}^d$ the space $A_E(\Omega)$ of $H^1(\Omega)$ -functions that vanish near E as follows: $u \in A_E(\Omega)$ if and only if $u \in H^1(\Omega)$ and there exists an open neighbourhood $V \subset \mathbb{R}^d$ of E such that u(x) = 0 for almost every $x \in V \cap \Omega$. Note that for any $E \subset \mathbb{R}^d$, we have $A_E(\Omega) = A_{E \cap \overline{\Omega}}(\Omega)$. A similar definition to the spaces $A_E(\Omega)$ can be found for instance in [26].

Then the Dirichlet Laplacian on Ω can be defined as the unique self-adjoint

operator that corresponds to the (closeable) quadratic form

$$\int_{\Omega} |\nabla u|^2 \quad \text{for } u \in A_{\partial\Omega}(\Omega).$$
(2.9)

Note that the closure of $A_{\partial\Omega}(\Omega)$ with respect to the $H^1(\Omega)$ -norm (which is in this case also the quadratic form norm) is equal to the closure of $C_c^{\infty}(\Omega)$ functions with respect to the $H^1(\Omega)$ -norm, which is usually denoted by $H_0^1(\Omega)$.

Using a quadratic form with form domain $A_E(\Omega)$ for different sets $E \subset \partial \Omega$ allows us to define Laplace operators with mixed boundary conditions. The closeable quadratic form $\int_{\Omega} |\nabla u|^2$ with $u \in A_E(\Omega)$ gives rise to a unique self-adjoint operator, which we refer to as having Dirichlet boundary conditions on E and Neumann boundary conditions on the rest of the boundary.

Denote for $\lambda \in \mathbb{R}$ and measurable $E \subset \mathbb{R}^d$ by $N_{\Omega}^E(\lambda)$ the number of eigenvalues (counted with multiplicity) below λ of the operator that corresponds to the quadratic form

$$\int_{\Omega} |\nabla u|^2 \quad \text{for } u \in A_E(\Omega), \tag{2.10}$$

and let $N_{\Omega}^{E}(\lambda) = \infty$ if λ is greater than or equal to the infimum of the essential spectrum of that operator.

Proposition 2.1 (Dirichlet-Neumann bracketing). Let $\Omega \subset \mathbb{R}^d$ be an open subset and let $\Omega_1, \Omega_2 \subset \Omega$ be open and disjoint such that $\Sigma = \Omega \setminus (\Omega_1 \cup \Omega_2)$ has zero Lebesgue measure. Then for any $\lambda \in \mathbb{R}$, we have a two-sided estimate for the Dirichlet Laplacian on Ω

$$N_{\Omega_1}^{\partial\Omega_1}(\lambda) + N_{\Omega_2}^{\partial\Omega_2}(\lambda) \leqslant N_{\Omega}^{\partial\Omega}(\lambda) \leqslant N_{\Omega_1}^{\partial\Omega}(\lambda) + N_{\Omega_2}^{\partial\Omega}(\lambda)$$
(2.11)

and a two-sided estimate for the Neumann Laplacian on Ω

$$N_{\Omega_1}^{\Sigma}(\lambda) + N_{\Omega_2}^{\Sigma}(\lambda) \leqslant N_{\Omega}^{\varnothing}(\lambda) \leqslant N_{\Omega_1}^{\varnothing}(\lambda) + N_{\Omega_2}^{\varnothing}(\lambda).$$
(2.12)

More generally, if $S \subset \partial \Omega$ denotes a measurable set, then we have a two-sided estimate for the Laplacian on Ω with Dirichlet boundary conditions on S and Neumann boundary conditions elsewhere

$$N_{\Omega_1}^{S\cup\Sigma}(\lambda) + N_{\Omega_2}^{S\cup\Sigma}(\lambda) \leqslant N_{\Omega}^S(\lambda) \leqslant N_{\Omega_1}^S(\lambda) + N_{\Omega_2}^S(\lambda).$$
(2.13)

(2.11) and (2.12) show that one can estimate the eigenvalues for a Laplacian with Dirichlet or Neumann boundary conditions on $\partial\Omega$ by the sums of the number of the corresponding eigenvalues on the two subdomains Ω_1 , Ω_2 after putting additional Dirichlet and Neumann boundary conditions on the interface Σ .

Proof. For concreteness, we stated the two most commonly used versions where we have Dirichlet or Neumann boundary conditions on the entire boundary of Ω separately, but of course, it suffices to show (2.13) and then set $S = \partial \Omega$ or $S = \emptyset$ to obtain (2.11) or (2.12).

First note that

$$\Sigma = \Omega \cap (\partial \Omega_1 \cup \partial \Omega_2). \tag{2.14}$$

The proof of this is a contradiction argument that involves the construction of small open balls completely contained in Σ , which is a contradiction to Σ having zero Lebesgue measure. We omit the details here.

Estimate with extra Dirichlet conditions on Σ . Let us first show that

$$N_{\Omega_1}^{S\cup\Sigma}(\lambda) + N_{\Omega_2}^{S\cup\Sigma}(\lambda) \leqslant N_{\Omega}^S(\lambda).$$
(2.15)

To this end, we will use the min-max characterisation of eigenvalues (2.3). For $k \in \{1, 2\}$, we denote $K_k := N_{\Omega_k}^{S \cup \Sigma}(\lambda)$ and note that by the version of (2.3) with a quadratic form core, there exists a subspace $M_k \subset A_{S \cup \Sigma}(\Omega_k)$ with

$$\int_{\Omega_k} |\nabla u_k|^2 < \lambda \int_{\Omega_k} |u_k|^2 \quad \text{for all } 0 \neq u_k \in M_k.$$
(2.16)

By extending the functions u_k by zero on $\Omega \setminus \Omega_k$, we can think of them as functions in $A_S(\Omega)$. Here we used that the u_k vanish in a neighbourhood of $S \cup \Sigma$. Define

$$M := \{ u_1 + u_2 \mid u_1 \in M_1, \, u_2 \in M_2 \},$$
(2.17)

and note that $K := \dim(M) = K_1 + K_2$. Writing every $0 \neq u \in M$ as $u = u_1 + u_2$ with $u_k \in M_k$, k = 1, 2 and $0 \neq u_1$ or $0 \neq u_2$, it follows that

$$\int_{\Omega} |\nabla u|^2 = \int_{\Omega_1} |\nabla u_1|^2 + \int_{\Omega_2} |\nabla u_2|^2 < \lambda \int_{\Omega_1} |u_1|^2 + \lambda \int_{\Omega_2} |u_2|^2 = \lambda \int_{\Omega} |u|^2, \quad (2.18)$$

so $N_{\Omega}^{S}(\lambda) \ge K = K_{1} + K_{2}$. Using the definition of K_{1} and K_{2} , we deduce (2.15).

Estimate with extra Neumann conditions on Σ . Next, let us show that

$$N_{\Omega}^{S}(\lambda) \leqslant N_{\Omega_{1}}^{S}(\lambda) + N_{\Omega_{2}}^{S}(\lambda).$$
(2.19)

In order to see this, we will use the second characterisation of min-max values, which we referred to as max-min values above. For $k \in \{1, 2\}$, we define $K_k := N_{\Omega_k}^S(\lambda)$. By (2.4), or more precisely its version with a quadratic form core and L a subspace of the corresponding Hilbert space, see for example [22, Theorem 1.26], there exists a subspace $L_k \subset L^2(\Omega_k)$ of dimension K_k such that for every $u_k \in A_S(\Omega_k)$ with $u_k \in L_k^{\perp}$, we have

$$\int_{\Omega_k} |\nabla u_k|^2 \ge \lambda \int_{\Omega_k} |u_k|^2.$$
(2.20)

Next, we identify functions $l_k \in L_k$ with functions in $L^2(\Omega)$ by extending them by zero on $\Omega \setminus \Omega_k$. Define

$$L := \{ l = l_1 + l_2 \mid l_1 \in L_1, \, l_2 \in L_2 \}$$

$$(2.21)$$

and note that L is a subspace of $L^2(\Omega)$ of dimension $K := K_1 + K_2$.

Let $u \in A_S(\Omega)$ with $u \in L^{\perp}$. In particular, for any $k \in \{1, 2\}$ and any $l_k \in L_k$, we have

$$\int_{\Omega_k} u l_k = \int_{\Omega} u l_k = 0, \qquad (2.22)$$

where we used that we extended the l_k by zero on $\Omega \setminus \Omega_k$. Thus, the restriction of u to Ω_k , which we denote by $u_k \in A_S(\Omega_k)$ satisfies $u_k \in L_k^{\perp}$. By (2.20), it follows that

$$\int_{\Omega_k} |\nabla u_k|^2 \ge \lambda \int_{\Omega_k} |u_k|^2.$$
(2.23)

Putting everything together and using that Σ has zero Lebesgue measure, we obtain

$$\int_{\Omega} |\nabla u|^2 = \int_{\Omega_1} |\nabla u_1|^2 + \int_{\Omega_2} |\nabla u_2|^2 \ge \lambda \int_{\Omega_1} |u_1|^2 + \lambda \int_{\Omega_2} |u_2|^2 = \lambda \int_{\Omega} |u|^2 \quad (2.24)$$

Thus, $N_{\Omega}^{S}(\lambda) \leq K$, which completes the proof of (2.19).

Chapter 3

Weyl formulae for some singular metrics with application to acoustic modes in gas giants

3.1 Overview of [9]

Seismology for gas planets plays an important role in understanding their interior. Astrophysicists use for instance ring seismology for Saturn when measuring eigenfrequencies of soundwaves inside the planet [38]. A particularity of gas planets is that the speed of sound goes to zero near the boundary of the planet. Physicists use a model for this behaviour that includes a parameter α depending on the density profile of the planet near the boundary, which depends on the chemical decomposition of the planet.

The propagation of sound waves in gas planets can be modelled by the Laplace-Beltrami operator on a compact Riemannian manifold X with boundary with a certain class of Riemannian metrics that become singular at the boundary. Near the boundary ∂X , these Riemannian metrics are of the form

$$g = u^{-\alpha} \bar{g}, \tag{3.1}$$

where u is a suitable coordinate transverse to the boundary with u = 0 at the boundary, \bar{g} is a smooth Riemannian metric on our manifold up until its boundary and $0 < \alpha < 2$. It has been shown in [30] that the spectrum of the corresponding Laplace-Beltrami operator is discrete.

Our work with Yves Colin de Verdière, Maarten de Hoop and Emmanuel Trélat

[9] provides a proof of the asymptotics of the eigenfrequencies of gas planets for any $0 < \alpha < 2$, which are the physically relevant cases [9, Theorem 1]. Furthermore, we show how most of the eigenfunctions are distributed inside the planet [9, Theorem 2].

When α is larger than a certain critical parameter, then the Hausdorff dimension of (X, g) is strictly larger than its topological dimension and we proved in [9] that the Weyl asymptotics are determined by α and the metric \bar{g} at the boundary. In this case, the Weyl measure, which describes the limiting averaged distribution of the eigenfunctions, is the uniform distribution on the boundary. Below this critical parameter, the Weyl asymptotics agree with the usual Weyl asymptotics on smooth Riemannian manifolds and the Weyl measure is a uniform distribution on X. We also covered the case of the critical value of α .

We have two different proofs; one of them uses heat kernel asymptotics and the other one uses Dirichlet-Neumann bracketing. Both in the critical and supercritical parameter range of α , it is crucial that our metric near the boundary is quasiisometric to a metric that has nice scaling properties.

3.2 Context for [9]

It is a classical result that for smooth compact Riemannian manifolds X of dimension $d \in \mathbb{N}$ with a smooth non-degenerate Riemannan metric g, the associated Laplace-Beltrami operator Δ_g has discrete spectrum and the eigenvalues satisfy the leading-order Weyl asymptotics

$$N(\lambda) = C_d \operatorname{vol}(X, g) \lambda^{d/2} + o(\lambda^{d/2}) \quad \text{as } \lambda \to \infty,$$
(3.2)

where $N(\lambda)$ denotes the number of eigenvalues of Δ_g that are less than λ counted with multiplicity, and vol(X, g) the corresponding Riemannian volume.

In [40], the authors gave a full asymptotic expansion of the corresponding heat kernel, which can be used to deduce (3.2) by a Karamata theorem [19, Chapter XIII, Theorem 2]. More generally, we still have (3.2) if X has a smooth boundary, for instance with Dirichlet boundary conditions, see for example [25] and references therein for a review of different techniques for heat kernel asymptotics in the case with and without boundary.

More generally, one can ask for eigenvalue asymptotics in the case of singular metrics, where the singularity could be both on the boundary of the manifold or in its interior.

In the case of domains in euclidean space with a certain power-type singularity at the boundary, Métivier [39] gave an extensive description and derived Weyl's law in different settings. In his proofs, he worked with weighted Sobolev spaces. Using his results, one can also derive the Weyl law in our setting (3.1) after suitably localising near the boundary and using local coordinates, though the constants in [39] are not explicit. We learnt about [39] after the completion of our work [9] from Bernard Helffer, who we would like to thank for this very helpful comment. We would also like to thank an anonymous referee for pointing out the reference [49], which also includes the Weyl law in our setting (3.1) in suitable coordinates.

More recently, in [4], the authors considered the example of a Grushin cylinder and a Grushin sphere where they computed the eigenvalues and corresponding eigenfunctions explicitly, thereby also establishing a Weyl law. The Weyl law in their example is a special case for our setting (3.1) with a critical value of α .

We would also like to mention the recent work [8], which considered singular Riemannian manifolds with a singularity at the boundary under different assumptions on the metric. The authors identify subcritical, critical and supercritical cases depending on the behaviour of the Riemannian volume of the set of points with distance at most $1/\sqrt{\lambda}$ for $\lambda \to \infty$ and derive a corresponding Weyl law in the subcritical and critical case. In the supercritical case, they derive corresponding bounds.

We would also like to refer to [12, 10, 11] for various results on asymptotic expansions of heat kernels in sub-Riemannian geometry and local Weyl laws. In particular, these papers treat the Grushin case, which corresponds to $\alpha = 1$ in (3.1).

In [9, Theorem 2], we showed that in the supercritical regime, most of the eigenfunctions accumulate at the boundary. In [18], I showed with Larry Read that this happens at scale $\lambda^{-1/(2-\alpha)}$, and we also identified a profile that emerges after zooming in at that scale.

In the following, I would like to give more details on the statement and the proof of [9, Theorem 1] using Dirichlet-Neumann bracketing [9, Remark 2]. Some of the proof ideas we explain here were rewritten in [18, Section 2] in the critical case for sums of eigenvalues and with a rescaled test function. After the completion of this work, we realised that very similar ideas were also used in [49].

3.3 Normal form and change of variables

We let $n \in \mathbb{N}$ such that the Riemannian manifold X is of dimension n + 1. The part of X near the boundary is diffeomorphic to $[0, 1] \times M$, where M is a smooth ndimensional Riemannian manifold that is diffeomorphic to the boundary; we identify $\{0\} \times M$ with ∂X . Near the boundary ∂X , we can choose our coordinates such that g can be expressed as

$$g = \frac{1}{u^{\alpha}} (du^2 + g_0(u)) \text{ for } (u, y) \in [0, 1] \times M,$$
(3.3)

which is referred to as a normal form, see [30, Lemma 5.2]. Here $(g_0(u))_{u \in [0,1]}$ is a family of smooth non-degenerate Riemannian metrics on M that depend continuously on u.

We can then perform a change of variables $x = 1/(1 - \alpha/2)u^{1-\alpha/2}$ and obtain

$$g = dx^2 + x^{-\beta}g_1(x)$$
 for $(x, y) \in [0, 1] \times M$, where $\beta := \frac{2\alpha}{2 - \alpha}$. (3.4)

Again, $(g_1(x))_{x \in [0,1]}$ is a family of smooth non-degenerate Riemannian metrics on M that depend continuously on x.

3.4 Statement of [9, Theorem 1] in the subcritical, critical and supercritical case

Denote by Δ_g the Laplace-Beltrami operator on X with Dirichlet boundary conditions on ∂X (consider the corresponding quadratic form for smooth compactly supported functions). We would like to understand the asymptotics for the number $N(\lambda)$ of eigenvalues of Δ_g that are less than λ for $\lambda \to \infty$. For the parameter $0 < \alpha < 2$ in (3.1), there is a subcritical regime $\alpha < 2/(n+1)$, a critical value $\alpha = 2/(n+1)$, and a supercritical regime $\alpha > 2/(n+1)$.

In the subcritical case $\alpha < 2/(n+1)$, we show in [9, Theorem 1] that we have the asymptotics

$$N(\lambda) = C_{n+1} \operatorname{vol}(X, g) \lambda^{(n+1)/2} + o\left(\lambda^{(n+1)/2}\right) \quad \text{as } \lambda \to \infty.$$
(3.5)

Here C_{n+1} denotes a universal semiclassical constant depending on the dimension n+1, and vol(X, g) is the Riemannian volume of X with respect to g. These asymptotics (3.5) agree with the leading order Weyl asymptotics for non-singular smooth

Riemannian metrics on compact Riemannian manifolds, possibly with boundary.

In the **critical case** $\alpha = 2/(n+1)$, we show in [9, Theorem 1] that

$$N(\lambda) = K_n \operatorname{vol}(M, g_1(0)) \lambda^{(n+1)/2} \log(\lambda) + o\left(\lambda^{(n+1)/2} \log(\lambda)\right) \quad \text{as } \lambda \to \infty$$
(3.6)

for some explicit constant K_n only depending on n. Here $vol(M, g_1(0))$ denotes the Riemannian volume of M with respect to $g_1(0)$.

In the supercritical case $\alpha > 2/(n+1)$, [9, Theorem 1] states that

$$N(\lambda) = K_{n,\alpha} \operatorname{vol}(M, g_1(0)) \lambda^{n/(2-\alpha)} + o\left(\lambda^{n/(2-\alpha)}\right) \quad \text{as } \lambda \to \infty,$$
(3.7)

where $K_{n,\alpha}$ is an explicit constant only depending on n and α .

3.5 Hausdorff dimension

The exponents appearing in (3.5), (3.6) and (3.7) are linked to the Hausdorff dimension of (X, g). For $\alpha \leq 2/(n+1)$, the Hausdorff dimension of (X, g) agrees with its topological dimension and is equal to n + 1. For $\alpha > 2/(n + 1)$, the Hausdorff dimension of (X, g) is given by $d := 2n/(2 - \alpha)$. Hence, up to the logarithmic factor in the critical case, we always have a behaviour like $\lambda^{d_H/2}$ if d_H denotes the Hausdorff dimension of (X, g). This is not surprising in view of the modified Weyl-Berry conjecture [35] (which is however false in general [36]) and that in our case, the Hausdorff dimension and the Minkowski dimension agree.

3.6 Discreteness of the spectrum

As we have mentioned previously, the discreteness of the spectrum of the Laplace-Beltrami operator Δ_g for g given in (3.1) with $0 < \alpha < 2$ was shown in [30, Proposition 29]. For critical and supercritical α , that is, $2 > \alpha \ge 2/(n+1)$, the Riemannian volume vol(X, g) is infinite. It might at first sight be a bit surprising that we nevertheless have discrete spectrum in this case. Morally speaking, I would say that the discrete spectrum for $\alpha < 2$ seems more linked to the fact that for $\alpha < 2$, every point in X has a finite distance to the boundary measured with respect to the metric g. By contrast, for $\alpha \ge 2$, every point in the interior has an infinite distance to the boundary. This infinite distance is helpful in the construction of Weyl sequences that can show the existence of essential spectrum. Note that $\alpha = 2$ corresponds to the case of hyperbolic geometry.

3.7 Quasi-isometric metric

Let us use the notation $X_{\tilde{\varepsilon}} = [0, \tilde{\varepsilon}]_x \times M_y$ for any $\tilde{\varepsilon} > 0$. Here the indices x and y indicate that we work with coordinates $(x, y) \in [0, \tilde{\varepsilon}] \times M$ as in (3.4), and we identify this set with the corresponding part of X. Since the family of metrics $g_1(x)$ on M is continuous in x (see (3.4)), we can find for any $\delta > 0$ an $\varepsilon > 0$ (which we fix in the following) and a smooth metric g_{ε} on X such that

$$g_{\varepsilon} = dx^2 + x^{-\beta}g_1(x=0) \quad \text{on } X_{3\varepsilon}$$

and

$$(1+\delta)^{-1}g \leqslant g_{\varepsilon} \leqslant (1+\delta)g \quad \text{on } X.$$
(3.8)

From (3.8), we obtain

$$(1+c_{\delta})^{-1}\Delta_g \leqslant \Delta_{g_{\varepsilon}} \leqslant (1+c_{\delta})\Delta_g$$

in the sense of quadratic forms for some $c_{\delta} > 0$ with $c_{\delta} \to 0$ as $\delta \to 0$. In particular, the leading order Weyl asymptotics for Δ_g and $\Delta_{g_{\varepsilon}}$ will only differ by a factor of at most $(1+c_{\delta})$. Since we can choose δ arbitrarily small, it suffices to prove the desired asymptotics for $\Delta_{g_{\varepsilon}}$. For simplicity of notation, we assume in the following without loss of generality that $g = g_{\varepsilon}$.

3.8 Change of function and change of measure

The change of functions consists in replacing any function $\tilde{f} \in L^2(X, dv_g)$ by a function f which is $x^{-\beta n/4}\tilde{f}$ on $X_{2\varepsilon}$ and is equal to \tilde{f} on $X \setminus X_{3\varepsilon}$, and f is some interpolation of the two behaviours on $X_{3\varepsilon} \setminus X_{2\varepsilon}$. We also make a change of measure such that this transformation becomes unitary. In particular, on $X_{2\varepsilon}$, the Riemannian volume measure $dv_g(x,y) = x^{-n\beta/2} dx dv_G(y)$ is replaced by $dx dv_G(y)$, where $G = g_1(x = 0)$ is a metric on M. In particular, for any $\tilde{f}, \tilde{h} \in L^2(X_{2\varepsilon}, dv_g)$, we have

$$\int_{X_{2\varepsilon}} \tilde{f}\tilde{h} \, dv_g(x,y) = \int_{X_{2\varepsilon}} fg \, dx dv_G(y). \tag{3.9}$$

On $X_{2\varepsilon}$ in local coordinates, the operator $\Delta_{g_{\varepsilon}}$ (which we from now on refer to as Δ_{g} since we assume without loss of generality $g = g_{\varepsilon}$) is of the form

$$-\partial_x^2 + \frac{C_\beta}{x^2} + x^\beta \Delta_M \quad \text{with} \quad C_\beta = \frac{\beta n}{4} \left(1 + \frac{\beta n}{4} \right). \tag{3.10}$$

Recall that β was defined in (3.4). Δ_M denotes the Laplace-Beltrami operator on M with respect to the metric $G = g_1(0)$. Notice that the operator in (3.10) allows to separate the variables x and y. This will be used in Section 3.9 below.

3.8.1 Splitting of the Riemannian manifold

The next step in the proof consists in splitting the Riemannian manifold X into a part close to the boundary X_{ε} and the rest $X \setminus X_{\varepsilon}$.

The number $N(\lambda) := N_X(\lambda)$ of negative eigenvalues of the Laplace-Beltrami operator Δ_g (which is a nonnegative operator for us) on X less than $\lambda > 0$ can then be estimated using Dirichlet-Neumann bracketing by the sum of the corresponding numbers of eigenvalues of the Laplace-Beltrami operator on the two parts:

$$N^{D}_{X_{\varepsilon}}(\lambda) + N^{D}_{X \setminus X_{\varepsilon}}(\lambda) \leqslant N(\lambda) \leqslant N^{N}_{X_{\varepsilon}}(\lambda) + N^{N}_{X \setminus X_{\varepsilon}}(\lambda)$$
(3.11)

Here the D or N stand for Dirichlet or Neumann boundary conditions at the boundary of X_{ε} in the interior of X, namely at $\{\varepsilon\} \times M$. Strictly speaking, we will first perform the change of function and change of measure from Section 3.8 and then put the Dirichlet or Neumann boundary conditions.

We then examine the asymptotics of the terms in (3.11) separately. The parts corresponding to the interior of X, namely $N_{X\setminus X_{\varepsilon}}^{D/N}(\lambda)$ can be treated using classical results since the metric g is non-singular there. We have

$$N_{X\setminus X_{\varepsilon}}^{D/N}(\lambda) = C_{n+1} \operatorname{vol}(X \setminus X_{\varepsilon}, g) \lambda^{(n+1)/2} + o\left(\lambda^{(n+1)/2}\right) \quad \text{as } \lambda \to \infty.$$
(3.12)

The main contribution is a precise understanding of the asymptotics for the term near the boundary $N_{X_{\varepsilon}}^{D/N}(\lambda)$.

3.9 Separation of variables and the one-dimensional operator P_{μ}

Note that when considering Δ_g on X_{ε} in local coordinates (both with Dirichlet or Neumann boundary conditions at $\{\varepsilon\} \times M$), we can find an orthonormal basis of eigenfunctions of Δ_g with eigenvalue $\eta \ge 0$ that are of the form

$$\Phi(x,y) = \varphi(x)\psi(y), \quad (x,y) \in [0,\varepsilon] \times M \tag{3.13}$$

where φ , ψ are L^2 -normalised, ψ is an eigenfunction of Δ_M with eigenvalue $\mu \ge 0$, and φ is an eigenfunction of

$$P_{\mu} := -\partial_x^2 + \frac{C_{\beta}}{x^2} + \mu x^{\beta} \tag{3.14}$$

on $[0, \varepsilon]$ with eigenvalue η . Here we used (3.10).

Using the notation $N_{\mu,[0,\varepsilon]}^{D/N}(\lambda)$ for the number of eigenvalues of P_{μ} less than λ with Dirichlet or Neumann boundary conditions at $x = \varepsilon$ and always Dirichlet boundary conditions at x = 0 (since we have Dirichlet boundary conditions for Δ_g at ∂X), we obtain

$$N_{X_{\varepsilon}}^{D/N}(\lambda) = \sum_{j=0}^{\infty} N_{\mu_j,[0,\varepsilon]}^{D/N}(\lambda).$$
(3.15)

Here we denote by $(\mu_j)_{j\in\mathbb{N}_0}$ the eigenvalues of Δ_M counted with multiplicity.

3.10 Application of the Weyl asymptotics for M

Note that for any $\mu \ge 0$, $P_{\mu} \ge -\partial_x^2$, and therefore, from the explicit theory for the one-dimensional Laplacian on an interval,

$$N^{D/N}_{\mu,[0,\varepsilon]}(\lambda) \leqslant 1 + \frac{1}{\pi} \varepsilon \sqrt{\lambda} = o(\lambda^{(n+1)/2}) \text{ as } \lambda \to \infty.$$
(3.16)

Thus, any of the terms in (3.15) individually is of subleading order.

At the same time, from the Weyl asymptotics on M, we know that

$$\mu_j = c_M j^{2/n} (1 + o(1)) \text{ as } j \to \infty,$$

where $c_M = (|B_1^n|(2\pi)^{-n}v_G(M))^{-2/n}$ and $|B_1^n|$ denotes the volume of the unit ball in dimension *n*. For any $\delta > 0$ we can find $K \in \mathbb{N}$ large enough such that

$$(1+\delta)^{-1}c_M j^{2/n} \leqslant \mu_j \leqslant (1+\delta)c_M j^{2/n}$$
(3.17)

for all $j \ge K$. Replacing μ_j by the estimates in (3.17) only changes the leading order asymptotics for $N_{X_{\varepsilon}}^{D/N}(\lambda)$ by at most a factor that converges to 1 as $\delta \to 0$. Since we can take δ arbitrarily small, we can without loss of generality assume that $\mu_j = c_M j^{2/n}$ for all $j \ge K$.

Combining these two aspects, we see that for the leading order asymptotics of

 $N_{X_{\varepsilon}}^{D/N}(\lambda)$, it suffices to find the leading-order asymptotics of

$$\sum_{j=K}^{\infty} N_{c_M j^{2/n}, [0,\varepsilon]}^{D/N}(\lambda).$$
(3.18)

3.11 Unitary equivalence of P_{μ} and scaling

For any $\mu > 0$ and a > 0, the operator P_{μ} on [0, a] is unitarily equivalent to $\mu^{\frac{2}{2+\beta}}P_1$ on $[0, \mu^{1/(2+\beta)}a]$, see [9, Proposition 5]. This can be seen through the unitary transformation

$$(Uf)(x) \mapsto \mu^{\frac{1}{2(2+\beta)}} f\left(\mu^{\frac{1}{2+\beta}}x\right).$$
(3.19)

It follows from this and (3.18) that we need to understand the leading-order asymptotics of

$$\sum_{j=K}^{\infty} N_{c_M j^{2/n}, [0,\varepsilon]}^{D/N}(\lambda) = \sum_{j=K}^{\infty} N_{1, [0,\varepsilon(c_M j^{2/n})^{1/(2+\beta)}]}^{D/N} \left((c_M j^{2/n})^{-2/(2+\beta)} \lambda \right).$$
(3.20)

The scaling we use here is one of the key steps in the proof. In particular, it is essential for the proof in the critical and supercritical case to obtain asymptotics instead of leading-order bounds with non-matching constants.

In the subcritical case, we will show that (3.20) is an arbitrarily small factor times the leading order term. In the critical and supercritical case, it will be the main order term and it does not depend on ε . By replacing ε by $\varepsilon(c_M)^{-1/(2+\beta)}$ and replacing λ by $(c_M)^{-2/(2+\beta)}\lambda$, on can check in those cases that the sum (3.20) is proportional to $v_G(M)$, as in (3.6) and (3.7), and one can also keep track of the constants. Therefore, it suffices to understand the asymptotics of

$$\sum_{j=K}^{\infty} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right), \qquad (3.21)$$

where we use the notation $d := n(1 + \beta/2)$. Note that we have d > n for all $\beta > 0$.

3.12 Removing zero and subleading contributions

Note that there exists some c > 0 (depending on β) such that

$$\frac{C_{\beta}}{x^2} + x^{\beta} \ge c \quad \text{for all } x > 0. \tag{3.22}$$

This expression appears in the definition of P_1 , see (3.14), and shows that $P_1 \ge c$, independently of the interval on which we consider P_1 or which boundary conditions we pick. Hence, there exists $C_0 > 0$ such that for all $j \ge C_0 \lambda^{d/2}$, we have

$$N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) = 0.$$
(3.23)

Thus, by (3.23), we can express (3.21) as

$$\sum_{j=K}^{\infty} N_{j^{n/2},[0,\varepsilon]}^{D/N}(\lambda) = \sum_{j=K}^{C_0 \lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right).$$
(3.24)

Furthermore, for any L > 0 we have by (3.16)

$$\sum_{j=K}^{L\lambda^{n/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \leqslant L\lambda^{n/2} \left(1 + \frac{1}{\pi} \varepsilon \sqrt{\lambda} \right) = \frac{1}{\pi} \varepsilon L\lambda^{(n+1)/2} + o(\lambda^{(n+1)/2}) \quad (3.25)$$

as $\lambda \to \infty$.

In the subcritical case, we will first choose L > 0 very large and then we only need to choose $\varepsilon = \varepsilon(L) > 0$ small enough (namely we want εL to be small). Then we get a term of leading order, but with arbitrarily small prefactor. In the critical and supercritical case, this term will be of subleading order, and we may choose $L = L(\varepsilon) > 0$ large depending on $\varepsilon > 0$.

Thus, for L > 0, we will need to understand the asymptotics of

$$\sum_{j=L\lambda^{n/2}}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) , \qquad (3.26)$$

which is a sum of counting functions for the operator P_1 , which was defined in (3.14), considered on intervals of different sizes. Continuing from this point, we give a separate end of proof in the subcritical, critical and supercritical case.
3.13 End of the proof in the critical case

In the critical case, we have $\beta = 2/n$. For some E > 0 small and L > 0 to be chosen, we split the sum (3.26) as follows:

$$\sum_{j=L\lambda^{n/2}}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right)$$

$$= \sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) + \sum_{j=E\lambda^{d/2}+1}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) =: T+S,$$
(3.27)

where we assume without loss of generality that $E\lambda^{d/2}, C_0\lambda^{d/2} \in \mathbb{N}$.

Let us first estimate T. We use the notation $\omega := j^{-2/d} \lambda$ and note that for $L\lambda^{n/2} \leq j \leq E\lambda^{d/2}$, we have

$$L^{-2/d}\lambda^{\beta/(2+\beta)} \geqslant \omega \geqslant E^{-2/d}.$$
(3.28)

If $L = L(\varepsilon) > 0$ is chosen large enough depending on ε , then we have $\varepsilon \sqrt{\lambda/\omega} > \omega^{1/\beta}$ for all j with $L\lambda^{n/2} \leq j \leq E\lambda^{d/2}$ and thus, by Dirichlet-Neumann bracketing,

$$N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \leqslant N_{1,[0,\omega^{1/\beta}]}^{N} \left(\omega \right) + N_{1,[\omega^{1/\beta},\varepsilon\sqrt{\lambda/\omega}]}^{N} \left(\omega \right) = N_{1,[0,\omega^{1/\beta}]}^{N} \left(\omega \right) , \quad (3.29)$$

where N stands for additional Neumann boundary conditions at $x = \omega^{1/\beta}$. Here we used that $P_1 \ge \omega$ on $[\omega^{1/\beta}, \varepsilon \sqrt{\lambda/\omega}]$ in the last step. Similarly, we get

$$N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \ge N_{1,[0,\omega^{1/\beta}]}^{D} \left(\omega \right).$$
(3.30)

Next, we claim that for any $\gamma > 0$, we can find E > 0 small enough such that for all $\omega := j^{-2/d} \lambda$ with $L\lambda^{n/2} \leq j \leq E\lambda^{d/2}$ and $L = L(\varepsilon) > 0$ chosen as above, we have

$$N_{1,[0,\omega^{1/\beta}]}^{D/N}(\omega) \in \left[(1+\gamma)^{-1} A \omega^{1/2+1/\beta}, (1+\gamma) A \omega^{1/2+1/\beta} \right],$$
(3.31)

where

$$A := \frac{1}{\pi} \int_0^1 \sqrt{1 - z^\beta} \, dz.$$
 (3.32)

The proof of (3.31) follows along the lines of the proof of [9, Proposition 4]. Using Dirichlet-Neumann bracketing, we can split the interval $[0, \omega^{1/\beta}]$ into smaller intervals. On each of those smaller intervals, we estimate the potential $C_{\beta}x^{-2} + x^{\beta}$ in the definition of P_1 by a constant from above and below. Then we use estimates for eigenvalues for the Dirichlet and Neumann Laplacian on intervals. We omit further details here.

By (3.31), we get

$$T \in \left[(1+\gamma)^{-1} A \sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} \left(j^{-2/d} \lambda \right)^{1/2+1/\beta}, (1+\gamma) A \sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} \left(j^{-2/d} \lambda \right)^{1/2+1/\beta} \right].$$
(3.33)

Since $\beta = 2/n$, we have $(1/2 + 1/\beta) = (n+1)/2$ and $2/d(1/2 + 1/\beta) = 1$. Thus,

$$\sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} \left(j^{-2/d}\lambda\right)^{1/2+1/\beta} = \lambda^{(n+1)/2} \sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} \frac{1}{j}.$$
(3.34)

Recall that

$$\sum_{j=1}^{E\lambda^{d/2}} \frac{1}{j} = \log(E\lambda^{d/2}) + \mathcal{O}(1) = \frac{d}{2}\log(\lambda) + \mathcal{O}(1) \text{ as } \lambda \to \infty$$
(3.35)

and similarly,

$$\sum_{j=1}^{L\lambda^{n/2}} \frac{1}{j} = \frac{n}{2} \log(\lambda) + \mathcal{O}(1) \quad \text{as } \lambda \to \infty.$$
(3.36)

Using d = n + 1, so d/2 - n/2 = 1/2, it follows from (3.35) and (3.36) that

$$\sum_{j=L\lambda^{n/2}}^{E\lambda^{d/2}} \frac{1}{j} = \frac{1}{2}\log(\lambda) + \mathcal{O}(1) \text{ as } \lambda \to \infty.$$
(3.37)

Combining (3.33), (3.34) and (3.37), we get

$$T \in \left[(1+\gamma)^{-1}, (1+\gamma) \right] \frac{A}{2} \lambda^{(n+1)/2} \log(\lambda) + \mathcal{O}\left(\lambda^{(n+1)/2}\right) \quad \text{as } \lambda \to \infty.$$
(3.38)

Recall that $\gamma > 0$ can be chosen arbitrarily small, and this will give us the leading-order asymptotics for (3.26) in the critical case.

In order to complete the proof of (3.6) in the critical case, it remains to show that S is of subleading order. To this end, note that if $j \ge E\lambda^{d/2}$ and $\lambda > 0$ is large enough (depending on ε and E), then we always have $E^{-2/(d\beta)} \le \varepsilon j^{1/d}$ and by Dirichlet-Neumann bracketing, we have

$$N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \leq N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(E^{-2/d} \right)$$
$$\leq N_{1,[0,E^{-2/(d\beta)}]}^{N} \left(E^{-2/d} \right) + N_{1,[E^{-2/(d\beta)},\varepsilon j^{1/d}]}^{N} \left(E^{-2/d} \right)$$
$$= N_{1,[0,E^{-2/(d\beta)}]}^{N} \left(E^{-2/d} \right) =: C(E),$$

where we used that $P_1 \ge E^{-2/d}$ on $[E^{-2/(d\beta)}, \varepsilon j^{1/d}]$. It follows using d = n + 1 that

$$S = \sum_{j=E\lambda^{d/2}+1}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \leqslant C_0 \lambda^{d/2} C(E) = C_0 C(E) \lambda^{(n+1)2}, \qquad (3.39)$$

and thus,

$$S = o\left(\lambda^{(n+1)/2}\log(\lambda)\right) \quad \text{as } \lambda \to \infty.$$
(3.40)

This completes the proof of (3.6) in the critical case.

3.14 End of the proof in the supercritical case

Recall that by the unitary equivalence of P_{μ} on [0, a] and $\mu^{\frac{2}{2+\beta}}P_1$ on $[0, \mu^{1/(2+\beta)}a]$ via (3.19),

$$N_{1,[0,\varepsilon j^{1/d}]}^{D/N}\left(j^{-2/d}\lambda\right) = N_{j^{2/n},[0,\varepsilon]}^{D/N}\left(\lambda\right),\tag{3.41}$$

which is monotone decreasing in j. Using this monotonicity and that each of those terms is of subleading order, see (3.16), we can interpret our sum as a Riemann sum and obtain

$$\lim_{\lambda \to \infty} \lambda^{-d/2} \sum_{j=L\lambda^{n/2}}^{C_0 \lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right)$$

=
$$\lim_{\lambda \to \infty} \lambda^{-d/2} \sum_{j=L\lambda^{n/2}}^{C_0 \lambda^{d/2}} N_{1,[0,\varepsilon \left(j\lambda^{-d/2} \right)^{1/d} \sqrt{\lambda}]}^{D/N} \left(\left(j\lambda^{-d/2} \right)^{-2/d} \right)$$

=
$$\lim_{\lambda \to \infty} \int_{L\lambda^{n/2-d/2}}^{C_0} N_{1,[0,\varepsilon s^{1/d} \sqrt{\lambda}]}^{D/N} \left(s^{-2/d} \right) ds.$$
 (3.42)

In a similar way to the proof of (3.31), one can also show that there exists a (non-optimal) constant C depending only on C_0 such that for any $\lambda > 0$, $\varepsilon > 0$ and for any $\omega \ge C_0^{-2/d}$, we have

$$N_{1,[0,\varepsilon\sqrt{\lambda/\omega}]}^{D/N}(\omega) \leqslant C\omega^{1/2+1/\beta}.$$
(3.43)

Applying (3.43) with $\omega = s^{-2/d}$ for $s \leq C_0$ and using $\beta > 2/n$, we obtain

$$N_{1,[0,\varepsilon s^{1/d}\sqrt{\lambda}]}^{D/N}\left(s^{-2/d}\right) \leqslant Cs^{-2/d(1/2+1/\beta)} = Cs^{-2/(n\beta)} \in L^{1}([0,C_{0}]).$$
(3.44)

Thus, the limit in (3.42) is finite and we have found a dominating function, which we will later use to apply the dominated convergence theorem in (3.42).

We claim that

$$\lim_{\lambda \to \infty} N_{1,[0,\varepsilon s^{1/d}\sqrt{\lambda}]}^{D/N} \left(s^{-2/d}\right) = N_{1,[0,\infty)} \left(s^{-2/d}\right) \text{ for almost every } s > 0.$$
(3.45)

Note that we cannot expect pointwise convergence since $N_{1,[0,\varepsilon s^{1/d}\sqrt{\lambda}]}^{D/N}$ $(s^{-2/d}) \in \mathbb{N}_0$. For (3.45), it suffices to show that for every $k \in \mathbb{N}$, the k-th eigenvalue of P_1 on [0, a] converges to the k-th eigenvalue of P_1 on $[0, \infty)$ as $a \to \infty$. In the case of Dirichlet boundary conditions, this can be seen using the min-max principle. In the case of Neumann boundary conditions, one can show an Agmon-type estimate to get the convergence. We will omit the details here.

Using (3.44), (3.45) and the dominated convergence theorem, we obtain

$$\lim_{\lambda \to \infty} \int_{L\lambda^{n/2-d/2}}^{C_0} N_{1,[0,\varepsilon s^{1/d}\sqrt{\lambda}]}^{D/N} \left(s^{-2/d} \right) = \int_0^{C_0} N_{1,[0,\infty)} \left(s^{-2/d} \right) \in (0,\infty), \tag{3.46}$$

and by (3.42), this completes the proof of (3.7).

3.15 End of the proof in the subcritical case

Using (3.44) and multiplying everything in (3.42) by $\lambda^{d/2}$, we get

$$\sum_{j=L\lambda^{n/2}}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d}\lambda\right) \leqslant \lambda^{d/2} \int_{L\lambda^{n/2-d/2}}^{C_0} Cs^{-2/(n\beta)} \, ds \, + o(\lambda^{d/2}) \quad \text{as } \lambda \to \infty.$$

$$(3.47)$$

Note that

$$\lambda^{d/2} \int_{L\lambda^{n/2-d/2}}^{C_0} s^{-2/(n\beta)} \, ds \leqslant C\lambda^{d/2} \left(\left(L\lambda^{n/2-d/2}\right)^{1-2/(n\beta)} \leqslant CL^{1-2/(n\beta)} \lambda^{(n+1)/2} \right)$$
(3.48)

and since $\beta < 2/n$, the factor $L^{1-2/(n\beta)}$ can be arbitrarily small if we choose L > 0 large enough. Since we have d < n + 1, we find that

$$\sum_{j=L\lambda^{n/2}}^{C_0\lambda^{d/2}} N_{1,[0,\varepsilon j^{1/d}]}^{D/N} \left(j^{-2/d} \lambda \right) \leqslant CL^{1-2/(n\beta)} \lambda^{(n+1)/2} + o(\lambda^{(n+1)/2}) \quad \text{as } \lambda \to \infty.$$
(3.49)

Combining this with (3.25) and (3.12), we obtain (3.5).

Chapter 4

Isoperimetric inequalities for inner parallel curves

4.1 Overview of [16]

Together with Ayman Kachmar and Vladimir Lotoreichik [16], we studied an isoperimetric problem that was motivated by a previous paper of theirs [32]. We proved the following statement: Let $\Omega \subset \mathbb{R}^2$ be a smooth simply connected domain. Let $B \subset \mathbb{R}^2$ be the ball centred at the origin with $|\partial B| = |\partial \Omega|$. Denote for all $t \ge 0$

$$S_t(\Omega) := \{ x \in \overline{\Omega} \mid \text{ dist } (x, \partial \Omega) = t \}$$

and similarly for $S_t(B)$. Define the centroid of $S_t(\Omega)$ by

$$c(t) := \left| S_t(\Omega) \right|^{-1} \int_{S_t(\Omega)} x d\mathcal{H}^1(x) \,. \tag{4.1}$$

Then

$$\int_{S_t(\Omega)} |x - c(t)|^2 d\mathcal{H}^1(x) \leqslant \int_{S_t(B)} |x|^2 d\mathcal{H}^1(x) \text{ for almost every } t \ge 0, \qquad (4.2)$$

where \mathcal{H}^1 denotes the one-dimensional Hausdorff measure. Note that $\int_{S_t(B)} |x|^2 d\mathcal{H}^1(x)$ is *not* the variance of x for $S_t(B)$ since we do not divide by $|S_t(B)|$.

(4.2) was the missing ingredient to extend an isoperimetric inequality for the lowest eigenvalue of the magnetic Robin Laplacian with negative boundary parameter for convex domains in [32] to all smooth simply connected domains in \mathbb{R}^2 with finite perimeter. The case t = 0 of (4.2) is a classical result due to Hurwitz [31] relying on Fourier techniques. In order to extend this result to t > 0, we used in [16] an explicit construction of a closed curve Σ_t with $S_t(\Omega) \subset \Sigma_t$ and $|\Sigma_t| \leq |S_t(B)|$ for almost every t > 0.

We also studied a related question: Fix p > 0. Let $\Gamma \subset \mathbb{R}^2$ be a closed Lipschitz curve with the origin as its centroid, and let $B \subset \mathbb{R}^2$ be a ball centred at the origin with $|\Gamma| = |\partial B|$. Then do we always have

$$\int_{\Gamma} |x|^p d\mathcal{H}^1(x) \leqslant \int_{\partial B} |x|^p d\mathcal{H}^1(x) ?$$

We could give a partial answer to this question, namely yes for all $p \leq 2$, and no for all p > 3. Using a Fuglede-type argument, we could show that if 2 in a $suitable sense, for nearly circular domains <math>\Omega$ locally near a ball, the answer is yes. The case 2 in its full generality remains an interesting open problem.

In the following, we focus on (4.2). We first consider the example of a dumbbell in Section 4.3. In Section 4.5 and Section 4.6, we explain the strategy of how to define the closed curve Σ_t with $S_t(\Omega) \subset \Sigma_t$ and $|\Sigma_t| \leq |S_t(B)|$ for almost every t > 0. In Section 4.4, we explain why this is enough to conclude (4.2).

4.2 Context for [16]

Using Fourier analysis, Hurwitz [31, pp. 396-397] showed that for any closed curve Γ of fixed length with the origin as its centroid, one has

$$\int_{\Gamma} |x|^2 \, d\mathcal{H}^1(x) \leqslant \frac{|\Gamma|^3}{(2\pi)^2} \tag{4.3}$$

with equality if and only if the curve is a circle around the origin. Recently, generalisations of (4.3) to higher dimensions and corresponding quantitative versions were shown in [33].

For inner parallel curves, Hartman [28, Corollary 6.1] showed that $|S_t(\Omega)| \leq |S_t(B)|$ for almost every t > 0.

Combining the two results [31, 28], one can show that if $S_t(\Omega)$ is a closed curve, then (4.2) holds, see for instance the proof of [32, Proposition 4.4] or our explanation in Section 4.4 below. For instance, $S_t(\Omega)$ is a closed curve for all t > 0 if Ω is convex, see [32, Proposition 4.4]. Our main contribution in [16, Theorem 1.1] is to generalise this result to general smooth, bounded, simply connected domains (the smoothness assumption is technical and can probably be relaxed).

Our work [16] was motivated by [32]. Assuming (4.2), the authors of [32] proved an isoperimetric inequality for the first eigenvalue of the magnetic Robin Laplacian [32, Theorem 4.8]. They left as an open question for which domains Ω one has (4.2). Using (4.2) for all smooth, bounded simply connected domains with finite perimeter, we could straightforwardly extend this isoperimetric inequality for the first eigenvalue of the magnetic Robin Laplacian [16, Theorem 1.5].

There are many results in the literature on weighted isoperimetric inequalities. In [3], the authors consider weighted integrals over the boundary of the domain for different domains with the same Lebesgue measure. In [1] a similar setting is considered, but the Lebesgue measure is replaced by some weighted Lebesgue measure with a power-type weight. Another result on weighted isoperimetric inequalities we would like to mention is [6].

4.3 The dumbbell example

Let us consider the example where Ω is a dumbbell, such as in Figure 4.1. Ayman Kachmar mentioned this example to me when we started working on the project. By symmetry of the dumbbell, we have c(t) = 0 for all $t \ge 0$. Let us without loss of generality assume that $|\partial \Omega| = 2\pi$, so we need to compare Ω with a ball of radius 1 centred at the origin.



Figure 4.1: This picture shows an example where Ω is a dumbbell-shaped region. The set $S_t(\Omega) =: S_t$ for some t > 0 is drawn in blue. Note that it consists of two connected components.

For t in a certain range, $S_t(\Omega)$ consists of two connected components that are far away from each other. We know that $|S_t(\Omega)| \leq |S_t(B)|$ for almost every t > 0 by [28, Corollary 6.1]. However, only using this information is not enough to conclude since the two parts of $S_t(\Omega)$ are very far away from the origin, so the $|x|^2$ for $x \in S_t(\Omega)$ in (4.2) can be larger than 1. Note that the two connected components of $S_t(\Omega)$ cannot be arbitrarily far away from the origin since they are a subset of Ω and we have $|\partial \Omega| = 2\pi$, so we always have $|x| \leq \pi$ for $x \in S_t(\Omega)$ (using the symmetry of the dumbbell around the origin). Also note that in the case of a dumbbell with a very long middle axis, $|S_t(\Omega)|$ is much smaller than $|S_t(B)|$, see Figure 4.1.

4.4 General proof strategy

In this section, we explain why for the proof of (4.2), it is sufficient to construct a closed curve Σ_t with $S_t(\Omega) \subset \Sigma_t$ and of length $|\Sigma_t| \leq |S_t(B)|$ for almost every t > 0. This result can be deduced from [31, pp. 397-397]. The following explanation can also be found in [16, Section 4].

Fix t > 0 and suppose we have already constructed such a curve Σ_t of length $L := |\Sigma_t| \leq |S_t(B)|$. Let $\sigma : [0, L] \to \mathbb{R}^2$ be a parametrisation of that curve by arc-length. Assume without loss of generality that the centroid of Σ_t is the origin, that is,

$$\int_0^L \sigma(s) \, ds = 0 \tag{4.4}$$

(otherwise shift the whole coordinate system). Then

$$\int_{S_t(\Omega)} |x - c(t)|^2 d\mathcal{H}^1(x) \leqslant \int_{S_t(\Omega)} |x|^2 d\mathcal{H}^1(x) \leqslant \int_{\Sigma_t} |x|^2 d\mathcal{H}^1(x) = \int_0^L |\sigma(s)|^2 ds.$$
(4.5)

Note that the first step can be shown by a short computation that uses the definition of c(t) as a centroid and the fact that the power is equal to 2. Next, we apply the Poincaré inequality using (4.4), see for example [27, Section 7.7], to get

$$\int_0^L |\sigma(s)|^2 \, ds \leqslant \frac{L^2}{(2\pi)^2} \int_0^L |\sigma'(s)|^2 \, ds = \frac{L^3}{(2\pi)^2} \,, \tag{4.6}$$

where we used the arc-length parametrisation of σ , that is, $|\sigma'(s)| = 1$ for all s, in the last step.

By scaling in (4.2), we may without loss of generality assume that $|\partial \Omega| = |\partial B| = 2\pi$, that is, *B* is a ball of radius 1 around the origin. In particular, we have $|S_t(B)| = 2\pi(1-t)$. Then

$$\int_{S_t(B)} |x|^2 d\mathcal{H}^1(x) = (1-t)^3 2\pi.$$
(4.7)

Thus, if

$$L \leqslant (1-t)2\pi = |S_t(B)|, \qquad (4.8)$$

then by (4.5), (4.6), (4.7) and (4.8), we obtain (4.2).

This shows that in order to complete the proof, it suffices to find a closed curve Σ_t with $S_t(\Omega) \subset \Sigma_t$ and of length $|\Sigma_t| \leq |S_t(B)|$.

4.5 A first naive try for the construction

Since Ω is assumed to be simply connected and we assume without loss of generality that $|\partial \Omega| = 2\pi$, we can parametrise its boundary by a curve $\gamma : [0, 2\pi] \to \mathbb{R}^2$ that is parametrised by arc-length. Moreover, we may assume that γ has positive orientation. We denote by $n : [0, 2\pi] \to \mathbb{R}^2$ the corresponding inward pointing normal vector

$$n(s) = (-\gamma_2'(s), \gamma_1'(s))^T = J\gamma'(s), \text{ where } J = \begin{pmatrix} 0 & -1\\ 1 & 0 \end{pmatrix}$$
(4.9)

and by $\kappa : [0, 2\pi] \to \mathbb{R}$ the curvature defined by

$$\gamma''(s) = \kappa(s)n(s). \tag{4.10}$$

Since γ is a positively oriented non-self-intersecting closed curve, we have by [2, Satz 2.2.10]

$$\int_0^{2\pi} \kappa(s) \, ds = 2\pi. \tag{4.11}$$

Using the above notation, we have [16, Lemma 2.3]

$$S_t(\Omega) \subset \{\gamma(s) + tn(s) \mid x \in [0, 2\pi]\}.$$
 (4.12)

In view of (4.12), a natural guess for a closed curve containing $S_t(\Omega)$ and with length less than or equal to $|S_t(B)|$ is given by the curve A_t parametrised by

$$\alpha : [0, 2\pi] \to \mathbb{R}^2, \quad \alpha(s) = \gamma(s) + tn(s). \tag{4.13}$$

The curve A_t for the dumbbell-shaped region that was shown in Figure 4.1 is illustrated in Figure 4.2.

The following results and explanations for A_t can be found in [16, Lemma 2.3].



Figure 4.2: This image shows an example where Ω is a dumbbell-shaped region. The curve A_t for some t > 0, which is parametrised by (4.13) is drawn in cyan. Note that we have $S_t(\Omega) \subset A_t(\Omega)$, compare with Figure 4.1. In this example we have $\kappa(s) \leq 1/t$ for all $s \in [0, 2\pi]$ and thus, by (4.16), we have $|A_t| = |S_t(B)|$.

We have

$$\alpha'(s) = \gamma'(s) + tn'(s) = \gamma'(s) + t\frac{\mathrm{d}}{\mathrm{d}s} \left(J\gamma'(s)\right) = \gamma'(s) + tJ\gamma''(s)$$

$$= \gamma'(s) + tJ\kappa(s)n(s) = \gamma'(s) + tJ\kappa(s)J\gamma'(s) = (1 - t\kappa(s))\gamma'(s)$$
(4.14)

Here we used (4.9) in the second step, (4.10) in the fourth step, and $J^2 = -Id$ in the last step. Thus, the length of the curve A_t is

$$|A_t| = \int_0^{2\pi} |\alpha'(s)| \, ds = \int_0^{2\pi} |1 - t\kappa(s)| \, ds \,, \tag{4.15}$$

where we used (4.14) and that γ is parametrised by arc-length.

Note that using (4.11), we have

$$\int_{0}^{2\pi} (1 - t\kappa(s)) \, ds = 2\pi(1 - t) \,, \tag{4.16}$$

so if $1 - t\kappa(s) > 0$ for all $s \in [0, 2\pi]$, then $|A_t| = 2\pi(1 - t) = |S_t(B)|$, which is the desired length, see (4.8). Thus, for t small enough, the construction of the curve A_t and the considerations from Section 4.4 yield (4.2).

By contrast, for t such that $|\{s \in [0, 2\pi] \mid 1 - t\kappa(s) < 0\}| > 0$, we see from (4.15) and (4.16) that $|A_t| > 2\pi(1-t) = |S_t(B)|$, which is not enough to conclude. Such an example is shown in Figure 4.3.



Figure 4.3: These images show the same example of a dumbbell Ω as in Figure 4.1 and Figure 4.2. Here we consider a larger value of t and these images show the corresponding inner parallel curve $S_t(\Omega) =: S_t$ on the left and the curve A_t on the right. In this example we have $\kappa(s) > 1/t$ for some $s \in [0, 2\pi]$ and thus, by (4.16) and (4.15), we deduce $|A_t| > |S_t(B)|$.

4.6 General ideas on the modification of the closed curve

In the case where $|\{s \in [0, 2\pi] \mid 1 - t\kappa(s) < 0\}| > 0$, that is, where $\kappa(s) > 1/t$ on a set of positive measure, a natural idea is to try to modify the curve A_t in such a way that its length decreases to at most $2\pi(1-t) = |S_t(B)|$ while still covering $S_t(\Omega)$.

Recall that the circle of radius $1/\kappa(s)$ that is tangent to the curve at $\gamma(s)$ is the circle that approximates the curve best at this point, namely up to second order in the Taylor expansion, see for example [16, Lemma 2.2]. If $\kappa(s) \leq \kappa_0$ for all $s \in [0, 2\pi]$, then for any $s \in [0, 2\pi]$ the ball of radius $1/\kappa_0$ that is tangent to the curve at $\gamma(s)$ is contained in Ω . On the other hand, if for some $s \in [0, 2\pi]$, we have $\kappa(s) > 1/t$, then we do not have $\gamma(s) + tn(s) \in S_t(\Omega)$ in general.

Thus, we would like to discard all points from the construction of our curve A_t , where $\kappa(s) > 1/t$. There are two restrictions we should keep in mind here: First, the resulting curve Σ_t should be a closed continuous curve, so in particular, we cannot allow for any jumps. Second, we need to make sure that the new curve Σ_t has a length of at most $2\pi(1-t) = |S_t(B)|$. For the second point, note that if $|\{s \in [0, 2\pi] \mid 1 - t\kappa(s) < 0\}| > 0$, then

$$\int_{\{s \in [0,2\pi] \mid \kappa(s) \le 1/t\}} |1 - t\kappa(s)| \, ds > 2\pi(1 - t) \tag{4.17}$$

by (4.16). Thus, just removing those points $\alpha(s)$ in the curve for which $\kappa(s) > 1/t$ is *not* enough.

4.7 Construction of the closed curve Σ_t

In order to deal with the two restrictions we explained in Section 4.6, we proceed as follows.

We will first remove all points $\alpha(s)$ from our curve that satisfy $\alpha(s) \notin S_t(\Omega)$. For almost every t > 0, we will obtain a finite number of curve segments, see [16, Proposition 2.1] and the references therein. More precisely, there is $m \in \mathbb{N}$ and there are $0 \leq a_1 < b_1 < a_2 < b_2 < \ldots < a_m < b_m \leq 2\pi$ such that

$$S_t(\Omega) = \{\gamma(s) + tn(s) \mid s \in [a_k, b_k] \text{ for some } 1 \leq k \leq m\}.$$

$$(4.18)$$

The result from [46, Theorem 4.4.1] we use here assumes smoothness of the curve, which could probably be relaxed.

Then, we will add some straight line segments to connect those curve segments $\gamma(s)+tn(s), s \in [a_k, b_k]$ and obtain a closed curve Σ_t . More precisely, we add straight lines connecting $\gamma(b_k) + tn(b_k)$ with $\gamma(a_{k+1}) + tn(a_{k+1})$, where we use the notation $a_{m+1} := a_1$. The details on the construction of Σ_t can be found in the proof of [16, Theorem 3.1]. An illustration of Σ_t is shown in Figure 4.4.



Figure 4.4: This picture shows the same setting as in Figure 4.3. The closed curve Σ_t is drawn in red. Note that the straight line in the middle is doubly covered by Σ_t . Four straight line segments from the construction of Σ_t just consist of a single point.

Finally, we need to show that $|\Sigma_t| \leq |S_t(B)|$, see [16, Theorem 3.1]. To this end, we use a lemma [16, Proposition 2.7] that estimates the length of a curve outside two disks starting and ending tangently to those by the distance of the centres of the two disks and the integral over the curvature. More precisely, [16, Proposition 2.7] states the following:

Suppose that we have two disks with radius t, possibly intersecting each other, with distance R of the two centres, and a smooth curve Γ parametrised by arc-length by $\gamma : [a, b] \to \mathbb{R}^2$ that always stays in the complement of the two disks and starts at $\gamma(a)$ tangently in a positively oriented way to the first disk and ends at $\gamma(b)$ tangently in a positively oriented way to the second disk, then

$$|\Gamma| \ge R + t \int_{a}^{b} \kappa(s) \, ds, \qquad (4.19)$$

where $\kappa(s)$ denotes the curvature of Γ at $\gamma(s)$. (4.19) is very intuitive, but until now, we only have proofs using relatively technical and powerful results from the literature [7, Theorem 3.3] (see the proof of [16, Proposition 2.7]) or tedious elementary proofs (see for example [15, Appendix A.2]).

With $|\Sigma_t| \leq |S_t(B)|$, we can conclude (4.2) by our explanations in Section 4.4, see [16, Section 4].

Chapter 5

Hardy-Sobolev interpolation inequalities

5.1 Overview of [17]

Recall that the classical Hardy inequality states that in any dimension $d \ge 3$, we have for all $u \in \dot{H}^1(\mathbb{R}^d)$

$$\int_{\mathbb{R}^d} |\nabla u|^2 \ge \frac{(d-2)^2}{4} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} dx.$$
(5.1)

Using the ground state representation, see for example [24, Eq. (2.14)], and radial decreasing arrangement, one can show that for $u \neq 0$, the inequality is strict.

One might thus ask for improvements of (5.1) that involve an error term. Note that by translation, from (5.1), one also gets for all $u \in \dot{H}^1(\mathbb{R}^d)$ with $u \neq 0$

$$\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} dx > 0.$$
(5.2)

Previous results estimated the left-hand side in (5.2) by a subcritical L^p -norm [5], see (5.6) below, or a subcritical Sobolev norm [21], while fixing the L^2 -norm of u.

In a joint work with Phan Thành Nam [17] we prove an inequality, see (5.3) below, that involves the left-hand side in (5.2) and is critical in the sense that all terms scale in the same way. More precisely, we show [17, Theorem 1] for d = 3 and

 $\theta = 1/3$ the following Hardy-Sobolev inequality: For all $u \in \dot{H}^1(\mathbb{R}^d)$, we have

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} dx\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} dx\right)^{1-\theta} \ge C ||u||_{L^{2^*}}^2,$$
(5.3)

where C > 0 only depends on d, θ and $2^* = 2d/(d-2)$. We also show that (5.3) does not hold if $d \ge 4$ or if $\theta \ne 1/3$. Furthermore, we show a more general analogue of (5.3) involving L^p -spaces for $p \in [2, d)$ and Lorentz spaces [17, Theorem 2]. The new point of (5.3) with respect to previous estimates, see for example [5] or (5.6) below, is the appearance of the critical exponent 2^* on the right-hand side.

Our proof relies on a Morrey-type inequality going back to Palatucci–Pisante [42], the ground state representation for the Hardy inequality, symmetric decreasing rearrangement and an optimal estimate in the radial case. In fact, we can determine all the optimisers of the corresponding radial problem explicitly [17, Theorem 3(i)], which are given by the family

$$u^{\eta}(x) = \frac{1}{(|x|^{1-\eta}(1+|x|^{2\eta}))^{(d-2)/2}}, \quad \eta \in (0,\infty),$$
(5.4)

up to dilation and multiplication by a constant. It remains an open problem to prove whether there exist minimisers for the full non-radial inequality (5.3).

5.2 Context of [17]

Brezis and Vázquez [5, Theorem 4.1 and Extension 4.3] showed that if

$$d \ge 3, \quad 2 < q < 2^* = \frac{2d}{d-2}, \quad \theta = d\left(\frac{1}{2} - \frac{1}{q}\right),$$
(5.5)

then there exists a constant C > 0 such that for all $u \in H^1(\mathbb{R}^d)$

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} \, dx\right)^{\theta} \|u\|_{L^2}^{2(1-\theta)} \ge C \|u\|_{L^q}^2, \tag{5.6}$$

Note that the q in (5.5) is subcritical with respect to the Sobolev critical exponent 2^* . More recently, a corresponding version involving the corresponding subcritical Sobolev norms was shown in [21, Theorem 1.2].

As we show below in Section 5.3, a corresponding version of (5.6) does *not* hold in the critical case. One of the main points of [17] is that we *do* have a corresponding inequality only involving quantities with critical scaling behaviour if we replace the L^2 -norm in (5.6) by a suitable Morrey-type norm.

A related Morrey-type inequality, which we also use in our proof, is the following inequality for all $d \ge 3$ and $1 - 2/d \le \theta \le 1$ that goes back to Palatucci and Pisante [42, Theorem 1]

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2\right)^{\theta} \left(\sup_{R>0, x\in\mathbb{R}^d} R^{-2} \int_{B_R(x)} |u|^2\right)^{1-\theta} \ge C \|u\|_{L^{2^*}}^2, \quad \text{for all } u\in\dot{H}^1(\mathbb{R}^d).$$
(5.7)

We would also like to refer to [48] for a recent simplified proof of (5.7) involving the use of maximal functions.

Below we explain a radial example from [17, Section 2] in Section 5.3 and a bubbling example from [17, Section 3.1] in Section 5.5 in detail. Both examples combined explain why (5.3) does *not* hold if $d \ge 4$ or if $\theta \ne 1/3$, see Section 5.4 and Section 5.5. Furthermore, the radial example is important for understanding why (5.6) can only work for subcritical norms. The bubbling example exhibits a key feature that makes it challenging to answer the question of existence of optimisers for (5.3) with d = 3 and $\theta = 1/3$, see Section 5.6.

5.3 Radial example

This section is devoted to an explanation of why the inequality estimating the first factor in (5.3) by a subcritical L^p -norm, see (5.6) [5, Theorem 4.1 and Extension 4.3], does *not* hold for critical L^p -norms. The understanding of the counterexample explained below was the starting point our work in [17].

For our counterexample, we will consider radial functions $u \in H^1(\mathbb{R}^d)$ and write them as

$$u(x) = \frac{f(|x|)}{|x|^{(d-2)/2}},$$
(5.8)

which corresponds to the ground state representation for Hardy's inequality, see for example [24, Eq. (2.14)]. With this notation, we have

$$\int_{\mathbb{R}^d} |u|^{2^*} = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f(r)|^{2^*}}{r} dr$$
$$\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} dx = \int_{\mathbb{R}^d} \frac{|\nabla f(x)|^2}{|x|^{d-2}} dx = |\mathbb{S}^{d-1}| \int_0^\infty |f'(r)|^2 r dr,$$
(5.9)

where $|\mathbb{S}^{d-1}|$ is the surface area of the unit sphere in \mathbb{R}^d .

For our counterexample, we choose $f = f_{\varepsilon}$ depending on a parameter $\varepsilon > 0$ (which we will later send to zero) as follows:

$$f_{\varepsilon}(r) = \begin{cases} r^{\varepsilon}, & r \in (0, 1], \\ r^{-\varepsilon}, & r \in [1, \infty). \end{cases}$$
(5.10)

The idea is to consider very flat profiles f_{ε} , which correspond to ε small. If $0 < \varepsilon \leq (d-2)/2$, then the function

$$u_{\varepsilon}(x) = f_{\varepsilon}(|x|)|x|^{-\frac{d-2}{2}} = \begin{cases} |x|^{\varepsilon - \frac{d-2}{2}} & , |x| \leq 1, \\ |x|^{-\varepsilon - \frac{d-2}{2}} & , |x| \geq 1 \end{cases}$$
(5.11)

is radially symmetric decreasing, and hence,

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_{\varepsilon}(x)|^2}{|x-y|^2} \, dx = \int_{\mathbb{R}^d} \frac{|u_{\varepsilon}(x)|^2}{|x|^2} \, dx.$$
(5.12)

With this choice of $f = f_{\varepsilon}$, we have for any p > 0

$$\int_0^\infty r |f_{\varepsilon}'(r)|^2 dr = \varepsilon, \quad \int_0^\infty \frac{|f_{\varepsilon}(r)|^p}{r} dr = \frac{2}{p\varepsilon}.$$
(5.13)

By (5.9) and (5.13), we get for any $\varepsilon > 0$ for the corresponding function u_{ε} defined by (5.11)

$$\int_{\mathbb{R}^d} |\nabla u_{\varepsilon}|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_{\varepsilon}(x)|^2}{|x-y|^2} \, dx = |\mathbb{S}^{d-1}|\varepsilon \tag{5.14}$$

and

$$\int_{\mathbb{R}^d} |u_{\varepsilon}|^{2^*} = |\mathbb{S}^{d-1}| \frac{2}{2^* \varepsilon}.$$
(5.15)

Sending $\varepsilon \to 0$, we find that (5.6) cannot hold with a constant C > 0 independent of u if $d \ge 3$, $q = 2^*$, $\theta = 1$.

5.4 Application of the radial example to (5.3)

Note that in our example defined by (5.11), we also have for $0 \leq \varepsilon \leq (d-2)/2$

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_{\varepsilon}(x)|^2}{|x-y|^2} = \int_{\mathbb{R}^d} \frac{|u_{\varepsilon}(x)|^2}{|x|^2} \, dx = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f_{\varepsilon}(r)|^2}{r} \, dr = |\mathbb{S}^{d-1}| \frac{1}{\varepsilon}, \qquad (5.16)$$

where we used (5.12) in the first step and (5.13) in the last step. Having noted this, it seems natural to ask if (5.3) holds for a suitable range of θ . Plugging in (5.14), (5.15) and (5.16) into (5.3), we get (for a different constant C)

$$\varepsilon^{\theta}\varepsilon^{-(1-\theta)} \geqslant C\varepsilon^{-2/2^*},\tag{5.17}$$

and since we may send $\varepsilon \to 0$, (5.3) can only hold if

$$\theta \leqslant \frac{1}{2} \left(1 - \frac{2}{2^*} \right) = \frac{1}{d}.$$
(5.18)

5.5 Bubbling example

In this section, I would like to explain a bubbling example from [17, Section 3.1]. Combined with $\theta \leq 1/d$, see (5.18), this example shows that (5.3) can only hold for d = 3, $\theta = 1/3$. Furthermore, this bubbling example seems to be the main obstruction in showing existence of an optimising function $u \neq 0$ for (5.3) with the optimal constant C > 0 for d = 3, $\theta = 1/3$, see Section 5.6 below.

We take $N \in \mathbb{N}$ identical bubbles travelling away from each other. Later we will let $N \to \infty$. Fix any $0 \neq \varphi \in C_c^{\infty}(B_1(0)), 0 \neq z \in \mathbb{R}^d$ and define

$$u_N(x) = \sum_{n=1}^{N} \varphi(x + nNz)$$
(5.19)

for every $N \in \mathbb{N}$. Note that for N large enough, the individual bubbles $\{\varphi(\cdot + nNz)\}_{n=1}^{N}$ have disjoint support and are far away from each other.

Thus, for N large enough, we have

$$\int_{\mathbb{R}^d} |u_N|^{2^*} = N \int_{\mathbb{R}^d} |\varphi|^{2^*},$$
(5.20)

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_N(x)|^2}{|x-y|^2} \, dx = \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|\varphi(x)|^2}{|x-y|^2} \, dx + o(1) \quad \text{as} \quad N \to \infty \tag{5.21}$$

and

$$\int_{\mathbb{R}^d} |\nabla u_N|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_N(x)|^2}{|x-y|^2} \, dx = N \int_{\mathbb{R}^d} |\nabla \varphi|^2 + \mathcal{O}(1) \quad \text{as} \quad N \to \infty \,.$$
(5.22)

The estimate in (5.21) indicates that a single term, which corresponds to one bubble, dominates the full integral, while the term in (5.22) is well approximated by the sum

of the corresponding terms for all the bubbles.

Thus, (5.3) can only hold if (for a different constant C), we have for all $N \in \mathbb{N}$ large enough

$$N^{\theta} \geqslant C N^{2/2^*}.\tag{5.23}$$

It follows that we need

$$\theta \ge \frac{2}{2^*} = 1 - \frac{2}{d}.$$
 (5.24)

Combining (5.24) with (5.18) for $d \ge 3$, we find that (5.3) can only possibly hold for d = 3, $\theta = 1/3$.

5.6 Outlook: Existence of optimisers?

One may ask if (5.3) for d = 3 and $\theta = 1/3$ admits nonzero-optimisers: Namely if C > 0 is the optimal constant in

$$\left(\int_{\mathbb{R}^3} |\nabla u|^2 - \frac{1}{4} \sup_{y \in \mathbb{R}^3} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x - y|^2} dx\right)^{1/3} \left(\sup_{y \in \mathbb{R}^3} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x - y|^2} dx \right)^{2/3} \ge C ||u||_{L^6}^2,$$
(5.25)

then is there a function $u \in \dot{H}^1(\mathbb{R}^3)$ with $u \neq 0$ such that we have equality in (5.25)?

We solved the corresponding radial problem in [17, Theorem 3(i)]. We showed that for the inequality

$$\left(\int_{\mathbb{R}^3} |\nabla u|^2 - \frac{1}{4} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|^2} dx\right)^{1/3} \left(\int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x|^2} dx\right)^{2/3} \ge C ||u||_{L^6}^2, \tag{5.26}$$

for radial $u \in \dot{H}^1(\mathbb{R}^3)$ with the optimal constant C > 0 (possibly a different one than in (5.25)), all optimisers (up to scaling and multiplication by constants) are given explicitly by (5.4) above. By contrast, the full non-radial problem (5.25) remains open.

A standard strategy for showing the existence of an optimiser is to take an optimising sequence $(u_n)_{n\in\mathbb{N}} \subset \dot{H}^1(\mathbb{R}^3)$ that is normalised in a suitable sense, for example $\int_{\mathbb{R}^3} |\nabla u|^2 = 1$, and to show that up to a subsequence, translation and scaling, it converges weakly to a candidate for an optimiser $\tilde{u} \neq 0$, and then to show that \tilde{u} is indeed an optimiser. A key difficulty in our case is to show that \tilde{u} is non-zero, or put differently, to exclude the vanishing case in the concentration-compactness method.

Let us elaborate a bit more on this problem in a related setting, namely the

inequality

$$\int_{\mathbb{R}^3} |\nabla u|^2 \left(\sup_{y \in \mathbb{R}^3} \int_{\mathbb{R}^3} \frac{|u(x)|^2}{|x - y|^2} dx \right)^2 \ge C \int_{\mathbb{R}^3} |u|^6 \tag{5.27}$$

for all $u \in \dot{H}^1(\mathbb{R}^3)$, which can be deduced from (5.7). Note that all three terms in (5.27) scale in the same way when replacing u(x) by au(b(x-c)) for a, b > 0 and $c \in \mathbb{R}^3$. Therefore, using a rescaling of u, it is only possible to fix the value of one of these three terms.

Moreover, if $\varphi \in C_c^{\infty}(\mathbb{R}^3)$ is an almost-optimiser, then taking an arbitrarily large number of copies of φ and sending them far away from each other by translation is also an almost-optimiser of (5.27). This bubbling phenomenon is the key challenge to overcome when trying to understand if (5.27) admits non-zero optimisers.

More precisely, for any given $\delta > 0$, there is $\delta \ge \varepsilon > 0$ and $0 \not\equiv \varphi \in C_c^{\infty}(\mathbb{R}^3)$ depending on ε such that

$$\int_{\mathbb{R}^3} |\nabla \varphi|^2 \left(\sup_{y \in \mathbb{R}^3} \int_{\mathbb{R}^3} \frac{|\varphi(x)|^2}{|x - y|^2} dx \right)^2 = (C - \varepsilon) \int_{\mathbb{R}^3} |\varphi|^6.$$
(5.28)

Furthermore, if we take $N \in \mathbb{N}$ copies of φ and send them far away from each other, namely to define u_N as in (5.19), then

$$\lim_{N \to \infty} \frac{\int_{\mathbb{R}^3} |\nabla u_N|^2 \left(\sup_{y \in \mathbb{R}^3} \int_{\mathbb{R}^3} \frac{|u_N(x)|^2}{|x-y|^2} dx \right)^2}{\int_{\mathbb{R}^3} |u_N|^6} = C - \varepsilon.$$
(5.29)

If we define $\tilde{u}_N := N^{-1/2} u_N$ in order to fix the \dot{H}^1 -norm, then $\int_{\mathbb{R}^3} |\nabla \tilde{u}_N|^2 = \int_{\mathbb{R}^3} |\nabla \varphi|^2$ for all $n \in \mathbb{N}$ large enough. At the same time,

$$\int_{\mathbb{R}^3} |\tilde{u}_N|^6 = N^{-2} \int_{\mathbb{R}^3} |\varphi|^6 \to 0 \quad \text{as} \quad N \to 0.$$
 (5.30)

Thus, any translated and rescaled subsequence of $(\tilde{u}_n)_{n\in\mathbb{N}}$ that is weakly convergent in $\dot{H}^1(\mathbb{R}^3)$ has zero as its weak limit.

Using the above example with different φ and a diagonal sequence argument, one can construct different minimising sequences of (5.27) that consist of an arbitrary number $N \in \mathbb{N}$ of bubbles that are smooth and compactly supported. Furthermore, these sequences, which are normalised in the $\dot{H}^1(\mathbb{R}^3)$ -norm, vanish in any $L^p(\mathbb{R}^3)$ norm for p > 2.

The main reason why this bubbling phenomenon occurs is a lack of strict convex-

ity in the problem. By contrast, for a strictly convex problem, such as the Sobolev inequality

$$\left(\int_{\mathbb{R}^3} |\nabla u|^2\right)^3 \ge C \int_{\mathbb{R}^3} |u|^6,\tag{5.31}$$

taking two or more copies of a compactly supported function are always far away from being an optimiser. In (5.31), the ratio of the left-hand side divided by the right-hand side will be at least N^2 if we take N disjoint bubbles. This shows that minimising sequences for (5.31) cannot exhibit a bubbling phenomenon as described above.

In our inequality (5.25), one can expect that the term

$$-\frac{1}{4}\sup_{y\in\mathbb{R}^3}\int_{\mathbb{R}^3}\frac{|u_N(x)|^2}{|x-y|^2}dx$$
(5.32)

in the first factor could prevent the bubbling phenomenon. However, it is less clear than in the case of strict convexity how to make this idea work in a proof. This remains a problem for future works.

Bibliography

- A. Alvino, F. Brock, F. Chiacchio, A. Mercaldo and M. R. Posteraro. "Some isoperimetric inequalities on ℝ^N with respect to weights |x|^α". In: Journal of Mathematical Analysis and Applications 451.1 (2017), pp. 280–318.
- [2] C. Bär. *Elementare Differentialgeometrie*. Walter de Gruyter, 2010.
- [3] M. F. Betta, F. Brock, A. Mercaldo and M. R. Posteraro. "A weighted isoperimetric inequality and applications to symmetrization". In: *Journal of Inequalities and Applications* 1999.3 (1999), p. 670245.
- [4] U. Boscain, D. Prandi and M. Seri. "Spectral analysis and the Aharonov-Bohm effect on certain almost-Riemannian manifolds". In: *Communications* in Partial Differential Equations 41.1 (2016), pp. 32–50.
- [5] H. Brezis and J. L. Vázquez. "Blow-up solutions of some nonlinear elliptic problems". In: *Rev. Mat. Univ. Complut. Madrid* 10.2 (1997), pp. 443–469.
- [6] X. Cabré, X. Ros-Oton and J. Serra. "Sharp isoperimetric inequalities via the ABP method." In: Journal of the European Mathematical Society (EMS Publishing) 18.12 (2016).
- [7] D. R. Chillingworth. "Winding numbers on surfaces, I". In: Mathematische Annalen 196.3 (1972), pp. 218–249.
- [8] Y. Chitour, D. Prandi and L. Rizzi. "Weyl's law for singular Riemannian manifolds". In: Journal de Mathématiques Pures et Appliquées 181 (2024), pp. 113–151.
- Y. Colin de Verdière, C. Dietze, M. V. de Hoop and E. Trélat. "Weyl formulae for some singular metrics with application to acoustic modes in gas giants". In: arXiv preprint arXiv:2406.19734 (2024).
- Y. Colin de Verdière, L. Hillairet and E. Trélat. "Small-time asymptotics of hypoelliptic heat kernels near the diagonal, nilpotentization and related results". In: Annales Henri Lebesgue 4 (2021), pp. 897–971.

- [11] Y. Colin de Verdière, L. Hillairet and E. Trélat. "Spectral asymptotics for sub-Riemannian Laplacians". In: arXiv preprint arXiv:2212.02920 (2022).
- [12] Y. Colin de Verdière, L. Hillairet and E. Trélat. "Spectral asymptotics for sub-Riemannian Laplacians, I: Quantum ergodicity and quantum limits in the 3-dimensional contact case". In: (2018).
- [13] R. Courant. "Über die Eigenwerte bei den Differentialgleichungen der mathematischen Physik". In: Mathematische Zeitschrift 7.1 (1920), pp. 1–57.
- [14] C. Dietze. "Semiclassical estimates for Schrödinger operators with Neumann boundary conditions on Hölder domains". In: arXiv preprint arXiv:2304.01587 (2023).
- [15] C. Dietze, A. Kachmar and V. Lotoreichik. "Isoperimetric inequalities for inner parallel curves". In: arXiv preprint arXiv:2311.18413 (2023).
- [16] C. Dietze, A. Kachmar and V. Lotoreichik. "Isoperimetric inequalities for inner parallel curves". In: J. Spectr. Theory 14.4 (2024), 1537–1562. DOI: 10.4171/ JST/534.
- C. Dietze and P. T. Nam. "Hardy-Sobolev interpolation inequalities". In: Calculus of Variations and Partial Differential Equations 63.7 (2024), p. 184. DOI: https://doi.org/10.1007/s00526-024-02800-x.
- [18] C. Dietze and L. Read. "Concentration of eigenfunctions on singular Riemannian manifolds". In: arXiv preprint arXiv:2410.20563 (2024).
- [19] W. Feller et al. "An introduction to probability theory and its applications". In: (1971).
- [20] E. Fischer. "Über quadratische Formen mit reellen Koeffizienten". In: Monatshefte für Mathematik und Physik 16 (1905), pp. 234–249.
- [21] R. L. Frank. "A simple proof of Hardy-Lieb-Thirring inequalities". In: Communications in Mathematical Physics 290 (2009), pp. 789–800.
- [22] R. L. Frank, A. Laptev and T. Weidl. Schrödinger operators: eigenvalues and Lieb-Thirring inequalities. Vol. 200. Cambridge University Press, 2022.
- [23] R. L. Frank, E. H. Lieb and R. Seiringer. "Equivalence of Sobolev inequalities and Lieb-Thirring inequalities". In: XVIth International Congress On Mathematical Physics: (With DVD-ROM). World Scientific. 2010, pp. 523–535.
- [24] R. L. Frank and R. Seiringer. "Non-linear ground state representations and sharp Hardy inequalities". In: *Journal of Functional Analysis* 255.12 (2008), pp. 3407–3430.

- [25] D. Grieser. "Notes on heat kernel asymptotics". In: Available on his website: http://www. staff. unioldenburg. de/daniel. grieser/wwwlehre/Schriebe/heat. pdf (2004).
- [26] R. Haller, H. Meinlschmidt and J. Rehberg. "H\" older Regularity for Domains of Fractional Powers of Elliptic Operators with Mixed Boundary Conditions". In: arXiv preprint arXiv:2210.03451 (2022).
- [27] G. H. Hardy, J. E. Littlewood and G. Pólya. *Inequalities*. Cambridge university press, 1952.
- [28] P. Hartman. "Geodesic parallel coordinates in the large". In: American Journal of Mathematics 86.4 (1964), pp. 705–727.
- [29] R. Hempel, L. A. Seco and B. Simon. "The essential spectrum of Neumann Laplacians on some bounded singular domains". In: *Journal of functional analysis* 102.2 (1991), pp. 448–483.
- [30] M. V. de Hoop, J. Ilmavirta, A. Kykkänen and R. Mazzeo. "Geometric inverse problems on gas giants". In: *arXiv preprint arXiv:2403.05475* (2024).
- [31] A. Hurwitz. "Sur quelques applications géométriques des séries de Fourier". In: Annales scientifiques de l'École Normale Supérieure. Vol. 19. 1902, pp. 357– 408.
- [32] A. Kachmar and V. Lotoreichik. "On the isoperimetric inequality for the magnetic Robin Laplacian with negative boundary parameter". In: *The Journal of Geometric Analysis* 32.6 (2022), p. 182.
- [33] D. A. La Manna and R. Sannipoli. "Some isoperimetric inequalities involving the boundary momentum". In: *arXiv preprint arXiv:2309.14191* (2023).
- [34] D. A. Labutin. "Embedding of Sobolev spaces on Holder domains". In: Trudy Matematicheskogo Instituta imeni VA Steklova 227 (1999), pp. 170–179.
- [35] M. L. Lapidus. "Fractal drum, inverse spectral problems for elliptic operators and a partial resolution of the Weyl-Berry conjecture". In: *Transactions of the American Mathematical Society* 325.2 (1991), pp. 465–529.
- [36] M. L. Lapidus and C. Pomerance. "Counterexamples to the modified Weyl-Berry conjecture on fractal drums". In: *Mathematical Proceedings of the Cambridge Philosophical Society*. Vol. 119. 1. Cambridge University Press. 1996, pp. 167–178.
- [37] E. H. Lieb and M. Loss. Analysis. Vol. 14. American Mathematical Soc., 2001.

- [38] C. R. Mankovich. "Saturn's rings as a seismograph to probe Saturn's internal structure". In: *AGU Advances* 1.2 (2020), e2019AV000142.
- [39] G. Métivier. "Comportement asymptotique des valeurs propres d'opérateurs elliptiques dégénérés". In: Journées équations aux dérivées partielles (1975), pp. 215–249.
- [40] S. Minakshisundaram and Å. Pleijel. "Some properties of the eigenfunctions of the Laplace-operator on Riemannian manifolds". In: *Canadian Journal of Mathematics* 1.3 (1949), pp. 242–256.
- [41] Y. Netrusov and Y. Safarov. "Weyl asymptotic formula for the Laplacian on domains with rough boundaries". In: *Communications in mathematical physics* 253 (2005), pp. 481–509.
- [42] G. Palatucci and A. Pisante. "Improved Sobolev embeddings, profile decomposition, and concentration-compactness for fractional Sobolev spaces". In: *Calculus of Variations and Partial Differential Equations* 50.3 (2014), pp. 799– 829.
- [43] W. Ritz. "Über eine neue Methode zur Lösung gewisser Variationsprobleme der mathematischen Physik." In: (1909).
- [44] G. Rozenbljum. "On the eigenvalues of the first boundary value problem in unbounded domains". In: *Mathematics of the USSR-Sbornik* 18.2 (1972), p. 235.
- [45] G. V. Rozenblum. "Distribution of the discrete spectrum of singular differential operators". In: *Izvestiya Vysshikh Uchebnykh Zavedenii*. Matematika 1 (1976), pp. 75–86.
- [46] K. Shiohama, T. Shioya and M. Tanaka. The geometry of total curvature on complete open surfaces. 159. Cambridge University Press, 2003.
- [47] B. Simon. Operator theory. Vol. 4. American Mathematical Soc., 2015.
- [48] M.-P. Tran and T.-N. Nguyen. "A remark on the interpolation inequality between Sobolev spaces and Morrey spaces". In: *Functional Analysis and Its Applications* 54.3 (2020), pp. 200–207.
- [49] I. Vulis and M. Solomjak. "Spectral asymptotic analysis for degenerate second order elliptic operators". In: *Izv. Akad. Nauk SSSR Ser. Mat* 38 (1974), pp. 1362– 1392.
- [50] H. Weyl. "Das asymptotische Verteilungsgesetz der Eigenwerte linearer partieller Differentialgleichungen (mit einer Anwendung auf die Theorie der Hohlraumstrahlung)". In: Mathematische Annalen 71.4 (1912), pp. 441–479.

Part II

Articles

In this part, the articles [14, 9, 16, 17] are printed in the version in which they appeared in the journal or on the preprint server arXiv.

SEMICLASSICAL ESTIMATES FOR SCHRÖDINGER OPERATORS WITH NEUMANN BOUNDARY CONDITIONS ON HÖLDER DOMAINS

CHARLOTTE DIETZE

ABSTRACT. We prove a universal bound for the number of negative eigenvalues of Schrödinger operators with Neumann boundary conditions on bounded Hölder domains, under suitable assumptions on the Hölder exponent and the external potential. Our bound yields the same semiclassical behaviour as the Weyl asymptotics for smooth domains. We also discuss different cases where Weyl's law holds and fails.

1. INTRODUCTION

The celebrated correspondence principle, which goes back to Niels Bohr in the early days of quantum mechanics, states that quantum systems exhibit classical behaviour in the limit of large quantum numbers. In the context of spectral analysis of Schrödinger operators, this leads to the semiclassical approximation, which suggests that any bound state can be related to a volume of size $(2\pi)^d$ in the phase space $\mathbb{R}^d \times \mathbb{R}^d$ [1, Section 4.1.1]. In particular, the number of negative eigenvalues $N(-\Delta_{\Omega} - \lambda)$ of $-\Delta_{\Omega} - \lambda$ on a domain $\Omega \subset \mathbb{R}^d$ with suitable boundary conditions can be approximated by its semiclassical analogue

$$N\left(-\Delta_{\Omega}-\lambda\right) \approx \frac{1}{(2\pi)^d} \left| \left\{ (p,x) \in \mathbb{R}^d \times \Omega \mid |p|^2 - \lambda < 0 \right\} \right| = \frac{|B_1(0)|}{(2\pi)^d} |\Omega| \lambda^{\frac{d}{2}} \tag{1}$$

in the large coupling limit $\lambda \to \infty$, with $|B_1(0)|$ the volume of the unit ball in \mathbb{R}^d . More generally, for a general potential $V : \Omega \to (-\infty, 0]$ one might expect that

$$N\left(-\Delta_{\Omega}+\lambda V\right)\approx\frac{1}{(2\pi)^{d}}\left|\left\{(p,x)\in\mathbb{R}^{d}\times\Omega\mid|p|^{2}+\lambda V(x)<0\right\}\right|=\frac{|B_{1}(0)|}{(2\pi)^{d}}\int_{\Omega}|\lambda V|^{\frac{d}{2}}.$$
 (2)

Rigorous justifications of (1) and (2) have been shown for a large class of smooth domains Ω and potentials V. On the other hand, in general, implementing the semiclassical approximation for rough domains and potentials is difficult. In this paper, we will discuss the validity of (2) for Hölder domains Ω and suitable L^p -integrable potentials V. We will focus on Neumann boundary conditions as Dirichlet boundary conditions have been well understood.

1.1. **Main results.** Let $d \in \mathbb{N}$, $d \geq 2$ and let $\Omega \subset \mathbb{R}^d$ be a domain, that is, an open bounded and connected subset of \mathbb{R}^d . For $\gamma \in (0, 1]$, a γ -Hölder domain is a domain $\Omega \subset \mathbb{R}^d$ which is locally the subgraph of a γ -Hölder continuous function f, that is, there exists a constant c > 0 such that

$$|f(x) - f(y)| \le c|x - y|^{\gamma} \tag{3}$$

Date: July 3, 2023.

 $^{2020\} Mathematics\ Subject\ Classification.\ 35P15,\ 35P20.$

Key words and phrases. Neumann Laplacian, Hölder domains, Cwikel-Lieb-Rozenblum inequality, Semiclassical asymptotics, Weyl's law.

CHARLOTTE DIETZE

for all x, y in the domain of f, which is an open subset of \mathbb{R}^{d-1} . In the case $\gamma = 1$, we call Ω a Lipschitz domain. See Section 2.1 for details.

Denote the Dirichlet Laplacian on Ω by $-\Delta_{\Omega}^{D}$ and the number of negative eigenvalues of $-\Delta_{\Omega}^{D} - \lambda$ by $N\left(-\Delta_{\Omega}^{D} - \lambda\right)$, where $\lambda > 0$. One of the fundamental results of spectral theory is Weyl's law [2, 3, 4, 5], which states that

$$N\left(-\Delta_{\Omega}^{D}-\lambda\right) = \frac{|B_{1}(0)|}{(2\pi)^{d}}|\Omega|\lambda^{\frac{d}{2}} + o\left(\lambda^{\frac{d}{2}}\right) \text{ as } \lambda \to \infty,$$
(4)

thus rigorously justifying (1), see also [6] for a second order result. This asymptotics also holds for the Neumann Laplacian¹ $-\Delta_{\Omega}^{N}$ for Lipschitz domains Ω , or more generally for extension domains, see [7] and also [8, Theorem 3.20].

In general, the Weyl asymptotics for the Neumann Laplacian does *not* hold for arbitrary domains Ω . It is well-known, see for example [9], that there are domains Ω such that zero is contained in the essential spectrum of $-\Delta_{\Omega}^{N}$. Interestingly, Netrusov and Safarov showed that the Weyl asymptotics (4) holds for the Neumann Laplacian $-\Delta_{\Omega}^{N}$, with any γ -Hölder domain Ω , if and only if $\gamma > (d-1)/d$ (see [10, Corollary 1.6 and Theorem 1.10]).

In the present paper, we are interested in $N(-\Delta_{\Omega}^{N}+V)$ with a potential $V: \Omega \to (-\infty, 0]$ on a Hölder domain Ω . Unlike the case of constant potentials, the problem with a general potential V is more subtle as the following theorem shows.

Theorem 1.1 (Example with non-semiclassical behaviour). Let $d \geq 2$. For every $\gamma \in \left(\frac{d-1}{d}, 1\right)$ there exists a γ -Hölder domain $\Omega \subset \mathbb{R}^d$ and $V : \Omega \to (-\infty, 0]$ with $V \in L^{\frac{d}{2}}(\Omega)$ such that

$$\limsup_{\lambda \to \infty} \lambda^{-d/2} N\left(-\Delta_{\Omega}^{N} + \lambda V\right) = \infty.$$
(5)

We prove Theorem 1.1 by constructing a γ -Hölder domain Ω in the same way as Netrusov and Safarov in [10, Theorem 1.10], and we choose the potential V to be growing to infinity near the boundary in such a way that $-\Delta_{\Omega}^{N} + \lambda V$ can support significantly more than $\lambda^{\frac{d}{2}}$ bound states, see Section 7.1 for more details.

Our next result is a universal bound on the number of negative eigenvalues of the Schrödinger operator $-\Delta_{\Omega}^{N} + V$ on $L^{2}(\Omega)$ with a suitable potential V on a Hölder domain Ω . The Cwikel-Lieb-Rozenblum inequality [11, 12, 13] states that for any open set $\Omega \subset \mathbb{R}^{d}$ for $d \geq 3$, there exists a constant C = C(d) > 0 such that for every $V : \Omega \to (-\infty, 0]$, we have

$$N\left(-\Delta_{\Omega}^{D}+V\right) \le C \int_{\mathbb{R}^{d}} |V|^{\frac{d}{2}}.$$
(6)

Here $||V||_p$ is the L^p norm of V. In the case of the Neumann Laplacian on a Lipschitz domain Ω , if $d \geq 3$, then

$$N\left(-\Delta_{\Omega}^{N}+V\right) \le C_{\Omega}\left(1+\int_{\mathbb{R}^{d}}|V|^{\frac{d}{2}}\right)$$

$$\tag{7}$$

for a finite constant $C_{\Omega} > 0$ independent of the potential V, see for example [8, Corollary 4.37]. In dimension d = 2, (6) and (7) still hold, provided that $\|V\|_{\frac{d}{2}}$ is replaced by the Orlicz norm $\|V\|_{\mathcal{B}}$ [14, 15].

¹Here $-\Delta_{\Omega}^{N}$ is the self-adjoint operator generated by the quadratic form $\int_{\Omega} |\nabla u|^{2}$ for all $u \in H^{1}(\Omega)$.

On the other hand, Theorem 1.1 implies that (7) cannot be extended to Hölder domains. Our main result is a replacement of (7) for Hölder domains under suitable assumptions on the potential V. We need to use a weighted L^p -norm with a weight that grows to infinity as one approaches $\partial \Omega$. A simplified version of our result reads as follows.

Theorem 1.2 (Cwikel-Lieb-Rozenblum type bound). Let $d \ge 2$. Let $\gamma \in \left[\frac{d-1}{d}, 1\right)$ and let Ω be a γ -Hölder domain. Then there exists a constant $C_{\Omega} = C_{\Omega}(d, \gamma, \Omega) > 0$ such that for every $V : \Omega \to (-\infty, 0]$ with $||V||| < \infty$, we have

$$N\left(-\Delta_{\Omega}^{N}+V\right) \leq C_{\Omega}\left(1+\||V|\|^{\frac{d}{2}}\right).$$
(8)

Here the norm $|||V||| = ||V||_{\tilde{p},\beta}$ is given in Definition 2.4, with β and \tilde{p} chosen as in (22). Moreover, if $\gamma \in \left[\frac{2(d-1)}{2d-1}, 1\right)$, then |||V||| can be replaced by $||V||_p$ where $p = p_{d,\gamma} > \frac{d}{2}$ is a constant depending only on d and γ satisfying

$$\lim_{\gamma \to 1} p_{d,\gamma} = \frac{d}{2}.$$
(9)

A more precise statement is given in Theorem 1.4 below. The norm |||V||| is stronger than $||V||_{\frac{d}{2}}$ (in particular, the potential V in Theorem 1.1 satisfies $|||V||| = \infty$). Nevertheless, by (8), we still get the correct semiclassical behaviour as soon as $|||V||| < \infty$, namely

$$N\left(-\Delta_{\Omega}^{N} + \lambda V\right) = \mathcal{O}\left(\lambda^{\frac{d}{2}}\right) \text{ as } \lambda \to \infty.$$
(10)

On the technical level, our norm |||V||| is chosen carefully to capture the correct leading order behavior of the number of bound states close to the boundary. By following Rozenblum's method [13], it is possible to obtain the following bound

$$N\left(-\Delta_{\Omega}^{N}+V\right) \lesssim 1 + \int_{\Omega} |V|^{\frac{d}{2}} + \int_{\text{close to }\partial\Omega} |V|^{\tilde{p}}$$
(11)

for some $\tilde{p} > \frac{d}{2}$ (this bound could also be obtained from the analysis in [16] and [17]). However, (11) is insufficient to deduce (10).

As a consequence of Theorem 1.2, we are able to come back to sharp semiclassics.

Theorem 1.3 (Weyl's law for Schrödinger operators on Hölder domains). Let $d \ge 2$. Let $\gamma \in \left[\frac{d-1}{d}, 1\right)$ and let $\Omega \subset \mathbb{R}^d$ be a γ -Hölder domain. Let $V : \Omega \to (-\infty, 0]$ be measurable and such that $|||V||| < \infty$, where the norm $||| \cdot |||$ is the same as in Theorem 1.2. Then

$$N\left(-\Delta_{\Omega}^{N}+\lambda V\right) = (2\pi)^{-d} |B_{1}(0)| \lambda^{\frac{d}{2}} \int_{\Omega} |V|^{\frac{d}{2}} + o\left(\lambda^{\frac{d}{2}}\right) \text{ as } \lambda \to \infty.$$
(12)

Let us remark that Weyl's law for constant potentials fails for $\gamma = \frac{d-1}{d}$ [10, Theorem 1.10]. This does not contradict Theorem 1.3 since $\|\|-1_{\Omega}\|\| = \infty$ for $\gamma = \frac{d-1}{d}$.

1.2. Main ingredients of Theorem 1.2. Now let us explain the proof strategy of our main result Theorem 1.2. Let us focus on the case $d \ge 3$.

Our general approach is inspired by the method of Rozenblum [13] where the number of bound states is bounded using techniques from microlocal analysis. The main idea is to first localize the Schrödinger operator $-\Delta + V$ in small domains such that it has at most one bound state in each domain, and then put these local bounds together by a covering lemma. In the present paper, since we have to deal with Hölder domains with Neumann boundary conditions, we need to deal with "oscillatory domains" when working close to the

CHARLOTTE DIETZE

boundary, and in particular we need to introduce a new Poincaré-Sobolev inequality and a new Besicovitch-type covering lemma for those domains.

To be precise, while Rozenblum [13] works with cubes $Q \subset \mathbb{R}^d$, we will work with "oscillatory domains" $D \subset \Omega$ which are given either by cubes if they are far enough away from $\partial\Omega$, or rectangles intersected with Ω if they are close to the boundary. Since Ω is a γ -Hölder domain, the classical Poincaré-Sobolev inequality fails in general for oscillatory domains D with $\overline{D} \cap \partial\Omega \neq \emptyset$. Therefore, we will develop a Poincaré-Sobolev inequality for those domains (see Corollary 3.3), which involves the ratio of the two side-lengths of the rectangle and a L^{p^*} -norm for some $p^* = p^*(d, \gamma) > \frac{d}{2}$. Consequently, we show that for oscillatory domains D that are small enough in a suitable sense, we have

$$N\left(-\Delta_D^N + KV1_D\right) \le 1. \tag{13}$$

Here, $K = K(d, \gamma) \in \mathbb{N}$ is the constant from our Besicovitch-type covering theorem for oscillatory domains (see Lemma 4.1). Then the total number of bound states is controlled by a counting argument, eventually leading to the weighted norm

$$|||V||| := ||V||_{\tilde{p},\beta,D} := ||V||_{\frac{d}{n},D} + |V|_{\tilde{p},\beta,D},$$
(14)

where $\tilde{p} = \tilde{p}(d,\gamma) > \frac{d}{2}$, $\beta = \beta(d,\gamma) > 0$ and $|V|_{\tilde{p},\beta}$ is a weighted $L^{\tilde{p}}(\Omega)$ -seminorm with a weight supported near the boundary of Ω that grows at a rate determined by β as one approaches the boundary, see (22) and Definition 2.4 for the precise definitions.

In (14) we need a $\tilde{p} > p^*$ for the following reason. Due to the Hölder-regularity of $\partial\Omega$, our oscillatory domains close to the boundary will in many cases look like very narrow rectangles intersected with Ω , so the ratio of the two side-lengths of these rectangles is very far away from one. This influences the constant in the Poincaré-Sobolev inequality (Corollary 3.3). Using Hölder's inequality, we get

$$\|V\|_{p^*,D} \le \|V\|_{\tilde{p},D} \, \|1\|_{\tilde{r},D} = \|V\|_{\tilde{p},D} \, |D|^{\frac{1}{\tilde{r}}} \tag{15}$$

for $\frac{1}{p^*} = \frac{1}{\tilde{p}} + \frac{1}{\tilde{r}}$. The quantity $|D|^{\frac{1}{\tilde{r}}}$, which is relatively small for these narrow rectangles, cancels the effect of a growing constant in the Poincaré-Sobolev inequality as the rectangle gets more narrow.

At this point, it seems natural to measure the "size" of oscillatory domains close to the boundary by $||V||_{\tilde{p},D}$. Following [13], we would get (11) but it does not capture the correct semiclassical behaviour. In order to get the desired semiclassical behaviour for the parts close to the boundary as well, the key idea is to count the number of oscillatory domains $\{D_j\}_{j\in J_3}$, with some index set J_3 , which are narrow rectangles intersected with Ω close to the boundary, by using a convexity argument. More precisely, we introduce coefficients $A_j > 0$ depending only on the larger side-length of the corresponding rectangle and the distance of the centre of the oscillatory domain D_j to the boundary measured in a suitable sense. Now for suitably chosen $s, s' \in (1, \infty)$ with $\frac{1}{s} + \frac{1}{s'} = 1$, we apply Hölder's inequality for sums of products of real numbers to get

$$|J_3| = \sum_{j \in J_3} A_j^{-1} A_j \le \left(\sum_{j \in J_3} A_j^{-s'}\right)^{\frac{1}{s'}} \left(\sum_{j \in J_3} A_j^s\right)^{\frac{1}{s}}.$$
 (16)

The coefficients A_j will be chosen in such a way that

$$A_j^s \lesssim |V|_{\tilde{p},\beta,D_j}^{\tilde{p}} \tag{17}$$

for all $j \in J_3$. Therefore,

$$\sum_{j \in J_3} A_j^s \lesssim \sum_{j \in J_3} |V|_{\tilde{p},\beta,D_j}^{\tilde{p}} \lesssim |V|_{\tilde{p},\beta}^{\tilde{p}} \lesssim \|V\|_{\tilde{p},\beta}^{\tilde{p}} .$$

$$(18)$$

In the proof of Rozenblum, the cubes Q were chosen such that $\|V\|_{\frac{d}{2},Q} \leq 1$. By contrast, here the oscillatory domains $\{D_j\}_{j\in J_3}$ are chosen such that $\|V\|_{\tilde{p},D_j}$ is significantly larger than one if the distance of the centre of the oscillatory domain D_j to the boundary is significantly larger than the largest side-length of the corresponding rectangle. This is possible, even though we need to ensure $N\left(-\Delta_{D_j}^N + KV1_{D_j}\right) \leq 1$ since in fact, we gain something in the Poincaré-Sobolev inequality for oscillatory domains because we consider an $L^{\tilde{p}}$ -norm instead of an L^{p^*} -norm. The fact that $\|V\|_{\tilde{p},D_j}$ can be significantly larger than one in certain circumstances combined with the weight that grows near the boundary in the definition of $|V|_{\tilde{p},\beta}$ allow us to choose A_j so large that

$$\sum_{j \in J_3} A_j^{-s'} \lesssim \|V\|_{\tilde{p},\beta}^{-\frac{1}{2}}$$
(19)

while (17) holds. We choose all parameters in such a way that

$$\frac{\tilde{p}}{s} - \frac{1}{2s'} = \frac{d}{2},\tag{20}$$

so by (18) and (19),

$$|J_3| \le \left(\sum_{j \in J_3} A_j^{-s'}\right)^{\frac{1}{s'}} \left(\sum_{j \in J_3} A_j^s\right)^{\frac{1}{s}} \lesssim \|V\|_{\tilde{p},\beta}^{-\frac{1}{2s'}} \|V\|_{\tilde{p},\beta}^{\frac{\tilde{p}}{s}} = \|V\|_{\tilde{p},\beta}^{\frac{d}{2}},$$
(21)

which has the desired semiclassical behaviour.

Combining these computations with the computations for oscillatory domains far enough away from the boundary, we obtain the following.

Theorem 1.4 (Precise version of Theorem 1.2). Let $d \in \mathbb{N}$ with $d \geq 2$, $\gamma \in \left[\frac{d-1}{d}, 1\right)$, $c > 0, 0 < h_{\Omega} < 1, L \in \mathbb{N}$ and let $\emptyset \neq \Omega \subset (0, 1)^d$ be a γ -Hölder domain with parameters c, h_{Ω}, L . Define

$$\beta := \frac{1}{d+1} \left(\frac{d-1}{\gamma} + 1 \right) \left[\frac{1}{d} \left(\frac{d-1}{\gamma} + 1 \right)^2 - d \right] > 0, \quad \tilde{p} := \frac{1}{2d} \left(\frac{d-1}{\gamma} + 1 \right)^2 > \frac{d}{2}.$$
(22)

Let $V: \Omega \to (-\infty, 0]$ be measurable and such that $\|V\|_{\tilde{p}, \beta} < \infty$. Define

$$\delta_0 := \delta_0(V) := \min\left(\frac{h_\Omega}{\sqrt{d}}, \|V\|_{\tilde{p},\beta}^{-\frac{1}{2}}\right) \le 1.$$
(23)

Then

$$N\left(-\Delta_{\Omega}^{N}+V\right) \lesssim \delta_{0}^{-d}, \qquad (24)$$

where the constant in the inequality may depend on $d, \gamma, c, h_{\Omega}, L$.

The definition of a " γ -Hölder domain with parameters c, h_{Ω}, L " will be given in Definition 2.1. The condition $\Omega \subset (0, 1)^d$ is not a restriction since we can recover the same result for all (bounded) γ -Hölder domains by scaling.

Note that from (24) in Theorem 1.4, we can deduce the first claim (8) of Theorem 1.2. Moreover, the second claim of Theorem 1.2 follows from the following corollary.

Corollary 1.5. If $\gamma \in \left[\frac{2(d-1)}{2d-1}, 1\right)$ and $p > \frac{\tilde{p}}{1-\beta}$, where \tilde{p} and β were defined in Theorem 1.4, then

$$N\left(-\Delta_{\Omega}^{N}+V\right) \lesssim 1 + \|V\|_{p}^{\frac{d}{2}}, \qquad (25)$$

where the constant in the inequality may depend on $d, \gamma, c, h_{\Omega}, L, p$.

The details of the proofs of Theorem 1.4 and Corollary 1.5 can be found in Section 5. Let us give the key ingredients of the proof of Theorem 1.4 below. First, we have

Lemma 1.6. There exists $K = K(d, \gamma, L) \in \mathbb{N}$ such that for any measurable $V : \Omega \rightarrow \mathbb{N}$ $(-\infty, 0]$ with $\|V\|_{\tilde{p}, \beta} < \infty$ there are families $\mathcal{F}_1, \ldots, \mathcal{F}_K$ of oscillatory domains $D \subset \Omega$ such that the following properties are satisfied:

(a) The oscillatory domains D in every \mathcal{F}_k with $k \in \{1, \ldots, K\}$ are disjoint and

$$\Omega = \bigcup_{k=1}^{K} \bigcup_{D \in \mathcal{F}_k} D.$$
(26)

(b) For every $k \in \{1, \ldots, K\}$ and every $D \in \mathcal{F}_k$ we have

$$N\left(-\Delta_D^N + V\right) \le 1. \tag{27}$$

(c) Moreover, for an implicit constant depending on $d, \gamma, c, h_{\Omega}, L$,

$$\sum_{k=1}^{K} |\mathcal{F}_k| \lesssim \delta_0^{-d}.$$
(28)

To prove Lemma 1.6, we need a new covering lemma (see Lemma 4.1 for details). From Lemma 1.6, we can deduce the following two lemmata.

Lemma 1.7 (Selfadjointness of the operator $-\Delta_{\Omega}^{N} + V$). Let $V : \Omega \to (-\infty, 0]$ be measurable and such that $\|V\|_{\tilde{p},\beta} < \infty$. Then the operator $-\Delta_{\Omega}^{N} + V$ is a selfadjoint operator, which is bounded from below and has the $H^1(\Omega)$ norm as quadratic form norm.

Lemma 1.8. Let $V : \Omega \to (-\infty, 0]$ be measurable and such that $\|V\|_{\tilde{p},\beta} < \infty$. Let $K = K(d, \gamma, L) \in \mathbb{N}$ and the families $\mathcal{F}_1, \ldots, \mathcal{F}_K$ of oscillatory domains be chosen as in Lemma 1.6. Then

$$N\left(-\Delta_{\Omega}^{N} + \frac{1}{K}V\right) \lesssim \delta_{0}^{-d}.$$
(29)

The proofs of Lemmas 1.6, 1.7 and 1.8 can be found in Section 5.1. We are now ready to prove Theorem 1.4 assuming those lemmata.

Proof of Theorem 1.4. Let $K = K(d, \gamma, L) \in \mathbb{N}$ be chosen as in Lemma 1.6. Now apply Lemma 1.8 to KV to get

$$N\left(-\Delta_{\Omega}^{N}+V\right) = N\left(-\Delta_{\Omega}^{N}+\frac{1}{K}KV\right) \lesssim \delta_{0}(KV)^{-d} \lesssim \delta_{0}(V)^{-d}, \qquad (30)$$

ed that K only depends on d, γ , L in the second last step.

where we used that K only depends on d, γ, L in the second last step.

Structure of the paper. Sections 2 to 5 are devoted to proving the Cwickel-Lieb-Rozenblum type bound (Theorem 1.2). In Section 2, we introduce some preliminaries, including the definition and some basic properties of oscillatory domains. We prove several estimates for oscillatory domains in Section 3. In Section 4, we prove a covering lemma for oscillatory domains. In Section 5, we combine the results from Sections 2 to 4 and prove Theorem 1.2. In Section 6, we prove Weyl's law for Schrödinger operators on Hölder

domains (Theorem 1.3). In Section 7, we construct an example with non-semiclassical behaviour and thereby prove Theorem 1.1.

Acknowledgements. The author would like to express her deepest gratitude to Phan Thành Nam for his continued support and very helpful advice. She would also like to thank Laure Saint-Raymond for her support and hospitality at Institut des Hautes Études Scientifiques and for inspiring discussions. She would like to thank László Erdős for useful remarks. She would like to thank Luca Maio for helping prepare the tex file. The author acknowledges the support from the Deutsche Forschungsgemeinschaft (DFG project Nr. 426365943), from the Jean-Paul Gimon Fund and from the Erasmus+ programme.

Notation. For two numbers $A, B \ge 0$, we write $A \le B$ if $A \le CB$ for some constant C > 0 which may depend on $d, \gamma, c, h_{\Omega}, L$ (see Definition 2.1). Similarly we write $A \sim B$ if $A \le B$ and $A \ge B$.

2. Preliminaries

In this section we collect some technical definitions and preliminary results.

2.1. **Hölder domains.** We often write elements in \mathbb{R}^d as $x = (x', x_d)$, where $x' \in \mathbb{R}^{d-1}$ and $x_d \in \mathbb{R}$. We denote the infinity norm on \mathbb{R}^{d-1} by $|\cdot|_{\infty}$. We have the following technical definition which agrees with the definition of Hölder domains given around (3).

Definition 2.1 (γ -Hölder domain with parameters c, h_{Ω}, L). Let $d \in \mathbb{N}$ with $d \geq 2$, $\gamma \in (0, 1], c > 0, h_{\Omega} > 0, L \in \mathbb{N}$ and let $\emptyset \neq \Omega \subset \mathbb{R}^d$ and be a bounded open set. We call Ω a γ -Hölder domain with parameters c, h_{Ω}, L if there exists a collection $\{O_l\}_{l=1}^L \subset \mathbb{R}^d$ of open sets covering $\partial\Omega$ with the following properties:

(i) For every $l \in \{1, ..., L\}$ there exists an orthogonal map R_l and a translation map T_l such that for $\Omega_l := O_l \cap \Omega$ we have

$$\Omega_l = T_l R_l \left\{ (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x \in \Omega_l^{(d-1)}, \ 0 < x_d < f_l(x') \right\}$$
(31)

and

$$O_l \setminus \Omega_l \subset T_l R_l \left\{ (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x \in \Omega_l^{(d-1)}, \ f_l(x') \le x_d \right\}$$
(32)

for some open sets $\Omega_l^{(d-1)} \subset \mathbb{R}^{d-1}$ and for a function $f_l : \Omega_l^{(d-1)} \to (3h_\Omega, \infty)$ with

$$|f_l(x') - f_l(y')| \le c |x' - y'|_{\infty}^{\gamma} \text{ for all } x', y' \in \Omega_l^{(d-1)}.$$
(33)

(ii) For every $l \in \{1, \ldots, L\}$ define

$$\hat{\Omega}_{l}^{(d-1)} := \left\{ x' \in \Omega_{l}^{(d-1)} \mid \operatorname{dist}\left(x', \partial \Omega_{l}^{(d-1)}\right) > 2h_{\Omega} \right\},\tag{34}$$

where dist $(x', \partial \Omega_l^{(d-1)})$ denotes the distance from x' to $\partial \Omega_l^{(d-1)}$ with respect to the Euclidean norm in \mathbb{R}^{d-1} , and

$$\hat{\Omega}_l := T_l R_l \left\{ (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x' \in \hat{\Omega}_l^{(d-1)}, \ 2h_\Omega < x_d < f_l(x') \right\}.$$
(35)

Then

$$\partial \Omega \subset \bigcup_{l=1}^{L} \overline{\hat{\Omega}_l} \,. \tag{36}$$

7

Definition 2.2 $(\tilde{\Omega}_l^{(d-1)} \text{ and } \tilde{\Omega}_l)$. For every $l \in \{1, \ldots, L\}$ define

$$\tilde{\Omega}_{l}^{(d-1)} := \left\{ x' \in \Omega_{l}^{(d-1)} \mid \operatorname{dist}\left(x', \partial \Omega_{l}^{(d-1)}\right) > h_{\Omega} \right\},\tag{37}$$

$$\tilde{\Omega}_l := T_l R_l \left\{ (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x' \in \tilde{\Omega}_l^{(d-1)}, \ h_\Omega < x_d < f_l(x') \right\},\tag{38}$$

$$\Omega^{b}_{\varepsilon} := \left\{ x \in \Omega \mid \operatorname{dist}\left(x, \partial \Omega\right) < \varepsilon \right\},\tag{39}$$

where dist $(x, \partial \Omega)$ denotes the distance with respect to the Euclidean norm in \mathbb{R}^d .

Lemma 2.3. Let $\tilde{\Omega}_l$ and Ω^b_{ε} be defined as in Definition 2.2. Then

$$\bigcup_{l=1}^{L} \tilde{\Omega}_l \supset \Omega_{h_{\Omega}}^b.$$
(40)

Proof. Let $x \in \Omega$ with dist $(x, \partial\Omega) < h_{\Omega}$. Then there exists $y \in \partial\Omega$ with $|x - y| = \text{dist}(x, \partial\Omega)$. By in Definition 2.1(ii), we know (36), so there exists $l \in \{1, \ldots, L\}$ with $y \in \partial\hat{\Omega}_l$. Without loss of generality, we may assume that T_l , R_l are the identity map. Hence, we have $y = (y', f_l(y'))$ with $y' \in \hat{\Omega}_l^{(d-1)}$. It follows that $\text{dist}(y', \partial\Omega_l^{(d-1)}) > 2h_{\Omega}$ by (34). By the triangle inequality, we obtain $\text{dist}(x', \partial\Omega_l^{(d-1)}) > h_{\Omega}$, so $x' \in \Omega_l^{(d-1)}$. Furthermore, by the definition of f_l , we have $f_l(y') > 3h_{\Omega}$. Again, by the triangle inequality, this implies $x_d > 2h_{\Omega} > h_{\Omega}$. It follows that $x \in \tilde{\Omega}_l$.

Definition 2.4 $(h_{x,l}, \|\cdot\|_{p,\beta})$ and $\|\cdot\|_{p,\beta}$. Let $d \geq 2$ and let $\Omega \subset \mathbb{R}^d$ be a γ -Hölder domain with constant c > 0 and parameters $h_{\Omega} > 0$ and $L \in \mathbb{N}$ for some $\gamma \in (0, 1]$. In the following, we use the notation from Definition 2.1.

(i) For any $l \in \{1, \ldots, L\}$ and $x = T_l R_l(x', x_d) \in \Omega_l$, define

$$h_x := h_{x,l} := f_l(x') - x_d > 0.$$
(41)

Moreover, for any $x \in \bigcup_{l=1}^{L} \Omega_l$, we let

$$h_{x,\min} := \min_{l \in \{1,\dots,L\} \text{ with } x \in \Omega_l} h_{x,l} \,. \tag{42}$$

(ii) For $\beta > 0$ and $p \in [1, \infty)$, define the seminorm $|\cdot|_{p,\beta}$ by

$$|f|_{p,\beta}^{p} := \int_{\bigcup_{l=1}^{L} \Omega_{l}} \mathrm{d}x \, h_{x,\min}^{-\beta} |f(x)|^{p} \tag{43}$$

for all measurable functions $f: \Omega \to \mathbb{C}$.

(iii) Define the norm $\|\cdot\|_{p,\beta}$ by

$$\|f\|_{p,\beta} := \|f\|_{\frac{d}{2},\Omega} + |f|_{p,\beta} \quad \text{if } d \ge 3 \,, \quad \|f\|_{p,\beta} := \|f\|_{\mathcal{B},\Omega} + |f|_{p,\beta} \quad \text{if } d = 2 \tag{44}$$

for all measurable functions $f: \Omega \to \mathbb{C}$. See [14, p. 1] for the definition of $||f||_{\mathcal{B},\Omega}$.

2.2. Oscillatory domains. In this subsection we define oscillatory domains and prove several properties of oscillatory domains.

Definition 2.5 $(c_0, c_1 \text{ and } c_2)$. Let c > 0 be as in Definition 2.4. We define

$$c_0 := \left[\min\left(\frac{1}{c_1}, \frac{2^{\gamma}}{64c}, \frac{1}{2^{\gamma+3}c}\right) \right]^{\frac{1}{\gamma}}, \quad c_1 := 16, \quad c_2 := c_0 c_1^{\frac{1}{\gamma}}.$$
(45)
Definition 2.6 (Oscillatory domain D). For every $l \in \{1, ..., L\}$, $x \in \Omega_l$ and $\delta \in (0, \delta_0]$ define

$$a := a_x := a_x(\delta) := a_{x,l}(\delta) := \min\left(\delta, c_0 \max\left(h_{x,l}, c_1\delta\right)^{\frac{1}{\gamma}}\right),$$
(46)

where c_0 , c_1 were defined in Definition 2.5. Define $D := D_x := D_x(\delta) := D_{x,l}(\delta)$ by

$$D := T_l R_l \left\{ (y', y_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid |y' - x'| < \frac{1}{2}a, \ |y_d - x_d| < \frac{1}{2}\delta, \ f_l(y') > y_d \right\}.$$
 (47)

Lemma 2.7. Let $l \in \{1, \ldots, L\}$, $x \in \tilde{\Omega}_l$ and $\delta \in (0, \delta_0]$. Then

- (i) $D = D_x$ is well-defined and $D \subset \Omega_l$.
- (ii) For all $y = T_l R_l(y', y_d) \in D$, we have

$$f_l(y') \ge x_d - \frac{\delta}{4} \,. \tag{48}$$

Proof. **Proof of (i).** Let $y \in D$. Then $y = T_l R_l(y', y_d)$ for some $(y', y_d) \in \mathbb{R}^{d-1} \times \mathbb{R}$ with $|y' - x'| < \frac{1}{2}a \leq \frac{1}{2}h_{\Omega}$, $|y_d - x_d| < \frac{1}{2}\delta \leq \frac{1}{2}h_{\Omega}$ and $f_l(y') > y_d$. Here we used that $a \leq \delta \leq \delta_0 \leq \frac{h_{\Omega}}{\sqrt{d}} \leq h_{\Omega}$. Since $x' \in \tilde{\Omega}_l^{(d-1)}$ and $|y' - x'| < \frac{1}{2}h_{\Omega}$, we get $y' \in \Omega_l^{(d-1)}$. Thus, $f_l(y')$ is well-defined and therefore, D is well-defined. Since $x \in \tilde{\Omega}_l$, we have $x_d > h_{\Omega}$, so using $|y_d - x_d| < \frac{1}{2}h_{\Omega}$, we get $y_d > 0$. To sum up, we have shown that $y' \in \Omega_l^{(d-1)}$ and $f_l(y') > y_d > 0$, so $y \in \Omega_l$ by (31).

Proof of (ii). Let $y = T_l R_l(y', y_d) \in D$. Then

$$|f_l(x') - f_l(y')| \le c |x' - y'|_{\infty}^{\gamma} < c \left(\frac{1}{2}a\right)^{\gamma} = \frac{c}{2^{\gamma}}a^{\gamma},$$
(49)

 \mathbf{SO}

$$f_l(y') \ge f_l(x') - |f_l(x') - f_l(y')| \ge f_l(x') - \frac{c}{2\gamma}a^{\gamma} = h + x_d - \frac{c}{2\gamma}a^{\gamma},$$
(50)

where we used (41). Thus, in order to show (48), it suffices to show that

$$\frac{\delta}{4} + h - \frac{c}{2^{\gamma}} a^{\gamma} \ge 0, \qquad (51)$$

If $a = c_0 h^{\frac{1}{\gamma}}$ or $a = c_0 (c_1 \delta)^{\frac{1}{\gamma}}$, then (51) holds by

$$\max\left(2^{\gamma+1}cc_0^{\gamma}, \frac{cc_0^{\gamma}c_1}{2^{\gamma}}\right) \le \frac{1}{4}.$$
(52)

If $a = \delta$, then $\delta \leq c_0 h^{\frac{1}{\gamma}}$ by the definition of a, and it reduces to the case $a = c_0 h^{\frac{1}{\gamma}}$. \Box Lemma 2.8. Let $l \in \{1, \ldots, L\}$, $x \in \tilde{\Omega}_l$ and $\delta_x \in (0, \delta_0]$. Then the following holds true:

(i) If $a_x = \delta_x$ or $a_x = c_0 h_x^{\frac{1}{\gamma}}$, then D_x is a cuboid, namely

$$D_x := T_l R_l \left\{ (w', w_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid |w' - x'| < \frac{1}{2} a_x, \ |w_d - x_d| < \frac{1}{2} \delta_x \right\}.$$
(53)

(ii) In addition, let $y, z \in \tilde{\Omega}_l$ with $\delta_y, \delta_z \in (0, \delta_0]$ and assume that

$$a_w = c_0 h_w^{\frac{1}{\gamma}} \text{ for all } w \in \{x, y, z\} .$$
(54)

Furthermore, assume that $\delta_x \leq 2\delta_y$, $\delta_z \leq 2\delta_y$, $D_x \cap D_y \neq \emptyset$ and $D_x \cap D_z \neq \emptyset$. Then

$$\frac{1}{2}h_y \le h_w \le 2h_y \text{ for all } w \in \{x, z\} .$$

$$(55)$$

9

(iii) If $a_x = c_0 h_x^{\frac{1}{\gamma}}$, then

$$\frac{1}{2}h_x \le h_w \le 2h_x \text{ for all } w \in D_x.$$
(56)

(iv) If $a_x = c_2 \delta_x^{\frac{1}{\gamma}}$, then $|h_w - h_x| \le \delta_x$ for all $w \in D_x$. (57) (v) If $a_x = c_0 \max(h_x, c_1 \delta)^{\frac{1}{\gamma}}$ and $f : \Omega \to \mathbb{C}$ is measurable, then $|f|_{\tilde{p}, \beta, D_x}^{\tilde{p}} \gtrsim \max(h_x, c_1 \delta_x)^{-\beta} \|f\|_{\tilde{p}, D_x}^{\tilde{p}}$. (58)

Proof. For simplicity of notation, we write $a := a_x$, $\delta := \delta_x$, $h := h_x$, $D := D_x$ in the proof. **Proof of (i).** Since $c_0 c_1^{\frac{1}{\gamma}} \leq 1$ and $\delta_0 \leq 1$, we know by the definition of a and by $a = \delta$ or $a = c_0 h^{\frac{1}{\gamma}}$ that $h \geq c_1 \delta$. For all $w = T_l R_l(w', w_d)$ with $|w' - x'| < \frac{1}{2}a$ and $|w_d - x_d| < \frac{1}{2}\delta$, we have

$$f_l(w') \ge h + x_d - \frac{c}{2\gamma} a^{\gamma} \ge h + x_d - \frac{c}{2\gamma} \left(c_0 h^{\frac{1}{\gamma}} \right)^{\gamma}$$

$$= x_d + h \left(1 - \frac{cc_0^{\gamma}}{2\gamma} \right) \ge x_d + \frac{1}{2} h \ge x_d + \frac{c_1}{2} \delta = x_d + 8\delta \ge w_d + 7\delta ,$$

$$(59)$$

where we used (50) in the first step, $a = \delta$ or $a = c_0 h^{\frac{1}{\gamma}}$ in the second step, (52) in the fourth step, and $c_1 = 16$ in the second last step. By (47), we get $w \in D$.

Proof of (ii). First note that by $a_w = c_0 h_w^{\overline{\gamma}}$, we have $h_w \ge c_1 \delta_w$ for all $w \in \{x, y, z\}$. Also note that in order to show (55), it suffices to show that

$$|h_w - h_y| \le \frac{1}{2} \max(h_w, h_y)$$
 (60)

Let us begin by showing (60) for w = x. Since $D_x \cap D_y \neq \emptyset$, we have

$$|x_d - y_d| < \frac{1}{2}\delta_x + \frac{1}{2}\delta_y \le \frac{1}{2}2\delta_y + \frac{1}{2}\delta_y \le 2\delta_y$$
(61)

and

$$|x' - y'| < \frac{1}{2}a_x + \frac{1}{2}a_y = \frac{c_0}{2}\left(h_x^{\frac{1}{\gamma}} + h_y^{\frac{1}{\gamma}}\right) \le c_0\left[\max\left(h_x, h_y\right)\right]^{\frac{1}{\gamma}}.$$
(62)

We obtain

$$|h_x - h_y| = |f_l(x') - x_d - (f_l(y') - y_d)| \le |f_l(x') - f_l(y')| + |x_d - y_d| \le c|x' - y'|^{\gamma} + 2\delta_y$$

$$\le cc_0^{\gamma} \max(h_x, h_y) + \frac{2}{c_1}h_y \le \left(cc_0^{\gamma} + \frac{2}{c_1}\right) \max(h_x, h_y) \le \frac{1}{2}\max(h_x, h_y) .$$

Here we used $h_y \ge c_1 \delta_y$ in the second last step and we used (52) and $c_1 = 16$ in the last step. This shows (60) for w = x.

Next, let us show (60) for w = z. Since $D_x \cap D_y \neq \emptyset$ and $D_x \cap D_z \neq \emptyset$, we have

$$|z_d - y_d| \le |z_d - x_d| + |x_d - y_d| < \frac{1}{2}\delta_z + \delta_x + \frac{1}{2}\delta_y \le \frac{1}{2}2\delta_y + 2\delta_y + \frac{1}{2}\delta_y \le 4\delta_y$$
(63)

and

$$|z' - y'| \le |z' - x'| + |x' - y'| < \frac{1}{2}a_z + a_x + \frac{1}{2}a_y \le 2\max(a_x, a_y, a_z)$$

= $2c_0 \max\left(h_x^{\frac{1}{\gamma}}, h_y^{\frac{1}{\gamma}}, h_z^{\frac{1}{\gamma}}\right) \le 2c_0 \max\left((2h_y)^{\frac{1}{\gamma}}, h_y^{\frac{1}{\gamma}}, h_z^{\frac{1}{\gamma}}\right) \le 2^{1 + \frac{1}{\gamma}}c_0 \max\left(h_y^{\frac{1}{\gamma}}, h_z^{\frac{1}{\gamma}}\right),$

where we used $h_x \leq 2h_y$ in the second last step. We obtain

$$\begin{aligned} |h_z - h_y| &= |f_l(z') - z_d - \left(f_l(y') - y_d\right)| \le |f_l(z') - f_l(y')| + |z_d - y_d| \\ &\le c|z' - y'|^{\gamma} + 4\delta_y \le c \left(2^{1 + \frac{1}{\gamma}} c_0 \left[\max\left(h_x, h_y\right)\right]^{\frac{1}{\gamma}}\right)^{\gamma} + 4\delta_y \\ &\le 2^{\gamma + 1} cc_0^{\gamma} \max\left(h_x, h_y\right) + \frac{4}{c_1} h_y \le \left(2^{\gamma + 1} cc_0^{\gamma} + \frac{4}{c_1}\right) \max\left(h_x, h_y\right) \le \frac{1}{2} \max\left(h_x, h_y\right) \,. \end{aligned}$$

where we used (52) and $c_1 = 16$ in the last step. This shows (60) for w = z. **Proof of (iii).** Let $w \in D_x$. By $a = c_0 h^{\frac{1}{\gamma}}$, we have $h \ge c_1 \delta$. As in the proof of (ii), we get

$$|h_w - h| = |f_l(w') - w_d - (f_l(x') - x_d)| \le |f_l(w') - f_l(x')| + |w_d - x_d|$$

$$\le c|w' - x'|^{\gamma} + \frac{1}{2}\delta \le c\left(\frac{1}{2}a\right)^{\gamma} + \frac{1}{2c_1}h = c\left(\frac{1}{2}c_0h^{\frac{1}{\gamma}}\right)^{\gamma} + \frac{1}{2c_1}h = \frac{cc_0^{\gamma}}{2^{\gamma}}h + \frac{1}{2c_1}h \le \frac{1}{2}h,$$

where we used (52) and $c_1 = 16$ in the last step. This shows (55).

Proof of (iv). Let $w \in D_x$. By $a = c_2 \delta^{\frac{1}{\gamma}}$, we have $h \leq c_1 \delta$. As in the proof of (iii), we get

$$\begin{aligned} |h_w - h| &= |f_l(w') - w_d - \left(f_l(x') - x_d\right)| \le |f_l(w') - f_l(x')| + |w_d - x_d| \\ &\le c|w' - x'|^{\gamma} + \frac{1}{2}\delta \le c\left(\frac{1}{2}a\right)^{\gamma} + \frac{1}{2}\delta = \frac{cc_0^{\gamma}c_1}{2^{\gamma}}\delta + \frac{1}{2}\delta \le \delta \,, \end{aligned}$$

where we used (52) and $c_1 = 16$ in the last step.

Proof of (v). Let $w \in D_x$. Then $h_w = |f_l(w') - w_d| \le h + |f_l(w') - w_d - (f_l(x') - x_d)| \le h + |f_l(w') - f_l(x')| + |w_d - x_d|$ $\le h + c|w' - x'|^{\gamma} + \frac{1}{2}\delta \le h + c\left(\frac{1}{2}c_0 \max{(h_x, c_1\delta)^{\frac{1}{\gamma}}}\right)^{\gamma} + \frac{1}{2}\delta \lesssim \max{(h, c_1\delta)},$

By the definition of $|f|_{\tilde{p},\beta,D}^{\tilde{p}}$, we have

$$|f|_{\tilde{p},\beta,D}^{\tilde{p}} = \int_{D} \mathrm{d}y \, h_{y,\min}^{-\beta} |f(y)|^{\tilde{p}} \gtrsim \max\left(h, c_{1}\delta\right)^{-\beta} \|f\|_{\tilde{p},D}^{\tilde{p}} \,. \tag{64}$$

3. Estimates for oscillatory domains

3.1. **Poincaré-Sobolev inequality for oscillatory domains.** In this subsection we prove a Poincaré-Sobolev inequality for oscillatory domains. The following Lemma and its proof is a version of [17] for oscillatory domains (Corollary 3.3).

Lemma 3.1 (Sobolev inequality for oscillatory domains). Let c > 0, $\gamma \in (0,1)$ and $\delta \in (0,1)$. Let $\tilde{a} > 0$ and let $f : [-\tilde{a}/2, \tilde{a}/2]^{d-1} \to (0, \delta)$ be such that

$$|f(x') - f(y')| \le c |x' - y'|_{\infty}^{\gamma} \text{ for all } x', y' \in \left[-\frac{\tilde{a}}{2}, \frac{\tilde{a}}{2}\right]^{d-1}.$$
(65)

Define $q^* \in (2,\infty)$ by

$$\frac{1}{q^*} := \frac{1}{2} - \frac{1}{\frac{d-1}{\gamma} + 1} \tag{66}$$

and define

$$\tilde{D} := \left\{ (x', x_d) \in \left[-\frac{\tilde{a}}{2}, \frac{\tilde{a}}{2} \right]^{d-1} \times (0, \delta) \mid f(x') > x_d \right\}.$$

$$(67)$$

Define the "straight part I'm nots of the boundary of D" by

$$\tilde{B} := \partial \tilde{D} \setminus \left\{ (x', x_d) \in \left[-\frac{\tilde{a}}{2}, \frac{\tilde{a}}{2} \right]^{d-1} \times (0, \delta) \mid f(x') = x_d \right\}.$$
(68)

Then there exists a constant $C_S = C_S(d, c, \gamma) > 0$ such that for all $u \in H^1(\tilde{D})$ with $u \upharpoonright_{\tilde{B}} \equiv 0$ in the trace sense, we have

$$\|u\|_{q^*,\tilde{D}}^2 \le C_S \|\nabla u\|_{2,\tilde{D}}^2 .$$
(69)

Proof. We first consider the case of smooth functions u and later deduce (69) for all $u \in H^1(\tilde{D})$ with $u \upharpoonright_{\tilde{B}} \equiv 0$ in the trace sense. Let $u \in C^{\infty}(\tilde{D}) \cap H^1(\tilde{D})$ with $u \upharpoonright_{\tilde{B}} \equiv 0$. Let

$$\psi: (0,\infty) \to \mathbb{R}, \quad \psi(s) = c^{-\frac{1}{\gamma}} s^{\frac{1}{\gamma}}.$$
 (70)

Fix $\tilde{x} \in \tilde{D}$. For simplicity of notation, we shift the coordinate system and reflect the last coordinate by replacing every $x \in \mathbb{R}^d$ by $(x_1 - \tilde{x}_1, \ldots, x_{d-1} - \tilde{x}_{d-1}, -x_d + \tilde{x}_d) \in \mathbb{R}^d$, so without loss of generality, we may assume $\tilde{x} = 0$. Define

$$K := \left\{ (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid 0 < x_d, \ 0 \le |x'| < \psi(x_d) \right\}.$$

$$\tag{71}$$

and note that u it is well-defined on K since we can extend it by zero on $K \setminus \tilde{D}$ due to $u \upharpoonright_{\tilde{B}} \equiv 0$ in the trace sense. Let $y' \in \mathbb{R}^{d-1}$ with |y'| < 1 and note that $(y'\psi(x_d), x_d) \in K$ for all $x_d \in (0, \delta)$. Since u is smooth, we can apply Newton's theorem and $u((y'\psi(\delta), \delta)) = 0$ to obtain

$$-u(0) = \int_0^\delta \mathrm{d}x_d \,\frac{\partial}{\partial x_d} u(y'\psi(x_d), x_d)$$

=
$$\int_0^\delta \mathrm{d}x_d \,\left(\sum_{j=1}^{d-1} (\partial_j u)(y'\psi(x_d), x_d)y'_j\psi'(x_d) + (\partial_d u)(y'\psi(x_d), x_d)\right) \,.$$

Here ψ' denotes the derivative of ψ while $y' \in \mathbb{R}^{d-1}$. Integrating over $y' \in B_1^{(d-1)}(0)$ and using the notation $\omega_{d-1} = |B_1^{(d-1)}(0)|$, we get

$$\begin{split} &-\omega_{d-1}u(0)\\ &= \int_{B_1^{(d-1)}(0)} \mathrm{d}y' \int_0^\delta \mathrm{d}x_d \left(\sum_{j=1}^{d-1} (\partial_j u) (y'\psi(x_d), x_d) \frac{y'_j\psi(x_d)}{\psi(x_d)} \psi'(x_d) + (\partial_d u) (y'\psi(x_d), x_d) \right) \\ &= \int_K \mathrm{d}x \, \frac{1}{\psi(x_d)^{d-1}} \left(\sum_{j=1}^{d-1} (\partial_j u) (x) \frac{x'_j}{\psi(x_d)} \psi'(x_d) + (\partial_d u) (x) \right) \,, \end{split}$$

where we used the change of variables $x = (y'\psi(x_d), x_d)$. Note that

$$\psi'(s) = \frac{1}{c^{\frac{1}{\gamma}}} \frac{1}{\gamma} s^{\frac{1}{\gamma}-1}.$$
(72)

Thus, if $x = (x', x_d) \in K$, then using $x_d < \delta \in (0, 1)$, we get

$$\left|\frac{x'_{j}}{\psi(x_{d})}\psi'(x_{d})\right| \leq \left|\psi'(x_{d})\right| = \frac{1}{c^{\frac{1}{\gamma}}}\frac{1}{\gamma}x_{d}^{\frac{1}{\gamma}-1} \leq \frac{1}{c^{\frac{1}{\gamma}}}\frac{1}{\gamma}\delta^{\frac{1}{\gamma}-1} \lesssim 1.$$
(73)

Therefore, by (73) and the Cauchy-Schwarz inequality on \mathbb{R}^d ,

$$\begin{split} \omega_{d-1}|u(0)| &\leq \int_{K} \mathrm{d}x \, \frac{1}{\psi(x_{d})^{d-1}} \sum_{j=1}^{d} |(\partial_{j}u)(x)| \lesssim \int_{K} \mathrm{d}x \, \frac{1}{\psi(x_{d})^{d-1}} \sqrt{\sum_{j=1}^{d} |(\partial_{j}u)(x)|^{2} \sqrt{d}} \\ &\lesssim \int_{K} \mathrm{d}x \, \frac{1}{\psi(x_{d})^{d-1}} |\nabla u(x)| \,. \end{split}$$

We get

$$|u(0)| \lesssim \int_{K} \mathrm{d}x \, \frac{1}{\psi(x_d)^{d-1}} |\nabla u(x)| \,, \tag{74}$$

where the constant in the inequality only depends on d, γ , c. Let us now undo the change of variables, which was convenient for the above computation. In the old coordinate system (74) reads

$$|u(\tilde{x})| \lesssim \int_{\tilde{D}} \mathrm{d}y \, \mathbf{1}_{K}(\tilde{x}-y) \frac{1}{\psi(\tilde{x}_{d}-y_{d})^{d-1}} |\nabla u(y)| = \left(\left(\mathbf{1}_{K} \frac{1}{\psi(\cdot_{d})^{d-1}} \right) * |\nabla u| \right) (\tilde{x})$$
(75)

for all $\tilde{x} \in D$. Here we used the rotational symmetry of K with respect to the first d-1 variables. We define $r \in (1, \infty)$ by

$$1 + \frac{1}{q^*} = 1 + \frac{1}{2} - \frac{1}{\frac{d-1}{\gamma} + 1} =: \frac{1}{2} + \frac{1}{r}.$$
(76)

By the weak Young inequality and (74), we get

$$\|u\|_{q^*} \lesssim \|\nabla u\|_2 \left\| 1_K \frac{1}{\psi(\cdot_d)^{d-1}} \right\|_{r,w}.$$
 (77)

Here $\|\cdot\|_{r,w}$ denotes the weak L^r -norm. From the definitions of ψ and K in (70) and (71), we have

$$\left\| 1_{K} \frac{1}{\psi(\cdot_{d})^{d-1}} \right\|_{r,w} \lesssim 1, \tag{78}$$

which completes the proof for smooth u. The claim for general u follows a standard density argument.

Lemma 3.2 (Poincaré inequality for oscillatory domains with Neumann boundary conditions). Let $\delta > 0$, let $\hat{a} \in (\frac{\delta}{3}, \delta)$ and let $f : [-\hat{a}/2, \hat{a}/2]^{d-1} \to [\delta/4, \delta)$ be continuous. Define

$$\hat{D} := \left\{ (x', x_d) \in \left[-\frac{\hat{a}}{2}, \frac{\hat{a}}{2} \right]^{d-1} \times \left(-\frac{\delta}{4}, \delta \right) \, \middle| \, f(x') > x_d \right\}.$$
(79)

Then there exists a constant $C_P = C_P(d) > 0$ such that for all $u \in H^1(\hat{D})$ with $\int_{\hat{D}} u = 0$, we have

$$C_P \frac{1}{\delta^2} \|u\|_{2,\hat{D}}^2 \le \|\nabla u\|_{2,\hat{D}}^2 .$$
(80)

Proof. The proof can be found in [10, Lemma 2.6(2)] for slightly different side lengths of the domain. \Box

Corollary 3.3 (Poincaré-Sobolev inequality for oscillatory domains). Let $M \ge 1$, $\delta \in (0,1)$ and let $f: [-\delta/(2M), \delta/(2M)]^{d-1} \to [\delta/4, \delta)$ be such that

$$|f(x') - f(y')| \le c |x' - y'|_{\infty}^{\gamma} \text{ for all } x', y' \in \left[-\frac{\delta}{2M}, \frac{\delta}{2M}\right]^{d-1}$$
(81)

for some $c > 0, \gamma \in (0,1)$. Define $q^* \in (2,\infty)$ by

$$\frac{1}{q^*} = \frac{1}{2} - \frac{1}{\frac{d-1}{\gamma} + 1} \,. \tag{82}$$

and define

$$D := \left\{ (x', x_d) \in \left(-\frac{\delta}{2M}, \frac{\delta}{2M} \right)^{d-1} \times (0, \delta) \mid f(x') > x_d \right\}.$$
(83)

Then there exists a constant $C_{PS} = C_{PS}(d, c, \gamma) > 0$ such that for all $u \in H^1(D)$ with $\int_D u = 0$, we have

$$\|u\|_{q^*,D}^2 \le C_{PS} M^{(d-1)\left(1-\frac{2}{q^*}\right)} \|\nabla u\|_{2,D}^2 .$$
(84)

Proof. We combine Lemma 3.1 and Lemma 3.2. Define \tilde{M} as the largest odd number such that $\tilde{M} \leq M$ and note that $\tilde{M} \leq M \leq 3\tilde{M}$, so

$$\hat{a} := \frac{\tilde{M}\delta}{M} \in \left(\frac{\delta}{3}, \delta\right) \,. \tag{85}$$

We define

$$\tilde{f}: \left[-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M}\right]^{d-1} \to \left[\frac{\delta}{4}, \delta\right)$$
(86)

by reflecting f: We can write every $x' \in \left[-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M}\right]^{d-1}$ as

$$x' = \frac{\delta}{M}z' + w' \tag{87}$$

with
$$z' = (z_1, \dots, z_{d-1}) \in \mathbb{Z}^{d-1}$$
 and $w' = (w_1, \dots, w_{d-1}) \in \left[-\frac{\delta}{2M}, \frac{\delta}{2M}\right]^{d-1}$. Define \tilde{f} by
 $\tilde{f}(x') := f\left(\left((-1)^{z_1}w_1, \dots, (-1)^{z_{d-1}}w_{d-1}\right)\right)$.
(88)

Note that \tilde{f} is well defined, continuous and

$$\left|\tilde{f}(x') - \tilde{f}(y')\right| \le c \left|x' - y'\right|_{\infty}^{\gamma} \text{ for all } x', y' \in \left[-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M}\right]^{d-1}.$$
(89)

Define

$$\tilde{D} := \left\{ (x', x_d) \in \left(-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M} \right)^{d-1} \times \left(-\frac{\delta}{4}, \delta \right) \left| \tilde{f}(x') > x_d \right\}$$
(90)

and

$$\hat{D} := \left\{ (x', x_d) \in \left(-\frac{\tilde{M}\delta}{2M}, \frac{\tilde{M}\delta}{2M} \right)^{d-1} \times (0, \delta) \left| \tilde{f}(x') > x_d \right\}.$$
(91)

Note that \hat{D} consists of \tilde{M}^{d-1} reflected copies of D. Moreover, \tilde{D} consists of less than $2(3\tilde{M})^{d-1}$ reflected copies of D in the sense that $\tilde{D} \cap \{x_d \geq 0\}$ consists of $(3\tilde{M})^{d-1}$ reflected copies of D but $\tilde{D} \cap \{x_d < 0\}$ is only contained in $(3\tilde{M})^{d-1}$ reflected copies of D. Let

$$\begin{split} \varphi \in C_c^\infty \left(\left(-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M} \right)^{d-1} \times \left(-\frac{\delta}{4}, \frac{5\delta}{4} \right) \right) \\ \text{with } 0 &\leq \varphi \leq 1 \,, \ \varphi \restriction_{\left[-\frac{\tilde{M}\delta}{2M}, \frac{\tilde{M}\delta}{2M} \right]^{d-1}} \times [0,\delta]} \equiv 1 \ \text{and} \ \|\nabla\varphi\|_\infty \lesssim \frac{1}{\delta} \,. \end{split}$$

It is possible to choose such a φ by scaling. Note that $\varphi \upharpoonright_{\hat{D}} \equiv 1$ by the definition of \hat{D} . Let $u \in H^1(D)$ with $\int_D u = 0$ and define the corresponding reflected version $\tilde{u} \in H^1(\tilde{D})$ by

$$\tilde{u}(x) := u\left(\left((-1)^{z_1} w_1, \dots, (-1)^{z_{d-1}} w_{d-1}, |w_d|\right)\right) \,. \tag{92}$$

for every $x = \frac{\delta}{M}z + w \in \tilde{D}$ with $z = (z_1, \ldots, z_{d-1}, 0) \in \mathbb{Z}^d$ and $w = (w_1, \ldots, w_d) \in \left[-\frac{\delta}{2M}, \frac{\delta}{2M}\right]^{d-1} \times \left(-\frac{\delta}{4}, \infty\right)$. Note that \tilde{u} is well-defined because H^1 functions are defined up to almost everywhere equality and since reflections of H^1 functions are again H^1 functions. Moreover, by $\tilde{u} \upharpoonright_D \equiv u$, we know that $\tilde{u} \upharpoonright_{\tilde{D}}$ consists of less than $2(3\tilde{M})^{d-1}$ reflected copies of

u, and $\tilde{u} \upharpoonright_{\hat{D}}$ consists of \tilde{M}^{d-1} reflected copies of u. Also note that $\int_{\hat{D}} \tilde{u} = 0$. Furthermore, $\varphi \tilde{u} \in H^1(\tilde{D}), \ \varphi \tilde{u} \upharpoonright_{\hat{D}} = \tilde{u}$ and $\varphi \tilde{u} \upharpoonright_{\tilde{B}} \equiv 0$, where

$$\tilde{B} := \partial \tilde{D} \setminus \left\{ (x', x_d) \in \left[-\frac{3\tilde{M}\delta}{2M}, \frac{3\tilde{M}\delta}{2M} \right]^{d-1} \times \left(-\frac{\delta}{4}, \delta \right] \middle| \tilde{f}(x') = x_d \right\}.$$
(93)

Since $\int_{\tilde{D}} \tilde{u} = 0$, $\hat{a} \in \left(\frac{\delta}{3}, \delta\right)$ and \tilde{D} satisfies the assumptions of Lemma 3.2, we can apply Lemma 3.2 to get

$$\begin{split} \|\nabla(\varphi\tilde{u})\|_{2,\tilde{D}}^2 &\lesssim \int_{\tilde{D}} |\varphi|^2 |\nabla\tilde{u}|^2 + \int_{\tilde{D}} |\tilde{u}|^2 |\nabla\varphi|^2 \lesssim \int_{\tilde{D}} |\nabla\tilde{u}|^2 + \frac{1}{\delta^2} \int_{\tilde{D}} |\tilde{u}|^2 \\ &\leq 2 \cdot 3^{d-1} \left(\int_{\hat{D}} |\nabla\tilde{u}|^2 + \frac{1}{\delta^2} \int_{\hat{D}} |\tilde{u}|^2 \right) \lesssim \int_{\hat{D}} |\nabla\tilde{u}|^2 = \tilde{M}^{d-1} \int_{D} |\nabla u|^2 \,. \end{split}$$

In the third step we used that $0 \leq \varphi \leq 1$ and $\|\nabla \varphi\|_{\infty} \lesssim \frac{1}{\delta}$, and in the second last step we used that $\tilde{u} \upharpoonright_{\hat{D}}$ consists of \tilde{M}^{d-1} reflected copies of u. On the other hand, $\varphi \tilde{u} \upharpoonright_{\tilde{B}} \equiv 0$ and \tilde{D} satisfies the assumptions of Lemma 3.1, so by Lemma 3.1, we obtain

$$\begin{split} \|\nabla(\varphi \tilde{u})\|_{2,\tilde{D}}^2 \gtrsim \|\varphi \tilde{u}\|_{q^*,\tilde{D}}^2 \ge \|\tilde{u}\|_{q^*,\hat{D}}^2 = \left(\int_{\hat{D}} |\tilde{u}|^{q^*}\right)^{\frac{2}{q^*}} = \left(\tilde{M}^{d-1} \int_{D} |\tilde{u}|^{q^*}\right)^{\frac{2}{q^*}} \\ &= \tilde{M}^{(d-1)\frac{2}{q^*}} \left(\int_{D} |\tilde{u}|^{q^*}\right)^{\frac{2}{q^*}} = \tilde{M}^{(d-1)\frac{2}{q^*}} \|u\|_{q^*,D}^2 \,. \end{split}$$

To sum up, we get

$$\tilde{M}^{d-1} \int_{D} |\nabla u|^2 \gtrsim \|\nabla(\varphi \tilde{u})\|_{2,\tilde{D}}^2 \gtrsim \tilde{M}^{(d-1)\frac{2}{q^*}} \|u\|_{q^*,D}^2 , \qquad (94)$$

 \mathbf{SO}

$$\tilde{M}^{(d-1)\left(1-\frac{2}{q^*}\right)} \int_D |\nabla u|^2 \gtrsim ||u||_{q^*,D}^2 .$$
(95)

Now recall that $M \sim 3\tilde{M}$, so we obtain the desired result.

3.2. Choice of the oscillatory domains. In this subsection, we choose depending on V for every $x \in \Omega$ close to $\partial\Omega$ an oscillatory domain D_x with centre x such that

$$N\left(-\Delta_{D_x}^N + V\right) \le 1.$$
(96)

For the proof, we use the Poincaré-Sobolev inequality for oscillatory domains (Corollary 3.3). At the same time, we choose the oscillatory domains D_x such that a certain norm of V on D_x is not too small. This will be needed in the following subsection.

Lemma 3.4. Let $l \in \{1, \ldots, L\}$ and let $x \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$. Let $\delta \in (0, \delta_0]$ and let $D := D_x(\delta)$. (i) Suppose $a_x(\delta) = \delta$ and $\|V\|_{\frac{d}{2},D} \lesssim 1$ if $d \ge 3$ and $\|V\|_{\mathcal{B},D} \lesssim 1$ if $d \ge 2$. Then

$$N\left(-\Delta_D^N + V\right) \le 1.$$
(97)

(ii) Suppose $a_x(\delta) = c_0 \max(h_x, c_1\delta)^{\frac{1}{\gamma}}$ and

$$\|V\|_{\tilde{p},D}^{\tilde{p}} \lesssim \max\left(\frac{h_x}{c_1\delta}, 1\right)^{\frac{d-1}{\gamma}} .$$
(98)

Then

$$N\left(-\Delta_D^N + V\right) \le 1.$$
⁽⁹⁹⁾

All the constants in \leq in this Lemma only depend on d, γ, c .

15

Proof. **Proof of (i).** Let $0 \neq u \in H^1(D)$ with $\int_D u = 0$. Then using Hölder's inequality and the Poincaré-Sobolev inequality for cubes, see [18, Theorem 8.12], we get for $d \geq 3$

$$\int_{D} |\nabla u|^{2} + \int_{D} V|u|^{2} \ge \int_{D} |\nabla u|^{2} - \|V\|_{\frac{d}{2},D} \|u\|_{\frac{2d}{d-2},D}^{2} \ge \|\nabla u\|_{2,D}^{2} \left(1 - C_{PS} \|V\|_{\frac{d}{2},D}\right).$$

Hence, since $\nabla u \neq 0$, we get $\int_D |\nabla u|^2 + \int_D V |u|^2 > 0$ if $||V||_{\frac{d}{2},D} < \frac{1}{C_{PS}}$. If d = 2, we use the Poincaré-Sobolev inequality for Orlicz norms, see [14, Proposition 2.1].

Proof of (ii). Choose $M := \delta/a$ and note $M \ge 1$. By Lemma 2.7(ii), we know that D is an oscillatory domain as in Corollary 3.3. In order to show $N\left(-\Delta_D^N+V\right) \leq 1$, it suffices to show that for all $0 \neq u \in H^1(D)$ with $\int_D u = 0$, we have

$$\int_{D} |\nabla u|^2 + \int_{D} V|u|^2 > 0.$$
(100)

To this end, let $0 \not\equiv u \in H^1(D)$ with $\int_D u = 0$. Define p^* by

$$1 = \frac{1}{p^*} + \frac{1}{\frac{q^*}{2}}, \quad \text{namely} \quad \frac{1}{p^*} := 1 - \frac{2}{q^*} = \frac{2}{\frac{d-1}{\gamma} + 1}.$$
 (101)

By Hölder's inequality and the Poincaré-Sobolev inequality for oscillatory domains, see Corollary 3.3, we get

$$\int_{D} |\nabla u|^{2} + \int_{D} V|u|^{2} \ge \int_{D} |\nabla u|^{2} - \|V\|_{p^{*},D} \|u\|_{q^{*},D}^{2} \ge \|\nabla u\|_{2,D}^{2} \left(1 - C_{PS}M^{\frac{d-1}{p^{*}}} \|V\|_{p^{*},D}\right).$$
Hence, since $\nabla u \ne 0$, the left-hand side is strictly positive if

Hence, since $\nabla u \not\equiv 0$, the left-hand side is strictly positive if

$$M^{\frac{d-1}{p^*}} \|V\|_{p^*,D} < \frac{1}{C_{PS}}.$$
(102)

We define $\tilde{r} \in (1, \infty)$ by

$$\frac{1}{\tilde{r}} := \frac{1}{p^*} - \frac{1}{\tilde{p}} = \frac{2(d-1)}{\left(\frac{d-1}{\gamma} + 1\right)^2} \left(\frac{1}{\gamma} - 1\right) \,, \tag{103}$$

where p^* and \tilde{p} are given in (101) and (22). Using Hölder's inequality with (103), $|D| \sim$ $\delta^d M^{-(d-1)}$ and $M = \delta/a$, it follows that

$$\begin{split} M^{\frac{d-1}{p^*}} \|V\|_{p^*,D} &\lesssim M^{\frac{d-1}{p^*}} \|V\|_{\tilde{p},D} |D|^{\frac{1}{\tilde{r}}} \lesssim M^{\frac{d-1}{p^*}} \|V\|_{\tilde{p},D} \left(\delta^d M^{-(d-1)}\right)^{\frac{1}{\tilde{r}}} \\ &\sim \|V\|_{\tilde{p},D} \,\delta^{\frac{d}{\tilde{r}}} \left(\delta^{1-\frac{1}{\gamma}} \max\left(\frac{h}{c_1\delta},1\right)^{-\frac{1}{\gamma}}\right)^{\frac{d-1}{\tilde{p}}} = \left(\|V\|_{\tilde{p},D}^{\tilde{p}} \max\left(\frac{h}{c_1\delta},1\right)^{-\frac{d-1}{\gamma}}\right)^{\frac{1}{\tilde{p}}} \lesssim 1\,, \end{split}$$

where we used that

$$\frac{d}{\tilde{r}} + \left(1 - \frac{1}{\gamma}\right)\frac{d-1}{\tilde{p}} = 0 \tag{104}$$

in the third step and the assumption (98) in the last step. Hence, if the constant in (98)is chosen small enough, we can deduce (102), which is what we wanted to show.

Lemma 3.5 (Choice of the oscillatory domains). Let $l \in \{1, ..., L\}$ and let $x \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$. Then there exists $\delta_x \in (0, \delta_0]$ such that for $D := D_x := D_x(\delta_x)$ we have

$$N\left(-\Delta_D^N + V\right) \le 1 \tag{105}$$

and at least one of the following properties is satisfied: (1) $\delta_x = \delta_0$.

(2) $a_x = \delta_x$ and

$$\|V\|_{\frac{d}{2},D}^{\frac{d}{2}} \gtrsim 1 \text{ if } d \ge 3 \quad and \quad \|V\|_{\mathcal{B},D} \gtrsim 1 \text{ if } d = 2.$$
 (106)

(3) $a_x = c_0 \max(h_x, c_1 \delta_x)^{\frac{1}{\gamma}}$ and

$$\|V\|_{\tilde{p},D}^{\tilde{p}} \gtrsim \max\left(\frac{h_x}{c_1\delta_x}, 1\right)^{\frac{d-1}{\gamma}}.$$
(107)

All the constants in \gtrsim in this Lemma only depend on d, γ , c.

Proof. Let us first explain how to choose δ if $c_0 \max(h, c_1 \delta_0)^{\frac{1}{\gamma}} > \delta_0$. It follows that $a(\delta) = \delta$ for all $\delta \leq \delta_0$. Pick $\delta \leq \delta_0$ such that

$$\|V\|_{\frac{d}{2},D(\delta)}^{\frac{d}{2}} \sim 1 \text{ if } d \ge 3 \text{ and } \|V\|_{\mathcal{B},D(\delta)} \sim 1 \text{ if } d = 2.$$
 (108)

holds, so (2) is satisfied. By (108) and Lemma 3.4(i), we get $N\left(-\Delta_D^N+V\right) \leq 1$.

Let us now assume that $c_0 \max(h, c_1 \delta_0)^{\frac{1}{\gamma}} \leq \delta_0$. Define $\delta_c := c_0 h^{\frac{1}{\gamma}} \leq \delta_0$, so $c_0 \max(h, c_1 \delta_c)^{\frac{1}{\gamma}} = \delta_c$. We have $a(\delta) = c_0 \max(h, c_1 \delta)^{\frac{1}{\gamma}}$ for all $\delta \in [\delta_c, \delta_0]$ and $a(\delta) = \delta$ is for all $\delta \leq \delta_c$. If

$$\|V\|_{\tilde{p},D_x(\delta_0)}^{\tilde{p}} \lesssim \max\left(\frac{h}{c_1\delta_0},1\right)^{\frac{d-1}{\gamma}},\tag{109}$$

pick $\delta := \delta_0$ and note that (1) is satisfied. Furthermore, by Lemma 3.4(ii), we have $N\left(-\Delta_D^N + V\right) \leq 1$. Else, if there exists $\delta \in [\delta_c, \delta_0]$ such that

$$\|V\|_{\tilde{p},D(\delta)}^{\tilde{p}} \sim \max\left(\frac{h}{c_1\delta},1\right)^{\frac{d-1}{\gamma}},\tag{110}$$

where the constant in ~ only depends on d, γ , c, pick this δ . By (110), (3) is satisfied and moreover, we have $N\left(-\Delta_D^N+V\right) \leq 1$ by Lemma 3.4(ii). Else, since the left hand side of (110) is increasing in δ and the right hand side of (110) is decreasing in δ , we have (107) for $\delta = \delta_c$. Now if $d \geq 3$ and $\|V\|_{\frac{d}{2}, D_x(\delta_c)} \leq 1$ or if d = 2 and $\|V\|_{\mathcal{B}, D_x(\delta_c)} \leq 1$, pick $\delta := \delta_c$ and note that (3) is satisfied and $N\left(-\Delta_D^N+V\right) \leq 1$ by Lemma 3.4(i). Else, pick $\delta < \delta_c$ such that (108) holds. Thus, (2) is satisfied and $N\left(-\Delta_D^N+V\right) \leq 1$ by Lemma 3.4(i).

4. Covering Lemmas

4.1. Covering of the part close to the boundary by oscillatory domains. In this subsection, we prove a Besicovitch type covering lemma for oscillatory domains. It is one of the key ingredients of the proof of Theorem 1.2 since it allows us to choose a family of oscillatory domains as in the previous subsection, which cover the part of Ω close to $\partial\Omega$, but which do not overlap too much. Using this result, we show that the number of oscillatory domains we choose is bounded by a constant times δ_0^{-d} .

Lemma 4.1 (Covering lemma). Let $l \in \{1, \ldots, L\}$ and suppose that for every $x \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$ we are given a $\delta_x \in (0, \delta_0]$. We define the oscillatory domains $D_x := D_x(\delta_x)$ as in Definition 2.6. Then there exists $K_l = K_l(d, \gamma) \in \mathbb{N}$ and subfamilies $\mathcal{F}_1, \ldots, \mathcal{F}_{K_l}$ of oscillatory domains $D_x := D_x(\delta_x) \subset \Omega_l \cap \Omega^b_{\sqrt{d\delta_0}}$ such that

(i) For every $k \in \{1, \ldots, K_l\}$ all oscillatory domains in \mathcal{F}_k are disjoint.

(ii)

$$\bigcup_{k=1}^{K_l} \bigcup_{D \in \mathcal{F}_k} D \supset \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d}\delta_0} \,. \tag{111}$$

Proof. Recall that for every $x \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d}\delta_0}$ we are given a $\delta_x \in (0, \delta_0]$ and the corresponding a_x is given by

$$a_x = \min\left(\delta_x, c_0 \max\left(h_{x,l}, c_1 \delta_x\right)^{\frac{1}{\gamma}}\right).$$
(112)

Furthermore, the oscillatory domain $D_x = D_x(\delta_x)$ is given by

$$D_x := T_l R_l \left\{ (y', y_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid |y' - x'| < \frac{1}{2}a, \ |y_d - x_d| < \frac{1}{2}\delta_x, \ f_l(y') > y_d \right\}.$$
 (113)

Without loss of generality, let us assume that $T_l R_l$ is the identity map. To begin with, we decompose $\tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$ into three parts

$$\tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}} = A_1 \cup A_2 \cup A_3 \tag{114}$$

such that

$$A_1 := \left\{ a_x = c_0 h_x^{\frac{1}{\gamma}} \right\}, \quad A_2 := \left\{ a_x = c_2 \delta_x^{\frac{1}{\gamma}} \right\}, \quad A_3 := \{ a_x = \delta_x \}$$

For each of these sets, we will prove a corresponding covering theorem with a constant K_1^l , K_2^l , K_3^l depending on d, γ . Combining these results, we will get the desired result with $K^l := K_1^l + K_2^l + K_3^l \in \mathbb{N}$.

The beginning of the proof for A_b with $b \in \{1, 2\}$ will be similar to the proof of Besicovich's covering theorem for cubes, see for example. Let $b \in \{1, 2\}$. We will denote families of oscillatory domains D_x with $x \in A_b$ by $(\mathcal{F}_k)_{k \in \mathbb{N}}$. At the beginning of our construction, the \mathcal{F}_k are all assumed to be empty. We put oscillatory domains inside those \mathcal{F}_k according to the following procedure:

First, choose $\tilde{x}_1 \in A_b$ with

$$\tilde{\delta}_1 := \delta_{\tilde{x}_1} \ge \frac{1}{2} \sup_{x \in A_b} \delta_x \tag{115}$$

and denote the corresponding domain by $\tilde{D}_1 := D_{\tilde{x}_1}$. Put \tilde{D}_1 in \mathcal{F}_1 . Then, if possible, choose $\tilde{x}_2 \in A_b \setminus \tilde{D}_1$ such that

$$\tilde{\delta}_2 := \delta_{\tilde{x}_2} \ge \frac{1}{2} \sup_{x \in A_b \setminus \tilde{D}_1} \delta_x \tag{116}$$

and denote the corresponding domain by $\tilde{D}_2 := D_{\tilde{x}_2}$. If $\tilde{D}_2 \cap \tilde{D}_1 = \emptyset$, put \tilde{D}_2 in \mathcal{F}_1 . Otherwise, put \tilde{D}_2 in \mathcal{F}_2 . More generally, if $\tilde{x}_1, \ldots, \tilde{x}_{n-1}$ have already been chosen for some $n \in \mathbb{N}$, we proceed as follows: If

$$\bigcup_{m=1}^{n-1} \tilde{D}_m \supset A_b \,, \tag{117}$$

then stop. Else, we can choose $\tilde{x}_n \in A_b \setminus \bigcup_{m=1}^{n-1} \tilde{D}_m$ such that

$$\tilde{\delta}_n := \delta_{\tilde{x}_n} \ge \frac{1}{2} \sup_{x \in A_b \setminus \bigcup_{m=1}^{n-1} \tilde{D}_m} \delta_x \tag{118}$$

and denote the corresponding domain by $\tilde{D}_n := D_{\tilde{x}_n}$. Put \tilde{D}_n in \mathcal{F}_k , where k is the lowest natural number such that $\tilde{D}_n \cap D = \emptyset$ for all $D \in \mathcal{F}_k$. Note that by construction, we know that all oscillatory domains in each \mathcal{F}_k are disjoint.

We are done if we can show that there exists $K_b^l = K_b^l(d, \gamma) \in \mathbb{N}$ such that $\mathcal{F}_k = \emptyset$ for all $k \ge K_b^l + 1$. For the moment, let us assume we had already shown that.

The oscillatory domains, which we have chosen in the above construction, cover A_b , namely,

$$\bigcup_{n \in \mathbb{N}} \tilde{D}_n \supset A_b \,, \tag{119}$$

where we use the convention that $\tilde{D}_n = \emptyset$ if we have to stop before the n^{th} step in the procedure above. In order to see this, note that (119) is clear from the construction above if we have to stop after a finite number of steps. If the number of steps is infinite, we claim

$$\lim_{n \to \infty} \tilde{\delta}_n = 0.$$
 (120)

By construction, we know that $\tilde{\delta}_n \leq 2\tilde{\delta}_m$ for all $n \geq m$. Thus, in order to show (120), it suffices to show that for every $\varepsilon > 0$ there exists $n \in \mathbb{N}$ with $\tilde{\delta}_n < \varepsilon$. Suppose this was wrong, that is, there exists $\varepsilon > 0$ such that for all $n \in \mathbb{N}$, we have $\tilde{\delta}_n \geq \varepsilon$. It follows that

$$\sum_{n \in \mathbb{N}} |\tilde{D}_n| = \infty \,. \tag{121}$$

On the other hand, we have $\tilde{D}_n \subset \Omega_l \cap \Omega^b_{\sqrt{d}\delta_0}$ for all $n \in \mathbb{N}$, $|\Omega_l \cap \Omega^b_{\sqrt{d}\delta_0}| \leq |\Omega| \leq 1$ and by assumption, all \mathcal{F}_k with $k \geq K_b^l + 1$ are empty. Since the oscillatory domains in each \mathcal{F}_k are disjoint, we get

$$\sum_{n \in \mathbb{N}} |\tilde{D}_n| \le \sum_{k=1}^{K_b^l} |\Omega_l \cap \Omega_{\sqrt{d\delta_0}}^b| \le K_b^l,$$
(122)

which contradicts (121). This proves (120). Now suppose (119) was wrong, that is, there exists $\tilde{x} \in A_b \setminus \bigcup_{n \in N} \tilde{D}_n$. By (120), there exists $n \in \mathbb{N}$ with $\tilde{\delta}_n \leq \frac{1}{4} \delta_{\tilde{x}}$, which contradicts (118), and thereby proves (119).

It remains to show that there exists $K_b^l = K_b^l(d, \gamma) \in \mathbb{N}$ such that $\mathcal{F}_k = \emptyset$ for all $k \geq K_b^l + 1$. Let $m \in \mathbb{N}$ with $\mathcal{F}_m = \emptyset$. Let $D_m \in \mathcal{F}_m$. By the above construction, for every $k \in \{1, \ldots, m-1\}$ there exists $D_k \in \mathcal{F}_k$ with $D_k \cap D_m \neq \emptyset$ and such that D_1, \ldots, D_{m-1} were all chosen before D_m in the construction. By relabelling the families \mathcal{F}_k and the corresponding domains D_k , we may without loss of generality assume that D_k was chosen before D_n for all $k, n \in \{1, \ldots, m\}$ with $k \leq n$. We denote the centres of D_1, \ldots, D_m by $x_1, \ldots, x_m \in A_b$, the corresponding δ by $\delta_1, \ldots, \delta_m$ and the corresponding h by h_1, \ldots, h_m . Note that by construction, we have

$$\delta_n \le 2\delta_k \text{ for all } k, n \in \{1, \dots, m\} \text{ with } k \le n.$$
 (123)

Note that the D_k do not agree with the \tilde{D}_k from the construction above but we have $D_k \in \bigcup_{n \in \mathbb{N}} \tilde{D}_n$ for all $k \in \{1, \ldots, m\}$.

Our goal is to show that $m \leq K_b^l$ for some $K_b^l = K_b^l(d, \gamma) \in \mathbb{N}$, and we will show this for b = 1 and b = 2 separately.

Proof of $m \leq K_1^l < \infty$ for A_1 . By Lemma 2.8(i), we note that the $D_k, k \in \{1, \ldots, m\}$

CHARLOTTE DIETZE

are all cuboids contained in Ω . Applying Lemma 2.8(ii) with $x = x_m$, $y = x_1$ and $z = x_k$, we get for all $k \in \{1, \ldots, m-1\}$

$$\left|x_{k}'-x_{m}'\right|_{\infty} < \frac{1}{2}c_{0}h_{k}^{\frac{1}{\gamma}} + \frac{1}{2}c_{0}h_{m}^{\frac{1}{\gamma}} \le \frac{1}{2}c_{0}\left(4h_{m}\right)^{\frac{1}{\gamma}} + \frac{1}{2}c_{0}h_{m}^{\frac{1}{\gamma}} \le \frac{\alpha}{2}c_{0}h_{m}^{\frac{1}{\gamma}}$$
(124)

with $\alpha := \lceil 4^{\frac{1}{\gamma}} + 1 \rceil \in \mathbb{N}$. Now define the lattice \mathcal{G} by

$$\mathcal{G} := \left\{ (g', g_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid \begin{array}{c} g_d - (x_m)_d \in \left\{ -\frac{\delta_m}{2}, -\frac{\delta_m}{4}, 0, \frac{\delta_m}{4}, \frac{\delta_m}{2} \right\} \text{ and} \\ g' - x'_m = \frac{j}{4\alpha} c_0 h_m^{\gamma} \text{ for some } j \in \mathbb{Z}^{d-1} \text{ with } |j|_{\infty} \le 2\alpha^2 \end{array} \right\}.$$

Here $(x_m)_d$ is the d^{th} coordinate of x_m . Note that

$$|\mathcal{G}| = 5 \cdot \left(2 \cdot 2\alpha^2 + 1\right)^{d-1}, \qquad (125)$$

which only depends on d, γ . For every $k \in \{1, \ldots, m-1\}$, we associate a $g \in \mathcal{G}$ to x_k , which is chosen such that

$$|g' - x'_k|_{\infty} + |g_d - (x_k)_d|$$
(126)

is minimal among all $g \in \mathcal{G}$. Note that by the choice of \mathcal{G} , if $g \in \mathcal{G}$ is associated to x_k , then by (124), $h_m \leq 4h_k$ and the definition of α , we obtain

$$|x'_{k} - g'|_{\infty} \le \frac{1}{8\alpha} c_{0} \left(4h_{k}\right)^{\frac{1}{\gamma}} \le \frac{1}{8} c_{0} h_{k}^{\frac{1}{\gamma}} .$$
(127)

Furthermore, we claim that

$$|g_d - (x_k)_d| < \frac{1}{2}\delta_k \,. \tag{128}$$

If $|(x_m)_d - (x_k)_d| \ge \frac{1}{2}\delta_m$, then by the triangle inequality and $D_m \cap D_k \ne \emptyset$, we get (128). If $|(x_m)_d - (x_k)_d| < \frac{1}{2}\delta_m$, we use the definition of \mathcal{G} and $2\delta_k \ge \delta_m$ to get (128). Combining (127) and (128), we obtain that if $g \in \mathcal{G}$ is associated to x_k , then $g \in D_k$.

Now, if $g \in \mathcal{G}$ is associated to both x_k and x_n for $k, n \in \{1, \ldots, m-1\}$ with k < n, then by (127) and $h_m \leq 4h_k$,

$$|x'_{k} - x'_{n}|_{\infty} \le |x'_{k} - g'|_{\infty} + |g' - x'_{n}|_{\infty} \le \frac{1}{8\alpha}c_{0}h_{m}^{\frac{1}{\gamma}} + \frac{1}{8\alpha}c_{0}h_{m}^{\frac{1}{\gamma}} \le \frac{1}{4}c_{0}h_{k}^{\frac{1}{\gamma}}$$
(129)

where we used $\alpha := \lfloor 4^{\frac{1}{\gamma}} + 1 \rfloor$ in the last step. Since k < n, we have $x_n \notin D_k$ by construction. Using (129) and $x_n \notin D_k$, we deduce that

$$|(x_n)_d - (x_k)_d| \ge \frac{1}{2}\delta_k.$$
 (130)

Claim. Fix $g \in \mathcal{G}$. Then there are at most two indices $k, n \in \{1, \ldots, m-1\}$ with $k \neq n$ such that $(x_k)_d \geq g_d$ and $(x_n)_d \geq g_d$ and such that g is associated to both x_k and x_n .

Proof of the claim. Suppose there were $k, n, j \in \{1, \ldots, m-1\}$ with k < n < j such that g is associated to x_k , x_n and x_j and such that $(x_k)_d \ge g_d$, $(x_n)_d \ge g_d$ and $(x_j)_d \ge g_d$. Without loss of generality, let us assume that $g_d = 0$. By $g \in D_i$ for all $i \in \{k, n, j\}$, we have

$$0 \le (x_i)_d < \frac{1}{2}\delta_i \tag{131}$$

By (130) and k < n, we have

$$(x_n)_d \ge (x_k)_d + \frac{1}{2}\delta_k$$
 (132)

Note that $0 \leq (x_n)_d \leq (x_k)_d$ is not possible since this would imply $x_n \in D_k$ by (129), $g \in D_k$ and the fact that D_k is a cuboid. Similarly, we find that

$$(x_j)_d \ge (x_n)_d + \frac{1}{2}\delta_n$$
 (133)

Combining (131), (132) and (133), we obtain

$$\frac{1}{2}\delta_j > (x_j)_d \ge (x_n)_d + \frac{1}{2}\delta_n \ge (x_k)_d + \frac{1}{2}\delta_k + \frac{1}{2}\delta_n \ge \frac{1}{2}\delta_j , \qquad (134)$$

which is a contradiction. In the last step we used that $2\delta_k \ge \delta_j$ and $2\delta_n \ge \delta_j$ since k, n < j. This finishes the proof of the claim.

Similarly, we can show that there are at most two indices $k, n \in \{1, \ldots, m-1\}$ with $k \neq n$ such that $(x_k)_d \leq g_d$ and $(x_n)_d \leq g_d$ and such that g is associated to both x_k and x_n . Hence, for every $g \in \mathcal{G}$ there exist at most four different indices $j_1, j_2, j_3, j_4 \in \{1, \ldots, m-1\}$ such that g is associated to x_{j_i} for all i = 1, 2, 3, 4. This shows that $m - 1 \leq 4|\mathcal{G}|$, where the right hand side only depends on d, γ . It follows that a possible choice is

$$K_1^l := 4|\mathcal{G}| + 1. \tag{135}$$

Proof of $m \leq K_2^l < \infty$ for A_2 . Define the lattice \mathcal{G} by

$$\mathcal{G} := \left\{ (g', g_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid \begin{array}{c} g_d - (x_m)_d \in \left\{ -\frac{\delta_m}{2}, -\frac{\delta_m}{4}, 0, \frac{\delta_m}{4}, \frac{\delta_m}{2} \right\} \text{ and} \\ g' - x'_m = \frac{j}{2\alpha} c_2 \delta_m^{\gamma} \text{ for some } j \in \mathbb{Z}^{d-1} \text{ with } |j|_{\infty} \le \alpha \end{array} \right\},$$

where $\alpha := \left\lceil 2^{\frac{1}{\gamma}} \right\rceil \in \mathbb{N}$. Note that $\mathcal{G} \subset \overline{D_m}$ and

$$\mathcal{G}| = 5 \cdot (2\alpha + 1)^{d-1} , \qquad (136)$$

which only depends on d, γ . For every $k \in \{1, \ldots, m-1\}$, we associate a $g \in \mathcal{G}$ to x_k , which is chosen such that

$$|g' - x'_k|_{\infty} + |g_d - (x_k)_d|$$
(137)

is minimal among all $g \in \mathcal{G}$. We may also say that x_k is associated to $g \in \mathcal{G}$. Note that if $g \in \mathcal{G}$ is associated to x_k , then

$$|g_d - (x_k)_d| < \frac{1}{2}\delta_k \,. \tag{138}$$

If $x_k \notin D_m$, this follows from $D_m \cap D_k \neq \emptyset$ and $\mathcal{G} \subset \overline{D_m}$. If $x_k \in D_m$, this follows from $\delta_m \leq 2\delta_k$ and the choice of \mathcal{G} . Furthermore, we have

$$|x'_k - g'|_{\infty} \le \frac{1}{2} c_2 \delta_k^{\frac{1}{\gamma}}.$$
 (139)

If $x_k \notin D_m$, this follows from $D_m \cap D_k \neq \emptyset$ and $\mathcal{G} \subset \overline{D_m}$. If $x_k \in D_m$, this follows from

$$\frac{1}{4\alpha}c_2\delta_m^{\frac{1}{\gamma}} \le \frac{1}{4}c_2\left(\frac{\delta_m}{2}\right)^{\frac{1}{\gamma}} \le \frac{1}{4}c_2\delta_k^{\frac{1}{\gamma}}.$$
(140)

Combining (138) and (139), we get $g \in D_k$.

Fix $g \in \mathcal{G}$ and consider all $k \in \{1, \ldots, m-1\}$ such that g is associated to x_k . Without loss of generality, we may assume that we chose our coordinate system in such a way that $g = (0, \ldots, 0)$. For each coordinate $i \in \{1, \ldots, d\}$, we distinguish between two cases:

(0)
$$(x_k)_i \ge 0$$
 and (1) $(x_k)_i < 0$,

and associate a $\sigma = (\sigma_1, \ldots, \sigma_d) \in \{0, 1\}^d$ to x_k such that condition (σ_i) is satisfied for $(x_k)_i$ for all $i \in \{1, \ldots, d\}$. Fix $\sigma \in \{0, 1\}^d$ and assume without loss of generality that

CHARLOTTE DIETZE

 $\sigma = (0, \ldots, 0)$. Otherwise, rotate and reflect the coordinate system accordingly. In the following, we would like to count the number of x_k such that x_k is associated to g and to $\sigma = (0, \ldots, 0)$.

Let $k_1 \in \{1, \ldots, m-1\}$ be the smallest number such that x_{k_1} is associated to g and to $\sigma = (0, \ldots, 0)$. For $n \in \mathbb{N}$, if it exists, let $k_n \in \{1, \ldots, m-1\}$ be the n^{th} smallest number such that x_{k_n} is associated to g and to $\sigma = (0, \ldots, 0)$. By construction, we have $\delta_{k_n} \leq 2\delta_{k_1}$ for all $n \geq 1$, and moreover, we also have $g \in D_{k_n}$ as we have noticed before. Hence, with the notation $S_{\sigma} := [0, \infty)^d$, we have by $\delta_{k_n} \leq 2\delta_{k_1}$

$$|D_{k_n} \cap S_{\sigma}| \le \left| \left[0, c_2 \delta_{k_n}^{\frac{1}{\gamma}} \right]^{d-1} \times [0, \delta_{k_n}] \right| \le 2^{\frac{d-1}{\gamma} + 1} \left(c_2 \delta_{k_1}^{\frac{1}{\gamma}} \right)^{d-1} \delta_{k_1} \tag{141}$$

for all $n \ge 1$. Let $n \ge 1$. We claim that $\delta_{k_n} \ge \delta_{k_1}$. To see this, recall that $x_{k_n} \in S_{\sigma}$ but $x_{k_n} \notin D_{k_1}$ since $k_n > k_1$. Therefore, there exists $i_0 \in \{1, \ldots, d\}$ with

$$(x_{k_n})_{i_0} \ge (x_{k_1})_{i_0} + \frac{1}{2}c_2\delta_{k_1}^{\frac{1}{\gamma}} \text{ if } i_0 \in \{1, \dots, d-1\}$$
(142)

and

$$(x_{k_n})_d \ge (x_{k_1})_d + \frac{1}{2}\delta_{k_1} \text{ if } i_0 = d.$$
 (143)

At the same time, since $g \in D_{k_n}$, we have

$$(x_{k_n})_i \le \frac{1}{2} c_2 \delta_{k_n}^{\frac{1}{\gamma}} \text{ for all } i \in \{1, \dots, d-1\}$$
 (144)

and $(x_{k_n})_d \leq \frac{1}{2}\delta_{k_n}$. We deduce that $\delta_{k_n} \geq \delta_{k_1}$. For all $n \geq 1$, we have by $\delta_{k_n} \geq \delta_{k_1}$ and $x_n \notin D_j$ for j < n,

$$\left| \left(D_{k_n} \cap S_{\sigma} \right) \setminus \bigcup_{j=1}^{n-1} D_{k_j} \right| \ge \frac{1}{2^d} \left(c_2 \delta_{k_1}^{\frac{1}{\gamma}} \right)^{d-1} \delta_{k_1} \,. \tag{145}$$

Now assume that x_{k_1}, \ldots, x_{k_1} are all associated to g and to $\sigma = (0, \ldots, 0)$. By (145),

$$\left|S_{\sigma} \cap \bigcup_{n=1}^{N} D_{k_n}\right| = \sum_{n=1}^{N} \left| (D_{k_n} \cap S_{\sigma}) \setminus \bigcup_{j=1}^{n-1} D_{k_j} \right| \ge N \frac{1}{2^d} \left(c_2 \delta_{k_1}^{\frac{1}{\gamma}} \right)^{d-1} \delta_{k_1}.$$
(146)

On the other hand, we have (141) and thus,

$$N \le 2^{\frac{d-1}{\gamma} + 1 + d}.$$
 (147)

Repeating the same argument for every $g \in \mathcal{G}$ and every $\sigma \in \{0, 1\}^d$, we find by (136) that

$$K_2^l := 5 \cdot (2\alpha + 1)^{d-1} \cdot 2^{\frac{d-1}{\gamma} + 1 + 2d} + 1 \in \mathbb{N}$$
(148)

is a possible choice. Note that the right-hand side only depends on d, γ .

Covering theorem for A_3 . This is simply the Besicovitch covering theorem for cubes and here $K_3^l = K_3^l(d) \in \mathbb{N}$.

Definition 4.2 (Choice of a subfamily of oscillatory domains). Let $l \in \{1, ..., L\}$ and for every $x \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$ let $\delta_x \in (0, \delta_0]$, and hence also $D_x := D_x(\delta_x)$, be chosen as in Lemma 3.5. Now apply Lemma 4.1 to this collection of oscillatory domains to get subfamilies $\mathcal{F}_1, \ldots, \mathcal{F}_{K_l}$ of oscillatory domains, where $K_l = K_l(d, \gamma) \in \mathbb{N}$. Write

$$\bigcup_{k=1}^{K_l} \mathcal{F}_k =: \{D_j\}_{j \in J} .$$
(149)

where $J = J_1 \cup J_2 \cup J_3$ is an index set such that for every $m \in \{1, 2, 3\}$ and all $j \in J_m$ the oscillatory domain D_j satisfies condition (m) in Lemma 3.5. In the following, for every $m \in \{1, 2, 3\}$ and every $j \in J_m$ the point $x_j \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$ is such that $D_j = D_{x_j}$.

Definition 4.3. Let $\gamma \in \left[\frac{d-1}{d}, 1\right)$. Define s, s', ω, ζ by

$$\frac{1}{s'} := \frac{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 - d}{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 + 1}, \qquad \frac{1}{s} := \frac{d+1}{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 + 1},$$
(150)

and

$$\omega := \left(\frac{d-1}{\gamma} + 1\right) \frac{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 - d}{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 + 1}, \quad \zeta := \frac{1}{s} \left(-\beta + \frac{d-1}{\gamma}\right). \tag{151}$$

We are now ready to prove that the number of oscillatory domains we choose is bounded by a constant times δ_0^{-d} .

Lemma 4.4. Let $l \in \{1, \ldots, L\}$ and let $\{D_j\}_{j \in J}$ be chosen as in Definition 4.2. Then

$$|J| \lesssim \delta_0^{-d} \,. \tag{152}$$

Proof. Recall that $J = J_1 \cup J_2 \cup J_3$, where each J_m , $m \in \{1, 2, 3\}$ is chosen such that all D_j with $j \in J_m$ satisfy condition (m) in Lemma 3.5. We will show for all $m \in \{1, 2, 3\}$ that $|J_m| \leq \delta_0^{-d}$.

Notation. If $j \in J$ and $D_j = D_{x_j}(\delta_{x_j})$ for some $x_j \in \tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d}\delta_0}$, write

$$\delta_j := \delta_{x_j}, \quad h_j := h_{x_j}, \quad a_j := a_{x_j} = \min\left(\delta_j, c_0 \max\left(h_j, c_1 \delta_j\right)^{\frac{1}{\gamma}}\right), \quad K := K_l.$$
 (153)

Estimate for J_1 . Recall that if $j \in J_1$, then $\delta_j = \delta_0$. We have

$$|J_1| \le |\{j \in J_1 \mid a_j = \delta_0\}| + \left|\left\{j \in J_1 \mid a_j = c_0 h_j^{\frac{1}{\gamma}}\right\}\right| + \left|\left\{j \in J_1 \mid a_j = c_2 \delta_0^{\frac{1}{\gamma}}\right\}\right|$$

=: $|J_{1,1}| + |J_{1,2}| + |J_{1,3}|$.

Estimate for $J_{1,1}$. Note that if $a_j = \delta_0$, then $|D_j| = \delta_0^d$ since by Lemma 2.7(i) and Lemma 2.8(i), D_j is a cuboid contained in Ω_l . Thus,

$$|J_{1,1}| = |\{j \in J_1 \mid a_j = \delta_0\}| = \delta_0^{-d} \sum_{j \in J_{1,1}} |D_j| \lesssim \delta_0^{-d} \left|\Omega_l \cap \Omega_{\sqrt{d}\delta_0}^b\right| \lesssim \delta_0^{-d},$$
(154)

where we used the covering lemma (Lemma 4.1) and $D_j \subset \Omega_l \cap \Omega^b_{\sqrt{d\delta_0}}$ for all $j \in J$ in the second last step. In the last step, we used that $\left|\Omega_l \cap \Omega^b_{\sqrt{d\delta_0}}\right| \leq |\Omega| \leq 1$.

Estimate for $J_{1,2}$. Recall that if $a_j = c_0 h_j^{\frac{1}{\gamma}}$, then $h_j \ge c_1 \delta_0$ and $|D_j| \sim \delta_0 h_j^{\frac{d-1}{\gamma}}$ by Lemma 2.8(i). Using these properties, Lemma 2.8(iii) and Lemma 4.1, we get

$$\begin{aligned} |J_{1,2}| &= \left| \left\{ j \in J_1 \mid a_j = c_0 h_j^{\frac{1}{\gamma}} \right\} \right| \sim \delta_0^{-1} \sum_{j \in J_{1,2}} |D_j| h_j^{-\frac{d-1}{\gamma}} \sim \delta_0^{-1} \sum_{j \in J_{1,2}} \int_{D_j} \mathrm{d}w \, h_w^{-\frac{d-1}{\gamma}} \\ &\lesssim \delta_0^{-1} \int_{\left\{ w \in \Omega_l \cap \Omega_{\sqrt{d}\delta_0}^b \, \Big| \, h_w \ge \frac{1}{2} c_1 \delta_0 \right\}} \mathrm{d}w \, h_w^{-\frac{d-1}{\gamma}} \, . \end{aligned}$$

Note that $\frac{d-1}{\gamma} > 1$ since $d \ge 2$ and $\gamma < 1$. We estimate using $\frac{d-1}{\gamma} \le d$ and $\delta_0 \le 1$

$$|J_{1,2}| \lesssim \delta_0^{-1} \int_{\left\{ w \in \Omega_l \cap \Omega_{\sqrt{d}\delta_0}^b \left| h_w \ge \frac{1}{2}c_1\delta_0 \right\}} \mathrm{d}w \, h_w^{-\frac{d-1}{\gamma}} \lesssim \delta_0^{-1} \int_{\frac{1}{2}c_1\delta_0}^{\infty} \mathrm{d}h \, h^{-\frac{d-1}{\gamma}} \sim \delta_0^{-\frac{d-1}{\gamma}} \le \delta_0^{-d} \, .$$

Estimate for $J_{1,3}$. By $a_j = c_2 \delta_0^{\frac{1}{\gamma}}$, we have $h_j \leq c_1 \delta_0$ for all $j \in J_{1,3}$. Furthermore,

$$|D_j| \sim \delta_0 a_j^{d-1} \sim \delta_0^{\frac{d-1}{\gamma}+1}.$$
 (155)

By Lemma 2.8(iv), we have $|h_w - h_j| \leq \delta_0$ for all $j \in J_{1,3}$ and all $w \in D_j$. Since $h_j \leq c_1 \delta_0$, we get

$$h_w \le h_j + |h_w - h_j| \le c_1 \delta_0 + \delta_0 = (c_1 + 1)\delta_0 \text{ for all } w \in D_j.$$
 (156)

We obtain

$$|J_{1,3}| = \left| \left\{ j \in J_1 \mid a_j = c_2 \delta_0^{\frac{1}{\gamma}} \right\} \right| = \delta_0^{-\left(\frac{d-1}{\gamma} + 1\right)} \sum_{j \in J_{1,3}} |D_j|$$

$$\lesssim \delta_0^{-\left(\frac{d-1}{\gamma} + 1\right)} \left| \left\{ w \in \Omega_l \cap \Omega_{\sqrt{d}\delta_0}^b \mid h_w \le (c_1 + 1)\delta_0 \right\} \right| \sim \delta_0^{-\frac{d-1}{\gamma}} \le \delta_0^{-d},$$

where were used (155) in the second step, the covering lemma (Lemma 4.1) and (156) in the third step, and $\frac{d-1}{\gamma} \leq d$ and $\delta_0 \leq 1$ in the last step.

Estimate for J_2 . Recall that if $j \in J_2$, then $||V||_{\frac{d}{2},D_j}^{\frac{d}{2}} \gtrsim 1$ if $d \geq 3$ and $||V||_{\mathcal{B},D_j} \gtrsim 1$ if d = 2. Using the covering lemma (Lemma 4.1), we obtain for $d \geq 3$

$$|J_2| = \sum_{j \in J_2} 1 \lesssim \sum_{j \in J_2} \|V\|_{\frac{d}{2}, D_j}^{\frac{d}{2}} \lesssim \|V\|_{\frac{d}{2}, \Omega_l \cap \Omega_{\sqrt{d}\delta_0}}^{\frac{d}{2}} \lesssim \|V\|_{\frac{d}{2}, \Omega}^{\frac{d}{2}} \lesssim \delta_0^{-d}.$$
 (157)

For d = 2, using [14, Lemma A.1] we get

$$|J_2| = \sum_{j \in J_2} 1 \lesssim \sum_{j \in J_2} \|V\|_{\mathcal{B}, D_j} \lesssim \|V\|_{\mathcal{B}, \Omega_l \cap \Omega^b_{\sqrt{d}\delta_0}} \lesssim \|V\|_{\mathcal{B}, \Omega} \lesssim \delta_0^{-d},$$
(158)

Estimate for J_3 . Recall that if $j \in J_3$, then $a_j = c_0 \max(h_j, c_1\delta_j)^{\frac{1}{\gamma}}$ and

$$\|V\|_{\tilde{p},D_j}^{\tilde{p}} \gtrsim \max\left(\frac{h_j}{c_1\delta_j}, 1\right)^{\frac{d-1}{\gamma}}.$$
(159)

In the following, we will use s', s, ω and ζ as defined in Definition 4.3. By

$$\frac{1}{s} + \frac{1}{s'} = 1$$
(160)

and Hölder's inequality, we have

$$|J_{3}| = \sum_{j \in J_{3}} \delta_{j}^{\omega} \max\left(\frac{h_{j}}{c_{1}\delta_{j}}, 1\right)^{-\zeta} \delta_{j}^{-\omega} \max\left(\frac{h_{j}}{c_{1}\delta_{j}}, 1\right)^{\zeta}$$

$$\leq \left(\sum_{j \in J_{3}} \delta_{j}^{\frac{d-1}{\gamma}+1} \max\left(\frac{h_{j}}{c_{1}\delta_{j}}, 1\right)^{-\zeta s'}\right)^{\frac{1}{s'}} \left(\sum_{j \in J_{3}} \delta_{j}^{-\beta} \max\left(\frac{h_{j}}{c_{1}\delta_{j}}, 1\right)^{-\beta+\frac{d-1}{\gamma}}\right)^{\frac{1}{s}}, \quad (161)$$

where we used

$$\omega s' = 1 + \frac{d-1}{\gamma}$$
 and $\omega s = \beta$ (162)

and the definition of ζ , see Definition 4.3, in the last step. By (159) and Lemma 2.8(v), we have for every $j \in J_3$

$$\delta_j^{-\beta} \max\left(\frac{h_j}{c_1\delta_j}, 1\right)^{-\beta + \frac{d-1}{\gamma}} \lesssim \max\left(h_j, c_1\delta_j\right)^{-\beta} \|V\|_{\tilde{p}, D_j}^{\tilde{p}} \lesssim |V|_{\tilde{p}, \beta, D_j}^{\tilde{p}} .$$
(163)

Using the covering lemma (Lemma 4.1), we get by (23)

$$\sum_{j \in J_3} \delta_j^{-\beta} \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\beta + \frac{d-1}{\gamma}} \lesssim \sum_{j \in J_3} |V|_{\tilde{p}, \beta, D_j}^{\tilde{p}} \lesssim \|V\|_{\tilde{p}, \beta}^{-\tilde{p}} \lesssim |V|_{\tilde{p}, \beta, \Omega}^{\tilde{p}} \lesssim \delta_0^{-2\tilde{p}}.$$
(164)

If we can show that

$$\sum_{j \in J_3} \delta_j^{\frac{d-1}{\gamma}+1} \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\zeta s'} \lesssim \delta_0 \,, \tag{165}$$

then we get by (161) and (164) that

$$|J_3| \lesssim \delta_0^{\frac{1}{s'}} \delta_0^{-2\tilde{p}\frac{1}{s}} = \delta_0^{-d} ,$$

where we used

$$-2\tilde{p}\frac{1}{s} + \frac{1}{s'} = -d \tag{166}$$

in the last step. Hence, it remains to show (165). First note that by $a_j = c_0 \max(h_j, c_1 \delta_j)^{\frac{1}{\gamma}}$ and by Lemma 2.7(ii), we have

$$|D_j| \sim \delta_0 a_j^{d-1} = \delta_0 \left(c_0 \max(h_j, c_1 \delta_j)^{\frac{1}{\gamma}} \right)^{d-1} \sim \delta_0^{\frac{d-1}{\gamma} + 1} \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{\frac{d-1}{\gamma}}$$
(167)

for all $j \in J_3$. Thus,

$$\sum_{j \in J_3} \delta_j^{\frac{d-1}{\gamma} + 1} \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\zeta s'} \sim \sum_{j \in J_3} |D_j| \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \\ = \sum_{j \in J_3, h_j \ge c_1 \delta_0} |D_j| \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} + \sum_{j \in J_3, h_j < c_1 \delta_0} |D_j| \max\left(\frac{h_j}{c_1 \delta_j}, 1\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \\ =: S_1 + S_2 \,.$$

Estimate for S_1 . If $j \in J_3$ with $h_j \ge c_1 \delta_0$, then we have $h_j \ge c_1 \delta_0 \ge c_1 \delta_j$ since $\delta_0 \ge \delta_j$. Thus,

$$a_j = c_0 \max(h_j, c_1 \delta_j)^{\frac{1}{\gamma}} = c_0 h_j^{\frac{1}{\gamma}}.$$
(168)

By Lemma 2.8(iii), it follows that

$$S_{1} \leq \sum_{j \in J_{3}, h_{j} \geq c_{1}\delta_{0}} |D_{j}| \left(\frac{h_{j}}{c_{1}\delta_{0}}\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \sim \sum_{j \in J_{3}, h_{j} \geq c_{1}\delta_{0}} \int_{D_{j}} \mathrm{d}w \left(\frac{h_{w}}{c_{1}\delta_{0}}\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)}$$
$$\leq \int_{\left\{w \in \Omega_{l} \cap \Omega^{b}_{\sqrt{d}\delta_{0}} \middle| h_{w} \geq \frac{1}{2}c_{1}\delta_{0}\right\}} \mathrm{d}w \left(\frac{h_{w}}{c_{1}\delta_{0}}\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \leq \int_{\frac{1}{2}c_{1}\delta_{0}}^{\infty} \mathrm{d}h \left(\frac{h}{c_{1}\delta_{0}}\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)}$$
$$= c_{1}\delta_{0} \int_{\frac{1}{2}}^{\infty} \mathrm{d}t \ t^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \leq \delta_{0} \ .$$

Here we used $h_j \ge c_1 \delta_0$, $\delta_0 \ge \delta_j$ and

$$\zeta s' + \frac{d-1}{\gamma} > 1 \tag{169}$$

in the second step. We explain (169) below. In the fourth step, we used Lemma 2.8(iii) to get $h_w \geq \frac{1}{2}h_j \geq \frac{1}{2}c_1\delta_0$ and moreover we applied the covering lemma (Lemma 4.1). In the second last step we used the change of variables $t = \frac{h_w}{c_1\delta_0}$ and in the last step we used (169) to deduce that the integral is finite.

In order to show (169), note that we have

$$\zeta s' + \frac{d-1}{\gamma} = \frac{1}{s} \left(-\beta + \frac{d-1}{\gamma} \right) s' + \frac{d-1}{\gamma} = 1 - 2 + \frac{d-1}{\gamma} \frac{d+1}{\frac{1}{d} \left(\frac{d-1}{\gamma} + 1 \right)^2 - d} \,. \tag{170}$$

Therefore, (169) is equivalent to

$$\frac{d-1}{\gamma} > \frac{2}{d+1} \left[\frac{1}{d} \left(\frac{d-1}{\gamma} + 1 \right)^2 - d \right] = \frac{2d}{d+1} \left[\frac{1}{d^2} \left(\frac{d-1}{\gamma} + 1 \right)^2 - 1 \right].$$
 (171)

Define

$$Y := \frac{1}{d} \left(\frac{d-1}{\gamma} + 1 \right) \tag{172}$$

and note that since $\gamma \in \left[\frac{d-1}{d}, 1\right)$, we have $\frac{d-1}{\gamma} \in (d-1, d]$ and therefore, $Y \in \left(1, \frac{d+1}{d}\right]$. The inequality (171) now reads

$$dY - 1 > \frac{2d}{d+1} \left[Y^2 - 1 \right] \,. \tag{173}$$

Using $\frac{d}{d+1} \leq \frac{1}{Y}$, $\frac{2d}{d+1} > 1$ and $d \geq 2$, we have

$$\frac{2d}{d+1}\left[Y^2 - 1\right] \le \frac{2}{Y}Y^2 - \frac{2d}{d+1} < 2Y - 1 \le dY - 1,$$
(174)

which shows (173) and hence, (169) holds.

Estimate for S_2 . By (169), we get

$$S_{2} = \sum_{j \in J_{3}, h_{j} < c_{1}\delta_{0}} |D_{j}| \max\left(\frac{h_{j}}{c_{1}\delta_{j}}, 1\right)^{-\left(\zeta s' + \frac{d-1}{\gamma}\right)} \le \sum_{j \in J_{3}, h_{j} < c_{1}\delta_{0}} |D_{j}|.$$
(175)

Let $j \in J_3$ with $h_j < c_1 \delta_0$. By the definition of J_3 , we have $a_j = c_0 h_j^{\frac{1}{\gamma}}$ or $a_j = c_2 \delta_j^{\frac{1}{\gamma}}$. If $a_j = c_0 h_j^{\frac{1}{\gamma}}$, then by assumption $h_j \le c_1 \delta_0$, so we get by Lemma 2.8(iii)

$$h_w \le 2h_j \le 2c_1 \delta_0 \quad \text{for all} \quad w \in D_j \,. \tag{176}$$

If $a_j = c_2 \delta_j^{\frac{1}{\gamma}}$, we get by Lemma 2.8(iv)

 $h_w \le h_j + |h_w - h_j| \le c_1 \delta_0 + \delta_j \le c_1 \delta_0 + \delta_0 \le 2c_1 \delta_0 \quad \text{for all} \quad w \in D_j.$ (177)

In both cases we get $h_w \leq 2c_1\delta_0$ for all $w \in D_j$. Using this fact, (175) and the covering lemma (Lemma 4.1), we obtain

$$S_2 \le \sum_{j \in J_3, h_j < c_1 \delta_0} |D_j| \lesssim \left| \left\{ w \in \Omega_l \cap \Omega^b_{\sqrt{d}\delta_0} \, \middle| \, h_w \le 2c_1 \delta_0 \right\} \right| \lesssim \delta_0 \,. \tag{178}$$

4.2. Covering of the interior by cubes. In this subsection, we consider the part of Ω far enough away from $\partial\Omega$ and show that we can choose a family of cubes D_x with centre $x \in \Omega$ far enough away from $\partial\Omega$ such that

$$N\left(-\Delta_{D_x}^N + V\right) \le 1 \tag{179}$$

and such that the number of cubes we choose is bounded by a constant times δ_0^{-d} . This part of the proof mimics the proof strategy of Rozenblum [13], [8, Section 4.5.1].

Definition 4.5. For all
$$x \in \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^{b}$$
, define $\delta_x \in (0, \delta_0]$ by
$$\delta_x := \sup \left\{ \tilde{\delta} \in (0, \delta_0] \, \middle| \, \|V\|_{\frac{d}{2}, D_x(\tilde{\delta})}^{\frac{d}{2}} \lesssim 1 \right\} \text{ if } d \ge 3$$
(180)

and

$$\delta_x := \sup\left\{\tilde{\delta} \in (0, \delta_0] \,\middle| \, \|V\|_{\mathcal{B}, D_x(\tilde{\delta})} \lesssim 1\right\} \ if \ d = 2$$
(181)

where

$$D_x(\tilde{\delta}) := \left\{ y \in \mathbb{R}^d \, \middle| \, |y - x|_\infty < \frac{1}{2} \tilde{\delta} \right\}$$
(182)

and where $|\cdot|_{\infty}$ denotes the ∞ -norm on \mathbb{R}^d . Here the constants in \leq have to be chosen small enough depending on d.

Lemma 4.6. Let $x \in \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^b$ and let $\delta_x \in (0, \delta_0]$ be as in Definition 4.5. Then $D := D_x := D_x(\delta_x) \subset \Omega$ and

$$N\left(-\Delta_D^N + V\right) \le 1. \tag{183}$$

Proof. Since $x \in \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^b$ and $\delta_x \leq \delta_0$, we have dist $(x, \partial\Omega) \geq \frac{1}{2}\sqrt{d\delta_0}$. By the definition of D, we obtain $D \subset \Omega$. The bound (183) can be proved as in Lemma 3.4(i).

Lemma 4.7 (Covering lemma for the interior of Ω). For every $x \in \Omega \setminus \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}$ let $\delta_x \in (0, \delta_0]$ and $D_x := D_x(\delta_x)$ be as in Definition 4.5.

(i) Then there exists $K_0 = K_0(d, \gamma) \in \mathbb{N}$ and subfamilies $\mathcal{F}_1, \ldots, \mathcal{F}_{K_0}$ of oscillatory domains $D_x = D_x(\delta_x) \subset \Omega_l \cap \Omega^b_{\sqrt{d\delta_0}}$ such that for every $k \in \{1, \ldots, K_0\}$ all oscillatory domains in \mathcal{F}_k are disjoint, and moreover,

$$\bigcup_{k=1}^{K_0} \bigcup_{D \in \mathcal{F}_k} D \supset \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^b.$$
(184)

(ii) Let J_0 be an index set and denote

$$\bigcup_{k=1}^{K_0} \mathcal{F}_k =: \{D_j\}_{j \in J_0} .$$
(185)

Then

$$|J_0| \lesssim \delta_0^{-d} \,. \tag{186}$$

Proof. **Proof of (i).** In order to get the desired result, it suffices to apply the Besicovitch covering lemma for cubes to the family $\{D_x\}_{x\in\Omega\setminus\Omega_{1,\sqrt{2\delta_{\alpha}}}^{b}}$.

Proof of (ii). For every $j \in J_0$, let δ_j be such that $D_j := D_{x_j}(\delta_j)$ for some $x_j \in \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^b$. Write $J_0 = J_{0,0} \cup J_{0,1}$, where $J_{0,0}$ and $J_{0,1}$ are chosen such that $\delta_j = \delta_0$ for all $j \in J_{0,0}$, $\|V\|_{\frac{d}{2},D_j} \sim 1$ for all $j \in J_{0,1}$ if $d \geq 3$, and $\|V\|_{\mathcal{B},D_j} \sim 1$ for all $j \in J_{0,1}$ if d = 2. This is possible by the definition of δ_x for $x \in \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d\delta_0}}^b$. By (i), $|D_j| = \delta_0^d$ and $|\Omega| \leq 1$ for all $j \in J_{0,0}$, so we get $|J_{0,0}| \lesssim \delta_0^{-d}$. Using (i), we obtain

$$|J_{0,1}| = \sum_{j \in J_{0,1}} 1 \lesssim \sum_{j \in J_{0,1}} \|V\|_{\frac{d}{2}, D_j}^{\frac{d}{2}} \lesssim \|V\|_{\frac{d}{2}, \Omega \setminus \Omega_{\frac{1}{2}\sqrt{d}\delta_0}}^{\frac{d}{2}} \lesssim \|V\|_{\frac{d}{2}, \Omega}^{\frac{d}{2}} \lesssim \delta_0^{-d}$$
(187)

if $d \geq 3$ and

$$|J_{0,1}| = \sum_{j \in J_{0,1}} 1 \lesssim \sum_{j \in J_{0,1}} \|V\|_{\mathcal{B},D_j} \lesssim \|V\|_{\mathcal{B},\Omega \setminus \Omega^b_{\frac{1}{2}\sqrt{d}\delta_0}} \lesssim \|V\|_{\mathcal{B},\Omega} \lesssim \delta_0^{-d}$$
(188)

if d = 2. Thus, we get (186).

5. Conclusion of Theorem 1.2 and Theorem 1.4

In this section, we conclude Theorem 1.4 and we also prove Corollary 1.5. We remarked in Section 1.2 that these two results imply Theorem 1.2.

5.1. **Proof of Theorem 1.4.** In this subsection, we combine the results from the previous subsections to prove Lemma 1.6. From this, we deduce Lemma 1.7 and Lemma 1.8. As we have already shown in Section 1.2, we obtain Theorem 1.4 from these lemmata.

Proof of Lemma 1.6. Define $K := K_0 + K_1 + \ldots K_L$, where K_0 was defined in Lemma 4.7(i) and K_l for $l \in \{1, \ldots, L\}$ was defined in Lemma 4.1. Note that K only depends on d, γ, L . Denote by \mathcal{F}_k^l with $l \in \{0, \ldots, L\}$ and $k \in \{1, \ldots, K_l\}$ the corresponding families of oscillatory domains. By Lemma 4.7(i) and Lemma 4.1(i), the oscillatory domains in each \mathcal{F}_k^l are disjoint and moreover,

$$\Omega \supset \bigcup_{l=0}^{L} \bigcup_{k=1}^{K_l} \bigcup_{D \in \mathcal{F}_k^l} D \supset \left(\Omega \setminus \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}\right) \cup \left(\bigcup_{l=1}^{L} \left(\tilde{\Omega}_l \cap \Omega^b_{\frac{1}{2}\sqrt{d\delta_0}}\right)\right) \supset \Omega,$$
(189)

where we used in the last step that by $h_{\Omega} \geq \sqrt{d}\delta_0$ and Lemma 2.3, we have $\bigcup_{l=1}^{L} \tilde{\Omega}_l \supset \Omega_{\frac{1}{2}\sqrt{d}\delta_0}^b$. This shows (a). For (b), note that by Lemma 4.6 and Lemma 3.4, we have

$$N\left(-\Delta_D^N+V\right) \le 1 \text{ for all } l \in \{0,\dots,L\}, \ k \in \{1,\dots,K_l\} \text{ and } D \in \mathcal{F}_k^l.$$
(190)

For (c), we obtain by Lemma 4.4 and Lemma 4.7(ii)

$$\sum_{l=0}^{L} \sum_{k=1}^{K_l} |\mathcal{F}_k| \lesssim \delta_0^{-d} \,. \tag{191}$$

Proof of Lemma 1.7. By Friedrich's extension, it suffices to prove that $-\Delta_{\Omega}^{N}+V$ is bounded from below with the quadratic form domain $H^{1}(\Omega)$. Let $K = K(d, \gamma, L) \in \mathbb{N}$ be as in Lemma 1.6. Denote $\tilde{V} := 4KV$ and recall $\|V\|_{\tilde{p},\beta} < \infty$. Let $\mathcal{F}_{1}, \ldots, \mathcal{F}_{K}$ be the families of oscillatory domains $D \subset \Omega$ which we got from Lemma 1.6 applied to \tilde{V} . We compute

$$-\Delta_{\Omega}^{N} + V \ge -\frac{1}{2}\Delta_{\Omega}^{N} + \frac{1}{2}\left(\frac{1}{K}\sum_{k=1}^{K}\left(-\Delta_{\Omega}^{N}\right) + \sum_{k=1}^{K}\sum_{D\in\mathcal{F}_{k}}2V1_{D}\right)$$

$$\ge -\frac{1}{2}\Delta_{\Omega}^{N} + \frac{1}{2K}\sum_{k=1}^{K}\sum_{D\in\mathcal{F}_{k}}\left(-\Delta_{D}^{N} + \frac{1}{2}\tilde{V}1_{D}\right).$$
(192)

Let $k \in \{1, \ldots, K\}$, $D \in \mathcal{F}_k$, $u \in H^1(\Omega)$ and $u_D := \frac{1}{|D|} \int_D u \in \mathbb{R}$. Note that for $v := u - u_D$, we have $v \in H^1(\Omega)$, $\int_D v = 0$ and $\int_D |\nabla u|^2 = \int_D |\nabla v|^2$. We get using $\tilde{V} \leq 0$

$$\int_{D} |\nabla u|^{2} + \int_{D} \frac{1}{2} \tilde{V} |u|^{2} \ge \int_{D} |\nabla v|^{2} + \int_{D} \frac{1}{2} \tilde{V} \left(2|v|^{2} + 2|u_{D}|^{2} \right)$$
$$= \int_{D} |\nabla v|^{2} + \int_{D} \tilde{V} |v|^{2} + \left| \frac{1}{|D|} \int_{D} u \right|^{2} \int_{D} \tilde{V} \ge \frac{1}{|D|} \int_{D} \tilde{V} \int_{D} |u|^{2}.$$

In the fourth step we used that $\int_D |\nabla v|^2 + \int_D \tilde{V} |v|^2 \ge 0$ by the choice of the family \mathcal{F}_k and $\int_D v = 0$ for the first two summands, and we used Jensen's inequality and $\tilde{V} \le 0$ for the last summand. We deduce that in the sense of quadratic forms,

$$-\Delta_D^N + \frac{1}{2} \mathbf{1}_D \tilde{V} \mathbf{1}_D \ge -\left(\frac{1}{|D|} \int_D |\tilde{V}|\right) \mathbf{1}_D.$$
 (193)

We obtain by (192) and (193),

$$-\Delta_{\Omega}^{N} + V \ge \frac{1}{2} \left(-\Delta_{\Omega}^{N} \right) - \left(\frac{1}{2K} \sum_{k=1}^{K} \sum_{D \in \mathcal{F}_{k}} \frac{1}{|D|} \int_{D} |\tilde{V}| \right) \mathbf{1}_{\Omega}, \tag{194}$$

where the constant in the last part is finite since $\sum_{k=1}^{K} |\mathcal{F}_k| \leq \delta_0(V)^{-d} < \infty$. It follows that the quadratic form for $-\Delta_{\Omega}^N + V$ with domain $H^1(\Omega)$ is well-defined, bounded from below and its form norm is given by the $H^1(\Omega)$ -norm. This finishes the proof of the self-adjointness of $-\Delta_{\Omega}^N + V$.

Proof of Lemma 1.8. As at the beginning of the proof of Lemma 1.7, in the sense of quadratic forms, we have

$$-\Delta_{\Omega}^{N} + \frac{1}{K}V \ge \frac{1}{K}\sum_{k=1}^{K}\sum_{D\in\mathcal{F}_{k}}\left(-\Delta_{D}^{N} + V1_{D}\right),\tag{195}$$

where we used $V \leq 0$ and $\Omega = \bigcup_{k=1}^{K} \bigcup_{D \in \mathcal{F}_k} D$ in the first step. In the second step, we used that for every $k \in \{1, \ldots, K\}$ the oscillatory domains $D \in \mathcal{F}_k$ are disjoint and therefore, $\int_{\Omega} |\nabla u|^2 \geq \sum_{D \in \mathcal{F}_k} \int_D |\nabla u|^2$ for all $u \in H^1(\Omega)$. By Lemma 1.6(b), we have

$$N\left(-\Delta_D^N + V\right) \le 1 \tag{196}$$

CHARLOTTE DIETZE

for every $k \in \{1, \ldots, K\}$ and every $D \in \mathcal{F}_k$. Hence, by the min-max principle [18, Theorem 12.1, version 2]², for every $k \in \{1, \ldots, K\}$ and $D \in \mathcal{F}_k$ there exists a function $u^D \in H^1(D) \subset L^2(\Omega)$ such that

$$\int_{D} |\nabla u|^{2} + \int_{D} V|u|^{2} \ge 0 \text{ for all } u \in H^{1}(\Omega) \text{ with } \int_{\Omega} \overline{u^{D}}u = 0.$$
(197)

It follows that if $u \in H^1(\Omega)$ is in the orthogonal complement in the $L^2(\Omega)$ sense of span $\{u^D \mid D \in \mathcal{F}_k, k = 1, \dots K\}$, then

$$\int_{\Omega} |\nabla u|^2 + \int_{\Omega} \frac{1}{K} V|u|^2 \ge \frac{1}{K} \sum_{k=1}^K \sum_{D \in \mathcal{F}_k} \int_D \left(|\nabla u|^2 + \int_D V|u|^2 \right) \ge 0.$$
(198)

Since the orthogonal complement of span $\{u^D \mid D \in \mathcal{F}_k, k = 1, ..., K\}$ is a subspace of $L^2(\Omega)$ of dimension at most $\sum_{k=1}^{K} |\mathcal{F}_k|$, we obtain by the min-max principle [18, Theorem 12.1, version 2] and (195)

$$N\left(-\Delta_{\Omega}^{N}+\frac{1}{K}V\right) \leq \sum_{k=1}^{K} |\mathcal{F}_{k}| \lesssim \delta_{0}^{-d}, \qquad (199)$$

where we used Lemma 1.6(c) in the last step.

5.2. Proof of Corollary 1.5. For the proof of Corollary 1.5, we need the following lemma. Lemma 5.1 (A subset of the γ with $\beta < 1$). Let $d \ge 2$.

(i) If
$$\gamma \in \left[\frac{2(d-1)}{2d-1}, 1\right]$$
, then

$$\beta = \beta(d, \gamma) = \frac{1}{d+1} \left(\frac{d-1}{\gamma} + 1\right) \left[\frac{1}{d} \left(\frac{d-1}{\gamma} + 1\right)^2 - d\right] < 1.$$
(200)

(ii) If $\gamma \in (0,1)$ is such that $\beta < 1$, then $\|\cdot\|_{\tilde{p},\beta} \lesssim \|\cdot\|_p$ for all $p > \frac{\tilde{p}}{1-\beta}$.

Proof. **Proof of (i).** Let $\gamma \in \left[\frac{2(d-1)}{2d-1}, 1\right]$. Note that

$$Y := \frac{1}{d} \left(\frac{d-1}{\gamma} + 1 \right) \le \frac{1}{d} \left(\frac{2d-1}{2} + 1 \right) = 1 + \frac{1}{2d},$$
(201)

so $Y^2 - 1 \leq \frac{1}{d} \left(1 + \frac{1}{4d} \right)$. Hence, we have

$$\beta = \frac{d^2}{d+1} Y \left[Y^2 - 1 \right] \le \frac{d^2}{d+1} \left(1 + \frac{1}{2d} \right) \frac{1}{d} \left(1 + \frac{1}{4d} \right) < 1.$$

Proof of (ii). Let $q := \frac{p}{\tilde{p}} > \frac{1}{1-\beta} > 1$ and note that $\frac{1}{q'} = 1 - \frac{1}{q} > 1 - (1-\beta) = \beta$, so $\beta q' < 1$. By Hölder's inequality, we get

$$\begin{split} |V|_{\tilde{p},\beta}^{\tilde{p}} &= \int_{\bigcup_{l=1}^{L}\Omega_{l}} \mathrm{d}x \, h_{x,\min}^{-\beta} |V(x)|^{\tilde{p}} \leq \left(\int_{\bigcup_{l=1}^{L}\Omega_{l}} \mathrm{d}x \, h_{x,\min}^{-\beta q'} \right)^{\frac{1}{q'}} \left(\int_{\bigcup_{l=1}^{L}\Omega_{l}} \mathrm{d}x \, |V(x)|^{\tilde{p}q} \right)^{\frac{1}{q}} \\ &\lesssim \|V\|_{p}^{\tilde{p}} \,, \end{split}$$

²As the proof [18, Theorem 12.1, version 2] shows, in fact, the subspace M need not be a subset of $H^1(\Omega)$ but it suffices to take $M \subset L^2(\Omega)$.

where we used $\beta q' < 1$ and $\tilde{p}q = p$ in the last step. Furthermore, since $|\Omega| \leq 1$ and $p > \tilde{p} \ge \frac{d}{2}$, we can apply Jensen's inequality to get

$$\|V\|_{p}^{p} = \int_{\Omega} |V|^{p} = |\Omega| \frac{1}{|\Omega|} \int_{\Omega} \left(|V|^{\frac{d}{2}} \right)^{\frac{2p}{d}} \ge |\Omega| \left(\frac{1}{|\Omega|} \int_{\Omega} |V|^{\frac{d}{2}} \right)^{\frac{2p}{d}} = |\Omega|^{1 - \frac{2p}{d}} \|V\|_{\frac{d}{2}}^{p} \ge \|V\|_{\frac{d}{2}}^{p}$$

For d = 2, we have $\|V\|_p \gtrsim \|V\|_{\mathcal{B},\Omega}$ by [19, Chapter 5.1, Theorem 3, p. 155] and $p > \tilde{p} \ge 1$ $\frac{d}{2} = 1$. Using Definition 2.4(iii), we obtain $\|V\|_{\tilde{p},\beta} \lesssim \|V\|_p$ for any $d \ge 2$.

Proof of Corollary 1.5. By Lemma 5.1, we have $\|\cdot\|_{\tilde{p},\beta} \lesssim \|\cdot\|_p$. Therefore, by Theorem 1.4,

$$N\left(-\Delta_{\Omega}^{N}+V\right) \lesssim \left[\min\left(\frac{h_{\Omega}}{\sqrt{d}}, \|V\|_{\tilde{p},\beta}^{-\frac{1}{2}}\right)\right]^{-d} = \max\left(\left(\frac{h_{\Omega}}{\sqrt{d}}\right)^{-d}, \|V\|_{\tilde{p},\beta}^{\frac{d}{2}}\right)$$
$$\lesssim 1 + \|V\|_{\tilde{p},\beta}^{\frac{d}{2}} \lesssim 1 + \|V\|_{p}^{\frac{d}{2}}.$$

6. Weyl's law for Schrödinger operators (Theorem 1.3)

In this section, we deduce Theorem 1.3 using Theorem 1.2. The main idea is to first reduce to Weyl's law for continuous compactly supported potentials, namely

$$N\left(-\Delta_{\Omega}^{N} + \lambda W\right) = (2\pi)^{-d} |B_{1}(0)| \lambda^{\frac{d}{2}} \int_{\Omega} |W|^{\frac{d}{2}} + o\left(\lambda^{\frac{d}{2}}\right) \text{ as } \lambda \to \infty.$$
(202)

for all $W \in C_c(\Omega)$ with $W \leq 0$. Using (202) combined with the Cwikel-Lieb-Rozenblum type bound (Theorem 1.2), we can then deduce Theorem 1.3.

6.1. Reduction to compactly supported potentials. Let V be as in Theorem 1.3. Assume that we have (202) for $0 \ge W \in C_c(\Omega)$. Since $\|\|\cdot\|\| := \|\cdot\|_{\tilde{p},\beta}$ is a weighted $L^{\tilde{p}}$ -norm on Ω with $\tilde{p} < \infty$, there exists a sequence $(V_n)_{n \in \mathbb{N}} \subset C_c(\Omega)$ with $V_n \leq 0$ such that

$$\|V_n - V\|_{\frac{d}{2}} \lesssim \|V - V_n\| \to 0 \text{ as } n \to \infty.$$

$$(203)$$

Let $\delta \in (0, 1)$. Recall that from the min-max principle [18, Theorem 12.1, version 2], one can deduce that

$$N(A+B) \le N(A) + N(B) \tag{204}$$

for any two self-adjoint operators A and B defined on the same Hilbert space with the same quadratic form domain. We have for every $n \in \mathbb{N}$ and $\lambda > 0$

$$N\left(-\Delta_{\Omega}^{N}+\lambda V\right) \leq N\left(\left(1-\delta\right)\left(-\Delta_{\Omega}^{N}\right)+\lambda V_{n}\right)+N\left(\delta\left(-\Delta_{\Omega}^{N}\right)+\lambda\left(V-V_{n}\right)\right)$$
$$=N\left(-\Delta_{\Omega}^{N}+\lambda\frac{V_{n}}{1-\delta}\right)+N\left(-\Delta_{\Omega}^{N}+\lambda\frac{V-V_{n}}{\delta}\right)$$
$$\leq N\left(-\Delta_{\Omega}^{N}+\lambda\frac{V_{n}}{1-\delta}\right)+C_{\Omega}\left(1+\delta^{-\frac{d}{2}}\lambda^{\frac{d}{2}}\|\|V-V_{n}\|\|^{\frac{d}{2}}\right),$$
(205)

31

where we used Theorem 1.2 in the last step. Using (203), (205), (202) for $W = \frac{V_n}{1-\delta}$ and the definition of $\|\cdot\|$, we get

$$\begin{split} &\lim_{\lambda \to \infty} \sup \lambda^{-\frac{d}{2}} N\left(-\Delta_{\Omega}^{N} + \lambda V\right) \\ &\leq \limsup_{\delta \to 0} \limsup_{n \to \infty} \limsup_{\lambda \to \infty} \sup \lambda^{-\frac{d}{2}} \left(N\left(-\Delta_{\Omega}^{N} + \lambda \frac{V_{n}}{1-\delta}\right) + C_{\Omega}\left(1 + \delta^{-\frac{d}{2}} \lambda^{\frac{d}{2}} \|\|V - V_{n}\|\|^{\frac{d}{2}}\right) \right) \\ &\leq \limsup_{\delta \to 0} \limsup_{n \to \infty} \left(\left(2\pi\right)^{-d} |B_{1}(0)| \int_{\Omega} \left|\frac{V_{n}}{1-\delta}\right|^{\frac{d}{2}} + C_{\Omega} \delta^{-\frac{d}{2}} \|\|V - V_{n}\|\|^{\frac{d}{2}} \right) \\ &= \left(2\pi\right)^{-d} |B_{1}(0)| \limsup_{\delta \to 0} \int_{\Omega} \left|\frac{V}{1-\delta}\right|^{\frac{d}{2}} = \left(2\pi\right)^{-d} |B_{1}(0)| \int_{\Omega} |V|^{\frac{d}{2}}, \end{split}$$

$$(206)$$

which is the desired upper bound. For the corresponding lower bound, we replace V by $(1-\delta)V_n$, and we replace V_n by $(1-\delta)V$ in (205) to get

$$N\left(-\Delta_{\Omega}^{N}+\lambda V\right) \geq N\left(-\Delta_{\Omega}^{N}+\lambda(1-\delta)V_{n}\right)-C_{\Omega}\left(1+\delta^{-\frac{d}{2}}(1-\delta)^{\frac{d}{2}}\lambda^{\frac{d}{2}}\||V-V_{n}\||^{\frac{d}{2}}\right), \quad (207)$$

and then proceeding as above.

6.2. Weyl's law for compactly supported potentials. Now we prove (202) for $0 \ge W \in C_c(\Omega)$. This result can also be found in [8, Theorem 4.29] for the Laplacian on \mathbb{R}^d . For the reader's convenience, we explain the proof below since our setting is slightly different. We follow the proof strategy of Weyl, namely we cover the support of W by small cubes of side-length independent of λ such that each cube is completely contained in Ω . We then apply Weyl's law for constant potentials on cubes.

Let $m_0 \in \mathbb{N}$ be such that

$$\sqrt{d}2^{-m_0} < \operatorname{dist}\left(\operatorname{supp} W, \partial\Omega\right).$$
(208)

Then every cube of side-length at most 2^{-m_0} intersecting supp W is contained in Ω . For every $m \in \mathbb{N}, j \in \mathbb{Z}^d$ let

$$Q_j^m := 2^{-m} \left(j + (0, 1)^d \right) \tag{209}$$

be the open cube of side-length 2^{-m} whose bottom left corner is at $2^{-m}j \in \mathbb{R}^d$. For every $m \in \mathbb{N}$ let

$$J_m := \left\{ j \in \mathbb{Z}^d \mid Q_j^m \cap \operatorname{supp} W \neq \emptyset \right\}.$$
(210)

Upper bound. We claim that for every $m \ge m_0$, we have

$$N\left(-\Delta_{\Omega}^{N}+\lambda W\right) \leq \sum_{j\in J_{m}} N\left(-\Delta_{Q_{j}^{m}}^{N}+\lambda W\right).$$
(211)

This can be seen as follows. By the min-max principle [18, Theorem 12.1, Version 2], we know that for every $j \in J_m$ there exists an $N(-\Delta_{Q_j}^N + \lambda W)$ -dimensional subspace of $L^2(Q_j^m)$, which we call M_j , such that

$$\int_{Q_j^m} |\nabla u|^2 + \int_{Q_j^m} \lambda W |u|^2 \ge 0$$
(212)

for all $u \in H^1(Q_j^m)$ that are in the orthogonal complement of M_j with respect to $L^2(Q_j^m)$. Let $M \subset L^2(\Omega)$ be the span of all M_j for $j \in J_m$, where we extend functions in $L^2(Q_j^m)$. by zero on $\Omega \setminus Q_i^m$. Note that M has dimension at most

$$\sum_{j \in J_m} N\left(-\Delta_{Q_j^m}^N + \lambda W\right).$$
(213)

By (212), $\Omega \supset \bigcup_{j \in J_m} \overline{Q_j} \supset \operatorname{supp} W$ and since the cubes Q_j are disjoint, we get for every $u \in H^1(\Omega)$ in the orthogonal complement of M with respect to $L^2(\Omega)$

$$\int_{\Omega} |\nabla u|^2 + \int_{\Omega} \lambda W |u|^2 \ge \sum_{j \in J_m} \left(\int_{Q_j^m} |\nabla u|^2 + \int_{Q_j^m} \lambda W |u|^2 \right) \ge 0.$$
(214)

By the min-max principle [18, Theorem 12.1, version 2], we obtain (211). Therefore, for every $m \in \mathbb{N}, m \geq m_0$

$$N\left(-\Delta_{\Omega}^{N}+\lambda W\right) \leq \sum_{j\in J_{m}} N\left(-\Delta_{Q_{j}^{m}}^{N}+\lambda W\right) \leq \sum_{j\in J_{m}} N\left(-\Delta_{Q_{j}^{m}}^{N}-\lambda \sup_{x\in Q_{j}^{m}}|W(x)|\right).$$
(215)

By Weyl's law for constant potentials with Neumann boundary conditions on cubes, see for example [8, Theorem 3.20],

$$\limsup_{\lambda \to \infty} \lambda^{-\frac{d}{2}} N\left(-\Delta_{\Omega}^{N} + \lambda W\right) \leq \sum_{j \in J_{m}} \limsup_{\lambda \to \infty} \lambda^{-\frac{d}{2}} N\left(-\Delta_{Q_{j}^{m}}^{N} - \lambda \sup_{x \in Q_{j}^{m}} |W(x)|\right)$$
$$= \sum_{j \in J_{m}} \left((2\pi)^{-d} |B_{1}(0)| \left|Q_{j}^{m}\right| \left(\sup_{x \in Q_{j}^{m}} |W(x)|\right)^{\frac{d}{2}} \right).$$

Since $W \in C_c(\Omega)$, the right hand side agrees with (202) as $m \to \infty$.

Lower bound. For every $m \in \mathbb{N}$, $m \ge m_0$, we have

$$N\left(-\Delta_{\Omega}^{N}+\lambda W\right) \geq \sum_{j\in J_{m}} N\left(-\Delta_{Q_{j}^{m}}^{D}+\lambda W\right).$$
(216)

For the proof of (216), note that by the min-max principle [18, Theorem 12.1, Version 3] for every $j \in J_m$ there exists an $N\left(-\Delta_{Q_j^m}^D + \lambda W\right)$ -dimensional subspace of $H_0^1(Q_j^m)$, which we call M_j , such that

$$\int_{Q_j^m} |\nabla u_j|^2 + \int_{Q_j^m} \lambda W |u_j|^2 < 0 \text{ for all } 0 \neq u_j \in M_j.$$

$$(217)$$

Since the cubes Q_j^m , $j \in J_m$ are disjoint and each $M_j \subset H_0^1(Q_j^m) \subset H^1(\Omega)$, if we denote by M the span of all M_j , then $M \subset H^1(\Omega)$ is a subspace of dimension

$$\sum_{j\in J_m} N\left(-\Delta_{Q_j^m}^D + \lambda W\right).$$
(218)

By supp $W \subset \bigcup_{j \in J_m} \overline{Q_j}$ and (217), we obtain for every $0 \neq u = \sum_{j \in J_m} u_j \in M$ with $u_j \in M_j$ for each $j \in J_m$,

$$\int_{\Omega} |\nabla u|^2 + \int_{\Omega} \lambda W |u|^2 = \sum_{j \in J_m} \left(\int_{Q_j^m} |\nabla u_j|^2 + \int_{Q_j^m} \lambda W |u_j|^2 \right) < 0.$$
(219)

CHARLOTTE DIETZE

By the min-max principle [18, Theorem 12.1, Version 3], we get (216). Using (216) and Weyl's law for constant potentials on cubes [2, 3], we get

$$\liminf_{\lambda \to \infty} \lambda^{-\frac{d}{2}} N\left(-\Delta_{\Omega}^{N} + \lambda W\right) \ge \liminf_{\lambda \to \infty} \lambda^{-\frac{d}{2}} \sum_{j \in J_{m}} N\left(-\Delta_{Q_{j}^{m}}^{D} + \lambda W\right)$$
$$\ge \sum_{j \in J_{m}} \liminf_{\lambda \to \infty} \lambda^{-\frac{d}{2}} N\left(-\Delta_{Q_{j}^{m}}^{D} - \lambda \inf_{x \in Q_{j}^{m}} |W(x)|\right)$$
$$= \sum_{j \in J_{m}} \left(\left(2\pi\right)^{-d} |B_{1}(0)| \left|Q_{j}^{m}\right| \left(\inf_{x \in Q_{j}^{m}} |W(x)|\right)^{\frac{d}{2}}\right).$$

Taking $m \to \infty$, we conclude (202). The proof of Theorem 1.3 is complete.

Remark 6.1. In dimension $d \ge 3$, we can obtain the lower bound in Theorem 1.3 for all potentials $V \in L^{\frac{d}{2}}(\Omega), V \le 0$ by comparing with $-\Delta_{\mathbb{R}^d} + \lambda V$ and using Weyl's law for Schrödinger operators on \mathbb{R}^d [8, Theorem 4.46]. However, this is not true in dimension d = 2, see [8, Remark after Theorem 4.46].

7. Example with non-semiclassical behaviour (Theorem 1.1)

In this section we prove Theorem 1.1. We explain the proof strategy in Section 7.1 and the details are given in Section 7.2.

7.1. General strategy. In this subsection, we explain for fixed $\gamma \in \left(\frac{d-1}{d}, 1\right)$ how to construct a γ -Hölder domain $\Omega \subset \mathbb{R}^d$ and a potential $V : \Omega \to (-\infty, 0]$ with $V \in L^{\frac{d}{2}}(\Omega)$ such that (5) holds. We construct the γ -Hölder domain Ω in the same way as Netrusov and Safarov [10, Theorem 1.10]. The potential V will be chosen such that it grows near the boundary of Ω . This will allow us for certain values of λ going to infinity to find significantly more than $\lambda^{\frac{d}{2}}$ negative eigenvalues of $-\Delta_{\Omega}^N + \lambda V$.

We start by fixing $M := 2^m$, where $m \in \mathbb{N}$ is chosen large enough depending on γ . The main part of Ω will be given by the subgraph

$$\left\{ \left(x', x_d\right) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x' \in Q^{(d-1)}, 0 < x_d < f(x') \right\}$$

$$(220)$$

of a γ -Hölder continuous function f on the (d-1)-dimensional unit cube $Q^{(d-1)}$ that vanishes on the boundary of $Q^{(d-1)}$. For every $j \in \mathbb{N}$ we can decompose $Q^{(d-1)}$ into $M^{(d-1)j}$ small cubes of side-length M^{-j} . The function f will be chosen in such a way that for any $j \in \mathbb{N}$ it oscillates on the order of magnitude $M^{-j\gamma}$ on each of the small cubes of side-length M^{-j} . Intuitively speaking, f looks no better than a γ -Hölder continuous function on each of the small cubes for every length scale M^{-j} , $j \in \mathbb{N}$.

The potential V will be chosen such that it is large close to the boundary of Ω . We define

$$V(x', x_d) := -c \left(f(x') - x_d \right)^{\frac{2}{d}(-1+\varepsilon)} \text{ for } (x', x_d) \in \Omega \subset \mathbb{R}^{d-1} \times \mathbb{R}$$
(221)

for a suitably chosen $0 < \varepsilon < (d-1)(1/\gamma - 1)$ and a constant $c = c(d, \gamma, \varepsilon) > 0$. Note that $V \in L^{\frac{d}{2}}(\Omega)$ since $\varepsilon > 0$. In the following, for fixed $j \in \mathbb{N}$, we can for simplicity think of f as a j-dependent constant c(j) > 0 plus a small spike of height $M^{-j\gamma}$ on each of the $M^{(d-1)j}$ (d-1)-dimensional small cubes of side-length M^{-j} . We denote these small cubes of side-length M^{-j} by $Q(j,k), k \in \{1, \ldots, M^{(d-1)j}\}$. Moreover, we define

$$\Omega_{j,k} := \left\{ \left(x', x_d \right) \in Q(j,k) \times \mathbb{R} \mid c(j) < x_d < f(x') \right\},\tag{222}$$

and $u_{j,k} \in H^1(\Omega)$ by

$$u_{j,k}(x', x_d) := \sin\left(M^{j\gamma}(x_d - c(j))\right) \mathbf{1}_{\Omega_{j,k}}(x', x_d).$$
(223)

Note that for fixed $j \in \mathbb{N}$, the interior of the support the $\{u_{j,k}\}_{k=1}^{M^{(d-1)j}}$ are disjoint. One can show that for

$$\lambda(j) := M^{2\gamma j \left(1 + \frac{1}{d}(-1+\varepsilon)\right)} \tag{224}$$

we have for every $k \in \{1, \ldots, M^{(d-1)j}\}$

$$\int_{\Omega} |\nabla u_{j,k}|^2 + \int_{\Omega} \lambda V |u_{j,k}|^2 < 0.$$
(225)

Hence, for every $j \in \mathbb{N}$

$$N\left(-\Delta_{\Omega}^{N} + \lambda(j)V\right) \ge M^{(d-1)j}.$$
(226)

A computation shows that since $\varepsilon < (d-1)(1/\gamma - 1)$,

$$\lim_{j \to \infty} \lambda(j)^{-\frac{d}{2}} M^{(d-1)j} = \infty.$$
(227)

It follows that

$$\limsup_{\lambda \to \infty} \lambda^{-\frac{d}{2}} N\left(-\Delta_{\Omega}^{N} + \lambda V\right) = \infty.$$
(228)

7.2. Details of the proof of Theorem 1.1. Let us now come to the details of the proof of Theorem 1.1. For the definition of Ω and of the orthogonal set of test functions, we closely follow [10, Theorem 1.10].

Definition 7.1. Let $d \ge 2$, $\gamma \in \left(\frac{d-1}{d}, 1\right)$ and let $m \in \mathbb{N}$ be large enough such that $m\gamma \ge 1$ and $m(1-\gamma) \ge 4$. Define $Q^{(d-1)} := (0, 1)^{d-1}$ and

$$\psi : \mathbb{R}^{d-1} \to [0, 1/2] , \ x' \mapsto \frac{1}{2} - \left| x' - \left(\frac{1}{2}, \dots, \frac{1}{2}\right) \right|_{\infty} \mathbb{1}_{Q^{(d-1)}}(x').$$

For every $j \in \mathbb{N}_0$ let

$$K_j := \left\{ 0, 1, 2, 3, \dots, 2^{jm} - 1 \right\}^{d-1}$$
(229)

and for $k \in K_j$ define

$$Q(j,k) := \left\{ x' \in \mathbb{R}^{d-1} \mid 2^{jm} x' - k \in Q^{(d-1)} \right\} \subset Q^{(d-1)}.$$
(230)

Define for $j \in \mathbb{N}_0$ the functions

$$g_j: Q^{(d-1)} \to [0, 1/2] , \ x' \mapsto \sum_{k \in K_j} \psi \left(2^{jm} x' - k \right),$$
 (231)

for $n \in \mathbb{N}_0 \cup \{-1\}$

$$f_n: Q^{(d-1)} \to [0,\infty) , \ x' \mapsto \sum_{j=0}^n 2^{-\gamma j m} g_j(x').$$
 (232)

We also denote $f_{-1} \equiv 0$ and $f = \lim_{n \to \infty} f_n$. Also define for every $n \in \mathbb{N}_0$, $k \in K_n$

$$a_{n,k} := \sup_{x' \in Q(n,k)} f_{n-1}(x').$$
(233)

Lemma 7.2. (i) $f: (Q^{(d-1)}, |\cdot|_{\infty}) \to [0, \infty)$ is γ -Hölder continuous with constant 3. (ii) For every $n \in \mathbb{N}_0$, $k \in K_n$ and $x', y' \in Q(n, k)$, we have

$$|f_{n-1}(x') - f_{n-1}(y')| \le \frac{1}{8} 2^{-\gamma mn}.$$
(234)

(iii) For every $n \in \mathbb{N}_0$, $k \in K_n$ and $x' \in Q(n,k)$ with $2^{nm}x' - k \in [1/4, 3/4]^{d-1}$, we have

$$f(x') - a_{n,k} \ge \frac{1}{8} 2^{-\gamma m n}.$$
 (235)

(iv) For every $n \in \mathbb{N}_0$, $k \in K_n$ and $x' \in Q(n,k)$, we have

$$f(x') - a_{n,k} \le 2^{-\gamma mn}.$$
 (236)

Proof. **Proof of (i).** Let $x', y' \in Q^{(d-1)}$ with $x' \neq y'$ and denote by n' the largest number in \mathbb{N}_0 such that $2^{-n'm} \geq |x' - y'|_{\infty}$. In particular,

$$2^{-(n'+1)m} < |x' - y'|_{\infty} \le 2^{-n'm}.$$
(237)

We have

$$|f(x') - f(y')| = \left| \sum_{j=0}^{\infty} 2^{-\gamma m j} g_j(x') - \sum_{j=0}^{\infty} 2^{-\gamma m j} g_j(y') \right|$$
$$\leq \sum_{j=0}^{n'} 2^{-\gamma m j} |g_j(x') - g_j(y')| + \sum_{j=n'+1}^{\infty} 2^{-\gamma m j} |g_j(x') - g_j(y')|$$

For the first term, we use the Lipschitz continuity of g_j , (237) and $m(1 - \gamma) \ge 4$ by Definition 7.1 to get

$$\sum_{j=0}^{n'} 2^{-\gamma m j} |g_j(x') - g_j(y')| \le \sum_{j=0}^{n'} 2^{-\gamma m j} \cdot 2^{jm} |x' - y'|_{\infty}$$
$$\le |x' - y'|_{\infty}^{\gamma} 2^{-m(1-\gamma)n'} \sum_{j=0}^{n'} 2^{m(1-\gamma)j} \le |x' - y'|_{\infty}^{\gamma} \sum_{j=0}^{n'} 2^{-m(1-\gamma)j} \le 2 |x' - y'|_{\infty}^{\gamma}.$$

For the second term, we use the Lipschitz continuity of g_i , (237) and $m\gamma \ge 1$ to get

$$\sum_{j=n'+1}^{\infty} 2^{-\gamma m j} |g_j(x') - g_j(y')| \le \frac{1}{2} \sum_{j=n'+1}^{\infty} 2^{-\gamma m j} = \frac{1}{2} 2^{-\gamma m (n'+1)} \sum_{j=0}^{\infty} 2^{-\gamma m j} \le |x' - y'|_{\infty}^{\gamma}.$$

Combining these two estimates, we obtain the claim.

Proof of (ii). Let $n \in \mathbb{N}_0$, $k \in K_n$ and $x', y' \in Q(n, k)$. We have

$$|f_{n-1}(x') - f_{n-1}(y')| = \left| \sum_{j=0}^{n-1} 2^{-\gamma m j} g_j(x') - \sum_{j=0}^{n-1} 2^{-\gamma m j} g_j(y') \right| \le \sum_{j=0}^{n-1} 2^{-\gamma m j} |g_j(x') - g_j(y')|$$
$$\le \sum_{j=0}^{n-1} 2^{-\gamma m j} \cdot 2^{m j} |x' - y'|_{\infty} \le 2^{-m n} \sum_{j=0}^{n-1} 2^{(1-\gamma)m j} \le \frac{1}{8} 2^{-\gamma m n},$$

where we used Lipschitz continuity of g_j in the third step, and $m(1-\gamma) \ge 4$ in the last step.

Proof of (iii). First note that for all $y' \in [1/4, 3/4]^{d-1}$, we have $\psi(y') \ge 1/4$. Let $n \in \mathbb{N}_0, k \in K_n$ and $x' \in Q(n, k)$ with $2^{nm}x' - k \in [1/4, 3/4]^{d-1}$. Then,

$$g_n(x') = \sum_{\tilde{k} \in K_n} \psi \left(2^{nm} x' - \tilde{k} \right) = \psi \left(2^{nm} x' - k \right) \ge \frac{1}{4}.$$
 (238)

36

Therefore, we have

$$f(x') - a_{n,k} \ge f_n(x') - a_{n,k} = f_n(x') - f_{n-1}(x') + f_{n-1}(x') - a_{n,k}$$
$$\ge 2^{-\gamma mn} g_n(x') - |f_{n-1}(x') - a_{n,k}| \ge \frac{1}{4} 2^{-\gamma mn} - \frac{1}{8} 2^{-\gamma mn} = \frac{1}{8} 2^{-\gamma mn},$$

where we used (238), (ii) and (233) in the fourth step.

Proof of (iv). Let $n \in \mathbb{N}_0$, $k \in K_n$, $x' \in Q(n,k)$. Then by (233), $a_{n,k} \ge f_{n-1}(x')$. By $g_j(x') \le \frac{1}{2}$ and $m\gamma \ge 1$, it follows that

$$f(x') - a_{n,k} \le f(x') - f_{n-1}(x') = \sum_{j=n}^{\infty} 2^{-\gamma m j} g_j(x') \le \frac{1}{2} 2^{-\gamma m n} \sum_{j=0}^{\infty} 2^{-\gamma m j} \le 2^{-\gamma m n}.$$

Definition 7.3 (Ω and $\Omega_{n,k}$). Define the γ -Hölder domain

 $\Omega := \left\{ x = (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x' \in Q^{d-1}, \ 0 \le x_d < f(x') \right\} \cup \left((-2, 2)^{d-1} \times (-2, 0) \right)$ and for all $n \in \mathbb{N}_0, \ k \in K_n$ define

$$\Omega_{n,k} := \left\{ x \in \Omega \mid x' \in Q(n,k), \ x_d \in \left(f_{n-1}(x'), f(x') \right) \right\}.$$
(239)

Definition 7.4 $(u_{n,k}, b_2, b_{\nabla}, b_V \text{ and } V)$. (i) For $n \in \mathbb{N}_0$, $k \in K_n$, let $u_{n,k} : \Omega \to \mathbb{R}$ with

$$u_{n,k}(x) := \begin{cases} \sin(2^{\gamma m n} (x_d - a_{n,k})) & \text{for } x' \in Q(n,k), \, x_d \ge a_{n,k} \\ 0 & \text{else.} \end{cases}$$
(240)

(ii) Define

$$b_2 := 2^{-(d-1)} \int_0^{\frac{1}{8}} \mathrm{d}t \,|\,\sin(t)|^2, \quad b_{\nabla} := \int_0^1 \mathrm{d}t \,|\,\cos(t)|^2 \quad and \quad b_V := 2\frac{b_{\nabla}}{b_2}. \tag{241}$$

(iii) Let
$$\varepsilon \in (0,1)$$
. Define $V : \Omega \to \mathbb{R}$ depending on ε by

$$V(x) := \begin{cases} -b_V (f(x) - x_d)^{\frac{2}{d}(-1+\varepsilon)} & \text{for } x' \in Q^{(d-1)}, \ 0 \le x_d < f(x') \\ 0 & \text{else.} \end{cases}$$
(242)

Lemma 7.5 (Estimates for $u_{n,k}$). Let $n \in \mathbb{N}_0$ and $k \in K_n$. Let $\varepsilon \in (0,1)$.

(i) Then $u_{n,k} \in H^1(\Omega)$. Moreover,

$$\int_{\Omega} |u_{n,k}|^2 \ge b_2 2^{-(d-1)mn} \cdot 2^{-\gamma mn}, \quad \int_{\Omega} |\nabla u_{n,k}|^2 \le b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn}$$

(ii) For all $\lambda > 0$, we have

$$\int_{\Omega} |\nabla u_{n,k}|^2 + \int_{\Omega} \lambda V |u_{n,k}|^2 \le b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn} \left(1 - 2\lambda 2^{-2\gamma mn} \cdot 2^{-\gamma mn \frac{2}{d}(-1+\varepsilon)}\right). \quad (243)$$

Proof. **Proof of (i).** A direct computation shows that $u_{n,k} \in H^1(\Omega)$. Next, we compute

$$\begin{split} \int_{\Omega} |u_{n,k}|^2 &= \int_{Q(n,k)} \mathrm{d}x' \int_0^{f(x')-a_{n,k}} \mathrm{d}s \,|\sin\left(2^{\gamma mn}s\right)|^2 \\ &\geq \int_{\left\{x' \in Q(n,k) \,|\, 2^{mn}x'-k \in [1/4,3/4]^{d-1}\right\}} \mathrm{d}x' \int_0^{\frac{1}{8}2^{-\gamma mn}} \mathrm{d}s \,|\sin(2^{\gamma mn}s)|^2 \\ &= 2^{-(d-1)mn} \cdot 2^{-(d-1)} \int_0^{\frac{1}{8}} \mathrm{d}t \,|\sin(t)|^2 2^{-\gamma mn} = b_2 2^{-(d-1)mn} \cdot 2^{-\gamma mn}, \end{split}$$

where we used Lemma 7.2(iii) in the third step. Moreover, by Lemma 7.2 (iv),

$$\int_{\Omega} |\nabla u_{n,k}|^2 = \int_{Q(n,k)} dx' \int_0^{f(x')-a_{n,k}} ds \, 2^{2\gamma mn} |\cos\left(2^{\gamma mn}s\right)|^2$$
$$\leq \int_{Q(n,k)} dx' \int_0^{2^{-\gamma mn}} ds \, 2^{2\gamma mn} |\cos\left(2^{\gamma mn}s\right)|^2 = b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn},$$

where we used the change of variables $t = 2^{\gamma mn} s$.

Proof of (ii). Let $\lambda > 0$. For all $x \in \Omega(n,k)$, we have $x_d \in (f_{n-1}(x'), f(x'))$, so by $0 \le g_j \le \frac{1}{2}$ for all $j \in \mathbb{N}_0$, we get as in the proof of Lemma 7.2 (iv)

$$f(x') - x_d \le f(x') - f_{n-1}(x') = \sum_{j=n}^{\infty} 2^{-\gamma m j} g_j(x') \le 2^{-\gamma m n}.$$
 (244)

For all $x \in \text{supp}(V|u_{n,k}|^2) \subset \Omega_{n,k}$, we have by (244) and $\varepsilon < 1$

$$|V(x)| = b_V \left(f(x') - x_d \right)^{\frac{2}{d}(-1+\varepsilon)} \ge b_V 2^{-\gamma m n \frac{2}{d}(-1+\varepsilon)}.$$
(245)

Using (ii) and (iii), we get by (241)

$$\int_{\Omega} |\nabla u_{n,k}|^2 + \int_{\Omega} \lambda V |u_{n,k}|^2 \le b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn} - \lambda b_V 2^{-\gamma mn} \frac{2}{d} (-1+\varepsilon) b_2 2^{-(d-1)mn} 2^{-\gamma mn} = b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn} \left(1 - 2\lambda 2^{-\gamma mn} \frac{2}{d} (-1+\varepsilon) \cdot 2^{-2\gamma mn} \right).$$

Remark 7.6. Lemma 7.5(ii) will be the starting point for the example that satisfies (5) we are looking for in this subsection. If we choose for $n \in \mathbb{N}_0$

$$\lambda := 2^{2\gamma mn} \cdot 2^{\gamma mn} \frac{2}{d} (-1+\varepsilon) \tag{246}$$

in Lemma 7.5(ii), then for all $k \in K_n$, we have

$$\int_{\Omega} |\nabla u_{n,k}|^2 + \int_{\Omega} \lambda V |u_{n,k}|^2 \le b_{\nabla} 2^{-(d-1)mn} \cdot 2^{\gamma mn} \left(1 - 2\lambda 2^{-\gamma mn} \frac{2}{d} (-1+\varepsilon) \cdot 2^{-2\gamma mn} \right) < 0$$

Now suppose $-\Delta_{\Omega}^{N} + \lambda V$ was a self-adjoint operator with quadratic form domain $H^{1}(\Omega)$. In fact, this will be shown under suitable assumptions in Lemma 7.7. Then, we deduce by the min-max principle [18, Theorem 12.1, version 3] and since the $\{u_{n,k}\}_{k \in K_{n}}$ have disjoint interior of their support that

$$N\left(-\Delta_{\Omega}^{N} + \lambda V\right) \ge |K_{n}| = 2^{(d-1)mn}.$$
(247)

If $\varepsilon = (d-1)(1/\gamma - 1)$, then $|K_n| = \lambda^{\frac{d}{2}}$. But we can apply Lemma 7.5 with $0 < \varepsilon < (d-1)(1/\gamma - 1) < 1$ and λ as in (246), then we get

$$\lambda^{-\frac{d}{2}}N\left(-\Delta_{\Omega}^{N}+\lambda V\right) \ge \lambda^{-\frac{d}{2}}|K_{n}| = \lambda^{-\frac{d}{2}}2^{(d-1)mn} \to \infty \text{ as } n \to \infty.$$
(248)

Since $\lambda \to \infty$ as $n \to \infty$, we have shown (5).

Lemma 7.7 (Self-adjointness of $-\Delta_{\Omega}^{N} + \lambda V$ and $\|V\|_{\tilde{p},\beta} = \infty$). Let $\gamma \in \left(\frac{d-1}{d}, 1\right)$.

- (i) Then there exists $0 < \varepsilon < (d-1)(1/\gamma 1)$ such that $V \in L^{p^*}(\Omega) \subset L^{\frac{d}{2}}(\Omega)$ and for every $\lambda > 0$ the operator $-\Delta_{\Omega}^N + \lambda V$ is bounded from below, has finitely many negative eigenvalues and it is self-adjoint with quadratic form domain $H^1(\Omega)$.
- (ii) For every $0 < \varepsilon < (d-1)(1/\gamma 1)$, we have $||V||_{\tilde{p},\beta} = \infty$.

Remark 7.8. Lemma 7.7(ii) shows that this example does not contradict Theorem 1.4.

Proof. **Proof of (i).** Let us first find $0 < \varepsilon < (d-1)(1/\gamma - 1)$ such that $V \in L^{p^*}(\Omega) \subset L^{\frac{d}{2}}(\Omega)$. Note that $L^{p^*}(\Omega) \subset L^{\frac{d}{2}}(\Omega)$ since Ω is bounded and $p^* > \frac{d}{2}$. By (242), the definition of V and the boundedness of Ω , it suffices to find $0 < \varepsilon < (d-1)(1/\gamma - 1)$ such that

$$p^* \frac{2}{d} \left(-1 + \varepsilon \right) > -1. \tag{249}$$

By a continuity argument, it suffices to show that (249) holds for $\varepsilon = (d-1)(1/\gamma - 1)$, namely $\mu(\mu - (d+1)) > -d$ with $\mu := \frac{d-1}{\gamma} + 1 \in (d, d+1)$, which is true.

Fix $\varepsilon = \varepsilon(d, \gamma)$ as above and let $\lambda > 0$ be arbitrary. In order to show the self-adjointness of the operator $-\Delta_{\Omega}^{N} + \lambda V$, we show that the corresponding quadratic form on $H^{1}(\Omega)$ is well-defined, bounded from below and that it has the $H^{1}(\Omega)$ -norm as its quadratic form norm, hence it is closed. The claim then follows from Friedrich's theorem.

For every $x \in \Omega$ with $x_d \ge 0$ and $\delta > 0$, let the oscillatory domain $D_x(\delta)$ and $a_x = a$ be defined as in Definition 2.6. Let $M := \delta/a$ as in the proof of Lemma 3.4(ii). Then

$$\|V\|_{p^*,D_x(\delta)}^{p^*} \le a^{d-1} \int_0^\delta \mathrm{d}t \, \left| b_V t^{\frac{2}{d}(-1+\varepsilon)} \right|^{p^*} = M^{-(d-1)} \delta^{d-1} |b_V|^{p^*} \int_0^\delta \mathrm{d}t \, t^{p^* \frac{2}{d}(-1+\varepsilon)}$$

$$= C(d,\gamma) M^{-(d-1)} \delta^{d+p^* \frac{2}{d}(-1+\varepsilon)} \to 0 \quad \text{as } \delta \to 0.$$
(250)

In the third step we used that p^* , b_V , ε only depend on d and γ . We also used (249) to ensure that the integral is finite.

Let $K = K(d, \gamma)$ be the constant in the covering theorem for oscillatory domains (Lemma 4.1)). By (250), we can choose $\delta > 0$ small enough such that

$$(4\lambda K)^{p^*}C(d,\gamma)\delta^{d+p^*\frac{2}{d}(-1+\varepsilon)} < \left(\frac{1}{C_{PS}}\right)^{p^*}.$$
(251)

Note that δ only depends on d, γ , λ but not on x. For each $x \in \Omega$ with $x_d \ge 0$ let $D_x := D_x(\delta)$ with δ defined in (251). By (251) and (250),

$$\|4\lambda KV\|_{p^*,D_x}^{p^*} M^{d-1} < \left(\frac{1}{C_{PS}}\right)^{p^*},\tag{252}$$

so for every $v \in H^1(\Omega)$ with $\int_{D_x} v = 0$, we have by (102)

$$\int_{D_x} |\nabla v|^2 + \int_{D_x} 4K\lambda V |v|^2 \ge 0.$$
(253)

Let $\mathcal{F}_1, \ldots, \mathcal{F}_K$ be the families of oscillatory domains we get from applying the covering theorem for oscillatory domains (Lemma 4.1) to $\{D_x\}_{x\in\Omega}$ with $x_d>0$. Note that

$$\bigcup_{k=1}^{K} \bigcup_{D \in \mathcal{F}_{k}} D \supset \operatorname{supp}(V).$$
(254)

Using (253) and (254), we obtain in exactly the same way as in (192) in the proof of Lemma 1.7

$$-\Delta_{\Omega}^{N} + \lambda V \ge \frac{1}{2}(-\Delta_{\Omega}^{N}) + \frac{1}{2K}\sum_{k=1}^{K}\sum_{D\in\mathcal{F}_{k}}(-\Delta_{D}^{N} + 2\lambda KV1_{D})$$

$$\geq \frac{1}{2}(-\Delta_{\Omega}^{N}) - 2\lambda \left(\sum_{k=1}^{K} \sum_{D \in \mathcal{F}_{k}} \frac{1}{|D|} \int_{D} |V|\right) \mathbf{1}_{\Omega}.$$

For each $k \in \{1, \ldots, K\}$ and $D \in \mathcal{F}_k$, we have

$$|D| \ge \frac{1}{4} \delta a^{d-1} \ge \frac{1}{4} \delta \left(c_0 (c_1 \delta)^{\frac{1}{\gamma}} \right)^{d-1} = \frac{1}{4} c_0^{d-1} c_1^{\frac{d-1}{\gamma}} \delta^{1+\frac{d-1}{\gamma}}.$$
 (255)

Since the domains $D \in \mathcal{F}_k$ for each $k \in \{1, \ldots, K\}$ are all disjoint and Ω is bounded, we deduce that $\sum_{k=1}^{K} |\mathcal{F}_k| < \infty$. Thus, since $V \in L^{p^*}(\Omega) \subset L^1(\Omega)$, we get

$$\sum_{k=1}^{K} \sum_{D \in \mathcal{F}_k} \int_D |V| \frac{1}{|D|} < \infty, \tag{256}$$

so $-\Delta_{\Omega}^{N} + \lambda V$ is a well-defined quadratic form on $H^{1}(\Omega)$ that is bounded from below and has the $H^{1}(\Omega)$ -norm as its quadratic form norm. We can deduce as in the proof of Lemma 1.8 that $-\Delta_{\Omega}^{N} + \lambda V$ has finitely many negative eigenvalues.

Proof of (ii). By the definition of V and of $||V||_{\tilde{p},\beta}$, it suffices to consider the case $\varepsilon := (d-1)(1/\gamma - 1)$. Recall that

$$\operatorname{supp}\left(V\right) = \left\{x = (x', x_d) \in \mathbb{R}^{d-1} \times \mathbb{R} \mid x_d \ge 0\right\} \cap \Omega.$$
(257)

For all $x \in \Omega$ with $x_d \ge 0$ we use the shorthand notation

$$h_x := f(x') - x_d. (258)$$

By the definition of $\|V\|_{\tilde{p},\beta}$, it suffices to show $|V|_{\tilde{p},\beta}^{\tilde{p}} = \infty$. Recall that

$$|V|_{\tilde{p},\beta}^{\tilde{p}} = \int_{\Omega} \mathrm{d}x \, |V(x)|^{\tilde{p}} h_x^{-\beta} = \int_{\Omega} |b_V|^{\tilde{p}} h_x^{\tilde{p}\frac{2}{d}(-1+\varepsilon)} h_x^{-\beta}.$$
(259)

Now, recall that by (238), we have

$$f(x') = \sum_{j=0}^{\infty} 2^{-\gamma j m} g_j(x') \ge g_0(x') \ge \frac{1}{4}$$
(260)

for all $x' \in [1/4, 3/4]^{d-1}$, so

$$\begin{aligned} |V|_{\tilde{p},\beta}^{\tilde{p}} &= |b_V|^{\tilde{p}} \int_{\Omega} h_x^{\tilde{p}_d^2(-1+\varepsilon)-\beta} = |b_V|^{\tilde{p}} \int_{Q^{(d-1)}} \mathrm{d}x' \int_0^{f(x')} \mathrm{d}x_d \ \left(f(x') - x_d\right)^{\tilde{p}_d^2(-1+\varepsilon)-\beta} \\ &\geq |b_V|^{\tilde{p}} \int_{\left[\frac{1}{4},\frac{3}{4}\right]^{d-1}} \mathrm{d}x' \int_0^{f(x')} \mathrm{d}t \ t^{\tilde{p}_d^2(-1+\varepsilon)-\beta} \geq |b_V|^{\tilde{p}} 2^{-(d-1)} \int_0^{\frac{1}{4}} \mathrm{d}t \ t^{\tilde{p}_d^2(-1+\varepsilon)-\beta}. \end{aligned}$$

Hence, if we can show that

$$\tilde{p}\frac{2}{d}(-1+\varepsilon) - \beta \le -1, \tag{261}$$

then we have $|V|_{\tilde{p},\beta}^{\tilde{p}} = \infty$, so $||V||_{\tilde{p},\beta} = \infty$. Recalling (22), the bound (261) is equivalent to

$$f(\mu) = \frac{1}{d^2} \mu^2 \left(\mu - (d+1)\right) - \frac{1}{d+1} \mu \left[\frac{1}{d} \mu^2 - d\right] \le -1$$
(262)

with $\mu := \frac{d-1}{\gamma} + 1 \in (d, d+1)$. Note that

$$f'(\mu) = \frac{3\mu^2}{d^2(d+1)} - \frac{2(d+1)}{d^2}\mu + \frac{d}{d+1}$$

REFERENCES

is a convex function with f'(d) < 0 and f'(d+1) < 0. Therefore, f is monotone decreasing in (d, d+1), and hence (262) follows from the fact that f(d) = -1. The proof of Lemma 7.7 is complete.

Proof of Theorem 1.1. Remark 7.6 combined with Lemma 7.7(i) show Theorem 1.1. \Box

References

- [1] E. H. Lieb and R. Seiringer. *The stability of matter in quantum mechanics*. Cambridge University Press, 2010.
- [2] H. Weyl. "Über die asymptotische Verteilung der Eigenwerte". In: Nachrichten von der Gesellschaft der Wissenschaften zu Göttingen, Mathematisch-Physikalische Klasse 1911 (1911), pp. 110–117.
- [3] H. Weyl. "Das asymptotische Verteilungsgesetz der Eigenwerte linearer partieller Differentialgleichungen (mit einer Anwendung auf die Theorie der Hohlraumstrahlung)". In: Mathematische Annalen 71.4 (1912), pp. 441–479.
- [4] G. V. Rozenblum. "On the distribution of eigenvalues of the first boundary value problem in unbounded regions". In: *Doklady Akademii Nauk*. Vol. 200. 5. Russian Academy of Sciences. 1971, pp. 1034–1036.
- [5] G. Rozenbljum. "On the eigenvalues of the first boundary value problem in unbounded domains". In: *Mathematics of the USSR-Sbornik* 18.2 (1972), p. 235.
- [6] V. Y. Ivrii. "Second term of the spectral asymptotic expansion of the Laplace-Beltrami operator on manifolds with boundary". In: *Functional Analysis and Its Applications* 14.2 (1980), pp. 98–106.
- [7] M. S. Birman and M. Z. Solomyak. "The principal term of the spectral asymptotics for "non-smooth" elliptic problems". In: *Funktsional. Anal. i Prilozhen* 4.4 (1970), pp. 1–13.
- [8] R. L. Frank, A. Laptev, and T. Weidl. Schrödinger operators: eigenvalues and Lieb-Thirring inequalities. Vol. 200. Cambridge University Press, 2022.
- [9] R. Hempel, L. A. Seco, and B. Simon. "The essential spectrum of Neumann Laplacians on some bounded singular domains". In: *Journal of Functional Analysis* 102.2 (1991), pp. 448–483.
- [10] Y. Netrusov and Y. Safarov. "Weyl asymptotic formula for the Laplacian on domains with rough boundaries". In: *Communications in Mathematical Physics* 253.2 (2005), pp. 481–509.
- [11] M. Cwikel. "Weak type estimates for singular values and the number of bound states of Schrodinger operators". In: Annals of Mathematics 106 (1977), pp. 93–100.
- [12] E. Lieb. "Bounds on the eigenvalues of the Laplace and Schrödinger operators". In: Bulletin of the American Mathematical Society 82.5 (1976), pp. 751–753.
- [13] G. Rozenbljum. "Distribution of the discrete spectrum of singular differential operators". In: Soviet Math. Doki. 202 (1972), pp. 1012–1015.
- [14] R Frank and A. Laptev. "Bound on the number of negative eigenvalues of twodimensional Schrödinger operators on domains". In: St. Petersburg Mathematical Journal 30.3 (2019), pp. 573–589.
- [15] M Solomyak. "Piecewise-polynomial approximation of functions from $H^{l}((0,1)^{d})$, 2l = d, and applications to the spectral theory of the Schrödinger operator". In: *Israel Journal of Mathematics* 86.1 (1994), pp. 253–275.
- [16] R. L. Frank, E. H. Lieb, and R. Seiringer. "Equivalence of Sobolev inequalities and Lieb-Thirring inequalities". In: XVIth International Congress On Mathematical Physics. World Scientific. 2010, pp. 523–535.

REFERENCES

- [17] D. A. Labutin. "Embedding of Sobolev spaces on Hölder domains". In: Trudy Matematicheskogo Instituta imeni VA Steklova 227 (1999), pp. 170–179.
- [18] E. H. Lieb and M. Loss. Analysis. 2nd ed. Vol. 14. American Mathematical Soc., 2001.
- [19] M. M. Rao and Z. D. Ren. Theory of Orlicz spaces. Vol. 146. Monographs and Textbooks in Pure and Applied Mathematics. Marcel Dekker, Inc., New York, 1991, pp. xii+449.

Department of Mathematics, LMU Munich, Theresienstr. 39, 80333 Munich, Germany Institut des Hautes Études Scientifiques, 35 route de Chartres, 91440 Bures-sur-Yvette, France

 $Email \ address: \ \tt{dietzeQmath.lmu.de}$

42

Weyl formulae for some singular metrics with application to acoustic modes in gas giants

Yves Colin de Verdière^{*}

Charlotte Dietze[†], Maarten V. de Hoop[‡], Emmanuel Trélat[§]

Abstract

This paper is motivated by recent works on inverse problems for acoustic wave propagation in the interior of gas giant planets. In such planets, the speed of sound is isotropic and tends to zero at the surface. Geometrically, this corresponds to a Riemannian manifold with boundary whose metric blows up near the boundary. Here, the spectral analysis of the corresponding Laplace-Beltrami operator is presented and the Weyl law is derived. The involved exponents depend on the Hausdorff dimension which, in the supercritical case, is larger than the topological dimension.

AMS classification: 11F72, 58C40.

1 Introduction

1.1 Seismology on gas giant planets

Seismology has played an important role in revealing the (deep) interiors of gas giant planets in our solar system [6, 24]. Indeed, the acoustic spectra and free oscillations have been studied for Saturn and Jupiter over the past few decades [33, 20, 18]. The excitation of acoustic modes in gas giant planets presumably occurs through convection in their interiors. The observation of acoustic eigenfrequencies, that is, the discrete spectrum can be realized, in principle, through visible photometry, thermal infrared photometry, Doppler spectrometry, and ring seismology for nonradial oscillations [25, 26] (in particular, in the case of Saturn). In ring seismology and with the Cassini mission, one measured the "resonances" in the inner C ring of Saturn with visual and infrared mapping spectrometer (VIMS) stellar occultations [22, 14, 17]. The rings are gravitationally coupled to the acoustic modes of the planet (taking self gravitation into account). Detection of Jupiter's acoustic eigenvalues has been attempted with ground-based imaging-spectrometry (seismographic imaging interferometer for monitoring of planetary atmospheres or SYMPA) by measuring line of sight velocity [32, 18]. Recently, Juno spacecraft gravity measurements have provided evidence for normal modes of Jupiter [15].

^{*}Université Grenoble-Alpes, Institut Fourier, Unité mixte de recherche CNRS-UGA 5582, BP 74, 38402-Saint Martin d'Hères Cedex (France) (yves.colin-de-verdiere@univ-grenoble-alpes.fr)

[†]Mathematisches Institut der Universität München (dietze@mathematik.uni-muenchen.de)

[‡]Simons Chair in Computational and Applied Mathematics and Earth Science, Rice University (mvd2@rice.edu) [§]Sorbonne Université, CNRS, Université Paris Cité, Inria, Laboratoire Jacques-Louis Lions (LJLL), F-75005

Paris, France (emmanuel.trelat@sorbonne-universite.fr)

1.2 Singular Riemannian metrics

On a gas giant planet, unlike a rocky planet, the speed of sound goes to zero at the boundary. In the geometric mathematical model that we employ hereafter, the rate at which this happens follows a power law which determines a specific conformal blow-up rate of a Riemannian metric, thus defining a singular metric. This rate happens to be slower than on asymptotically hyperbolic manifolds and the boundary is at a finite distance from interior points. The rate is implied by an equation of state in the upper part of the planet, in general, in the sense of a fit. (For some models of the speed of sound of Jupiter and Saturn showing this behavior, see [20, Figure 1] and [23, Figure 1].) Only for a polytrope is the rate exact. Polytropes, for which the pressure is proportional to a power of the density of mass, have been viewed as relevant simplifications; models with variable polytrope index have indeed been applied to planet and material models [34]. Typically, an equation of state is computed numerically using density functional molecular dynamics simulations with mixtures of chemical elements: The dominant elements in terms of mass fraction are hydrogen and helium, but also heavy elements are important. The equation of state is different for the upper part and the deep interior as the helium fraction can be higher in the interior due to helium rain (helium becoming immiscible with hydrogen at high pressure). Equations of state play a vital role in the evolution and realization of structure of gas giant planets [28, 29].

More specifically, if g_e is the Euclidean Riemannian metric on a smooth domain $X \subset \mathbb{R}^{n+1}$, then the speed of sound c can be encoded by the conformally Euclidean Riemannian metric $g = c^{-2}g_e$. In local coordinates where the boundary of X is (locally) described by u = 0, the polytropic model suggests that $c \sim u^{1/2}$. Indeed, the natural generalization is $c \sim u^{\alpha/2}$, that is, $c^{-2} \sim u^{-\alpha}$; through previous analysis [13] it appears that restricting α according to $\alpha \in (0, 2)$ guarantees the presence of a discrete spectrum as it has been observed. Thus, the Riemannian geometry lies between standard geometry with boundary and asymptotically hyperbolic geometry. Some of the phenomena in this geometry are unlike those seen at either end. The extreme case $\alpha = 0$ corresponds physically to solid bodies and mathematically to manifolds with boundary, and the other extreme $\alpha = 2$ corresponds to asymptotically hyperbolic geometry but is far from all planetary models.

Therefore, following [13, Section 1.1], we model a gas giant planet as a smooth manifold X with a boundary, endowed with a Riemannian metric g on $X \setminus \partial X$ such that, near ∂X , we have $g = \bar{g}/u^{\alpha}$ where \bar{g} is a well-defined Riemannian metric up to the boundary, and $\partial X = \{u = 0\}$ locally. The fact that \bar{g} is neither zero nor infinite at ∂X implies a specific blow-up rate for g near ∂X . This conformal power-law blow-up is the key geometric feature of gas giant metrics. The speed of sound might contain jump discontinuities where phase transitions occur (see [27]), that is, the metric can contain conormal singularities while the manifold consists of multiple "layers". A key interior boundary in gas giants corresponds with the transition from molecular to metallic hydrogen. Accounting for discontinuities in an asymptotic formalism for gas giant seismology was developed a few decades ago (see [30]).

The mathematical study of the spectrum associated with gas giants' acoustic modes was initiated in [13]. In this paper, we analyze the relevant Laplace-Beltrami operator and we compute the Weyl law. The study of Weyl asymptotics, which reflects some properties of the singular metric, is a preliminary step towards analyzing some inverse problems, in view of reconstructing some features of the internal structure of gas giant planets.
2 Mathematical model and main results

2.1 Mathematical model

Let X be a smooth compact manifold of dimension n + 1 with a boundary ∂X . Near ∂X , X is diffeomorphic to $[0, 1) \times M$, where M is a smooth compact manifold of dimension $n \ge 1$ and ∂X is identified with $\{0\} \times M$ and also with u = 0 where u is a transverse coordinate, locally near ∂X , ranging over [0, 1). As discussed in Section 1.2, we consider on X a singular Riemannian metric g that is a smooth metric on $X \setminus \partial X$, written near ∂X as

$$g = \bar{g}/u^{\alpha}$$

where $0 < \alpha < 2$ and \bar{g} is a smooth (non-singular) Riemannian metric on X, up to the boundary. Following [13, Proposition 2], which uses a normal form for the metric near the boundary, due to [19, Lemma 5.2], we have

$$g = u^{-\alpha} (du^2 + g_0(u))$$

where $g_0(u)$ is a smooth Riemannian metric on M (pulled back to the level set u = Cst) depending smoothly on $u \in [0, 1)$.

We make a change of variable. Setting $x = x(u) = \int_0^u s^{-\alpha/2} ds = (1 - \frac{\alpha}{2})^{-1} u^{1-\frac{\alpha}{2}}$, we get

$$g = dx^2 + x^{-\beta}g_1(x) \text{ where } \beta = \frac{2\alpha}{2-\alpha}$$
(1)

and $g_1 = g_1(x)$ is a smooth Riemannian metric on M (pulled back to the level set x = Cst) depending smoothly on $x \in [0, 1)$. We note that, since $\alpha \in (0, 2)$, β can take any positive value. We also note that a polytrope (for any index) corresponds to $\beta = \beta_{\text{poly}} = 2$. We have that $g_1(x) = C(\alpha)g_0(u)$ for some constant $C(\alpha) > 0$.

For any $x \in [0, 1)$, denoting by dv_1^x the volume measure on M associated to the metric $g_1(x)$, the g-volume is $dv_g = x^{-\beta n/2} |dx| dv_1^x$. The volume is finite if and only if $\beta < \beta_c$, where

$$\beta_c = \frac{2}{n} \tag{2}$$

is a critical value of β . We will see later that this critical value plays a role in the Weyl asymptotics. At this point, we can note that $\beta_{\text{poly}} > \beta_c$ for n = 2.

The following three propositions were proved in [13]. The first proposition concerns the Haussdorff dimension.

Proposition 1. The Hausdorff dimension of (X, g) is

$$d_H = \max\left(n+1, n\left(1+\frac{\beta}{2}\right)\right).$$

We define $\delta_H = n(1 + \frac{\beta}{2})$, and note that $d_H > n + 1$ (n + 1) is the topological dimension of X) if and only if $\beta > \beta_c$. We give in Appendix A.5 a sketch of the proof of Proposition 1, in which we also show that d_H coincides with the Minkowski dimension of (X, g).

Proposition 2. The Laplace-Beltrami operator \triangle_g , with core $C_0^{\infty}(X \setminus \partial X)$, is essentially selfadjoint if and only if $\beta \ge \beta_c$.

For $\beta < \beta_c$, there exist several extensions of Δ_g , with core $C_0^{\infty}(X \setminus \partial X)$. In the further analysis, we consider its Friedrichs extension (that is, "Dirichlet extension").

Proposition 3. For every $\beta > 0$, the spectrum of \triangle_g is discrete.

We denote the eigenvalues of \triangle_g by $0 < \lambda_1 \leq \lambda_2 \leq \cdots \leq \lambda_j \leq \cdots$ with associated eigenfunctions ϕ_j , $j \in \mathbb{N}^*$, normalized in $L^2(X, dv_g)$. We note that, if the volume of X is infinite, i.e., if $\beta \geq \beta_c$, we have $\lambda_1 > 0$, in contrast to the usual Riemannian case. The Weyl counting function is defined by

$$N(\lambda) = \#\{j \in \mathbb{N}^* \mid \lambda_j \leq \lambda\}$$

where the eigenvalues are counted with their multiplicity. Our objective is to derive a Weyl law describing the asymptotics of $N(\lambda)$ as $\lambda \to +\infty$.

Remark 1. The following fact will be used in Section 3.5. For any $\varepsilon > 0$, there exists $\delta > 0$ such that the metric g is ε -quasi-isometric (see Appendix A.1) to a singular Riemannian metric \tilde{g} on X, smooth on $X \setminus \partial X$ and given by $\tilde{g} = dx^2 + x^{-\beta}g_1(0)$ on $(0, \delta) \times M$. In order to derive a Weyl law for (X, g) it suffices to derive the corresponding Weyl law for \tilde{g} for any $\varepsilon > 0$ (see, again, Appendix A.1 for details). This remark is important, because it implies that we mainly have to work within the so-called *separable* case.

Separable case. We say that we are in the *separable* case if the metric $g_1(x)$ on M (defined by (1)) does not depend on x, i.e., $g_1(x) = g_1(0)$ for any $x \in (0, 1)$; we still denote this metric by g_1 . In the sequel, we consider $[0, 1) \times M$ instead of $[0, \delta) \times M$ for simplicity of notation, while the proofs are similar in both cases.

We denote by Δ_M the Laplace-Beltrami operator on (M, g_1) . We denote the eigenvalues of Δ_M by $0 \leq \omega_1 \leq \omega_2 \leq \cdots \leq \omega_j \leq \cdots$ with an associated orthonormal basis of eigenfunctions $(\psi_j)_{j \in \mathbb{N}^*}$. The Weyl counting function for Δ_M is defined by

$$N_M(\omega) = \#\{k \in \mathbb{N}^* \mid \omega_k \leqslant \omega\}.$$

Since g_1 is a smooth Riemannian metric on M, the classical Weyl law for (M, g_1) yields that $N_M(\omega) = \gamma_n \operatorname{Vol}_{g_1}(M) \omega^{n/2} + O(\omega^{n/2})$ as $\omega \to +\infty$ where

$$\gamma_n = \frac{1}{(4\pi)^{n/2} \Gamma\left(\frac{n}{2} + 1\right)}$$
(3)

(see [3, Chapter 3E] for the heat trace and then apply the Karamata tauberian theorem, i.e., Theorem 3 in Appendix A.2).

Denoting by dv_1 the volume measure on M associated to the metric $g_1 = g_1(0)$, the g-volume is $dv_g = x^{-\beta n/2} |dx| dv_1$. Making the change of function $f \mapsto x^{-\beta n/4} f$, we get the new volume form $|dx| dv_1$; the Laplace-Beltrami operator on $X_1 = (0, 1) \times M$ is now given by

$$\triangle_g = -\partial_x^2 + \frac{C_\beta}{x^2} + x^\beta \triangle_M$$

where $x \in (0, 1)$ and

$$C_{\beta} = \frac{\beta n}{4} \left(\frac{\beta n}{4} + 1 \right).$$

The proof is straightforward by performing an integration by parts with respect to x in the Dirichlet form defining the Laplace-Beltrami operator, using the Dirichlet boundary condition at x = 0. We note that $C_{\beta} \ge 3/4$ if and only if the volume of X is infinite. Using the Weyl criterion (see Appendix A.3), this inequality also implies that P_{ω} defined by (5) below is essentially self-adjoint for any $\omega > 0$, but not for $\omega = 0$. We will need to work first on the non-compact conic manifold $X_{\infty} = (0, +\infty) \times M$ endowed with the metric $g = dx^2 + x^{-\beta}g_1$. Let Δ_{∞} stand for the Laplace-Beltrami operator on (X_{∞}, g) . Invoking a separation of variables, we have

$$\Delta_{\infty} = \bigoplus_{k=1}^{+\infty} (\mathrm{id} \otimes \pi_k) \left(P_{\omega_k} \otimes \mathrm{id} \right) \left(\mathrm{id} \otimes \pi_k \right) \tag{4}$$

where

$$P_{\omega} = -\partial_x^2 + \frac{C_{\beta}}{x^2} + \omega x^{\beta} \tag{5}$$

is a Schrödinger operator on $L^2((0, +\infty), dx)$ for any $\omega \ge 0$, and where π_k is the orthogonal projection of $L^2(M, dv_1)$ onto the subspace generated by ψ_k and id denotes the identity operator on $L^2(X_{\infty}, |dx| dv_1)$ (resp., on $L^2(M, dv_1)$). Hence, Δ_{∞} is unitarily equivalent to $\bigoplus_{k=1}^{+\infty} P_{\omega_k}$.

2.2 Main results

Recalling that g_1 is defined by (1), we set $G = g_1(0)$ and denote by v_G the corresponding volume form on M. We also recall that β_c is defined in (2) and that γ_n is defined in (3).

Theorem 1. (Weyl asymptotics)

• If $\beta > \beta_c$ then

$$N(\lambda) \sim A(\beta, n) v_G(M) \,\lambda^{d_H/2}$$

as $\lambda \to +\infty$, with

$$A(\beta, n) = \frac{n\gamma_n(\beta+2)}{4\Gamma(1+d_H/2)} \int_0^{+\infty} \mathcal{Z}_1(\tau)\tau^{\frac{d_H}{2}-1} d\tau$$

where $\mathcal{Z}_1(\tau) = \text{Tr}(\exp(-\tau P_1))$ and P_1 is the Schrödinger operator on $L^2((0, +\infty), dx)$ defined by (5).

• If $\beta = \beta_c = 2/n$ then

$$N(\lambda) \sim C_n v_G(M) \,\lambda^{(n+1)/2} \ln \lambda$$

as $\lambda \to +\infty$, with

$$C_n = \frac{1}{(n+1)(4\pi)^{(n+1)/2}\Gamma((n+1)/2)}$$

In particular, $C_1 = 1/8\pi$.

• If $\beta < \beta_c$ then

$$N(\lambda) \sim \gamma_{n+1} v_g(X) \,\lambda^{(n+1)/2}$$

as $\lambda \to +\infty$.

Remark 2.

- When $M = \mathbb{R}/2\pi\mathbb{Z}$ and X is diffeomorphic to the hemisphere, endowed with the so-called Grushin metric, the authors of [4] derived the Weyl law using an explicit computation of the spectrum. We recover their result as a particular case with n = 1 and $\beta = \beta_c = 2$.
- To prove Theorem 1, we make use of heat kernels. Alternatively, it is possible to use Dirichlet-Neumann bracketing. Both methods allow to treat conormal jump singularities of the metric \bar{g} inside X that model layering in the gas planet.

Remark 3. A natural question is whether β can be determined from the Weyl asymptotics. Indeed, when *n* is known, β can be determined in the case where $\beta \ge \beta_c$. When $\beta < \beta_c$ the question remains open. To shed light on this, it would be useful and interesting to get the next term in the small-time heat trace expansion (see Section 3.2.2) when $\beta \le \beta_c$.

We next compute the Weyl measures, which are the probability measures w_g on X, defined, if the limit exists, by

$$\int_X f dw_g = \lim_{\lambda \to +\infty} \frac{1}{N(\lambda)} \sum_{\lambda_j \leq \lambda} \int_X f |\phi_j|^2 dv_g$$

for any function $f: X \to \mathbb{R}$ that is continuous up to the boundary of X. Such measures have been introduced in [9, 11] in the framework of sub-Riemannian geometry in order to provide an account of how the high-frequency eigenfunctions concentrate.

Theorem 2. (Weyl measures)

- If $\beta \ge \beta_c$ then the Weyl measure is $\delta_{x=0} \otimes dv_G/v_G(M)$.
- If $\beta < \beta_c$ then the Weyl measure is the uniform probability distribution given by the normalized volume of (X, g), that is $dv_g/v_g(X)$.

Using [11, Corollary 7.1], we obtain the following consequence.

Corollary 1. If $\beta \ge \beta_c$ then there exists a density-one subsequence $(\phi_{j_k})_{k \in \mathbb{N}^*}$ of the sequence of eigenfunctions that concentrates on ∂X , meaning that for any compact subset $K \subset X \setminus \partial X$, we have

$$\lim_{k \to +\infty} \int_K |\phi_{j_k}|^2 \, dv_g = 0.$$

3 Proofs of Theorems 1 and 2

Our strategy of proof is the following. We first treat the separable case (Sections 3.1 to 3.4). As a preliminary, we perform in Section 3.1 a spectral study of the 1D Schrödinger operator P_{ω} defined by (5), deriving exponential estimates for truncated heat traces. Then, in Section 3.2, we estimate the small-time asymptotics of the truncated heat trace of Δ_g , near the boundary (actually, on a cone); the three cases $\beta > \beta_c$, $\beta = \beta_c$, $\beta \leq \beta_c$, must be treated in different ways. In Section 3.3, using a heat parametrix, we glue together the heat kernel near the boundary and the Riemannian heat kernel far from the boundary. Finally in Section 3.4 we prove Theorem 1 in the separable case.

In Section 3.5, we show how to pass from the separable to the general case by using the fact that the metric g is quasi-isometric to a separable metric. In Section 3.6, we prove Theorem 2. Our approach uses again heat traces.

3.1 Spectral study of the 1D Schrödinger operators

We consider the family of Schrödinger operators,

$$P_1 = -\partial_x^2 + q_{C,\beta}(x)$$

where $q_{C,\beta}(x) = Cx^{-2} + x^{\beta}$, C > 0 and $\beta > 0$, acting on $L^2((0, +\infty), dx)$. The operators P_1 are essentially self-adjoint if and only if $C \ge 3/4$; when C < 3/4 we consider the Friedrichs extension of P_1 with core $C_0^{\infty}((0, +\infty))$ (see Appendix A.3). The spectrum of P_1 is discrete; we denote it by $0 < \mu_1 \leq \mu_2 \leq \mu_3 \leq \cdots$. We derive precise semi-classical asymptotics for the associated truncated heat trace.

Let $\chi : [0, +\infty) \to [0, 1]$ be a smooth decreasing function with $\chi \equiv 1$ on [0, a] with a > 0 and $\chi' \leq 0$ everywhere. We note that $\chi \equiv 1$ is included. We define the corresponding truncated heat trace by

$$\mathcal{Z}_{\chi}(\tau) = \operatorname{Tr}\left(e^{-\tau P_1}\chi\right) \qquad \forall \tau > 0.$$

Let $\gamma = \max(1/\beta, 1/2)$.

Proposition 4. Given any $0 < \tau \leq 1$, we have

$$\mathcal{Z}_{\chi}(\tau) = \frac{1}{\sqrt{4\pi\tau}} \int_{0}^{+\infty} e^{-\tau x^{\beta}} \chi(x) dx + \mathcal{O}\left(\tau^{-\gamma}\right)$$

and for $\tau \ge 1$,

$$\mathcal{Z}_{\chi}(\tau) = \mathcal{O}\left(e^{-\mu_{1}\tau}\right)$$

uniformly with respect to χ in both cases.

The counting function $N_1(\mu) = \#\{j \in \mathbb{N}^* \mid \mu_j \leq \mu\}$ satisfies $N_1(\mu) \sim A\mu^{\frac{1}{2} + \frac{1}{\beta}}$ as $\mu \to +\infty$ with $A = \sqrt{\frac{2}{\pi}} \frac{1}{\beta} B(3/2, 1 + 1/\beta)$ where B is the Beta function.

Proof of Proposition 4. We first establish an elementary lemma. We denote by Q^N (resp., Q^D) the self-adjoint operator $-\partial_x^2$ on an interval of length 1 with Neumann (resp., Dirichlet) boundary condition.

Lemma 1. For
$$0 < \tau \leq 1$$
 and $\star \in \{D, N\}$, we have $\operatorname{Tr}(\exp(-\tau Q^{\star})) = (4\pi\tau)^{-1/2} + O(1)$.

Proof of Lemma 1. The estimate does not depend on the chosen interval. The spectrum of Q^N is $\{n^2\pi^2 \mid n \in \mathbb{N}\}$ and the spectrum of Q^D is $\{n^2\pi^2 \mid n \in \mathbb{N}^*\}$. Hence both traces differ by 1, and it suffices to prove the estimate for Q^N . Writing

$$\operatorname{Tr}(\exp(-\tau Q^N)) = \frac{1}{2} \left(1 + \sum_{n \in \mathbb{Z}} e^{-\tau n^2 \pi^2} \right)$$

and applying the Poisson summation formula gives the result.

We now prove the proposition. We first consider the case where $\tau \leq 1$. We are going to apply Dirichlet-Neumann bracketing with the decomposition $(0, +\infty) = \bigcup_{j=0}^{+\infty} J_k$ where the intervals J_k are defined below.

Let x_0 be defined by $q_{C,\beta}(x_0) = \min q_{C,\beta}(x)$. Then $q_{C,\beta}(x) = Cx^{-2} + x^{\beta}$ is increasing on $[x_0, +\infty)$. Let $J_k = [x_0 + k, x_0 + k + 1]$ with $k \ge 1$ and $J_0 = [0, x_0 + 1]$. We have the following estimates for the Dirichlet and Neumann heat traces $\mathcal{Z}_{k,\chi}^{\star}$ on J_k : for $k \ge 1$,

$$\mathcal{Z}_{k,1}^{N}(\tau) \leqslant \left(\frac{1}{\sqrt{4\pi\tau}} + \mathcal{O}(1)\right) e^{-\tau q(x_0+k)} \leqslant \left(\frac{1}{\sqrt{4\pi\tau}} + \mathcal{O}(1)\right) \int_{x_0+k-1}^{x_0+k} e^{-\tau q_{C,\beta}(x)} dx$$

and

$$\begin{aligned} \mathcal{Z}_{k,\chi}^{D}(\tau) \geqslant \left(\frac{1}{\sqrt{4\pi\tau}} + \mathcal{O}(1)\right) e^{-\tau q(x_0+k+1)} \chi(x_0+k+1) \\ \geqslant \left(\frac{1}{\sqrt{4\pi\tau}} + \mathcal{O}(1)\right) \int_{x_0+k+1}^{x_0+k+2} e^{-\tau q_{C,\beta}(x)} \chi(x) dx, \end{aligned}$$
(6)

$$\mathcal{Z}_0^N(\tau) = \mathcal{O}(\tau^{-1/2}).$$

The minimax principle implies that each eigenvalue μ_j is larger than the j^{th} -eigenvalue of the union for all k of the Neumann problem on the intervals J_k . In this way, we obtain the following upper bound for the trace with $\chi = 1$:

$$\mathcal{Z}_{1}(\tau) \leqslant \mathcal{O}\left(\frac{1}{\sqrt{\tau}}\right) + \left(\frac{1}{\sqrt{4\pi\tau}} + \mathcal{O}(1)\right) \int_{x_{0}}^{+\infty} e^{-\tau q_{C,\beta}(x)} dx.$$
(7)

Noting that

$$\left| \int_0^{+\infty} \left(e^{-\tau q_{C,\beta}(x)} - e^{-\tau x^{\beta}} \right) dx \right| = \mathcal{O}(1),$$

we infer that

$$\mathcal{Z}_1(\tau) \leqslant \frac{1}{\sqrt{4\pi\tau}} \int_0^{+\infty} e^{-\tau x^{\beta}} dx + \mathcal{O}\left(\tau^{-\gamma}\right).$$

Similarly, using the fact that the Dirichlet heat kernel is smaller than the global kernel (see [12, Theorem 2.1.6]), we get from (6) the following lower bound:

$$\mathcal{Z}_{\chi}(\tau) \ge \frac{1}{\sqrt{4\pi\tau}} \int_{0}^{+\infty} e^{-\tau x^{\beta}} \chi(x) dx + \mathcal{O}\left(\tau^{-\gamma}\right).$$
(8)

Note that the same lower bound is valid when replacing χ by $1 - \chi$.

Now, we use a variant of the fact that

$$(A+B=A'+B',\ A\geqslant A',\ B\geqslant B')\quad\Rightarrow\quad (A=A',\ B=B').$$

We take $A = \mathcal{Z}_{\chi}, B = \mathcal{Z}_{1-\chi}, A' = (4\pi\tau)^{-\frac{1}{2}} \int_{0}^{+\infty} e^{-\tau x^{\beta}} \chi(x) dx$ and $B' = (4\pi\tau)^{-\frac{1}{2}} \int_{0}^{+\infty} e^{-\tau x^{\beta}} (1 - \chi(x)) dx$. By (7), we have $A + B = A' + B' + O(\tau^{-\gamma})$

and, from (8),

$$A \geqslant A' + \mathcal{O}\left(\tau^{-\gamma}\right), \ B \geqslant B' + \mathcal{O}\left(\tau^{-\gamma}\right).$$

It follows that $A = A' + O(\tau^{-\gamma})$ and $B = B' + O(\tau^{-\gamma})$. In particular,

$$\mathcal{Z}_1(\tau) \sim \frac{1}{(4\pi\tau)^{\frac{1}{2}}} \int_0^{+\infty} e^{-\tau x^{\beta}} dx.$$

Using the Karamata tauberian Theorem (recalled in Appendix A.2), we get

 $N_1(\mu) \sim A\mu^{\frac{1}{2} + \frac{1}{\beta}}$

as $\mu \to +\infty$, with the constant A defined in Proposition 4.

We now prove the exponential upper bound for $\tau \ge 1$. We note that $\mathcal{Z}_{\chi}(\tau) \le \mathcal{Z}_{1}(\tau)$. Moreover all eigenvalues μ_{j} of P_{1} are larger than the minimum $q(x_{0})$ of q. Then, for $\tau \ge 1$, we have

$$\mathcal{Z}_1(\tau) = e^{-\tau\mu_1} \sum_{j=1}^{+\infty} e^{-\tau(\mu_j - \mu_1)} \leqslant e^{-\tau\mu_1} \sum_{j=1}^{+\infty} e^{-(\mu_j - \mu_1)}.$$

The Weyl law applied to P_1 implies that the sum at the right-hand side converges and thus $\mathcal{Z}_{\chi}(\tau) \leq c e^{-\tau \mu_1}$ for some c > 0.

Corollary 2. There exists $C_1 > 0$ only depending on C and β but not on χ , such that, setting $J = \int_0^{+\infty} \chi(x) dx$, we have, for every $\tau > 0$,

$$\mathcal{Z}_{\chi}(\tau) \leqslant \frac{C_1}{\sqrt{\tau}} \min\left(J, \tau^{-1/\beta}\right).$$

Proof. The bound involving J follows from the estimate $e(\tau, x, x) \leq (4\pi\tau)^{-\frac{1}{2}}$ which is valid for any positive potential q (see [12, Theorem 2.1.6]). The other bound follows from Proposition 4 and from the estimate $\int_0^{+\infty} e^{-\tau x^{\beta}} dx = O(\tau^{-1/\beta})$.

Given any $\omega > 0$, we set

$$P_{\omega} = -\partial_x^2 + Cx^{-2} + \omega x^{\beta}. \tag{9}$$

Proposition 5. For any $\omega > 0$, the operator P_{ω} is unitarily equivalent to $\omega^{2/(2+\beta)}P_1$. In particular, the spectrum of P_{ω} is $\omega^{2/(2+\beta)}$ times the spectrum of P_1 .

Proof. Considering the unitary map $U: L^2(\mathbb{R}^+, dx) \to L^2(\mathbb{R}^+, dx)$ defined by

$$Uf(x) = \omega^{\frac{1}{2(2+\beta)}} f(\omega^{\frac{1}{2+\beta}}x),$$

we find that $U^* P_{\omega} U = \omega^{2/(2+\beta)} P_1$.

3.2 Truncated heat asymptotics for the cone X_{∞}

In this subsection, we compute the small-time asymptotics of the truncated heat trace,

$$Z_{\infty,\chi}(t) = \operatorname{Tr}\left(e^{-t\Delta_g}\chi\right)$$

where χ is as in Section 3.1 and moreover is compactly supported in $[0, +\infty)$, and g is the metric $g = dx^2 + x^{-\beta}g_1$ on the cone $X_{\infty} = (0, +\infty) \times M$. The manifold M is equipped with the metric g_1 that is independent of x. Here, we do not assume that ∂M is empty: this will be useful in the proof of Theorem 2. We will only use the Weyl asymptotics on M.

Using the direct sum decomposition given in (4), we have

$$Z_{\infty,\chi}(t) = \sum_{k=1}^{+\infty} \operatorname{Tr}\left(e^{-tP_{\omega_k}}\chi\right)$$
(10)

where we recall that the ω_k are the eigenvalues of Δ_M and P_{ω} is defined by (9).

We make the following two preliminary observations:

- For k fixed and $t \to 0^+$, we have $\operatorname{Tr}(e^{-tP_{\omega_k}}\chi) = O(t^{-1/2})$. This term will be negligible in the sequel because the global trace is not less than C/t for some C > 0 (since dim $(X) \ge 2$).
- For t > 0 fixed, the smooth function $f : \omega \mapsto \operatorname{Tr}(e^{-tP_{\omega}}\chi)$ has a fast decay at infinity: by Proposition 5,

$$f(\omega) \leqslant \operatorname{Tr}(e^{-tP_{\omega}}) = \operatorname{Tr}(e^{-t\omega^{2/(2+\beta)}P_1})$$

The claim then follows from the second assertion given in Proposition 4.

We split the sum (10) into two parts,

$$Z_{\infty,\chi}(t) = \sum_{\omega_k < 1} + \sum_{\omega_k \ge 1} = Z^0_{\infty,\chi}(t) + Z^1_{\infty,\chi}(t).$$

The first part, $Z^0_{\infty,\chi}(t)$, is $O(t^{-1/2})$ by the first preliminary observation and we thus only have to estimate the second part, $Z^1_{\infty,\chi}(t)$. Using Proposition 5 and its proof, we have

$$Z_{\infty,\chi}^{1}(t) = \sum_{\omega_{k} \geqslant 1} \operatorname{Tr}\left(e^{-t\omega_{k}^{2/(2+\beta)}P_{1}}\chi(\cdot/\omega_{k}^{1/(2+\beta)}\right) = \sum_{\omega_{k} \geqslant 1} \mathcal{Z}_{\chi(\cdot/\omega_{k}^{1/(2+\beta)})}(t\omega_{k}^{2/(2+\beta)})$$

and we remark that, for $\omega \ge 1$, the function $\chi(\cdot/\omega^{1/(2+\beta)})$ is identically equal to 1 on [0, a] (where a was introduced Section 3.1), so that we can use the estimate of Section 3.1.

Converting this sum into an integral (see Appendix A.2), using the Weyl law on M, we obtain

 $\#\{\omega_k \leq \omega\} \sim \gamma_n \operatorname{Vol}(M) \omega^{n/2} \quad \text{as} \quad \omega \to +\infty.$

Using Proposition 7 and the definition of f, we get

$$Z^{1}_{\infty,\chi}(t) \sim \frac{n\gamma_n \operatorname{Vol}(M)}{2} \int_{1}^{+\infty} \mathcal{Z}_{\chi(./\omega^{1/(2+\beta)})}(t\omega^{2/(2+\beta)})\omega^{\frac{n}{2}-1} d\omega.$$

Making the change of variable $\tau = t\omega^{2/(2+\beta)}$, we arrive at the following lemma, recalling the δ_H was introduced below Proposition 1.

Lemma 2. The following holds,

$$Z^1_{\infty,\chi}(t) \sim \frac{n\gamma_n(\beta+2)\mathrm{Vol}(M)}{4t^{\delta_H/2}} \int_t^{+\infty} \mathcal{Z}_{\chi(\cdot\sqrt{t/\tau})}(\tau)\tau^{\frac{\delta_H}{2}-1}d\tau \ as \ t \to 0^+ \ .$$

The integral,

$$I(t) = \int_{t}^{+\infty} \mathcal{Z}_{\chi(\cdot\sqrt{t/\tau})}(\tau) \tau^{\frac{\delta_{H}}{2} - 1} d\tau$$
(11)

is convergent at $\tau = \infty$ for all $\beta > 0$ but, in general, not at $\tau = 0$ because if $\beta \leq \beta_c$ then $\frac{\delta_H}{2} - 1 \leq \frac{1}{2} + \frac{1}{\beta} - 1$. We can compare this with the estimate in Corollary 2.

3.2.1 Case $\beta > \beta_c$

We estimate the small-time behavior of I(t) defined by (11). By the monotone convergence theorem, we have

$$\lim_{\varepsilon \to 0^+} \int_0^{+\infty} e(\tau, x, x) \chi(\varepsilon x) \, dx = \int_0^{+\infty} e(\tau, x, x) \, dx$$

where e is the heat kernel of P_1 . Hence, for any $\tau > 0$,

$$\lim_{t \to 0^+} \mathcal{Z}_{\chi(.\sqrt{t/\tau})}(\tau) = \mathcal{Z}_1(\tau).$$

Using, again, the monotone convergence theorem, we conclude that

$$\lim_{t \to 0^+} I(t) = \int_0^{+\infty} \mathcal{Z}_1(\tau) \tau^{\frac{\delta_H}{2} - 1} \, d\tau.$$

From Corollary 2, we get

$$\mathcal{Z}_1(\tau) \tau^{\delta_H/2 - 1} \leqslant C \tau^{\frac{n}{2}(1 + \frac{\beta}{2}) - 1 - \frac{1}{2} - \frac{1}{\beta}} e^{-\mu_1 \tau}$$

and $\beta > \beta_c = 2/n$ implies $\frac{n}{2}(1 + \frac{\beta}{2}) - 1 - \frac{1}{2} - \frac{1}{\beta} > -1$. Thus, the corresponding limit is finite:

$$\lim_{t \to 0^+} I(t) = \int_0^{+\infty} Z_1(\tau) \tau^{\frac{\delta_H}{2} - 1} \, d\tau < +\infty.$$

3.2.2 Case $\beta \leq \beta_c$

By the second estimate in Proposition 4, the contribution to the integral from 1 to $+\infty$ in the expression for I(t) (which was introduced in Lemma 2) is O(1) uniformly with respect to t and, hence, the corresponding part of $Z^1_{\infty,\chi}(t)$ is $O(t^{-\delta_H/2})$, which will be negligible. We only need to estimate the asymptotics of

$$J(t) = \int_t^1 \mathcal{Z}_{\chi(.\sqrt{t/\tau})}(\tau) \tau^{\frac{\delta_H}{2} - 1} \, d\tau.$$

Sub-case $\beta < \beta_c$. We prove that there exists a $\delta > 0$ such that

$$\operatorname{Tr}\left(e^{-t\Delta_{g}}\chi\right) = \mathcal{O}\left(J^{\delta}t^{-(n+1)/2}\right)$$
(12)

as $t \to 0^+$. We split the integral,

$$J(t) = \int_{t}^{1} \operatorname{Tr}\left(e^{-\tau P_{1}}\chi\left(\sqrt{t/\tau}\right)\right) \tau^{\frac{\delta_{H}}{2}-1} d\tau = J_{1}(t) + J_{2}(t) = \int_{t}^{\tau_{0}} + \int_{\tau_{0}}^{1} d\tau$$

where τ_0 satisfies $\tau_0^{-1/\beta} = J\sqrt{\tau_0/t}$, i.e., $\tau_0 = (t/J^2)^{\beta/(2+\beta)}$ with $J = \int_0^{+\infty} \chi(x) dx$ as in Corollary 2. We get upper bounds for J_1 and J_2 using the upper bounds given in Corollary 2 as follows. Using the first argument in the minimum, we have

$$J_1(t) \leqslant C \frac{J}{\sqrt{t}} \tau_0^{\delta_H/2} = C J^{1 - (n\beta/2)} t^{(n\beta/4) - (1/2)}$$

Similarly, using the second argument in the minimum, we find that

$$J_2(t) \leqslant C \int_{\tau_0}^1 \tau^{-1/\beta} \tau^{(\delta_H/2) - (3/2)} d\tau \leqslant C J^{1 - (n\beta/2)} t^{(n\beta/4) - (1/2)}.$$

Finally,

$$t^{-\delta_H/2}I(t) \leqslant CJ^{1-(n\beta/2)}t^{-(n+1)/2}$$

so that we can take $\delta = 1 - (n\beta/2)$. We will use this further in Section 3.4 by choosing J small.

Sub-case $\beta = \beta_c$. When $\beta = 2/n$, we have to estimate the asymptotics of

$$J(t) = \int_t^1 \operatorname{Tr}\left(e^{-\tau P_1}\chi(\cdot\sqrt{t/\tau})\right) \tau^{(n-1)/2} d\tau$$

Using the estimate of Proposition 4, we get

$$J(t) \sim \frac{1}{\sqrt{4\pi}} \int_{t}^{1} \tau^{\frac{n}{2}-1} d\tau \int_{0}^{\infty} e^{-\tau x^{2/n}} \chi(x\sqrt{t/\tau}) dx$$

modulo terms of smaller order in τ . Using the change of variable $y = \tau x^{2/n}$, we get

$$J(t) \sim \frac{n}{4\sqrt{\pi}} \int_{t}^{1} \frac{d\tau}{\tau} \int_{0}^{\infty} e^{-y} y^{\frac{n}{2}-1} \chi\left(\sqrt{t}\tau^{-(n+1)/2} y^{n/2}\right) \, dy = \frac{n}{4\sqrt{\pi}} \int_{t}^{1} \frac{d\tau}{\tau} F\left(\tau^{(n+1)/2} / t^{1/2}\right)$$

where the function F, defined by

$$F(X) = \int_0^{+\infty} e^{-y} y^{\frac{n}{2} - 1} \chi\left(y^{n/2}/X\right) \, dy,$$

is smooth and satisfies $F(X) = O_{X\to 0}(X)$ and $\lim_{X\to+\infty} F(X) = \Gamma(n/2)$. Using the new variable $u = \tau^{(n+1)/2}/t^{1/2}$, we get

$$J(t) \sim \frac{n}{2(n+1)\sqrt{\pi}} \int_{t^{n/2}}^{t^{-1/2}} F(u) \frac{du}{u}$$
$$J(t) \sim \frac{n\Gamma(n/2)}{4(n+1)\sqrt{\pi}} |\ln t|$$

and finally

as $t \to 0^+$.

3.3 The heat parametrix in the separable metric case

We adapt the method of [5] and we use Appendix A.4. We denote by z, z' some generic points of X and by z = (x, m), z' = (x', m') generic points of $[0, 1) \times M \subset X$. Let χ be as in the previous sections, vanishing near x = 1 and extended by 0 inside X. Let $\eta \in C_0^{\infty}([0, 1))$ so that $\eta = 1$ near the support of χ , and $\eta_0 \in C_0^{\infty}(X)$, vanishing near ∂X and equal to 1 near the support of $1 - \chi$. We choose a > 0 so that η_0 vanishes for $x \leq 2a$. We claim that

$$p(t;z,z') = \eta(x)e_{\infty}(t;z,z')\chi(x') + \eta_0(x)e_0(t;z;z')(1-\chi(x'))$$

where e_{∞} is the heat kernel on the cone X_{∞} and e_0 the Riemannian heat kernel generated by the Laplacian Δ_g on $X \setminus \{x \leq a\}$ with Dirichlet boundary conditions, is a good approximation of the heat kernel on X as $t \to 0$.

Proposition 6. Let P(t) be the operator of Schwartz kernel $p(t, \cdot, \cdot)$. We have

$$\operatorname{Tr}\left(P(t) - e^{-t\Delta_g}\right) = \mathcal{O}\left(t^{\infty}\right)$$

as $t \to 0^+$.

Proof. We set $r(t, z, z') = (\partial_t + (\Delta_g)_z) p(t, z, z')$. The kernel r vanishes if x is small enough. By the local nature of the small-time asymptotics of Riemannian heat kernels (see Appendix A.4), $e_{\infty}(t, \cdot, \cdot)$ and $e_0(t, \cdot, \cdot)$ are $O(t^{\infty})$ close in C^{∞} topology on $[0, 1) \times M$. Moreover, if $x \in \operatorname{supp}(\eta')$ or $x \in \operatorname{supp}(\eta'_0)$, $p(t, \cdot, \cdot)$ and $e_0(t, \cdot, \cdot)$ are $O(t^{\infty})$ in the C^{∞} topology, because $z \neq z'$. It follows that $r(t, \cdot, \cdot) = O(t^{\infty})$ in C^{∞} topology. Therefore, denoting by R(t) the operator of Schwartz kernel $r(t, \cdot, \cdot)$, the trace norm of R(t) is a $O(t^{\infty})$. By the Duhamel formula, using that $P(t) \to \operatorname{id}$ as $t \to 0^+$, we have

$$P(t) - e^{-t\Delta_g} = \int_0^t e^{-(t-s)\Delta_g} r(s) \, ds$$

The result follows because the operator norm of $e^{-t\Delta_g}$ is not greater than 1.

3.4 Completion of the proof of Theorem 1 in the separable metric case

Thanks to the previous section, we only have to estimate the trace of P(t). We use the local nature of the heat asymptotics to show that the contribution of the term $\eta_0 e_0(t)(1-\chi)$ is equivalent to $(4\pi t)^{-(n+1)/2} \int_X (1-\chi) dv_g$. We are left to estimate the term that corresponds to the truncated cone as in Section 3.2. This gives the conclusion when $\beta \ge \beta_c$.

When $\beta < \beta_c$, the first term can be made smaller than $\varepsilon t^{-(n+1)/2}$ for any $\varepsilon > 0$ by choosing $J = \int_0^{+\infty} \chi(x) dx$ small enough as mentioned in Section 3.2.2.

3.5 From the separable to the general case

We prove that, for any given $\varepsilon > 0$, the metric g on X is ε -quasi-isometric to a separable metric g_s . We choose $\delta > 0$ so that

$$|g_0(u) - g_0(0)| \leqslant \varepsilon (du^2 + g_0(u))$$

for any $u \in [0, \delta]$. Then, we choose $\eta \in C_0^{\infty}([0, \delta))$, identically equal to 1 near u = 0. We consider the separable metric g_s which coincides with g outside $u \leq \delta$ and is given near ∂X by

$$g_s = \eta u^{-\alpha} \left(du^2 + g_0(0) \right) + (1 - \eta) g_s$$

Then

$$\left|\frac{g_s}{g} - 1\right| = \left|\eta \frac{g_0(u) - g_0(0)}{du^2 + g_0(u)}\right| \leqslant \varepsilon$$

Using Appendix A.1, this concludes the proof of Theorem 1 in the general (non-separable) case.

3.6 Proof of Theorem 2

3.6.1 Case $\beta < \beta_c$

We consider the heat traces $Z_f(t) = \text{Tr}\left(e^{-t\Delta_g}f\right)$ where $f: X \to \mathbb{R}$ is continuous. Let $\varepsilon > 0$. We choose a smooth function $\chi: X \to [0, 1]$ that is identically equal to 1 near ∂X and such that $\int_X \chi \, dv_g \leq \varepsilon$. Writing $g = \chi f + (1 - \chi)f$, we get

$$\operatorname{Tr}\left(e^{-t\Delta_g}f\right) = \operatorname{Tr}\left(e^{-t\Delta_g}(1-\chi)f\right) + \operatorname{Tr}\left(e^{-t\Delta_g}\chi f\right) = J(t) + K(t).$$

By the local nature of the heat trace asymptotics (see Appendix A.4), we have

$$J(t) \sim \frac{1}{(4\pi t)^{(n+1)/2}} \int_X (1-\chi) f \, dv_g$$

as $t \to 0^+$. Besides,

$$K(t) \leq ||f||_{\infty} \operatorname{Tr} \left(e^{-t \Delta_g} \chi \right)$$

By (12), and since we can choose $\epsilon > 0$ arbitrarily small, it follows that

$$Z_f(t) \sim \frac{1}{(4\pi t)^{(n+1)/2}} \int_X f \, dv_g$$

as $t \to 0^+$, which gives the expected result.

3.6.2 Case $\beta \ge \beta_c$

We give the proof in the case $\beta > \beta_c$. The case $\beta = \beta_c$ is treated similarly. Let us first prove that the support of the Weyl measure is contained in ∂X . If $\operatorname{supp}(f) \cap \partial X = \emptyset$, we get again by the local nature of the heat asymptotics that

$$\operatorname{Tr}\left(e^{-t\Delta_g}f\right) = \mathcal{O}\left(t^{-(n+1)/2}\right)$$

while, by the Weyl law given in Theorem 1, we have

$$t^{-(n+1)/2} = o\left(\operatorname{Tr}\left(e^{-t\Delta_g}\right)\right)$$

as $t \to 0^+$. Hence, it suffices to consider functions f of the form $f = \mathbb{1}_D$ where $D = [0, a] \times D_1$ with D_1 a piecewise smooth domain in M.

We recall that, according to [12, Chapter 5, Theorem 2.1.6], we have

$$0 \leqslant e_D(t, m, m) \leqslant e_X(t, m, m) \qquad \forall m \in D$$
(13)

where e_D is the Dirichlet heat kernel on D.

We set $D' = [0, a] \times (M \setminus D)$ and $D'' = X \setminus [0, a] \times M$. For any domain K, we denote by Z_K the Dirichlet heat trace and by $Z'_K(t) = \int_K Z_X(t, m, m) dv_g(m)$. We have

$$Z_D(t) \sim C \operatorname{Vol}(D_1) t^{-d_H/2}, \qquad Z_{D'}(t) \sim C \operatorname{Vol}(M \setminus D_1) t^{-d_H/2}$$
(14)

and $Z_{D''}(t) = o(t^{-d_H/2})$ as $t \to 0^+$. Note that for (14), we used the proofs in Section 3.2 for the case where the *n*-dimensional manifold M (here: D_1 and $M \setminus D_1$) can have a boundary. The sum $(Z'_D + Z'_{D'} + Z'_{D''})(t) = Z_X(t)$ is equivalent to $C \operatorname{Vol}(M) t^{-d_H/2}$ by Theorem 1. Hence, $(Z'_D + Z'_{D'})(t) \sim C \operatorname{Vol}(M) t^{-d_H/2}$. On the other hand, thanks to (13) we have

$$Z_D \leqslant Z'_D, \qquad Z_{D'} \leqslant Z'_{D'}, \qquad Z_{D''} \leqslant Z'_{D''},$$

It follows that

$$Z'_D(t) \sim C \operatorname{Vol}(D_1) t^{-d_H/2}$$

as $t \to 0^+$, which yields the desired result.

4 Discussion and open problems

In this article, motivated by the propagation of acoustic waves in gas giant planets, we derived the Weyl law for the Laplace-Beltrami operator on a smooth compact Riemannian (n+1)-dimensional manifold X with boundary whose metric blows up near the boundary. Many new questions emerge. We present some of them.

Quantum Ergodicity and Quantum Limits. We have seen in Corollary 1 that, if $\beta \ge \beta_c$, then a density-one subsequence of eigenfunctions concentrates on ∂X . This is a preliminary result towards Quantum Ergodicity (QE).

Recall that, on a locally compact space U endowed with a probability Radon measure μ , given a self-adjoint nonnegative operator T on $L^2(U,\mu)$, of discrete spectrum $\lambda_1 \leq \lambda_2 \leq \cdots \leq \lambda_j \leq \cdots \rightarrow +\infty$ associated with an orthonormal eigenbasis $\Phi = (\phi_j)_{j \in \mathbb{N}^*}$ of $L^2(U,\mu)$, a Quantum Limit (QL) of Φ is a probability Radon measure ν on U that is a weak limit of a subsequence of the probability measures $|\phi_j|^2 \mu$, i.e., there exists a subsequence $(j_k)_{k \in \mathbb{N}^*}$ such that

$$\int_{U} f |\phi_{j_k}|^2 d\mu \xrightarrow[k \to +\infty]{} \int_{U} f d\nu \qquad \forall f \in C^0_c(U).$$
(15)

We say that QE holds for (T, Φ) if there exists a QL ν on U and a subsequence $(j_k)_{k \in \mathbb{N}^*}$ of density one such that (15) holds.

One may wonder whether, when $\beta \ge \beta_c$, a QE property on M would imply a QE property on X. Proving this fact certainly requires fine spectral properties of Schrödinger operators (see [1]). Besides, inspired by [9, Theorem B], we wonder what can be said on QLs supported on $\partial X = \{0\} \times M$: are they invariant under the geodesic flow of (M, G) (where $G = g_1(0)$)? Defining QLs on T^*M will already be a challenge.

Inverse problems on spectra. A natural question is: does the spectrum of X determine the spectrum of M? Attacking this problem certainly requires developing appropriate trace formulas, as in [8].

Closed geodesics. Recalling that $G = g_1(0)$ where g_1 is defined by (1), it is natural to view geodesics on (M, G) as limits, in an appropriate sense, of geodesics on (X, g). A natural question is then: do there exist some closed geodesics of X accumulating on (converging to) closed geodesics of $\partial X = M$? We refer to [7] for a similar question investigated in the framework of contact sub-Riemannian 3D manifolds. Here again, having appropriate trace formulas might be useful.

Observability properties. The study of the Weyl asymptotics is a first step towards solving some inverse problems. As explained in Section 1, the knowledge of spectrum properties can already be used to check the validity of some models, but the main objective in the physical context would be the ability to reconstruct some features of the internal structure of the planets, based on the observation of acoustic waves. The feasibility of such an inverse problem is mathematically modeled by an observability inequality, which can be settled as follows for half-waves. Given any T > 0and any subset ω of X, we say that the observability property holds true for (ω, T) if there exists a positive constant $C_T(\omega)$ such that

$$\int_0^T \left\| \mathbb{1}_{\omega} e^{it\sqrt{\Delta_g}} \phi \right\|_{L^2(X, dv_g)}^2 dt \ge C_T(\omega) \|\phi\|_{L^2(X, dv_g)}^2 \qquad \forall \phi \in L^2(X, dv_g).$$
(16)

- When $\beta < \beta_c$, we expect that (16) holds as soon as ω is open and (ω, T) satisfies the Geometric Control Condition (GCC, see [2]), like in the classical case of a non-singular Riemannian metric. - When $\beta \ge \beta_c$, an obvious necessary condition for (16) to hold is that ω contain an open neigh-

- When $\beta \ge \beta_c$, an obvious necessary condition for (16) to hold is that ω contain an open neighborhood of a subset of ∂X . Indeed, take ϕ in (16) to be a highfrequency eigenfunction and apply Corollary 1. We think that this condition is sufficient if moreover (ω, T) satisfies GCC.

We note that, when X is a closed ball in \mathbb{R}^{n+1} (an idealized situation for an exactly round planet), GCC is never satisfied unless ω contains an open neighborhood of the *whole* boundary of X, which is certainly not relevant for applications from the physical point of view. In this case where X is a round ball, it is more interesting to take a small observation subset ω , containing a small open subset of ∂X . But, as soon as ω is a proper subset of a half-ball, GCC (and thus (16)) obviously fails due to trapped rays, propagating along a diameter never meeting ω . In this deteriorated context, we wonder, however, whether (16) is anyway satisfied if we restrict the inequality to radial waves or to surface waves, which are the most physically meaningful waves to be observed.

Metrics that are singular on larger codimension submanifolds. In this paper, we have considered a class of singular metrics blowing up at the boundary of X, where the boundary can be seen as a codimension-one submanifold of X.

In more general, let X be a smooth compact manifold and let Z be a submanifold of X of codimension $m \in \mathbb{N}^*$, and consider the class of singular metrics g on X that are smooth on $X \setminus Z$ and that, near Z, are written as

$$g = h + g_Z(x)r^{-\beta}$$

in a neighborhood of Z assumed to be diffeomorphic to $Z \times B_m$ where B_m is the unit ball of \mathbb{R}^m_x (this holds if the normal bundle of Z is trivial) equipped with the Euclidean metric h and the polar coordinates (r, σ) , and $g_Z(x)$ is a metric on Z, parametrized by $x \in B_m$ and depending smoothly on x. The techniques developed in our paper can certainly be extended to compute the Weyl asymptotics in such cases.

A Appendix

A.1 Quasi-isometries

Let X be a smooth manifold of dimension n + 1, with boundary. Two metrics g_1 and g_2 , smooth on $X \setminus \partial X$, are said to be ε -quasi-isometric if

$$\left|\frac{g_1}{g_2} - 1\right| \leqslant \varepsilon$$

uniformly on $X \setminus \partial X$. For $i \in \{1, 2\}$, let \triangle_{g_i} be the Friedrichs extension of the Laplace-Beltrami operator on (X, g_i) with core $C_0^{\infty}(X \setminus \partial X)$.

If \triangle_{g_1} has a discrete spectrum $(\lambda_j^1)_{j \in \mathbb{N}^*}$ then \triangle_{g_2} has also a discrete spectrum $(\lambda_j^2)_{j \in \mathbb{N}^*}$ and, for $\varepsilon \leq \frac{1}{2}$ (this condition is to get bounds on the inverse of g_i), there exists C(n) > 0 such that, for every $j \in \mathbb{N}^*$,

$$\left|\frac{\lambda_j^1}{\lambda_j^2} - 1\right| \leqslant C(n)\varepsilon.$$

Indeed, this estimate follows from the minimax characterization of the eigenvalues and from the comparison of the Rayleigh quotients, i.e., of the volumes and co-metrics.

A.2 Karamata tauberian theorem and converse

We recall the Karamata tauberian theorem (see [16, Chapter XIII, Theorem 2]).

Theorem 3. Let μ be a positive Radon measure on \mathbb{R}^+ . If there exists $\alpha > 0$ such that

$$\int_0^{+\infty} e^{-t\lambda} d\mu(\lambda) \sim At^{-\alpha} \quad (resp., \ A|\ln t|t^{-\alpha})$$

as $t \to 0^+$, then

$$\mu([0,\lambda]) \sim \frac{A}{\Gamma(\alpha+1)} \lambda^{\alpha} \quad \left(\text{resp., } \frac{A}{\Gamma(\alpha+1)} \lambda^{\alpha} \ln \lambda \right)$$

as $\lambda \to +\infty$.

We need a converse of Theorem 3. Let $f : \mathbb{R}^+ \to \mathbb{R}^+$ be a nonincreasing function of class C^1 , such that f and f' have a fast decay at infinity. Let $(\lambda_j)_{j \in \mathbb{N}^*}$ be a nondecreasing sequence of positive real numbers. We define the counting function $N(\lambda) = \#\{j \in \mathbb{N}^* \mid \lambda_j \leq \lambda\}$, for any $\lambda \in \mathbb{R}$. The objective is to estimate the sum

$$S = \sum_{j=1}^{+\infty} f(\lambda_j)$$

Proposition 7. Assume that there exist C > 0 and $\alpha > 0$ such that $N(\lambda) \sim C\lambda^{\alpha}$ as $\lambda \to +\infty$. For any $\varepsilon > 0$, there exists $K(\varepsilon) > 0$, depending on the counting function N but not on f, such that

$$\left| S - C\alpha \int_{\lambda_1}^{+\infty} f(\lambda) \lambda^{\alpha - 1} \, d\lambda \right| \leq K(\varepsilon) f(\lambda_1) + \varepsilon \int_{\lambda_1}^{+\infty} f(\lambda) \lambda^{\alpha - 1} \, d\lambda$$

Proof. Given any $\varepsilon > 0$, let $\Lambda_0 > 0$ such that, for every $\lambda \ge \Lambda_0$,

$$(1-\varepsilon)C\lambda^{\alpha} \leqslant N(\lambda) \leqslant (1+\varepsilon)C\lambda^{\alpha}.$$
(17)

Noting that $dN(\lambda) = \sum_{j=1}^{+\infty} \delta_{\lambda_j}$, using the Stieltjes integral, we have

$$S = \sum_{\lambda_j < \Lambda_0} f(\lambda_j) + \int_{\Lambda_0}^{+\infty} f(\lambda) \, dN(\lambda).$$

Now, since $\sum_{\lambda_j < \Lambda_0} f(\lambda_j) \leq N(\Lambda_0) f(\lambda_1)$, we get by integration by parts, using the fast decay of f at infinity, that

$$S \leq N(\Lambda_0) \left(f(\lambda_1) - f(\Lambda_0) \right) - \int_{\Lambda_0}^{+\infty} f'(\lambda) N(\lambda) d\lambda.$$

We derive an upper bound for S. A lower bound is obtained similarly. Using (17), integrating by parts and using that $f(\Lambda_0) \leq f(\lambda_1)$, we obtain

$$-\int_{\Lambda_0}^{+\infty} f'(\lambda) N(\lambda) \, d\lambda \leqslant -(1+\varepsilon) C \int_{\Lambda_0}^{+\infty} f'(\lambda) \lambda^{\alpha} \, d\lambda$$
$$\leqslant (1+\varepsilon) C \left(f(\lambda_1) \Lambda_0^{\alpha} + \alpha \int_{\Lambda_0}^{+\infty} f(\lambda) \lambda^{\alpha-1} \, d\lambda \right).$$

Therefore,

$$S \leqslant f(\lambda_1) \left(N(\Lambda_0) + C(1+\varepsilon)\Lambda_0^{\alpha} \right) + (1+\varepsilon)C\alpha \int_{\lambda_1}^{+\infty} f(\lambda)\lambda^{\alpha-1} d\lambda$$

and the result follows with $K(\varepsilon) = N(\Lambda_0) + C(1 + \varepsilon)\Lambda_0^{\alpha}$.

A.3 Weyl circle-point limit criterion

We consider the Schrödinger operator $P = -\partial_x^2 + q(x)$ on $C_0^{\infty}((0, +\infty))$, where q is a smooth function on $(0, +\infty)$. According to the Weyl circle-point limit criterion (see [31, Theorem X.7]), P is essentially self-adjoint if and only if there exists at least one solution of Pu = 0 that is not square integrable at 0 and at least one solution of Pu = 0 that is not square integrable at $+\infty$.

When $q(x) = Cx^{-2} + \omega x^{\beta}$ for some $C \ge 0$, $\beta > 0$ and $\omega > 0$, there is only one solution of Pu = 0 that is square integrable at $+\infty$. Near 0, the solutions of Pu = 0 are equivalent to linear combinations of x^{γ_+} and x^{γ_-} where γ_+ and γ_- are the two solutions of $-\gamma(\gamma - 1) + C = 0$. It follows that P is essentially self-adjoint if and only if $\gamma_- \leq -\frac{1}{2}$, that is, if and only if $C \ge 3/4$.

A.4 Local nature of the small-time asymptotics of heat kernels

Let (U, g) be a smooth Riemannian manifold and let \triangle be the Laplace-Beltrami operator. For our needs (see Section 3.3), $U = X \setminus \partial X$ with the metric g.

Let e_1 and e_2 be two solutions of $(\partial_t + \Delta_x)e_i(t, x, y) = 0$ for t > 0, satisfying $e_i(t, x, y) = e_i(t, y, x)$ for all t > 0 and $(x, y) \in U \times U$ and

$$\lim_{t\to 0^+} \int_U e_i(t,x,y) f(y) \, dv_g(y) = f(x) \qquad \forall x \in U \qquad \forall f \in C_0^\infty(U),$$

for $i \in \{1, 2\}$.

Lemma 3. We have $e_1(t, \cdot, \cdot) - e_2(t, \cdot, \cdot) = O(t^{\infty})$ as $t \to 0^+$ in C^{∞} topology on $U \times U$. Moreover, denoting by D the diagonal of $U \times U$, for $i \in \{1, 2\}$, we have $e_i(t, \cdot, \cdot) = O(t^{\infty})$ as $t \to 0^+$ in C^{∞} topology on $U \times U \setminus D$.

This result reflects Kac's principle of "not feeling the boundary", showing that the small-time asymptotic behavior of heat kernels is purely local. A detailed proof can be found in [10, Section 3.2.1]. The idea comes from the paper [21]. The proof uses the fact that the Hörmander operator $P = 2\partial_t + \Delta_x + \Delta_y$ is hypoelliptic. Extending the kernels e_i by 0 for t < 0, we have $Pe_i = 0$ on $\mathbb{R} \times U \times U \setminus D$ and $P(e_1 - e_2) = 0$ on $\mathbb{R} \times U \times U$, in the distributional sense. The result then follows by hypoellipticity.

A.5 \triangle_g as a nonsmooth Hörmander operator

Based on the mathematical model provided in Section 2.1, near any point of the boundary of X we have $X \simeq [0,1) \times \mathbb{R}^n$ with a local system of coordinates (x,y), with $x \in [0,1)$ and $y = (y_1, \ldots, y_n) \in \mathbb{R}^n$, and we can write (locally)

$$\Delta_g = -\sum_{i=0}^n X_i^* X_i + V \tag{18}$$

where $V(x,y) = \frac{C(x,y)}{x^2}$ is a potential and the X_i 's are vector fields given by

$$X_0 = a_0(x, y) \,\partial_x, \qquad X_i = x^{\beta/2} a_i(x, y) \,\partial_{y_i}, \quad i \in \{1, \dots, n\}.$$

The functions C and a_i , $i \in \{0, ..., n\}$ are smooth on $\mathbb{R} \times \mathbb{R}^n$ and $C(0, \cdot) = C_\beta$ and $a_i(0, \cdot) = 1$ (they can be expressed in terms of the coefficients of the smooth Riemannian metric $g_1(x)$ on Mdefined by (1)). The separable case corresponds to $a_0 = 1$ and a_i not depending on x.

Expressed as (18), the operator \triangle_g is then a Hörmander operator, however nonsmooth unless $\beta \in 2\mathbb{N}^*$. Because of this lack of smoothness, many classical results cannot be applied here.

When $\beta \in 2\mathbb{N}^*$, the above vector fields are smooth and define an almost-Riemannian geometry, in which the Weyl asymptotics of the almost-Riemannian Laplacian $\triangle_{aR} = -\sum_{i=0}^{n} X_i^* X_i$ (i.e., (18) with V = 0), of Grushin type, has been established in [11].

With these preliminary remarks in mind, we then mention a few interesting facts hereafter.

Homogeneity. In the above local coordinates, given any $\varepsilon > 0$, we define the *dilation*

$$\delta_{\varepsilon}(x,y) = (\varepsilon x, \varepsilon^{1+\beta/2}y) \qquad \forall (x,y) \in [0,1) \times \mathbb{R}^n.$$

In the separable case where $a_0 = 1$ and a_i does not depend on x, for any $i \in \{1, \ldots, n\}$, we define $\widehat{X}_i = \lim_{\varepsilon \to 0} \varepsilon \delta_{\varepsilon}^* X_i = x^{\beta/2} a_i(0) \partial_{y_i}$, and we have

$$\varepsilon \delta_{\varepsilon}^* X_0 = X_0$$
 and $\varepsilon \delta_{\varepsilon}^* \widehat{X}_i = \widehat{X}_i \quad \forall i \in \{1, \dots, n\}.$

In sR geometry, \hat{X}_i is the *nilpotentization* of the vector field X_i at the point identified with (0,0). Extrapolating results of sub-Riemannian geometry that one can find in [10] to the case of $\beta > 0$, denoting by d_g the g-distance on X, one can show that $d_g((0,0), (x,y))$ divided by $|x| + \sum_{i=1}^{n} |y_i|^{1/(1+\beta/2)}$ is bounded above and below by some positive constants in a neighborhood of (0,0). Noting that $1/(1+\beta/2) = 1 - \alpha/2$, we thus recover [13, Proposition 13] and thus the result of Proposition 1 and the fact that Hausdorff and Minkowski dimensions coincide. In the non-separable case, we obtain the result by using quasi-isometries.

Weyl law when $\beta \in 2\mathbb{N}^*$. When $\beta \in 2\mathbb{N}^*$, we always have $\beta \ge \beta_c$, and $\beta = \beta_c$ if and only if n = 1. Since the potential $1/x^2$ is homogeneous, combining results of [10, 11], we recover the Weyl law established in Theorem 1.

Weyl law when $\beta \notin 2\mathbb{N}^*$. To establish the Weyl law in general sub-Riemannian cases, the approach developed in [11] consists of estimating singular integrals involving the heat kernel, by performing the so-called (J + K)-decomposition. Applying this approach to the nonsmooth operator in (18) cannot be done directly because we miss a general hypoellipticity theory, valid for nonsmooth vector fields as above, and a generalization of Lemma 3 (see Appendix A.4) to that context.

Acknowledgements. The authors thank Anton Ermakov for many insightful discussions on gas giants and ring seismology. C. Dietze expresses her deepest gratitude to Phan Thành Nam and Laure Saint-Raymond for their continued support. She acknowledges the support by the European Research Council via ERC CoG RAMBAS, Project No. 101044249 and the Engie Foundation. M.V. de Hoop carried out the work while he was an invited professor at the Centre Sciences des Données at École Normale Supérieure, Paris. He acknowledges the support of the Simons Foundation under the MATH + X Program and the sponsors of the GeoMathematical Imaging Group at Rice University. E. Trélat acknowledges the support of the grant ANR-23-CE40-0010-02 (Einstein-PPF).

References

- [1] N. Anantharaman. Quantum Ergodicity and Delocalization of Schrödinger Eigenfunctions. Zurich Lectures in Advanced Mathematics. EMS Press, 1 edition, July 2022.
- [2] C. Bardos, G. Lebeau, and J. Rauch. Sharp sufficient conditions for the observation, control, and stabilization of waves from the boundary. SIAM Journal on Control and Optimization, 30(5):1024–1065, September 1992.
- [3] M. Berger, P. Gauduchon, E. Mazet, M. Berger, P. Gauduchon, and E. Mazet. Le Spectre d'une Variété Riemannienne. Springer, 1971.
- [4] U. Boscain, D. Prandi, and M. Seri. Spectral analysis and the Aharonov-Bohm effect on certain almost-Riemannian manifolds. *Communications in Partial Differential Equations*, 41(1):32–50, January 2016.
- [5] J. Cheeger. Spectral geometry of singular Riemannian spaces. Journal of differential geometry, 18(4):575-657, 1983.
- [6] J. Christensen-Dalsgaard. Helioseismology. Reviews of Modern Physics, 74(4):1073–1129, November 2002.
- [7] Y. Colin de Verdière. Periodic geodesics for contact sub-Riemannian 3D manifolds, February 2022.
- [8] Y. Colin de Verdière. A proof of a trace formula by Richard Melrose. Advanced Nonlinear Studies, 23(1):20220054, March 2023.
- [9] Y. Colin de Verdière, L. Hillairet, and E. Trélat. Spectral asymptotics for sub-Riemannian Laplacians, I: Quantum ergodicity and quantum limits in the 3-dimensional contact case. *Duke Mathematical Journal*, 167(1), January 2018.
- [10] Y. Colin de Verdière, L. Hillairet, and E. Trélat. Small-time asymptotics of hypoelliptic heat kernels near the diagonal, nilpotentization and related results. *Annales Henri Lebesgue*, 4:897–971, August 2021.

- [11] Y. Colin de Verdière, L. Hillairet, and E. Trélat. Spectral asymptotics for sub-Riemannian Laplacians, 2022.
- [12] E. B. Davies. Heat Kernels and Spectral Theory. Number 92. Cambridge university press, 1989.
- [13] M. V. de Hoop, J. Ilmavirta, A. Kykkänen, and R. Mazzeo. Geometric inverse problems on gas giants, March 2024.
- [14] J. W. Dewberry, C. R. Mankovich, J. Fuller, D. Lai, and W. Xu. Constraining Saturn's interior with ring seismology: Effects of differential rotation and stable stratification. *The Planetary Science Journal*, 2(5):198, October 2021.
- [15] D. Durante, T. Guillot, L. Iess, D. J. Stevenson, C. R. Mankovich, S. Markham, E. Galanti, Y. Kaspi, M. Zannoni, L. Gomez Casajus, G. Lari, M. Parisi, D. R. Buccino, R. S. Park, and S. J. Bolton. Juno spacecraft gravity measurements provide evidence for normal modes of Jupiter. *Nature Communications*, 13(1):4632, August 2022.
- [16] W. Feller. An Introduction to Probability Theory and Its Applications. A Wiley Publication in Mathematical Statistics. Wiley, New York, 2d ed edition, 1971.
- [17] R. G. French, B. Bridges, M. M. Hedman, P. D. Nicholson, C. Mankovich, and C. A. McGhee-French. Kronoseismology V: A panoply of waves in Saturn's C ring driven by high-order internal planetary oscillations. *Icarus*, 370:114660, December 2021.
- [18] P. Gaulme, B. Mosser, F.-X. Schmider, T. Guillot, and J. Jackiewicz. Seismology of giant planets. pages 189–202. June 2015.
- [19] C. R. Graham and J. M. Lee. Einstein metrics with prescribed conformal infinity on the ball. Advances in Mathematics, 87(2):186–225, June 1991.
- [20] T. Gudkova and V. Zharkov. The free oscillations of Jupiter. Planetary and Space Science, 47(10-11):1211-1224, October 1999.
- [21] D. S. Jerison and A. Sánchez-Calle. Estimates for the heat kernel for a sum of squares of vector fields. *Indiana University Mathematics Journal*, 35(4):835–854, 1986.
- [22] C. R. Mankovich. Saturn's rings as a seismograph to probe Saturn's internal structure. AGU Advances, 1(2):e2019AV000142, June 2020.
- [23] C. R. Mankovich, M. S. Marley, J. J. Fortney, and N. Movshovitz. Cassini ring seismology as a probe of Saturn's interior. I. rigid rotation. *The Astrophysical Journal*, 871(1):1–15, January 2019.
- [24] S. R. Markham. Gas Planet Seismology and Cooling. PhD thesis, California Institute of Technology, October 2021.
- [25] M. Marley. Nonradial oscillations of Saturn: Implications for ring system structure. Ph.D. thesis, The University of Arizona., 1990.
- [26] M. Marley. Nonradial oscillations of Saturn. Icarus, 94(2):420-435, 1991.
- [27] B. Militzer and W. B. Hubbard. Study of Jupiter's interior: Comparison of 2, 3, 4, 5, and 6 layer models. *Icarus*, 411:115955, March 2024.

- [28] B. Militzer, F. Soubiran, S. M. Wahl, and W. Hubbard. Understanding Jupiter's interior. Journal of Geophysical Research: Planets, 121(9):1552–1572, September 2016.
- [29] B. Militzer, S. M. Wahl, and W. B. Hubbard. Models of Saturn's interior constructed with an accelerated concentric Maclaurin spheroid method. *The Astrophysical Journal*, 879(2):78, July 2019.
- [30] J. Provost, B. Mosser, and G. Berthomieu. A new asymptotic formalism for Jovian seismology. Astronomy and Astrophysics, 274:595–611, 1993.
- [31] M. Reed and B. Simon. Methods of Modern Mathematical Physics. 2: Fourier Analysis, Self-Adjointness. Acad. Pr, San Diego, 1980.
- [32] F.-X. Schmider, J. Gay, P. Gaulme, C. Jacob, L. Abe, M. Alvarez, J. A. Belmonte, E. Fossat, B. Gelly, T. Guillot, F. Jeanneaux, B. Mosser, and J.-C. Valtier. SYMPA, a dedicated instrument for Jovian seismology: I. Principle and performance. Astronomy and Astrophysics, 474(3):1073–1080, November 2007.
- [33] S. V. Vorontsov, V. N. Zharkov, and V. M. Lubimov. The free oscillations of Jupiter and Saturn. *Icarus*, 27(1):109–118, January 1976.
- [34] S. P. Weppner, J. P. McKelvey, K. D. Thielen, and A. K. Zielinski. A variable polytrope index applied to planet and material models. *Monthly Notices of the Royal Astronomical Society*, 452(2):1375–1393, September 2015.

Isoperimetric inequalities for inner parallel curves

Charlotte Dietze, Ayman Kachmar, and Vladimir Lotoreichik

Abstract. We prove weighted isoperimetric inequalities for smooth, bounded, and simplyconnected domains. More precisely, we show that the moment of inertia of inner parallel curves for domains with fixed perimeter attains its maximum for a disk. This inequality, which was previously only known for convex domains, allows us to extend an isoperimetric inequality for the magnetic Robin Laplacian to non-convex centrally symmetric domains. Furthermore, we extend our isoperimetric inequality for moments of inertia, which are second moments, to p-th moments for all p smaller than or equal to two. We also show that the disk is a strict local maximiser in the nearly circular, centrally symmetric case for all p strictly less than three, and that the inequality fails for all p strictly bigger than three.

1. Introduction

Let $\Omega \subset \mathbb{R}^2$ be a smooth, bounded and simply-connected domain. For any $t \ge 0$, we define the corresponding inner parallel curve S_t by

$$S_t := S_t(\Omega) = \{ x \in \overline{\Omega} : \operatorname{dist}(x, \partial \Omega) = t \}.$$
(1.1)

The systematic study of the geometric structure and regularity of inner parallel curves was initiated in [4, 11, 16], see also [27, 28] and references therein.

By [28, Theorem 4.4.1] and [27, Proposition A.1], the inner parallel curve S_t is a finite union of piecewise smooth simple curves for almost every $t \ge 0$. Hartman [16, Corollary 6.1] showed that

$$|S_t| \le |\partial \Omega| - 2\pi t$$
 for almost every $t \ge 0$ with $S_t \ne \emptyset$, (1.2)

where $|S_t|$ and $|\partial \Omega|$ denote the length of S_t and $\partial \Omega$, respectively.

Mathematics Subject Classification 2020: 49Q10 (primary); 51M15, 28A75, 35P15, 58J50 (secondary).

Keywords: inner parallel curves, isoperimetric inequalities, moments of inertia, spectral inequalities.

The moment of inertia of the inner parallel curve S_t computed with respect to its centroid

$$c(t) := \frac{1}{|S_t|} \int_{S_t} x \,\mathrm{d}\,\mathcal{H}^1(x) \tag{1.3}$$

is given by

$$\int_{S_t} |x - c(t)|^2 \,\mathrm{d}\,\mathcal{H}^1(x),\tag{1.4}$$

where \mathcal{H}^1 denotes the one-dimensional Hausdorff measure. In this paper, we address the following question.

Question. Fixing the perimeter of Ω , what shape of Ω maximises the moment of inertia of the inner parallel curve S_t as defined in (1.4) for given $t \ge 0$?

Our main result states that the optimal shape is attained for a disk.

Theorem 1.1 (An isoperimetric inequality for moments of inertia). Let $\Omega \subset \mathbb{R}^2$ be a smooth, bounded and simply-connected domain. Then, for almost every $t \ge 0$,

$$\int_{S_t(\Omega)} |x - c(t)|^2 \, \mathrm{d}\,\mathcal{H}^1(x) \leq \int_{S_t(\mathcal{B})} |x|^2 \, \mathrm{d}\,\mathcal{H}^1(x), \tag{1.5}$$

where \mathcal{B} is the disk centred at the origin and with the same perimeter as Ω . Here $S_t(\cdot)$ and c(t) are defined in (1.1) and (1.3), respectively. When $t \in [0, \frac{|\partial \Omega|}{2\pi})$, the equality in (1.5) is attained if and only if Ω is a disk.

Note that c(t) = 0 if Ω is a disk centred at the origin. In the setting where S_t is a closed curve, the statement of Theorem 1.1 can be deduced from a result due to Hurwitz [18, pp. 396–397] combined with (1.2). For instance, this argument applies for convex domains Ω , see [19, p. 12] for further details. In general, S_t can consist of several connected components, see for example Figure 1.1, and Theorem 1.1 is novel in this case. The classical result by Hurwitz itself provides an isoperimetric inequality for the moment of inertia of the boundary of a planar domain under fixed perimeter constraint and essentially coincides with the statement of Theorem 1.1 in the special case t = 0. The recent contribution [23] proves a quantitative version of the inequality by Hurwitz and addresses the higher-dimensional setting.

The smoothness assumption for the domain is not optimal. It is used in the proof for Proposition 2.1 below, see [16], see also [28, Theorem 4.4.1] and [27, Proposition A.1]. For piecewise C^2 -domains satisfying a similar result to Proposition 2.1 below, we can also obtain (1.5) using the same rest of the proof. In general, Theorem 1.1 fails for non-simply-connected domains; see Remark 4.1 below.



Figure 1.1. A schematic representation of the Figure 1.2. A schematic illustration of the coninner parallel curve S_t in a dumbbell-like domain. Note that S_t can be disconnected.



nected curve Σ_t in the case of a dumbbell-like domain. The proof of Theorem 1.2 will show that the segment connecting the two connected components of S_t is doubly covered.

Our proof of Theorem 1.1 relies on an explicit construction of a *closed* curve Σ_t . An illustration for Σ_t is shown in Figure 1.2 (where Σ_t inherits the symmetry of Ω).

Theorem 1.2 (Covering inner parallel curves with a closed curve). Let $\Omega \subset \mathbb{R}^2$ be a smooth, bounded and simply-connected domain. Then, for almost every $t \ge 0$ with $S_t \neq \emptyset$, there exists a closed and piecewise smooth curve Σ_t with

$$S_t \subset \Sigma_t$$
 and $|\Sigma_t| \leq |\partial \Omega| - 2\pi t$,

where S_t was defined in (1.1).

The technical result Theorem 1.2 is of independent interest as we obtain an improved version of (1.2) taking the distance between different connected components of S_t into account, see Corollary 7.1 below.

More generally, we can consider p-th moments and ask for which $p \in (0, \infty)$ we have

$$\int_{S_t(\Omega)} |x - c(t)|^p \, \mathrm{d}\,\mathcal{H}^1(x) \le \int_{S_t(\mathcal{B})} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x) \quad \text{for almost every } t \ge 0, \qquad (1.6)$$

where c(t) is the centroid of $S_t(\Omega)$ and \mathcal{B} is a disk centred at the origin with $|\partial \Omega| =$ $|\partial \mathcal{B}|$. For $t \geq 0$ small enough, we have $|S_t(\Omega)| = |S_t(\mathcal{B})|$ and $S_t(\Omega)$ is a closed curve, see Lemma 2.3 (ii) below, so (1.6) reduces to

$$\frac{(2\pi)^p}{|S_t(\Omega)|^{p+1}} \int_{S_t(\Omega)} |x - c(t)|^p \,\mathrm{d}\,\mathcal{H}^1(x) \le 1.$$

So we may ask if we have

$$\frac{(2\pi)^p}{|\Gamma|^{p+1}} \int\limits_{\Gamma} |x|^p \,\mathrm{d}\,\mathcal{H}^1(x) \le 1 \tag{1.7}$$

for all closed Lipschitz curves Γ with the origin as its centroid.

Note that the centroid c(t) is independent of t for all centrally symmetric domains Ω , or for example for domains with two not necessarily orthogonal axes of symmetry. To keep things simple, we focus on the centrally symmetric case.

Theorem 1.3 (An isoperimetric inequality for *p*-th moments). *The followings statements hold.*

- (i) The statement of Theorem 1.1 extends to (1.6) in the case $p \in (0, 2]$.
- (ii) For p < 3, the boundary of a disk is a strict local maximiser among nearly circular, centrally symmetric closed Lipschitz curves Γ of the left-hand side in the inequality (1.7).
- (iii) For p > 3, (1.7) does not hold, not even locally near boundary of the disk. More precisely, there exists a sequence of nearly circular, centrally symmetric closed Lipschitz curves $(\Gamma_n)_{n \in \mathbb{N}}$ converging uniformly to the boundary of the disk for which

$$\frac{(2\pi)^p}{|\Gamma_n|^{p+1}} \int_{\Gamma_n} |x|^p \,\mathrm{d}\,\mathcal{H}^1(x) > 1.$$

This naturally leads to the following conjecture.

Conjecture 1.4. (1.7) holds for all $p \leq 3$ and all closed Lipschitz curves Γ with the origin as its centroid.

In the case $p \in (0, 2]$, Theorem 1.3 follows from Theorem 1.1 and Theorem 1.2 using Jensen's inequality. For the local optimality for p < 3 in Theorem 1.3 (ii), we follow a Fuglede-type argument [12]. From these computations, we also obtain Theorem 1.3 (iii), where symmetry breaking occurs for p > 3.

Theorem 1.1 and Theorem 1.3 are of general interest as (weighted) isoperimetric inequalities have recently received great attention [1, 3, 7, 10], see also [5, 9, 13, 24, 25] on quantitative isoperimetric inequalities. In the present paper, we consider the moment of inertia of the inner parallel curves S_t and compare it with the corresponding quantity for a disk of the same perimeter. This is a relatively unusual setting as our constraints do not involve the area of the domain Ω , but only its perimeter. Note that under fixed area constraint the *p*-th moment of the boundary is minimised by the disk for all $p \ge 1$ (cf. [3, Theorem 2.1]), which is in contrast to our result under fixed perimeter. In the case of centrally symmetric domains, or more generally for domains Ω for which the centroid c(t) defined in (1.3) is independent of t, we can deduce from Theorem 1.1 a result going back to Hadwiger [14], see (7.1) by integrating over t.

As an application of Theorem 1.1, we obtain an isoperimetric inequality for the magnetic Robin Laplacian. More precisely, considering the magnetic Robin Laplacian with a negative boundary parameter β and a sufficiently small constant magnetic field *b*, the ground state energy is expressed as follows:

$$\lambda_1^{\beta,b}(\Omega) = \inf_{\substack{u \in H^1(\Omega) \\ \|u\|_{L^2(\Omega)} = 1}} \left(\int_{\Omega} |(-i\nabla - b\mathbf{A})u|^2 + \beta \int_{\partial\Omega} |u|^2 \, \mathrm{d}\,\mathcal{H}^1(x) \right).$$

where **A** is a vector field in Ω with curl **A** = 1. It was shown in [19, Theorem 4.8] that the corresponding ground state energies for convex and centrally symmetric domains Ω and a disk \mathcal{B} of the same perimeter satisfy $\lambda_1^{\beta,b}(\Omega) \leq \lambda_1^{\beta,b}(\mathcal{B})$. Using Theorem 1.1, we can remove the convexity assumption on Ω .

Theorem 1.5 (An isoperimetric inequality for the magnetic Robin Laplacian). Let $\Omega \subset \mathbb{R}^2$ be a smooth, bounded and simply-connected domain. Assume that Ω is centrally symmetric or, more generally, that the centroid of $S_t(\Omega)$ is independent of t for all $t \geq 0$ with $S_t(\Omega) \neq \emptyset$. Let $\beta < 0$ be the negative Robin parameter, and let $0 < b < b_0(|\partial \Omega|, \beta)$, where $b_0(|\partial \Omega|, \beta)$ depends on $|\partial \Omega|$ and β . Then the lowest eigenvalue of the magnetic Robin Laplacian on Ω with constant magnetic field of strength b and Robin boundary conditions with parameter β satisfies

$$\lambda_1^{\beta,b}(\Omega) \le \lambda_1^{\beta,b}(\mathcal{B}),\tag{1.8}$$

where $\mathcal{B} \subset \mathbb{R}^2$ is the disk having the same perimeter as Ω . Equality in (1.8) occurs if and only if Ω is a disk.

If $|\partial \Omega| = 2\pi$, then we have the explicit expression $b_0(|\partial \Omega|, \beta) = \min(1, 4\sqrt{-\beta})$.

Structure of the paper. The rest of the paper is organised as follows. In Section 2, we introduce some notation and auxiliary results on inner parallel curves. In Section 3, we prove Theorem 1.2. We use this in Section 4 to prove Theorem 1.1. In Section 5, we show Theorem 1.3. More precisely, the proof and the precise statement of Theorem 1.3 (i) can be found in Corollary 5.1 and for Theorem 1.3 (ii), (iii) we refer to Proposition 5.6. Some background material on the magnetic Robin Laplacian and the proof of Theorem 1.5 are given in Section 6. In Section 7, we present two simple applications of Theorem 1.1 and Theorem 1.2, namely Corollary 7.1 on an improved version of (1.2), and Section 7.2 on moments of inertia of domains.

2. Preliminaries

2.1. Notation

We introduce for a piecewise- C^1 mapping $\gamma: [0, L] \to \mathbb{R}^2$ the length of the closed and not necessarily simple curve parametrised by γ ,

$$\ell(\gamma) := \int_{0}^{L} |\gamma'(s)| \,\mathrm{d} s.$$

We also use the notation $\gamma([a, b]) = \{\gamma(s): s \in [a, b]\}$ for $a, b \in [0, L], a < b$. We say that $\gamma_1, \gamma_2: [0, L] \to \mathbb{R}^2$ parametrise the same curve if there exists a continuous bijection $\psi: [0, L] \to [0, L]$ such that $\gamma_1 = \gamma_2 \circ \psi$. A subset $U \subset \mathbb{R}^2$ is said to be *centrally symmetric* if it coincides with its reflection $\{-x: x \in U\}$ with respect to the origin.

2.2. Inner parallel curves

Let $\Omega \subset \mathbb{R}^2$ be a bounded, simply-connected smooth domain. In this subsection we recall some properties of the inner parallel curves of Ω .

Parametrisation of the boundary. Let us denote by $L = |\partial \Omega|$ the perimeter of Ω . Consider the arc-length parametrisation of $\partial \Omega$ oriented in the counter-clockwise direction,

$$s \in \mathbb{R}/(L\mathbb{Z}) \mapsto \gamma(s) = (\gamma_1(s), \gamma_2(s))^\top \in \mathbb{R}^2,$$

which identifies $\partial \Omega$ with $\mathbb{R}/(L\mathbb{Z}) \simeq [0, L)$; the function γ is smooth which matches with the smoothness hypothesis we imposed on $\partial \Omega$.

The vector $\gamma'(s) = (\gamma'_1(s), \gamma'_2(s))^{\top}$ is the unit tangent vector to $\partial \Omega$ at $\gamma(s)$ and points in the counter-clockwise direction. The unit normal vector at $\gamma(s)$ pointing inwards the domain Ω is given by

$$\mathbf{n}(s) = (-\gamma_2'(s), \gamma_1'(s))^\top.$$

We introduce the curvature

$$\kappa(s) := \gamma_2''(s)\gamma_1'(s) - \gamma_1''(s)\gamma_2'(s)$$
(2.1)

of $\partial \Omega$ at the point $\gamma(s)$. In particular, the Frenet formula

$$\gamma''(s) = \kappa(s)\mathbf{n}(s), \qquad (2.2)$$

holds. Recall that, since $\partial \Omega$ is a smooth closed simple curve, the total curvature identity [20, Corollary 2.2.2] yields

$$\int_{0}^{L} \kappa(s) \,\mathrm{d}\, s = 2\pi. \tag{2.3}$$

We remark that within the chosen sign convention the curvature of a convex domain is non-negative.

Properties of inner parallel curves. We define the *in-radius* of Ω by

$$r_{\rm i}(\Omega) := \max_{x \in \Omega} \rho(x),$$

where ρ is the distance function given by

$$\rho: \Omega \to \mathbb{R}_+, \quad \rho(x) := \inf_{y \in \partial \Omega} |x - y|.$$
(2.4)

Recall that the inner parallel curve for Ω is the level set of the distance function

$$S_t = \{x \in \overline{\Omega} : \rho(x) = t\}, \quad t \in [0, r_i(\Omega)).$$

For almost every $t \in (0, r_i(\Omega))$, the inner parallel curve S_t is a finite union of disjoint piecewise smooth simple closed curves, and the curve S_t admits a parametrisation as in Proposition 2.1 below, which was proved in [16], see also [28, Theorem 4.4.1] and [27, Proposition A.1] for more modern presentations and further refinements.

Proposition 2.1. There exists a subset $\mathcal{L} \subset [0, r_i(\Omega))$, whose complement is of Lebesgue measure zero, such that for any $t \in \mathcal{L}$, there exist $m \in \mathbb{N}$ and

$$0 \le a_1 < b_1 < a_2 < b_2 < \dots < a_m < b_m \le L,$$

such that the inner parallel curve S_t consists of the union of finitely many smooth curves parametrised by

$$[a_k, b_k] \ni s \mapsto \gamma(s) + t \mathbf{n}(s), \quad k \in \{1, 2, \dots, m\},\$$

which forms a union of finitely many piecewise-smooth simple closed curves.

Consider the mapping

$$(s,t) \in \mathbb{R}/(L\mathbb{Z}) \times (0,r_{i}(\Omega)) \mapsto \Phi(s,t) := \gamma(s) + t\mathbf{n}(s) \in \mathbb{R}^{2}$$
 (2.5)

According to [21, Theorem 5.25], there exists $t_{\star} \in (0, r_i(\Omega))$ such that the restriction of the mapping Φ in (2.5) to the set $\mathbb{R}/(L\mathbb{Z}) \times (0, t_{\star})$ is a smooth diffeomorphism

onto its range. The range of this restriction is then given by a tubular neighbourhood of $\partial \Omega$. It is not difficult to verify that $S_t = \Phi(\mathbb{R}/(L\mathbb{Z}), t)$ for all $t \in (0, t_*)$. However, for $t \ge t_*$ the same property, in general, does not hold. It can also be easily checked that for all $t \in (0, t_*)$, the inner parallel curve S_t is connected and $|S_t| = L - 2\pi t$.

Lemma 2.2. Let $t \in \mathcal{L}$ and the associated numbers $m \in \mathbb{N}$, $\{a_k\}_{k=1}^m$, $\{b_k\}_{k=1}^m$, be as in Proposition 2.1. Then, for any $k \in \{1, 2, ..., m\}$ and any $s_0 \in [a_k, b_k]$, it holds that $\kappa(s_0) \leq \frac{1}{t}$.

Proof. Let us introduce the notation

$$\mathbf{m}(t) := \gamma(s_0) + t \mathbf{n}(s_0).$$

By the Frenet formula (2.2), one has $\gamma''(s_0) = \kappa(s_0)\mathbf{n}(s_0)$ and by Taylor's formula near s_0 we get

$$\gamma(s) = \gamma(s_0) + (s - s_0)\gamma'(s_0) + \frac{1}{2}(s - s_0)^2\kappa(s_0)\mathbf{n}(s_0) + \mathcal{O}(|s - s_0|^3), \quad s \to s_0.$$

Consequently, using orthogonality of $\gamma'(s_0)$ and $\mathbf{n}(s_0)$, we get

dist
$$(\gamma(s), \mathsf{m}(t))^2 = t^2 + (1 - t\kappa(s_0))(s - s_0)^2 + \mathcal{O}(|s - s_0|^3), \quad s \to s_0.$$

Since $m(t) \in S_t$, then $dist(\gamma(s), m(t)) \ge t$ for s in a neighbourhood of s_0 , which is possible only when $1 - t\kappa(s_0) \ge 0$.

In the next lemma we provide a simple construction of a closed but not necessarily simple curve which contains S_t . The geometric bound as in Theorem 1.2 on its length will only hold for t not larger than the inverse of the maximum of the curvature for the curve γ :

$$\kappa_{\max}(\Omega) := \max_{s \in \mathbb{R}/(L\mathbb{Z})} \kappa(s)$$

Lemma 2.3. For $t \in (0, r_i(\Omega))$, the mapping

$$s \in \mathbb{R}/(L\mathbb{Z}) \mapsto \alpha_t(s) := \gamma(s) + t\mathbf{n}(s) \in \mathbb{R}^2$$

parametrises a smooth closed, not necessarily simple curve such that

- (i) $S_t \subset \alpha_t([0, L])$ for all $t \in \mathcal{L}$;
- (ii) $\ell(\alpha_t) = L 2\pi t$ for all $t \le \frac{1}{\kappa_{\max}(\Omega)}$;
- (iii) $\ell(\alpha_t) > L 2\pi t$ for all $\frac{1}{\kappa_{\max}(\Omega)} < t < r_i(\Omega)$.

Remark 2.4. According to [26], the domain Ω contains a disk of radius $\frac{1}{\kappa_{\max}(\Omega)}$. In other words, it holds that

$$r_{\rm i}(\Omega) \ge \frac{1}{\kappa_{\rm max}(\Omega)}.$$
 (2.6)

The equality occurs for some special types of domains such as a disk or (going beyond smooth domains) for a convex hull of two disjoint disks of equal radius. The original work [26] is hardly available and the complete proof can be found in [17, Proposition 2.1]. Inequality (2.6) shows that, in general, a more sophisticated method than in Lemma 2.3 is needed to construct for any $t \in (0, r_i(\Omega))$ a closed curve of length not larger than $L - 2\pi t$, which contains the inner parallel curve S_t .

Proof of Lemma 2.3. Notice that smoothness of γ on $\mathbb{R}/(L\mathbb{Z})$ ensures that α_t is smooth on $\mathbb{R}/(L\mathbb{Z})$ as well. It is also clear from Proposition 2.1 that $S_t \subset \alpha_t([0, L])$ for all $t \in \mathcal{L}$.

Using the identity (2.3), we get for any $t \leq \frac{1}{\kappa_{\max}(\Omega)}$

$$\ell(\alpha_t) = \int_0^L |\dot{\alpha}(s)| \, \mathrm{d}\, s = \int_0^L |1 - t\kappa(s)| \, \mathrm{d}\, s = \int_0^L (1 - t\kappa(s)) \, \mathrm{d}\, s = L - 2\pi t.$$

Analogously, we get for any $t \in \left(\frac{1}{\kappa_{\max}(\Omega)}, r_{i}(\Omega)\right)$

$$\ell(\alpha_t) = \int_0^L |\dot{\alpha}(s)| \, \mathrm{d}\, s = \int_0^L |1 - t\kappa(s)| \, \mathrm{d}\, s > \int_0^L (1 - t\kappa(s)) \, \mathrm{d}\, s = L - 2\pi t.$$

In the remainder of this subsection we will discuss the properties of S_t for a centrally symmetric domain Ω . The central symmetry of Ω is inherited by $\partial \Omega$. Consequently, if $y = \gamma(s) \in \partial \Omega$, we know that $-y \in \partial \Omega$ too; moreover the centroid of $\partial \Omega$ is the origin, so

$$\int_{0}^{L} \gamma(s) \, \mathrm{d} \, s = 0$$

Lemma 2.5. Let $\Omega \subset \mathbb{R}^2$ be a bounded, simply-connected, centrally symmetric smooth domain. Then, for all $t \in (0, r_i(\Omega))$, the inner parallel curve $S_t \subset \Omega$ is centrally symmetric.

Proof. Let $x \in S_t$ be fixed. Then $\rho(x) = t$ and there exists a point $y \in \partial\Omega$ such that |x - y| = t. Observe now that $\rho(-x) \le t$, because $-y \in \partial\Omega$. In the case that $\rho(-x) < t$, there would exist a point $z \in \partial\Omega$ such that |z + x| < t. Since $-z \in \partial\Omega$, we would get that $\rho(x) < t$, leading to a contradiction. Thus, we infer that $\rho(-x) = t$ and hence $-x \in S_t$.

2.3. An auxiliary geometric inequality

The aim of this subsection is to provide a geometric inequality, which will be used in the proof of Theorem 1.2.

Hypothesis 2.6. Let $c_1, c_2 \in \mathbb{R}^2$ and t > 0 be fixed. Let a smooth simple non-closed curve $\Gamma \subset \mathbb{R}^2$ be parametrised by the arc-length via the mapping $\gamma: [s_1, s_2] \to \mathbb{R}^2$, $s_1 < s_2$. Assume that the following properties hold:

- (i) $p_j := \gamma(s_j) \in \partial \mathcal{B}_t(c_j)$ for j = 1, 2;
- (ii) $\gamma'(s_j)$ is tangent to $\partial \mathcal{B}_t(c_j)$ in the counterclockwise direction for j = 1, 2.y;
- (iii) Γ can be extended up to a closed simple curve so that $\mathcal{B}_t(c_1) \cup \mathcal{B}_t(c_2)$ is surrounded by this extension.

Proposition 2.7. Under Hypothesis 2.6 the following geometric inequality holds:

$$|\Gamma| \ge |c_1 - c_2| + t \int_{s_1}^{s_2} \kappa(s) \,\mathrm{d}\,s, \tag{2.7}$$

where κ is the curvature of Γ defined as in (2.1).

Proof. The proof relies on an abstract result due to Chillingworth [8, Theorem 3.3], which states that two closed homotopic curves are regularly homotopic if the curves are direct, that is, the corresponding curves in the covering space are simple.

The curve Γ is homotopic to its projection Σ on the convex hull of the two disks. After modifying Σ suitably so we avoid nullhomotopic loops, we can extend Γ and Σ to closed direct curves $\tilde{\Gamma}$, $\tilde{\Sigma}$ using the same extension. By [8, Theorem 3.3], $\tilde{\Gamma}$ and $\tilde{\Sigma}$ are regularly homotopic, so the integral over their curvatures agree: $\int_{\tilde{\Gamma}} \kappa = \int_{\tilde{\Sigma}} \kappa$, see for example [30]. Since Γ and Σ were extended using the same extension, we also get $\int_{\Gamma} \kappa = \int_{\Sigma} \kappa$. Finally, one can check that Σ satisfies (2.7) and by $|\Gamma| \ge |\Sigma|$, we obtain (2.7) for Γ .

3. Proof of Theorem 1.2 – Covering inner parallel curves

The aim of this section is to prove the following theorem, which yields Theorem 1.2.

Theorem 3.1. Let $\Omega \subset \mathbb{R}^2$ be a smooth, bounded and simply-connected domain and let S_t for $t \ge 0$ be the corresponding inner parallel curves defined in (1.1). Then there exists a subset $\mathcal{L} \subset [0, r_i(\Omega))$ such that $(0, r_i(\Omega)) \setminus \mathcal{L}$ is of Lebesgue measure zero, and for any $t \in \mathcal{L}$, there exists a piecewise smooth continuous mapping $\sigma_t: \mathbb{R}/(L\mathbb{Z}) \to \mathbb{R}^2$ such that

- (i) $S_t \subset \sigma_t([0, L]);$
- (ii) $\ell(\sigma_t) \leq L 2\pi t;$
- (iii) for centrally symmetric domains Ω , the curve $\sigma_t([0, L])$ is centrally symmetric too.

In the following, let the set \mathcal{L} be as in Proposition 2.1. Let $t \in \mathcal{L}$ be fixed. Then, there exist $m \in \mathbb{N}$ and

$$0 \le a_1 < b_1 < a_2 < b_2 < \dots < a_m < b_m \le L$$

such that the inner parallel curve S_t is given by

$$S_t = \bigcup_{k=1}^m \{\gamma(s) + t\mathbf{n}(s) : s \in [a_k, b_k]\}.$$

Without loss of generality, we can always reparametrise the boundary of Ω so that $a_1 = 0$. In the following we assume that such a re-parametrisation is performed and that $a_1 = 0$. Moreover, for the sake of convenience we also set $a_{m+1} := L$ and $b_{m+1} := b_1$.

The inner parallel curve S_t , $t \in \mathcal{L}$, is not necessarily connected and, in general, it consists of finitely many piecewise-smooth, simple, closed curves. Note that, while S_t consists of finitely many piecewise smooth simple closed curves, the pieces $C_k :=$ $\{\gamma(s) + t\mathbf{n}(s): s \in [a_k, b_k]\}$ are not necessarily closed curves. Our aim is to construct a piecewise smooth, closed, not necessarily simple curve, which contains S_t and whose length is not larger than $L - 2\pi t$. The idea is to connect the terminal point of $\{\gamma(s) + t\mathbf{n}(s): s \in [a_k, b_k]\}$ with the starting point of $\{\gamma(s) + t\mathbf{n}(s): s \in [a_{k+1}, b_{k+1}]\}$ for all $k \in \{1, 2, ..., m\}$.

Using the computation in the proof of Lemma 2.3, and that $\kappa(s) \leq \frac{1}{t}$ for all $s \in [a_k, b_k]$ by Lemma 2.2, we get the following expression for the length of S_t :

$$|S_t| = \sum_{k=1}^m \int_{a_k}^{b_k} |1 - t\kappa(s)| \, \mathrm{d}\, s = \sum_{k=1}^m \int_{a_k}^{b_k} (1 - t\kappa(s)) \, \mathrm{d}\, s. \tag{3.1}$$

Let us define the points

 $\mathfrak{p}_k = \gamma(a_k) + t \mathbf{n}(a_k), \quad \mathfrak{q}_k := \gamma(b_k) + t \mathbf{n}(b_k), \quad \text{for } k \in \{1, 2, \dots, m+1\},$

and the line segments connecting them

$$\mathcal{I}_k := \{ (1-s)\mathfrak{q}_k + s\mathfrak{p}_{k+1} : s \in [0,1] \}, \quad k \in \{1, 2, \dots, m\}.$$

In the case that $b_m = L$, the line segment \mathcal{I}_m reduces to a single point. Note that, even when $b_k < a_{k+1}$, the line segment \mathcal{I}_k could reduce to a single point. The piecewise smooth continuous mapping $\sigma_t : \mathbb{R}/(L\mathbb{Z}) \to \mathbb{R}^2$ defined by

$$\sigma_t(s) = \begin{cases} \gamma(s) + t\mathbf{n}(s), & s \in [a_k, b_k] & \text{for } k \in \{1, \dots, m\}, \\ \frac{a_{k+1} - s}{a_{k+1} - b_k} \mathfrak{q}_k + \frac{s - b_k}{a_{k+1} - b_k} \mathfrak{p}_{k+1}, & s \in [b_k, a_{k+1}] & \text{for } k \in \{1, \dots, m\}, \end{cases}$$

parametrises a closed, not necessarily simple curve in \mathbb{R}^2 . The property (i) in the formulation of Theorem 3.1 follows from Proposition 2.1 and the construction of the mapping σ_t .

Note that the curve $\gamma([b_k, a_{k+1}])$, $k \in \{1, ..., m\}$, satisfies Hypothesis 2.6 with $c_1 = \mathfrak{q}_k$ and $c_2 = \mathfrak{p}_{k+1}$. Thus, it follows from Proposition 2.7 and since γ is parametrised by arc-length that for any $k \in \{1, 2, ..., m\}$

$$|\mathcal{I}_k| \le |\gamma([b_k, a_{k+1}])| - t \int_{b_k}^{a_{k+1}} \kappa(s) \, \mathrm{d}\, s = a_{k+1} - b_k - t \int_{b_k}^{a_{k+1}} \kappa(s) \, \mathrm{d}\, s$$

Combining formula (3.1) for the length of S_t with the upper bounds on the lengths of line segments $\{\mathcal{I}_k\}_{k=1}^m$ we get

$$\ell(\sigma_t) = |S_t| + \sum_{k=1}^m |\mathcal{I}_k| \le \sum_{k=1}^m \int_{a_k}^{b_k} (1 - t\kappa(s)) \, \mathrm{d}\, s + \sum_{k=1}^m \int_{b_k}^{a_{k+1}} (1 - t\kappa(s)) \, \mathrm{d}\, s$$
$$\le L - t \int_0^L \kappa(s) \, \mathrm{d}\, s = L - 2\pi t,$$

where we used the total curvature identity (2.3) in the last step. Hence, we get the property (ii) in the formulation of Theorem 3.1.

Finally, if Ω is centrally symmetric, then by Lemma 2.5 so is S_t . Consequently, to every piece C_k of S_t joining \mathfrak{p}_k to \mathfrak{q}_k , there corresponds a curve C_{k^*} which is the symmetric of C_k about the origin. This forces the number *m* of the curves C_k to be even, unless it is equal to one, and therefore we get that the corresponding joining segments $(\mathcal{I}_k)_{1 \le k \le m}$ constitute a centrally symmetric set. This proves that the image of σ_t is centrally symmetric, thereby establishing (iii) in the formulation of Theorem 3.1. This proves Theorem 1.2.

4. Proof of Theorem 1.1 – An isoperimetric inequality for moments of inertia

Let $t \in \mathcal{L}$ and the mapping σ_t be as constructed in the proof of Theorem 3.1, which defines a closed curve Σ_t . Since the moment of inertia of a curve about a point pis minimal when p is the centroid of the curve, it suffices to prove (1.5) with c(t)the centroid of Σ_t . Let us introduce the notation $L_t := \ell(\sigma_t)$ and re-parametrise the curve Σ_t by the arc-length via the mapping $\tilde{\sigma}_t : \mathbb{R}/(L_t\mathbb{Z}) \to \mathbb{R}^2$. Clearly, we have $\tilde{\sigma}_t \in$ $H^1(\mathbb{R}/(L_t\mathbb{Z}))$ thanks to the regularity of σ_t . Furthermore, by centring the coordinates at the centroid c(t) of Σ_t , we can assume that c(t) = 0, and consequently

$$\int_{0}^{L_{t}} \tilde{\sigma}_{t}(s) \,\mathrm{d}\,s = 0. \tag{4.1}$$

Using the inclusion $S_t \subset \tilde{\Sigma}_t$ and applying the Wirtinger inequality [15, Section 7.7],

$$\int_{0}^{L_{t}} |\tilde{\sigma}_{t}(s)|^{2} \,\mathrm{d}\,s \le \frac{|L_{t}|^{2}}{4\pi^{2}} \int_{0}^{L_{t}} |\tilde{\sigma}_{t}'(s)|^{2} \,\mathrm{d}\,s, \tag{4.2}$$

we get

$$\int_{S_t} |x|^2 \, \mathrm{d}\,\mathcal{H}^1(x) \le \int_{0}^{L_t} |\tilde{\sigma}_t(s)|^2 \, \mathrm{d}\,s \le \frac{|L_t|^3}{4\pi^2} \le \frac{(L-2\pi t)^3}{4\pi^2},\tag{4.3}$$

where we employed that $L_t \leq L - 2\pi t$ in the last step. Therefore, (1.5) is proved.

Assuming that there is equality in (1.5), then we get from (4.3) that $L_t = L - 2\pi t$ and there is equality is (4.2). Under the conditions (4.1) and $|\tilde{\sigma}'_t(s)| = 1$, equality happens in (4.2) if and only if $\tilde{\sigma}_t(s) = \frac{L_t}{2\pi} e^{\pm i 2\pi (s-s_0)/L_t}$ for some $s_0 \in \mathbb{R}$ and Σ_t is a circle (here we identify \mathbb{C} and \mathbb{R}^2). Moreover, knowing that $L_t = L - 2\pi t$ and $S_t = \Sigma_t$, we get that S_t is the circle of centre 0 and radius $\frac{L}{2\pi} - t$, and consequently, the domain Ω with perimeter L contains the disk \mathcal{B} of radius $\frac{L}{2\pi}$, hence $\Omega = \mathcal{B}$ is a disk, thanks to the geometric isoperimetric inequality.

Finally, if Ω is a disk, then S_t is a circle of radius $\frac{L}{2\pi} - t$ and equality in (1.5) occurs.

Remark 4.1. The hypothesis of simple connectivity is necessary in Theorem 1.1 as we demonstrate in the following example. Let $a \in (0, \frac{1}{4})$ be a parameter, which we will send to zero later. Consider the annulus $\Omega_a := B_{1-a}(0) \setminus \overline{B_a(0)}$, where we denote by $B_r(x)$ the open disk centred at $x \in \mathbb{R}^2$ of radius r > 0. Then the corresponding \mathcal{B} with the same perimeter as Ω_a is $\mathcal{B} := B_1(0)$. Furthermore, for all $t < \frac{1}{4}$, we have $S_t(\Omega_a) = \partial B_{a+t}(0) \cup \partial B_{1-(a+t)}(0)$, and $S_t(\mathcal{B}) = \partial B_{1-t}(0)$. Also note that the centroid of each $S_t(\Omega_a)$ is the origin. Then

$$\frac{1}{2\pi} \int_{S_t(\Omega_a)} |x|^2 \, \mathrm{d}\,\mathcal{H}^1(x) = (a+t)^3 + (1-(a+t))^3$$

and

$$\frac{1}{2\pi} \int_{S_t(\mathcal{B})} |x|^2 \, \mathrm{d}\,\mathcal{H}^1(x) = (1-t)^3.$$

Letting $a \to 0$, we find that

$$\frac{1}{2\pi} \int_{S_t(\Omega_a)} |x|^2 \,\mathrm{d}\,\mathcal{H}^1(x) \to t^3 + (1-t)^3 > \frac{1}{2\pi} \int_{S_t(\mathcal{B})} |x|^2 \,\mathrm{d}\,\mathcal{H}^1(x),$$

where the convergence is uniform for $t \in [t_1, t_2]$ for any $0 < t_1 < t_2 < \frac{1}{4}$. This shows that Theorem 1.1 cannot hold in this case.

5. Proof of Theorem 1.3 – An isoperimetric inequality for *p*-th moments

In this section, we study *p*-th moments of inner parallel curves and prove Theorem 1.3. We show Theorem 1.3 (i) (extension of Theorem 1.1 to *p*-th moments, for $0 \le p \le 2$) in Corollary 5.1 with the help of Jensen's inequality. Proposition 5.6 yields Theorem 1.3 (ii) and (iii).

Corollary 5.1. Let $p \in [0, 2]$ and suppose that $\Omega \subset \mathbb{R}^2$ is a smooth, bounded and simply-connected domain. Then, for almost every $t \ge 0$,

$$\int_{S_t(\Omega)} |x - c(t)|^p \, \mathrm{d}\,\mathcal{H}^1(x) \le \int_{S_t(\mathcal{B})} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x), \tag{5.1}$$

where $c(t) \in \mathbb{R}^2$ is the centroid of $S_t(\Omega)$ and where \mathcal{B} is the disk centred at the origin and with the same perimeter as Ω . Here $S_t(\cdot)$ is defined in (1.1), and \mathcal{H}^1 is the one-dimensional Hausdorff measure. For $p \neq 0$ and $t \in [0, \frac{|\partial \Omega|}{2\pi})$, the equality in (5.1) is attained if and only if Ω is a disk

Proof. For p = 0, (5.1) reduces to the well-known bound $|S_t| \le L - 2\pi t$, see (1.2). Let us take $p \in (0, 2]$. Since $\frac{2}{p} \ge 1$, we write by Jensen's inequality,

$$\left(\int_{S_t(\Omega)} |x-c(t)|^p \,\mathrm{d}\,\mathcal{H}^1(x)\right)^{2/p} \leq |S_t(\Omega)|^{\frac{2}{p}-1} \int_{S_t(\Omega)} |x-c(t)|^2 \,\mathrm{d}\,\mathcal{H}^1(x).$$

To finish the proof, we use that $|S_t| \leq L - 2\pi t$, apply Theorem 1.1, and note that

$$\int_{S_t(\mathcal{B})} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x) = 2\pi \left(\frac{L}{2\pi} - t\right)^{1+p}.$$

Since the inequality in Theorem 1.1 is strict for all $t \in \left[0, \frac{|\partial \Omega|}{2\pi}\right)$ when Ω is not a disk, this also holds for (5.1).

Secondly, we formulate the corresponding variational problem to Theorem 1.3. **Definition 5.2.** Given p > 0, we define

$$C_p := \sup_{\Gamma} \frac{\int_{\partial \mathcal{B}} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x)}{\int_{\partial \mathcal{B}} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x)} = \sup_{\Gamma} \frac{(2\pi)^p}{|\Gamma|^{p+1}} \int_{\Gamma} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x), \tag{5.2}$$

where the supremum is taken over all centrally symmetric, closed Lipschitz curves Γ and \mathcal{B} is a disk centred at the origin with $|\partial \mathcal{B}| = |\Gamma|$.

Remark 5.3. By scaling, we find that $C_p \in (0, \infty)$. By testing with $\Gamma = \partial \mathcal{B}$, we find $C_p \ge 1$ for all p. We have already shown that $C_p = 1$ for all $p \in (0, 2]$, see the end of the proof of Theorem 1.1 (or alternatively the result by Hurwitz [18]) combined with Jensen's inequality as in the proof of Corollary 5.1. In fact, using Jensen's inequality as in the proof of Corollary 5.1, one can show that C_p is non-decreasing in p.

Remark 5.4 (Existence of an optimising curve in (5.2)). We only sketch a way of proving the existence of an optimising curve in (5.2) here. First note that by scaling, we can restrict ourselves to smooth centrally symmetric curves satisfying $|\Gamma| = 1$. We can approximate these curves by piecewise linear centrally symmetric curves with $|\Gamma| = 1$, so we can take the supremum over such curves instead. Using a finite-step mirroring argument, the supremum stays the same if we only consider convex piecewise linear centrally symmetric curves with $|\Gamma| = 1$. More precisely, in this context we say that a piecewise linear curve is convex if it is the boundary of a convex polygon.

Now, consider a sequence of such curves $(\Gamma_n)_{n \in \mathbb{N}}$ such that the corresponding expression in the supremum in (5.2) converges to C_p . Assume that the Γ_n are parametrised by arc-length by $\gamma_n: [0, 1] \to \mathbb{R}^2$. By $|\Gamma_n| = 1$ and the central symmetry, we have $\Gamma_n \subset B_1(0)$. Due to the arc-length parametrisation of the γ_n , we obtain that up to a subsequence, they converge uniformly to a Lipschitz continuous function $\gamma: [0, 1] \to \mathbb{R}^2$ with Lipschitz constant at most one. To see this, one can for instance use the Arzelà–Ascoli theorem. Since the Γ_n are convex, the corresponding curve Γ is convex, too.

Furthermore, we have $|\Gamma| = 1$. Proving this is the key step and it uses the convexity. It can be seen by approximating Γ by piecewise linear convex centrally symmetric curves with the same length as Γ , and then showing that the lengths of these piecewise linear convex curves have to be close to $|\Gamma_n| = 1$. Finally, using that γ is Lipschitz continuous with Lipschitz constant at most one and $|\Gamma| = 1$, we get $|\gamma'(s)| = 1$ almost everywhere, so

$$\lim_{n \to \infty} \int_{\Gamma_n} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x) = \lim_{n \to \infty} \int_0^1 |\gamma_n(s)|^p \, \mathrm{d}\,s = \int_0^1 |\gamma(s)|^p \, \mathrm{d}\,s = \int_{\Gamma} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x).$$

Together with $|\Gamma| = 1$, this proves the optimality of the convex curve Γ . Also note that any optimal curve needs to be convex.

The next proposition shows that C_p is not constant.

Proposition 5.5. We have $C_p > 1$ for all p large enough.

Proof. Consider the curve Γ parametrised by $\gamma: [0, 1] \to \mathbb{R}^2$, $\gamma(s) = (\gamma_1(s), \gamma_2(s))^{\top}$ with $\gamma_2(s) = 0$ for all $s \in [0, 1]$ and

$$\gamma_1(s) = \begin{cases} -\frac{1}{4} + s & \text{for } s \in \left[0, \frac{1}{2}\right], \\ \frac{3}{4} - s & \text{for } s \in \left[\frac{1}{2}, 1\right] \end{cases}$$

 Γ is a closed curve that is parametrised by arc-length, centrally symmetric, its shape is the doubly covered interval $\left[-\frac{1}{4}, \frac{1}{4}\right] \times \{0\}$, and its length is $|\Gamma| = 1$. We have

$$\lim_{p \to \infty} \left(\int_{\Gamma} |x|^p \, \mathrm{d} \, \mathcal{H}^1(x) \right)^{1/p} = \lim_{p \to \infty} \left(\int_{0}^{1} |\gamma_1(s)|^p \, \mathrm{d} \, s \right)^{1/p} = \sup_{s \in [0,1]} |\gamma_1(s)| = \frac{1}{4},$$

and therefore, by $|\Gamma| = 1$,

$$\lim_{p \to \infty} \left(\frac{(2\pi)^p}{|\Gamma|^{p+1}} \int_{\Gamma} |x|^p \, \mathrm{d} \, \mathcal{H}^1(x) \right)^{1/p} = \frac{2\pi}{4} > 1.$$

This proves $C_p > 1$ for all *p* large enough.

This leads to the question of determining the critical value

$$p_* := \sup\{p : p > 0 \text{ and } C_p = 1\}.$$

The following proposition shows that $p_* \leq 3$, see (ii), which we conjecture to be optimal (see Conjecture 1.4) since the disk is a local optimiser for p < 3, see (i) below.

Proposition 5.6. *The following statements hold.*

(i) Let p < 3. Then the disk is a local optimiser among centrally symmetric curves in the following sense. If r: R/2πZ → R is continuous with r(θ) = r(θ + π) for all θ, and the curve Γ_ε is parametrised by γ_ε: R/2πZ → R² with

$$\gamma_{\varepsilon}(\theta) = \begin{pmatrix} R_{\varepsilon}(\theta)\cos(\theta) \\ R_{\varepsilon}(\theta)\sin(\theta) \end{pmatrix}, \quad \text{where } R_{\varepsilon}(\theta) = 1 + \varepsilon r(\theta), \quad (5.3)$$

then

$$\frac{(2\pi)^p}{|\Gamma_{\varepsilon}|^{p+1}} \int_{\Gamma_{\varepsilon}} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x) \le 1$$

for all $\varepsilon > 0$ small enough; furthermore, the inequality is strict if r is non-constant.

(ii) Let p > 3. Then the disk is not optimal, not even locally: There exists a sequence of nearly circular, centrally symmetric closed Lipschitz curves $(\Gamma_n)_{n \in \mathbb{N}}$ converging uniformly to the boundary of the disk for which

$$\frac{(2\pi)^p}{|\Gamma_n|^{p+1}} \int_{\Gamma_n} |x|^p \,\mathrm{d}\,\mathcal{H}^1(x) > 1.$$

Proof. Consider the curve Γ_{ε} defined by the parametrisation of Γ_{ε} in (5.3). Since we assume that $r(\theta) = r(\theta + \pi)$, the curve Γ_{ε} is centrally symmetric.

It is straightforward to check that with γ_{ε} defined as in (5.3),

$$\begin{aligned} |\gamma_{\varepsilon}(\theta)|^{p} &= 1 + \varepsilon p r(\theta) + \frac{\varepsilon^{2}}{2} p(p-1) |r(\theta)|^{2} + \mathcal{O}(\varepsilon^{3}), \\ |\gamma_{\varepsilon}'(\theta)| &= 1 + \varepsilon r(\theta) + \frac{\varepsilon^{2}}{2} |r'(\theta)|^{2} + \mathcal{O}(\varepsilon^{3}), \\ |\Gamma_{\varepsilon}| &= 2\pi + \varepsilon \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta + \frac{\varepsilon^{2}}{2} \int_{0}^{2\pi} |r'(\theta)|^{2} \,\mathrm{d}\,\theta + \mathcal{O}(\varepsilon^{3}). \end{aligned}$$
With the above formulas in hand, we get

$$\frac{|\Gamma_{\varepsilon}|^{p+1}}{(2\pi)^{p}} = 2\pi + \varepsilon(p+1) \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\theta + \varepsilon^{2}(p+1) \left(\frac{1}{2} \int_{0}^{2\pi} |r'(\theta)|^{2} \,\mathrm{d}\theta + \frac{p}{4\pi} \left(\int_{0}^{2\pi} r(\theta) \,\mathrm{d}\theta\right)^{2}\right) + \mathcal{O}(\varepsilon^{3}),$$

and

$$\int_{\Gamma_{\varepsilon}} |x|^{p} \,\mathrm{d}\,\mathcal{H}^{1}(x) - \frac{|\Gamma_{\varepsilon}|^{p+1}}{(2\pi)^{p}} = \frac{\varepsilon^{2} p}{2} (\mathcal{F}(r) + \mathcal{O}(\varepsilon)), \tag{5.4}$$

where

$$\mathcal{F}(r) = (p+1) \int_{0}^{2\pi} |r(\theta)|^2 \,\mathrm{d}\,\theta - \frac{p+1}{2\pi} \left(\int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta \right)^2 - \int_{0}^{2\pi} |r'(\theta)|^2 \,\mathrm{d}\,\theta$$
$$= \int_{0}^{2\pi} \left[(p+1) \left(r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta \right)^2 - |r'(\theta)|^2 \right] \mathrm{d}\,\theta.$$

We expand r as a Fourier series and notice that the coefficients of the odd indices will vanish, thanks to the symmetry condition on r. More precisely, we have

$$r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta = \sum_{n \ge 2} a_n \cos(n\theta) + \sum_{n \ge 2} b_n \sin(n\theta),$$
$$r'(\theta) = -\sum_{n \ge 2} n a_n \sin(n\theta) + \sum_{n \ge 2} n b_n \cos(n\theta).$$

By Parseval's identity, we write

$$\int_{0}^{2\pi} \left(r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \, \mathrm{d}\, \theta \right)^{2} \mathrm{d}\, \theta = \pi \sum_{n \ge 2} \left(|a_{n}|^{2} + |b_{n}|^{2} \right),$$
$$\int_{0}^{2\pi} |r'(\theta)|^{2} \, \mathrm{d}\, \theta = \pi \sum_{n \ge 2} n^{2} \left(|a_{n}|^{2} + |b_{n}|^{2} \right).$$

Hence,

$$\int_{0}^{2\pi} |r'(\theta)|^2 \,\mathrm{d}\,\theta \ge 4 \int_{0}^{2\pi} \left(r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta \right)^2 \mathrm{d}\,\theta, \tag{5.5}$$

and consequently, for p < 3

$$\mathcal{F}(r) \le (p-3) \int_{0}^{2\pi} \left(r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta \right)^{2} \,\mathrm{d}\,\theta \le 0.$$

For p < 3, $\mathcal{F}(r)$ vanishes if and only if the function r is constant. To conclude the proof of (i), we take $\varepsilon \to 0$ and note that the disk is a strict local maximiser if and only if we have for all non-constant r and ε small enough

$$\int_{\Gamma_{\varepsilon}} |x|^p \, \mathrm{d}\,\mathcal{H}^1(x) - \frac{|\Gamma_{\varepsilon}|^{p+1}}{(2\pi)^p} < 0.$$

Taking $\varepsilon \to 0$ and using (5.4), we obtain the desired result.

For (ii), note that choosing $r(\theta) := \sin(2\theta)$, we have equality in (5.5), which leads to

$$\mathcal{F}(r) = (p-3) \int_{0}^{2\pi} \left(r(\theta) - \frac{1}{2\pi} \int_{0}^{2\pi} r(\theta) \,\mathrm{d}\,\theta \right)^{2} \,\mathrm{d}\,\theta.$$

And for p > 3, we get $\mathcal{F}(r) > 0$, which yields the claim by (5.4) for ε small enough.

Remark 5.7. Proposition 5.6 does not address the case p = 3 since it is a degenerate case: the quantity $\mathcal{F}(r)$ is zero for the optimal choice $r(\theta) := \sin(2\theta)$.

Note that, by Remark 5.3, C_p is non-decreasing in p. Furthermore, we deduce that C_p is left-continuous by the continuity in p of the expression inside the supremum in (5.2) for every fixed Γ . Hence,

$${p \in [0, \infty) : C_p = 1} = [0, p_*].$$

So in order to prove Conjecture 1.4, namely that $p_* = 3$, it suffices to show $C_p = 1$ for all p < 3.

Remark 5.8 (Optimal curve for large p). The curve Γ considered in the proof of Proposition 5.5 is the optimal curve in the case $p = \infty$ when replacing the L^p norms by the corresponding supremum norms. For values of $p \in (3, \infty)$, we conjecture the optimal curve to be a deformed circle that degenerates into the curve from the proof of Proposition 5.5 as p increases, compare also with the proof of Proposition 5.6 (ii) above.

6. Proof of Theorem 1.5 – Applications to the magnetic Robin Laplacian

In this section, we show that Theorem 1.1 can be used to relax the assumptions on the domain in the isoperimetric inequality for the lowest eigenvalue of the magnetic Robin Laplacian on a bounded domain with a negative boundary parameter, recently obtained in [19] by the second and the third authors of the present paper.

The operator we study involves the vector potential (magnetic potential)

$$\mathbf{A}(x) := \frac{1}{2} (-x_2, x_1)^{\top} \quad (x = (x_1, x_2)).$$

and two parameters, $b \ge 0$ standing for the intensity of the magnetic field and $\beta \le 0$, the Robin parameter, appearing in the boundary condition. Let $\Omega \subset \mathbb{R}^2$ be a bounded simply-connected smooth domain. Our magnetic Robin Laplacian, $H_{\Omega}^{\beta,b}$, is the self-adjoint operator defined by the closed, symmetric, densely defined and lower semibounded quadratic form

$$\mathfrak{q}_{\Omega}^{\beta,b}[u] := \| (\nabla - \mathbf{i}b\mathbf{A})u \|_{L^{2}(\Omega;\mathbb{C}^{2})}^{2} + \beta \| u \|_{L^{2}(\partial\Omega)}^{2}, \quad \operatorname{dom} \mathfrak{q}_{\Omega}^{\beta,b} := H^{1}(\Omega),$$

and it is characterised by

dom
$$\mathsf{H}_{\Omega}^{\beta,b} = \{ u \in H^1(\Omega) : \text{there exists } w \in L^2(\Omega) \text{ such that}$$

 $\mathfrak{q}_{\Omega}^{\beta,b}[u,v] = (w,v)_{L^2(\Omega)} \text{ for all } v \in \operatorname{dom} \mathfrak{q}_{\Omega}^{\beta,b} \},$
 $\mathsf{H}_{\Omega}^{\beta,b}u := -(\nabla - \mathrm{i}b\mathbf{A})^2 u = w.$

Denoting by ν the unit inward normal vector on $\partial\Omega$, we observe that functions in dom $H_{\Omega}^{\beta,b}$ satisfy the (magnetic) Robin boundary condition

$$v \cdot (\nabla - \mathrm{i}b\mathbf{A})u = \beta u \quad \text{on } \partial\Omega.$$

The isoperimetric inequality obtained in [19] concerns the lowest eigenvalue of $H_{\Omega}^{\beta,b}$, which we express in the variational form as follows:

$$\lambda_1^{\beta,b}(\Omega) := \inf_{u \in H^1(\Omega) \setminus \{0\}} \frac{\mathfrak{q}_{\Omega}^{\beta,b}[u]}{\|u\|_{L^2(\Omega)}^2}.$$

Denoting by \mathcal{B} the disk in \mathbb{R}^2 centred at the origin, with radius R and having the same perimeter $2\pi R = |\partial \Omega|$ as the domain Ω , it is known that the following inequality holds (see [19, Theorem 4.8, Corollary 4.9]):

$$\lambda_1^{\beta,b}(\Omega) \leq \lambda_1^{\beta,b}(\mathcal{B}),$$

provided that

- (i) $\beta < 0$ and $0 < b < \min(R^{-2}, 4\sqrt{-\beta} R^{-3/2})$ (i.e., the magnetic field's intensity *b* is of moderate strength); and
- (ii) the *inner parallel curves* of Ω obey the condition

$$\int_{S_t} |x - x_0|^2 \, \mathrm{d}\,\mathcal{H}^1(x) \le \frac{(L - 2\pi t)^3}{4\pi^2} \tag{6.1}$$

for some fixed point $x_0 \in \mathbb{R}^2$ and almost all $t \in (0, r_i(\Omega))$. This condition holds for instance, when $\Omega \subset \mathcal{B}$ or when Ω is *convex* and *centrally symmetric* (see [19, Proposition 4.4])

Proof of Theorem 1.5. In view of Theorem 1.1, the condition in (6.1) holds with $x_0 = 0$ for all bounded centrally symmetric simply-connected smooth domains or, more generally, with x_0 being the centroid of all $S_t(\Omega)$ for all simply-connected smooth domains Ω such that the centroid of the inner parallel curve $S_t(\Omega)$ is independent of t. Thus, we relaxed the convexity assumption on the domain Ω . We obtain Theorem 1.5 with the choice $b_0(|\partial \Omega|, \beta) = \min\{R^{-2}, 4\sqrt{-\beta}R^{-3/2}\}$, where $R := \frac{|\partial \Omega|}{2\pi}$.

7. Some direct consequences of Theorems 1.1 and 1.2

7.1. A refined bound on the length of the disconnected inner parallel curve

In this subsection we use Theorem 3.1 to get a refined upper bound on the length of the inner parallel curve S_t in the situation when S_t consists of several connected components.

Let $\Omega \subset \mathbb{R}^2$ be a bounded simply-connected smooth domain with perimeter L > 0. Let the inner parallel curve $S_t \subset \Omega$ be as in (1.1). For any $t \in \mathcal{L}$, we have by Proposition 2.1 for some $N \in \mathbb{N}$

$$S_t = \bigcup_{n=1}^N \Gamma_n, \quad |S_t| \le L - 2\pi t,$$

where $\{\Gamma_n\}_{n=1}^N$ are piecewise-smooth closed simple curves that are pairwise disjoint. In the case that S_t is connected, one has N = 1. However, in general, N can be an arbitrarily large integer number. In the case that N = 2, we immediately get as a consequence of Theorem 3.1

$$|S_t| + 2\operatorname{dist}(\Gamma_1, \Gamma_2) \le L - 2\pi t.$$

This observation can be generalised to the case of arbitrary $N \in \mathbb{N}$ to improve Hartman's bound (1.2), see [16], on the length of the inner parallel curve S_t . **Corollary 7.1.** For all $t \in \mathcal{L}$, it holds that

$$|S_t| + \sum_{n=1}^N \operatorname{dist}(\Gamma_n, S_t \setminus \Gamma_n) \leq L - 2\pi t.$$

Proof. The length of the closed piecewise-smooth curve parametrised by the mapping σ_t constructed in the proof of Theorem 3.1 is given by

$$\ell(\sigma_t) = |S_t| + \sum_{k=1}^m |\mathcal{I}_k|.$$

Every \mathcal{I}_k connects some $\Gamma_{n(k)}$ with some $\Gamma_{n(k+1)}$, and for each $n \in \{1, ..., N\}$ there is at least one $k \in \{1, ..., m\}$ such that n = n(k). Hence, we get that

$$\sum_{k=1}^{m} |\mathcal{I}_k| \geq \sum_{k=1}^{m} \operatorname{dist}(\Gamma_{n(k)}, \Gamma_{n(k+1)}) \geq \sum_{n=1}^{N} \operatorname{dist}(\Gamma_n, S_t \setminus \Gamma_n).$$

Thus, we conclude that

$$|S_t| + \sum_{n=1}^N \operatorname{dist}(\Gamma_n, S_t \setminus \Gamma_n) \le \ell(\sigma_t) \le L - 2\pi t.$$

7.2. Moments of inertia of domains

In this subsection, we apply Theorem 1.1 to recover an isoperimetric upper bound on the moment of inertia for the domain Ω itself leading to an alternative proof of a result due to Hadwiger [14].

Assume that $\Omega \subset \mathbb{R}^2$ is a bounded, simply-connected, centrally symmetric domain. Then

$$\int_{\Omega} |x|^2 \,\mathrm{d}\, x \le \int_{\mathscr{B}} |x|^2 \,\mathrm{d}\, x,\tag{7.1}$$

where $\mathcal{B} \subset \mathbb{R}^2$ is a disk centred at the origin with the same perimeter as Ω .

Remark 7.2. The isoperimetric inequality (7.1) can be derived from the inequality by Hadwiger [14], where only convex domains were considered. Let $\mathcal{K} \subset \mathbb{R}^2$ be a bounded convex domain with a Lipschitz boundary. We can translate the domain \mathcal{K} so that the origin becomes the centroid of \mathcal{K} in the sense that

$$\int_{\mathcal{K}} x \, \mathrm{d} \, x = 0.$$

Let $\mathcal{B}' \subset \mathbb{R}^2$ be the disk centred at the origin of the same perimeter as \mathcal{K} . It is proved in [14] that

$$\int_{\mathcal{K}} |x|^2 \, \mathrm{d}\, x \le \int_{\mathcal{B}'} |x|^2 \, \mathrm{d}\, x. \tag{7.2}$$

Let us define the domain \mathcal{K} as the convex hull of the bounded simply-connected centrally symmetric smooth $\Omega \subset \mathbb{R}^2$. Then, the perimeter of \mathcal{K} does not exceed the perimeter of Ω . This is a well-known fact, whose proof can be found, e.g., in [29]. Moreover, the convex domain \mathcal{K} is centrally symmetric as well. Therefore, the origin is the centroid of \mathcal{K} . Hence, we get from (7.2)

$$\int_{\Omega} |x|^2 \, \mathrm{d} \, x \leq \int_{\mathcal{K}} |x|^2 \, \mathrm{d} \, x \leq \int_{\mathcal{B}'} |x|^2 \, \mathrm{d} \, x \leq \int_{\mathcal{B}} |x|^2 \, \mathrm{d} \, x$$

where we used that the perimeter of Ω is larger than or equal to the perimeter of \mathcal{K} , so the radius of \mathcal{B}' does not exceed the radius of \mathcal{B} .

Remark 7.3. If we furthermore assume that Ω is smooth, we have the following proof of (7.1) using Theorem 1.1. Recall that $\Omega \subset \mathbb{R}^2$ is a bounded simply-connected smooth domain with the perimeter L > 0 and the origin being the centroid of S_t for almost every $t \in (0, r_i(\Omega))$, and that $\mathcal{B} \subset \mathbb{R}^2$ is the disk of radius $R = \frac{L}{2\pi}$, having thus the same perimeter as Ω . By the geometric isoperimetric inequality, we have $|\Omega| \leq |\mathcal{B}|$ and therefore it holds that $R \geq r_i(\Omega)$.

Recall the co-area formula in two dimensions (see [2, Theorem 4.20] and [22]). If $\mathcal{A} \subset \mathbb{R}^2$ is an open set, $f : \mathcal{A} \to \mathbb{R}$ is a Lipschitz continuous real-valued function, and $g : \mathcal{A} \to \mathbb{R}$ is an integrable function, then we have

$$\int_{\mathcal{A}} g(x) |\nabla f(x)| \, \mathrm{d} \, x = \int_{\mathbb{R}} \int_{f^{-1}(t)} g(x) \, \mathrm{d} \, \mathcal{H}^{1}(x) \, \mathrm{d} \, t.$$
(7.3)

Applying the co-area formula (7.3) with $\mathcal{A} = \Omega$, $g(x) = |x|^2$ and $f(x) = \rho(x)$ (the distance function to the boundary of Ω defined in (2.4)) we get using the inequality in Theorem 1.1,

$$\int_{\Omega} |x|^2 \, \mathrm{d} \, x = \int_{0}^{r_i(\Omega)} \int_{S_t} |x|^2 \, \mathrm{d} \, \mathcal{H}^1(x) \le \int_{0}^{r_i(\Omega)} \frac{(L - 2\pi t)^3}{4\pi^2} \, \mathrm{d} \, t$$
$$\le \int_{0}^{R} \frac{(L - 2\pi t)^3}{4\pi^2} \, \mathrm{d} \, t = 2\pi \int_{0}^{R} (R - t)^3 \, \mathrm{d} \, t = \frac{\pi R^4}{2} = \int_{\mathcal{B}} |x|^2 \, \mathrm{d} \, x.$$

Acknowledgements. C. Dietze would like to express her deepest gratitude to Phan Thành Nam for his continued support and very helpful advice. She would also like to thank Laure Saint-Raymond for her support, inspiring discussions and hospitality at Institut des Hautes Études Scientifiques, where large parts of this work were carried out.

Special thanks go to Mirek Olšák for his ideas and discussions on an alternative elementary proof of Proposition 2.7 different from the proof presented in this paper. Dietze would like to thank Prasuna Bandi, Sami Fersi, Matthias Paulsen, and Shuddhodan Kadattur Vasudevan, and Leonard Wetzel for helpful discussions on Theorem 1.2. Furthermore, she would like to thank Bo Berndtsson, Jan Dereziński, Dmitry Faifman, Rupert Frank, Jonas Peteranderl, Heinz Siedentop, and Jakob Stern for helpful discussions on Theorem 1.3 and further directions. Dietze thanks Giovanni Brigati for helpful discussions on Remark 5.4. She would also like to thank Larry Read for helpful discussions and his help in creating the images for the introduction. A. Kachmar and Dietze wish to thank Petru Mironescu for discussions around [6]. The authors would like to thank the referees for many helpful remarks.

Funding. C. Dietze acknowledges the support from the Deutsche Forschungsgemeinschaft (DFG project Nr. 426365943), from the Jean-Paul Gimon Fund, and from the Erasmus+ programme. A. Kachmar acknowledges the support from The Chinese University of Hong Kong, Shenzhen (grant UDF01003322 and UF02003322), and from the Knut and Alice Wallenberg Foundation (grant KAW 2021.0259). V. Lotoreichik acknowledges the support by the grant No. 21-07129S of the Czech Science Foundation (GAČR).

References

- [1] A. Alvino, F. Brock, F. Chiacchio, A. Mercaldo, and M. R. Posteraro, Some isoperimetric inequalities on ℝ^N with respect to weights |x|^α. J. Math. Anal. Appl. 451 (2017), no. 1, 280–318 Zbl 1367.49005 MR 3619238
- [2] A. Baernstein, II, Symmetrization in analysis. With D. Drasin and R. S. Laugesen, With a foreword by W. Hayman. New Math. Monogr. 36, Cambridge University Press, Cambridge, 2019 Zbl 1509.32001 MR 3929712
- [3] M. F. Betta, F. Brock, A. Mercaldo, and M. R. Posteraro, A weighted isoperimetric inequality and applications to symmetrization. J. Inequal. Appl. 4 (1999), no. 3, 215–240 Zbl 1029.26018 MR 1734159
- [4] G. Bol, Isoperimetrische Ungleichungen f
 ür Bereiche auf Fl
 ächen. Jber. Deutsch. Math.-Verein. 51 (1941), 219–257 Zbl 0026.08901 MR 0018858
- [5] T. Bonnesen, Über das isoperimetrische Defizit ebener Figuren. Math. Ann. 91 (1924), no. 3-4, 252–268 JFM 50.0487.03 MR 1512192

- [6] H. Brezis and P. Mironescu, Sobolev maps to the circle—from the perspective of analysis, geometry, and topology. Progress in Nonlinear Differential Equations and their Applications 96, Birkhäuser/Springer, New York, 2021 Zbl 1501.46001 MR 4390036
- [7] X. Cabré, X. Ros-Oton, and J. Serra, Sharp isoperimetric inequalities via the ABP method. J. Eur. Math. Soc. (JEMS) 18 (2016), no. 12, 2971–2998 Zbl 1357.28007 MR 3576542
- [8] D. R. J. Chillingworth, Winding numbers on surfaces. I. Math. Ann. 196 (1972), 218–249
 Zbl 0221.57001 MR 0300304
- [9] M. Cicalese and G. P. Leonardi, A selection principle for the sharp quantitative isoperimetric inequality. *Arch. Ration. Mech. Anal.* 206 (2012), no. 2, 617–643 Zbl 1257.49045 MR 2980529
- [10] E. Cinti, F. Glaudo, A. Pratelli, X. Ros-Oton, and J. Serra, Sharp quantitative stability for isoperimetric inequalities with homogeneous weights. *Trans. Amer. Math. Soc.* 375 (2022), no. 3, 1509–1550 Zbl 1483.49053 MR 4378069
- [11] F. Fiala, Le problème des isopérimètres sur les surfaces ouvertes à courbure positive. Comment. Math. Helv. 13 (1941), 293–346 Zbl 0025.23003 MR 0006422
- B. Fuglede, Stability in the isoperimetric problem for convex or nearly spherical domains in Rⁿ. Trans. Amer. Math. Soc. 314 (1989), no. 2, 619–638 Zbl 0679.52007
 MR 0942426
- [13] N. Fusco, The quantitative isoperimetric inequality and related topics. Bull. Math. Sci. 5 (2015), no. 3, 517–607 Zbl 1327.49076 MR 3404715
- [14] H. Hadwiger, Konkave Eikörperfunktionale und höhere Trägheitsmomente. Comment. Math. Helv. 30 (1956), 285–296 Zbl 0073.17305 MR 0080942
- [15] G. H. Hardy, J. E. Littlewood, and G. Pólya, *Inequalities*. 2nd edition. Cambridge University Press, Cambridge, MA, 1952 Zbl 0047.05302 MR 0046395
- [16] P. Hartman, Geodesic parallel coordinates in the large. *Amer. J. Math.* 86 (1964), 705–727
 Zbl 0128.16105 MR 0173222
- [17] R. Howard and A. Treibergs, A reverse isoperimetric inequality, stability and extremal theorems for plane curves with bounded curvature. *Rocky Mountain J. Math.* 25 (1995), no. 2, 635–684 Zbl 0909.53002 MR 1336555
- [18] A. Hurwitz, Sur quelques applications géométriques des séries de Fourier. Ann. Sci. École Norm. Sup. (3) 19 (1902), 357–408 Zbl 33.0599.02 MR 1509016
- [19] A. Kachmar and V. Lotoreichik, On the isoperimetric inequality for the magnetic Robin Laplacian with negative boundary parameter. J. Geom. Anal. 32 (2022), no. 6, article no. 182 Zbl 1487.35260 MR 4411745
- [20] W. Klingenberg, A course in differential geometry. Grad. Texts in Math. 51, Springer, Berlin etc., 1978 Zbl 0366.53001 MR 0474045
- [21] J. M. Lee, Introduction to Riemannian manifolds. Grad. Texts in Math. 176, Springer, Cham, 2018 MR 3887684
- [22] J. Malý, D. Swanson, and W. P. Ziemer, The co-area formula for Sobolev mappings. *Trans. Amer. Math. Soc.* 355 (2003), no. 2, 477–492 Zbl 1034.46032 MR 1932709
- [23] D. A. L. Manna and R. Sannipoli, Some isoperimetric inequalities involving the boundary momentum. 2023, arXiv:2309.14191v2

- [24] R. Osserman, The isoperimetric inequality. Bull. Amer. Math. Soc. 84 (1978), no. 6, 1182–1238 Zbl 0411.52006 MR 0500557
- [25] R. Osserman, Bonnesen-style isoperimetric inequalities. Amer. Math. Monthly 86 (1979), no. 1, 1–29 Zbl 0404.52012 MR 0519520
- [26] G. Pestov and V. Ionin, On the largest possible circle imbedded in a given closed curve. Dokl. Akad. Nauk SSSR 127 (1959), 1170–1172 Zbl 0086.36104 MR 0107214
- [27] A. Savo, Lower bounds for the nodal length of eigenfunctions of the Laplacian. Ann. Global Anal. Geom. 19 (2001), no. 2, 133–151 Zbl 1010.58025 MR 1826398
- [28] K. Shiohama, T. Shioya, and M. Tanaka, *The geometry of total curvature on complete open surfaces*. Cambridge Tracts in Math. 159, Cambridge University Press, Cambridge, 2003 Zbl 1086.53056 MR 2028047
- [29] A. Treibergs, Inequalities that imply the isoperimetric inequality. 2002, http://www.math. utah.edu/~treiberg/isoperim/isop.pdf visited on 24 September 2024
- [30] H. Whitney, On regular closed curves in the plane. *Compositio Math.* 4 (1937), 276–284
 Zbl 63.0647.01 MR 1556973

Received 15 March 2024; revised 29 July 2024.

Charlotte Dietze

Department of Mathematics, LMU Munich, Theresienstrasse 39, 80333 Munich, Germany; dietze@math.lmu.de

Ayman Kachmar

School of Science and Engineering, Chinese University of Hong Kong, Shenzhen, 2001 Longxiang Boulevard, Longgang District, 518172 Shenzhen, P. R. China; akachmar@cuhk.edu.cn

Vladimir Lotoreichik

Department of Theoretical Physics, Nuclear Physics Institute, Czech Academy of Sciences, Hlavní 130, 25068 Řež, Czech Republic; lotoreichik@ujf.cas.cz



Hardy–Sobolev interpolation inequalities

Charlotte Dietze¹ · Phan Thành Nam¹

Received: 21 December 2023 / Accepted: 24 June 2024 / Published online: 16 July 2024 © The Author(s) 2024

Abstract

We derive a family of interpolation estimates which improve Hardy's inequality and cover the Sobolev critical exponent. We also determine all optimizers among radial functions in the endpoint case and discuss open questions on nonrestricted optimizers.

Mathematics Subject Classification 35A23 · 35Q55

1 Introduction

The classical Hardy inequality states that for every dimension $d \ge 3$,

$$h[u] := \int_{\mathbb{R}^d} |\nabla u(x)|^2 \, \mathrm{d}x - \frac{(d-2)^2}{4} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} \, \mathrm{d}x \ge 0, \quad \forall u \in \dot{H}^1(\mathbb{R}^d).$$
(1)

In this short article we are interested in interpolation inequalities involving the quadratic form h[u] and L^p -norms of u. A classical result in this direction is the Gagliardo–Nirenberg type inequality

$$h[u]^{\theta} \|u\|_{L^{2}}^{2(1-\theta)} \ge C \|u\|_{L^{q}}^{2}, \quad \forall u \in H^{1}(\mathbb{R}^{d})$$
(2)

for a constant C > 0 independent of u, which holds for every

$$d \ge 3, \quad 2 < q < 2^* = \frac{2d}{d-2}, \quad \theta = d\left(\frac{1}{2} - \frac{1}{q}\right).$$

The inequality (2) can be deduced from the results of Brezis and Vázquez [3, Theorem 4.1 and Extension 4.3]; see [23] for related results. The bound (2) can be also derived from Sobolev's embedding theorem and the kinetic estimate

$$h[u]^{\theta} \|u\|_{2}^{2(1-\theta)} \ge C \|(-\Delta)^{s/2}u\|_{2}^{2}, \quad \forall u \in H^{1}(\mathbb{R}^{d}),$$
(3)

for $s \in (0, 1)$ and $\theta = \theta(s)$, which was proved by Frank [6, Theorem 1.2]. Both of (2) and (3) have been extended to the fractional Laplacian in [6], motivated by applications in the

Communicated by Susanna Terracini.

Charlotte Dietze dietze@math.lmu.de Phan Thành Nam nam@math.lmu.de

¹ Department of Mathematics, LMU Munich, Theresienstrasse 39, 80333 Munich, Germany

asymptotic behavior of large Coulomb systems [20] and the stability of relativistic matter [6, 8].

Note that the restriction $q < 2^*$ in (2) is necessary, namely the quadratic form h[u] is really weaker than $\|\nabla u\|_{L^2}^2$. Here we are interested in a replacement of (2) which covers the critical power $q = 2^*$, with the expense that the L^2 -norm is replaced by the energy associated with the inverse square potential. We have

Theorem 1 (*Hardy-Sobolev interpolation inequality*) If d = 3 and $\theta = 1/3$, then the inequality

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} \, \mathrm{d}x\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} \, \mathrm{d}x\right)^{1-\theta} \ge C \|u\|_{L^{2^*}}^2$$
(4)

holds with a constant $C = C(d, \theta) > 0$ independent of $u \in \dot{H}^1(\mathbb{R}^d)$. Moreover, (4) does not hold if $d \ge 4$ or if $\theta \ne 1/3$.

Remark 1 The bound (4) is invariant under translations and dilations. Note that for the first term on the left-hand side, Hardy's inequality (1) is equivalent to

$$\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} \, \mathrm{d}x \ge 0, \quad \forall u \in \dot{H}^1(\mathbb{R}^d).$$

For the second term, it is important to include $\sup_{y \in \mathbb{R}^d}$ since otherwise this term can be made arbitrarily small by translation $u \mapsto u(\cdot - z)$ with $|z| \to \infty$.

Remark 2 For all $d \ge 3$ and $1 - 2/d \le \theta \le 1$ we have

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x - y|^2} \,\mathrm{d}x\right)^{1 - \theta} \ge C \|u\|_{L^{2^*}}^2, \quad \forall u \in H^1(\mathbb{R}^d).$$
(5)

This is a consequence of the improved Sobolev inequality involving Morrey norms

$$\left(\int_{\mathbb{R}^d} |\nabla u|^2\right)^{\theta} \left(\sup_{R>0, x\in\mathbb{R}^d} R^{-2} \int_{B(x,R)} |u|^2\right)^{1-\theta} \ge C \|u\|_{L^{2^*}}^2, \quad \forall u\in\dot{H}^1(\mathbb{R}^d), \tag{6}$$

which was proved by Palatucci–Pisante [17, Theorem 1], using subtle weighted L^p -estimates for Riesz potentials in [19] and Calderón-Zygmund type techniques in the spirit of the Fefferman–Phong argument [5]. The bound (6) is helpful to obtain the compactness of minimizing sequences of the critical Sobolev inequality; see [17, Theorem 3] for details. In contrast, our inequality (4) is stronger than (5) and it only holds for the special case $d = 1/\theta = 3$.

In the next result, we extend (4) by replacing the gradient term $\|\nabla u\|_{L^2}$ by $\|\nabla u\|_{L^p}$, as well as replacing the L^{2^*} -norm by the $L^{p^*,r}$ -Lorentz norm. Recall that (see [9, Definition 1.4.6 and Proposition 1.4.9])

$$\|u\|_{L^{p,r}} = \|u\|_{p,r} = \begin{cases} \left(p \int_0^\infty s^{r-1} |\{|u| > s\}|^{r/p} \, \mathrm{d}s\right)^{1/r}, & 0 < r < \infty, \\ \sup_{s > 0} s |\{|u| > s\}|^{1/p}, & r = \infty. \end{cases}$$

Theorem 2 [Hardy-Sobolev inequalities with Lorentz norms] Let $d \ge 2$, $p \in [2, d)$, $p^* = pd/(d-p)$, $r \in [p, \infty]$ and

$$\theta \in \left[\frac{p}{\min(r, p^*)}, \frac{1}{p} - \frac{1}{r}\right].$$
(7)

Then

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p - \left(\frac{d-p}{p}\right)^p \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x\right)^\theta \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x\right)^{1-\theta} \ge C \|u\|_{L^{p^*,r}}^p$$
(8)

with a constant $C = C(d, p, r, \theta) > 0$ independent of $u \in \dot{W}^{1,p}(\mathbb{R}^d)$. The bound (8) does not hold if $\theta < p/\min(r, p^*)$ (with arbitrary $p \ge 2$), or if $\theta > 1/p - 1/r$ and p = 2. In particular, when p = 2, the range of θ in (7) is optimal.

Theorems 1 and 2 naturally lead to the question of determining optimizers of the relevant inequalities. We expect that in the non-endpoint cases

$$\frac{p}{\min(r, p^*)} < \theta < \frac{1}{p} - \frac{1}{r},$$

the existence of optimizers of (8) follows from the standard concentration compactness method. Below we focus on the endpoint cases. While the existence of optimizers in this case is open in general, we are able to give a partial answer under the restriction to radial functions. We will limit ourselves to the choice p = 2 and $r \in \{2^*, \infty\}$, for which the righthand side of (8) becomes either the usual L^{2^*} -norm or the $L^{2^*,\infty}$ -weak norm. The relevant functional space is

$$\dot{H}^{1}_{\text{rad}}(\mathbb{R}^{d}) = \{ u \in \dot{H}^{1}(\mathbb{R}^{d}) : u \text{ is radially symmetric} \}.$$

In the radial case, we can work directly with the quadratic form h[u] in (1). We have

Theorem 3 (*Radial optimizers*) Let $d \ge 3$ and p = 2.

(i) Let $r = 2^* = 2d/(d-2)$ and $\theta = 1/p - 1/r = 1/d$. Then all optimizers of the inequality

$$h[u]^{\theta} \left(\int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} \, \mathrm{d}x \right)^{1-\theta} \ge C_{\mathrm{rad},2^*} \|u\|_{L^{2^*}}^2, \quad \forall u \in \dot{H}^1_{\mathrm{rad}}(\mathbb{R}^d)$$
(9)

are given by the family

$$u_{\eta}(x) = \frac{1}{(|x|^{1-\eta}(1+|x|^{2\eta}))^{(d-2)/2}}, \quad \eta \in (0,\infty),$$
(10)

up to dilation $u_{\eta}(x) \mapsto au_{\eta}(bx)$ with $a \in \mathbb{C}$, b > 0. Furthermore, (9) does not hold if $\theta \neq 1/d$.

(ii) Let $r = \infty$ and $\theta = 1/p - 1/r = 1/2$. Then all optimizers of the inequality

$$h[u]^{\theta} \left(\int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} \,\mathrm{d}x \right)^{1-\theta} \ge C_{\mathrm{rad},\infty} \|u\|_{L^{2^*,\infty}}^2, \quad \forall u \in \dot{H}^1_{\mathrm{rad}}(\mathbb{R}^d)$$
(11)

are given by the family

$$u_c(x) = \begin{cases} |x|^{c-d/2+1} &, |x| \le 1, \\ |x|^{-c-d/2+1} &, |x| > 1, \end{cases} \quad c \in (0, d/2 - 1], \tag{12}$$

🙆 Springer

up to dilation.

Remark 3 The classification of all optimizers in Theorem 3 (i) is consistent with Terracini's study in [21] where all radial positive solution of the Euler-Lagrange equation

$$-\Delta u(x) - \frac{(d-2)^2}{4}(1-\eta^2)\frac{u(x)}{|x|^2} = u^{2^*-1}(x), \quad x \in \mathbb{R}^d,$$
(13)

with a given constant $\eta > 0$, were derived. In particular, according to [21, Eq. (4.6)], the only regular solutions (i.e., belonging to L^{2^*}), up to rescaling, are of the form $(d(d-2)\eta^2)^{(d-2)/4}u_\eta$ with u_η given in (10).

Remark 4 We leave the following **open questions**: Do optimizers of (4) exist? And if exist, are they radial? The same questions for the simpler inequality (5) in the endpoint case $\theta = 1 - 2/d$ remain unsolved. Note that both in (4) and (5), all quantities scale in the same way. Moreover, by taking several bubbles travelling far from each other, one can construct optimizing sequences that do not converge weakly to a nonzero limit after any choice of dilations and translations. It seems that a novel concentration-compactness argument will be needed to resolve the existence problem of optimizers for these inequalities.

We prove Theorem 3 in Sect. 2, and then prove Theorems 1 and 2 in Sect. 3.

2 Radial case

In this section we prove Theorem 3. Let $u \in H^1_{rad}(\mathbb{R}^d)$ with $d \ge 3$.

.....

Proof of (i): Using the ground state representation for Hardy's inequality (see e.g. [7, Eq. (2.14)]), we denote

$$u(x) = \frac{f(|x|)}{|x|^{(d-2)/2}}$$
(14)

and rewrite

$$\int_{\mathbb{R}^d} |u|^{2^*} = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f(r)|^{2^*}}{r} \, \mathrm{d}r, \quad \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} \, \mathrm{d}x = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f(r)|^2}{r} \, \mathrm{d}r,$$
$$\int_{\mathbb{R}^d} |\nabla u|^2 - \frac{(d-2)^2}{4} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2} \, \mathrm{d}x = \int_{\mathbb{R}^d} \frac{|\nabla f(x)|^2}{|x|^{d-2}} \, \mathrm{d}x = |\mathbb{S}^{d-1}| \int_0^\infty |f'(r)|^2 r \, \mathrm{d}r.$$

Here $|\mathbb{S}^{d-1}|$ is the surface area of the unit sphere in \mathbb{R}^d . Thus (9) is equivalent to

$$\left(\int_{0}^{\infty} r|f'(r)|^{2} \mathrm{d}r\right)^{\theta} \left(\int_{0}^{\infty} \frac{|f(r)|^{2}}{r} \mathrm{d}r\right)^{1-\theta} \ge C_{\mathrm{rad},2^{*}} |\mathbb{S}^{d-1}|^{2/2^{*}-1} \left(\int_{0}^{\infty} \frac{|f(r)|^{2^{*}}}{r} \mathrm{d}r\right)^{2/2^{*}}$$
(15)

The bound (15) can be interpreted as a Caffarelli–Kohn–Nirenberg type inequality [4], namely

$$\||x|^{1/2} \nabla f\|_{L^{2}(\mathbb{R}_{+})}^{\theta} \||x|^{-1/2} f\|_{L^{2}(\mathbb{R}_{+})}^{1-\theta} \ge \sqrt{C_{\mathrm{rad},2^{*}}} |\mathbb{S}^{d-1}|^{1/2^{*}-1/2} \||x|^{\gamma} f\|_{L^{r}(\mathbb{R}_{+})}$$
(16)

with $r = 2^* = 2d/(d-2)$, $\gamma = -1/r$. Actually it is a limiting case as $1/r + \gamma/n = 0$ in dimension n = 1, which does not seem available from the literature (see [13, Theorem 3.1], [14, Theorem 1.2] and [15, Theorem 2.2] for recent results in the limiting case).

Inspired by [21], we make the following changes of variables

$$f(r) = \psi(\log r), \quad s = \log r \in \mathbb{R}, \quad \mathrm{d}s = \frac{\mathrm{d}r}{r},$$

which give

$$\int_0^\infty \frac{|f(r)|^2}{r} \, \mathrm{d}r = \int_{\mathbb{R}} |\psi(s)|^2 \, \mathrm{d}s, \quad \int_0^\infty \frac{|f(r)|^{2^*}}{r} \, \mathrm{d}r = \int_{\mathbb{R}} |\psi(s)|^{2^*} \, \mathrm{d}s,$$
$$\int_0^\infty |f'(r)|^2 r \, \mathrm{d}r = \int_{\mathbb{R}} |\psi'(s)|^2 \, \mathrm{d}s.$$

Therefore, (15) is equivalent to the Gagliardo–Nirenberg interpolation inequality

$$\left(\int_{\mathbb{R}} |\psi'(s)|^2 \,\mathrm{d}s\right)^{\theta} \left(\int_{\mathbb{R}} |\psi(s)|^2 \,\mathrm{d}s\right)^{1-\theta} \ge C_{\mathrm{rad},2^*} |\mathbb{S}^{d-1}|^{2/2^*-1} \left(\int_{\mathbb{R}} |\psi(s)|^{2^*} \,\mathrm{d}s\right)^{2/2^*}.$$
(17)

The optimal constant of the one-dimensional inequality (17) was already obtained by Nagy in 1941 [16], with $2^* = 2d/(d-2)$ replaced by a general positive power. The existence and uniqueness of optimizers of the analogue of (17) in higher dimensions are also well-known; we refer to the classical works of Weinstein [24] and Kwong [11] for instance. The uniqueness of optimizers of (17) can be translated straightforwardly to the classification of optimizers of (9) as stated in Theorem 3 (i); we refer to [21, Eq. (4.6)] for a similar analysis.

Remark 5 In the special case d = 3 (which is relevant to Theorem 1), the interpolation inequality (17) with $2^* = 6$ goes back to the (1D, one-body) Lieb–Thirring inequality [12] as well as Keller's lower bound on the lowest eigenvalue of the Schrödinger operator $-d^2/dx^2 + V(x)$ on $L^2(\mathbb{R})$ [10]; see also [1, Section 2] for a simple derivation of the optimal constant in this special case.

Unique choice of θ for (9): Consider (15) with the trial function

$$f(r) = \begin{cases} r^{\varepsilon}, & r \in (0, 1], \\ r^{-\varepsilon}, & r \in [1, \infty), \end{cases}$$
(18)

where $\varepsilon > 0$ is a parameter. Then we have

$$\int_0^\infty r |f'(r)|^2 \, \mathrm{d}r = \varepsilon, \quad \int_0^\infty \frac{|f(r)|^2}{r} \, \mathrm{d}r = \frac{1}{\varepsilon}, \quad \int_0^\infty \frac{|f(r)|^{2^*}}{r} \, \mathrm{d}r = \frac{2}{2^*\varepsilon}.$$
 (19)

Therefore, (15) requires to have

$$\varepsilon^{\theta}\varepsilon^{-(1-\theta)} \ge C\varepsilon^{-2/2^*} \tag{20}$$

for all $\varepsilon > 0$. By letting $\varepsilon \to 0$ and $\varepsilon \to \infty$, we find that

$$\theta = \frac{1 - 2/2^*}{2} = \frac{1}{d}.$$

Thus, (9) holds only if $\theta = 1/d$.

Proof of (ii): Let us consider (11). Using again the ground state representation (14), we have

$$\|u\|_{L^{2^*,\infty}} = \sup_{t>0} t |\{x : |u(x)| > t\}|^{1/2^*} = \sup_{t>0} t |\{x : |f(x)| > t|x|^{d/2^*}\}|^{1/2^*}$$

$$\leq \sup_{t>0} t |\{x : \|f\|_{L^{\infty}} > t |x|^{d/2^*}\}|^{1/2^*} = |B(0,1)|^{1/2^*} \|f\|_{L^{\infty}}.$$
 (21)

Here |B(0, 1)| is the volume of the unit ball in \mathbb{R}^d . Therefore, (11) holds if we can show that

$$\left(\int_0^\infty r|f'(r)|^2 \,\mathrm{d}r\right) \left(\int_0^\infty \frac{|f(r)|^2}{r} \,\mathrm{d}r\right) \ge C_{\mathrm{rad},\infty}^2 |\mathbb{S}^{d-1}|^{-2} |B(0,1)|^{4/2^*} ||f||_{L^\infty}^4. \tag{22}$$

From (14) and $u \in \dot{H}^1(\mathbb{R}^d)$, we deduce that the function $f : (0, \infty) \to [0, \infty)$ is continuous and satisfies

$$\lim_{r \to 0} f(r) = 0, \quad \lim_{r \to \infty} f(r) = 0.$$

Therefore, up to dilation, we may without loss of generality assume that

$$f(1) = \|f\|_{L^{\infty}}.$$

By the Cauchy-Schwarz inequality and the fundamental theorem of calculus, we have

$$\left(\int_{0}^{\infty} r|f'(r)|^{2} dr\right) \left(\int_{0}^{\infty} \frac{|f(r)|^{2}}{r} dr\right) \ge \left(\int_{0}^{\infty} |f'(r)f(r)| dr\right)^{2}$$

$$= \left(\frac{1}{2} \left(\int_{0}^{1} \left|\frac{\partial}{\partial r} (|f(r)|^{2})\right| dr + \int_{1}^{\infty} \left|\frac{\partial}{\partial r} (|f(r)|^{2})\right| dr\right)\right)^{2} \ge |f(1)|^{4} = \|f\|_{L^{\infty}}^{4}.$$
(23)

Thus (22) holds, and consequently (11) holds, with

$$C_{\rm rad,\infty}^2 |\mathbb{S}^{d-1}|^{-2} |B(0,1)|^{4/2^*} = 1.$$
(24)

To have the equality in (11), we need to ensure all equalities in (21) and (23). The bound (23) contains two inequalities where the first equality occurs if there exists a constant c > 0 such that

$$r|f'(r)|^2 = c \frac{|f(r)|^2}{r}, \quad \text{a.e. } r \in (0, \infty),$$

while the second equality occurs if $|f|^2$ is monotone increasing on (0, 1) and monotone decreasing on $(1, \infty)$. Thus, we have all equalities in (23) if and only if

$$f(r) = \begin{cases} r^c &, r \in (0, 1], \\ r^{-c} &, r \in [1, \infty). \end{cases}$$
(25)

It remains to determine the range of c in (25) to get the equality in (21). If $0 < c \le (d-2)/2$, then the function

$$u(x) = f(|x|)|x|^{-\frac{d-2}{2}} = \begin{cases} |x|^{c-\frac{d-2}{2}} &, |x| \le 1, \\ |x|^{-c-\frac{d-2}{2}} &, |x| \ge 1 \end{cases}$$
(26)

is radially symmetric decreasing, and the equality in (21) occurs since

$$\|u\|_{L^{2^*,\infty}} = \sup_{t>0} t |\{x : |u(x)| > t\}|^{1/2^*} \ge |\{x : |u(x)| > 1\}|^{1/2^*} = |B(0,1)|^{1/2^*}.$$
 (27)

On the other hand, if c > (d - 2)/2, then the inequality in (21) is strict: since u defined in (26) is bounded by 1, we have

$$||u||_{L^{2^*},\infty}^{2^*} = \sup_{0 < t < 1} t^{2^*} |\{x : |u(x)| > t\}|$$

Dispringer

$$= \sup_{0 < t < 1} t^{2^*} \left(|\{|x| \le 1 : |u(x)| > t\}| + |\{|x| > 1 : |u(x)| > t\}| \right)$$

$$= \sup_{0 < t < 1} t^{2^*} \left(|\{x : 1 \ge |x| > t^{\frac{2}{2c - (d - 2)}}\}| + |\{x : t^{-\frac{2}{2c + d - 2}} > |x| > 1\}| \right)$$

$$= |B(0, 1)| \sup_{0 < t < 1} t^{2^*} \left(t^{-\frac{2d}{2c + d - 2}} - t^{\frac{2d}{2c - (d - 2)}} \right) < |B(0, 1)|,$$

where the latter estimate can be easily seen using the fact that

$$2^* = \frac{2d}{d-2} > \frac{2d}{2c+d-2}$$

Thus, in summary, (11) holds with the optimal constant $C_{\text{rad},\infty}$ given in (24), and all optimizers are uniquely characterized up to dilation by (26) with $c \in (0, (d-2)/2]$.

The proof of Theorem 3 is complete.

3 General case

In this section we prove Theorem 1 and Theorem 2.

3.1 Proof of Theorem 1

We divide the proof into two parts. First, we prove (4) for $d = 3 = 1/\theta$. Then we show that the condition $d = 3 = 1/\theta$ is necessary.

Proof (Proof of (4) for $d = 3 = 1/\theta$) Let $u \in \dot{H}^1(\mathbb{R}^d)$ and denote

$$A = \int_{\mathbb{R}^d} |\nabla u|^2, \quad B = \frac{(d-2)^2}{4} \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x-y|^2} \, \mathrm{d}x.$$

We consider two cases.

Case 1: $(1 - \theta)A \ge B$. Then using

$$B \geq \frac{(d-2)^2}{4} \sup_{r>0, y \in \mathbb{R}^d} \int_{B(y,r)} \frac{|u(x)|^2}{|x-y|^2} \, \mathrm{d}x \gtrsim \sup_{r>0, y \in \mathbb{R}^d} \frac{1}{r^2} \int_{B(y,r)} |u|^2,$$

and $A - B \ge \theta A$, we conclude from (6) (see [17, Theorem 1]) that

$$(A-B)^{\theta}B^{1-\theta} \gtrsim \|\nabla u\|_{L^2}^{\theta} \left(\sup_{r>0, y \in \mathbb{R}^d} \frac{1}{r^2} \int_{B(y,r)} |u|^2 \right)^{1-\theta} \gtrsim \|u\|_{L^{2^*}}^2.$$

Case 2: $(1 - \theta)A < B$. Then $B \mapsto (A - B)^{\theta}B^{1-\theta}$ is monotone decreasing since

$$\frac{d}{dB}((A-B)^{\theta}B^{1-\theta}) = ((1-\theta)A - B)(A-B)^{\theta-1}B^{-\theta} < 0.$$

Moreover, by the Hardy-Littlewood and Pólya-Szegö rearrangement inequalities (see e.g. [2, Lemma 1.6 and Theorem 4.7]) we have

$$B^* = \frac{(d-2)^2}{4} \int_{\mathbb{R}^d} \frac{|u^*(x)|^2}{|x|^2} \, \mathrm{d}x \ge B, \quad A^* = \|\nabla u^*\|_{L^2}^2 \le A,$$

where u^* denotes the radially symmetric decreasing rearrangement of u. Therefore,

$$(A - B)^{\theta} B^{\theta} \ge (A - B^{*})^{\theta} (B^{*})^{1 - \theta} \ge (A^{*} - B^{*})^{\theta} (B^{*})^{1 - \theta}$$

Thus it remains to consider (4) in the case when u is radially symmetric decreasing. In this case,

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x - y|^2} \, \mathrm{d}x = \int_{\mathbb{R}^d} \frac{|u(x)|^2}{|x|^2}.$$
(28)

by the Hardy-Littlewood rearrangement inequality. Now the desired bound has been already proved in Theorem 3 (i).

Thus in all cases, (4) holds for $d = 3 = 1/\theta$.

Proof (Proof of the necessity of $d = 3 = 1/\theta$) Let us show that (4) fails if $d \ge 4$ or if $\theta \ne 1/3$. First, the necessity of $\theta \le 1/d$ can be seen from the radial case as explained in Theorem 3 (i). To be precise, we consider the example in (18) with $\varepsilon > 0$ small. In this case, u is radially symmetric decreasing, and hence (28) holds. Therefore, (4) requires (20) for $\varepsilon > 0$ small, which implies that $\theta \le 1/d$.

In order to complete the proof, we consider another (non-radial) example. Fix $\varphi \in C_c^{\infty} \setminus \{0\}, z \in \mathbb{R}^d \setminus \{0\}$ and choose

$$u_N(x) = \sum_{n=1}^N \varphi(x + nNz)$$

with $N \to \infty$. Then by replacing *u* by u_N , we find that

$$A \sim N, \quad B \sim 1, \quad \|u\|_{L^{2^*}}^{2^*} \sim N.$$

for large N. Therefore, (4) requires

$$N^{ heta} \gtrsim N^{2/2^*},$$

which implies that $\theta \ge 2/2^* = 1 - 2/d$. Combining with the upper bound $\theta \le 1/d$ and the constraint $d \ge 3$, we find that the only possibility is d = 3 and $\theta = 1/3$.

The proof of Theorem 1 is complete.

3.2 Proof of Theorem 2

We first prove (8) and then explain the necessity of the constraint of θ .

Proof (Proof of (8)) Let $d \ge 2$ and $p \in [2, d)$. Assume that $r \in [p, \infty]$ and

$$\frac{p}{\min(r, p^*)} \le \theta \le \frac{1}{p} - \frac{1}{r}.$$
(29)

Fix a small constant $\varepsilon = \varepsilon(d, p, r, \theta) \in (0, 1)$. We consider two cases.

Case 1: Assume

$$(1-\varepsilon)\int_{\mathbb{R}^d} |\nabla u|^p \ge \left(\frac{d-p}{p}\right)^p \sup_{y\in\mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \,\mathrm{d}x.$$
(30)

🖉 Springer

From [17, Theorem 1], see also [18, Eq. (1.4)], we have

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p\right)^{p/p^*} \left(\sup_{y \in \mathbb{R}^d, R > 0} \frac{1}{R^p} \int_{B(y,R)} |u(x)|^p \, \mathrm{d}x\right)^{1-p/p^*} \gtrsim \|u\|_{L^{p^*}}^p.$$
(31)

A simplified proof of (31) based on sharp maximal functions can be obtained by following the analysis in [22]. We can extend this bound to the Lorentz norm on $L^{p^*,r}$ with $r \in [p, p^*]$, namely

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p\right)^{p/r} \left(\sup_{y \in \mathbb{R}^d, R>0} \frac{1}{R^p} \int_{B(y,R)} |u(x)|^p \,\mathrm{d}x\right)^{1-p/r} \gtrsim \|u\|_{L^{p^*,r}}^p. \tag{32}$$

To prove (32), let us use the standard dyadic decomposition: recalling that $\varphi : \mathbb{R} \to \mathbb{R}$ is a smooth function supported on the annulus $1/2 \le |t| \le 2$ such that

$$\sum_{j\in\mathbb{Z}}\varphi(2^{-j}t)=1,\quad\forall t\in\mathbb{R}\backslash\{0\},$$

we write

$$u = \sum_{j \in \mathbb{Z}} u_j, \quad u_j = u\varphi(2^{-j}|u|).$$

Then

$$\|u\|_{L^{p^{*},r}}^{p^{*}} \sim \left(\sum_{j\in\mathbb{Z}} \|u_{j}\|_{L^{p^{*}}}^{r}\right)^{p^{*}/r} \\ \lesssim \left(\sum_{j\in\mathbb{Z}} \left(\int_{\mathbb{R}^{d}} |\nabla u_{j}|^{p}\right)^{r/p^{*}} \left(\sup_{y\in\mathbb{R}^{d},R>0} \frac{1}{R^{p}} \int_{B(y,R)} |u_{j}(x)|^{p} dx\right)^{r/d}\right)^{p^{*}/r},$$

where we used (31) for u_j and the fact that $(1 - p/p^*)r/p = r/d$ (as $1/p - 1/d = 1/p^*$). On the other hand, using

$$\sup_{y\in\mathbb{R}^d,R>0}\frac{1}{R^p}\int_{B(y,R)}|u_j(x)|^p\,\mathrm{d}x\lesssim\min\left\{\int_{\mathbb{R}^d}|\nabla u_j|^p,\sup_{y\in\mathbb{R}^d,R>0}\frac{1}{R^p}\int_{B(y,R)}|u(x)|^p\,\mathrm{d}x\right\}$$

and splitting the power

$$r/d = r(1/p - 1/p^*) = (1 - r/p^*) + (r/p - 1),$$

we find that

$$\left(\sup_{y\in\mathbb{R}^d,R>0}\frac{1}{R^p}\int_{B(y,R)}|u_j(x)|^p\,\mathrm{d}x\right)^{r/d}$$

$$\lesssim \left(\int_{\mathbb{R}^d}|\nabla u_j|^p\right)^{1-r/p^*}\left(\sup_{y\in\mathbb{R}^d,R>0}\frac{1}{R^p}\int_{B(y,R)}|u(x)|^p\,\mathrm{d}x\right)^{r/p-1}.$$

Here we used the constraint $r \in [p, p^*]$ to ensure that both $(1 - r/p^*)$ and (r/p - 1) are nonnegative. Thus, we obtain

$$\begin{split} \|u\|_{L^{p^*,r}}^{p^*} \lesssim \left(\sum_{j \in \mathbb{Z}} \left(\int_{\mathbb{R}^d} |\nabla u_j|^p \right) \left(\sup_{y \in \mathbb{R}^d, R > 0} \frac{1}{R^p} \int_{B(y,R)} |u(x)|^p \, \mathrm{d}x \right)^{r/p-1} \right)^{p^*/r} \\ \lesssim \left(\left(\int_{\mathbb{R}^d} |\nabla u|^p \right) \left(\sup_{y \in \mathbb{R}^d, R > 0} \frac{1}{R^p} \int_{B(y,R)} |u(x)|^p \, \mathrm{d}x \right)^{r/p-1} \right)^{p^*/r}, \end{split}$$

which is equivalent to (32).

From (31), (32) and the obvious bound $||u||_{L^{p^*}}^p \gtrsim ||u||_{p^*,r}^p$ for $r \ge p^*$, we get

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p\right)^{\frac{p}{\min(r,p^*)}} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \,\mathrm{d}x\right)^{1-\frac{p}{\min(r,p^*)}} \gtrsim \|u\|_{p^*,r}^p.$$
(33)

By Hardy's inequality and the condition $\theta \ge p/\min(r, p^*)$, we also get

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x - y|^p} \,\mathrm{d}x\right)^{1 - \theta} \gtrsim \|u\|_{p^*, r}^p.$$
(34)

Combining (30) and (34), we obtain

$$\left(\int_{\mathbb{R}^d} |\nabla u|^p - \left(\frac{d-p}{p}\right)^p \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x\right)^{1-\theta} \\ \ge \left(\varepsilon \int_{\mathbb{R}^d} |\nabla u|^p\right)^{\theta} \left(\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x\right)^{1-\theta} \gtrsim \|u\|_{p^*,r}^p.$$
(35)

Case 2: Assume

$$(1-\varepsilon)\int_{\mathbb{R}^d}|\nabla u|^p \leq \left(\frac{d-p}{p}\right)^p \sup_{y\in\mathbb{R}^d}\int_{\mathbb{R}^d}\frac{|u(x)|^p}{|x-y|^p}\,\mathrm{d}x.$$

Then by a monotonicity argument as in the proof of Theorem 1, we can reduce to the case when u is radially symmetric decreasing. We have for radially symmetric decreasing functions u,

$$\|u\|_{q,s}^{s} \sim \int_{\mathbb{R}^d} \frac{|u(x)|^s}{|x|^{\alpha}} \,\mathrm{d}x \tag{36}$$

if $s/q = 1 - \alpha/d$ (see e.g. [7, Lemma 4.3] for the case $q = p^*$, s = p). In particular, for $q = p^* = dp/(d-p)$ and $s = \alpha = d$, we obtain

$$\|u\|_{p^*,s}^s \sim \int_{\mathbb{R}^d} \frac{|f(x)|^s}{|x|^d} \, \mathrm{d}x \sim \int_0^\infty \frac{|f(r)|^s}{r} \, \mathrm{d}r \tag{37}$$

with $u(x) = |x|^{1-d/p} f(x)$. Moreover,

$$\int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x|^p} \, \mathrm{d}x = \int_{\mathbb{R}^d} \frac{|f(x)|^p}{|x|^d} \, \mathrm{d}x \sim \int_0^\infty \frac{|f(r)|^p}{r} \, \mathrm{d}r$$

and by the ground state representation [7, Eq. (2.14)] (here we use that $p \ge 2$)

$$\int_{\mathbb{R}^d} |\nabla u|^p - \left(\frac{d-p}{p}\right)^p \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x|^p} \,\mathrm{d}x \gtrsim \int_{\mathbb{R}^d} \frac{|\nabla f(x)|^p}{|x|^{d-p}} \,\mathrm{d}x \sim \int_0^\infty |f'(s)|^p s^{p-1} \,\mathrm{d}s.$$

By Hölder's inequality, we have, with 1/p + 1/p' = 1,

$$\left(\int_{0}^{\infty} |f'(s)|^{p} s^{p-1} ds\right)^{1/p} \left(\int_{0}^{\infty} \frac{|f(s)|^{p}}{s} ds\right)^{1/p'} \\ \ge \int_{0}^{\infty} |f'(s)| |f(s)|^{p-1} = p^{-1} \int_{0}^{\infty} |(f^{p}(s))'| ds \gtrsim_{p} ||f||_{\infty}^{p}.$$
(38)

Under the constraint (29), there exists $\tilde{r} \in [p, r]$ such that $\theta = 1/p - 1/\tilde{r}$. Let $\beta \ge 0$ such that $\tilde{r} = p(1 + \beta)$. Then

$$\left(\int_{0}^{\infty} |f'(s)|^{p} s^{p-1} \, \mathrm{d}s\right)^{\beta/p} \left(\int_{0}^{\infty} \frac{|f(s)|^{p}}{s} \, \mathrm{d}s\right)^{1+\beta/p'}$$

$$\geq p^{-1} \|f\|_{\infty}^{\beta p} \int_{0}^{\infty} \frac{|f(s)|^{p}}{s} \, \mathrm{d}s = p^{-1} \int_{0}^{\infty} \frac{|f(s)|^{\tilde{r}}}{s} \, \mathrm{d}s \sim \|u\|_{p^{*},\tilde{r}}^{\tilde{r}} \gtrsim \|u\|_{p^{*},r}^{\tilde{r}}, \qquad (39)$$

where we used (37) with $s = \tilde{r}$ and $\tilde{r} \le r$. This implies (8) by the choice of \tilde{r} .

Proof (Proof of the necessity of the range of θ) Let us explain why (8) fails for certain values of θ as indicated in Theorem 2. First we consider $\theta < p/\min(r, p^*)$ for general $p \ge 2$ where we split into two cases $r \ge p^*$ and $r < p^*$, and then we focus on the case p = 2.

Counterexample for the case $r \ge p^*$ and $\theta < p/\min(r, p^*) = p/p^*$. The idea is to consider $N \in \mathbb{N}$ identical bubbles that travel away from each other and to let $N \to \infty$. Let $0 \ne \varphi \in C_c^{\infty}(B(0, 1)), 0 \ne z \in \mathbb{R}^d$ and choose

$$u_N(x) = \sum_{n=1}^{N} \varphi(x + nNz)$$
(40)

for every $N \in \mathbb{N}$. The translation by Nz ensures that the functions $\{\varphi(\cdot + nNz)\}_{n=1}^{N}$ have disjoint support for N large. We have

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_N(x)|^p}{|x-y|^p} \, \mathrm{d}x = \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|\varphi(x)|^p}{|x-y|^p} \, \mathrm{d}x + o(1) \quad \text{as} \quad N \to \infty$$
(41)

and

$$\int_{\mathbb{R}^d} |\nabla u_N|^p - \left(\frac{d-p}{p}\right)^p \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u_N(x)|^p}{|x-y|^p} \, \mathrm{d}x = N \int_{\mathbb{R}^d} |\nabla \varphi|^p + O(1) \quad \text{as} \quad N \to \infty.$$
(42)

Moreover, it is straightforward to see that for N large

$$\|u_N\|_{p^*,r} = p^* \int_0^\infty s^{r-1} |\{|u_N| > s\}|^{r/p^*} \,\mathrm{d}s = N^{r/p^*} p^* \int_0^\infty s^{r-1} |\{|\varphi| > s\}|^{r/p^*} \,\mathrm{d}s,$$
(43)

if $r < \infty$, and

$$\|u_N\|_{p^*,\infty} = \sup_{s>0} s|\{|u_N| > s\}|^{1/p^*} = N^{1/p^*} \sup_{s>0} s|\{|\varphi| > s\}|^{1/p^*}.$$
(44)

n

if $r = \infty$. Hence, if (8) holds, then by taking $u = u_N$, we get

$$N^{\theta} \gtrsim N^{\frac{p}{p^*}} \tag{45}$$

for N large, which implies that $\theta \ge p/p^*$.

Counterexample for the case $r < p^*$ and $\theta < p/\min(r, p^*) = p/r$. Let $0 \neq \varphi \in C_c^{\infty}(B(0, 1)), 0 \neq z \in \mathbb{R}^d$ and define

$$v_N(x) = \sum_{j=1}^N 2^j \varphi(2^{p^* j/d} (x + jNz))$$
(46)

for every $N \in \mathbb{N}$. The scaling is chosen such that $\|2^{j}\varphi(2^{p^{*}j/d}\cdot)\|_{L^{p^{*}}(\mathbb{R}^{d})} = \|\varphi\|_{L^{p^{*}}(\mathbb{R}^{d})}$ for all *j*, and similarly all relevant terms in (8) are invariant when changing $\varphi \mapsto 2^{j}\varphi(2^{p^{*}j/d}\cdot)$. Again, the translation by Nz ensures that the supports of the functions $\{2^{j}\varphi(2^{p^{*}j/d}(\cdot + jNz))\}_{j=1}^{N}$ are far away from each other for *N* large. Then

$$\int_{\mathbb{R}^d} |\nabla v_N|^p = N \int_{\mathbb{R}^d} |\nabla \varphi|^p, \quad \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|v_N(x)|^p}{|x - y|^p} dx \sim \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|\varphi(x)|^p}{|x - y|^p} dx \sim 1$$

and

$$\|v_N\|_{L^{p^*,r}}^p \sim \left(\sum_j \|2^j \varphi(2^{p^*j/d} \cdot)\|_{L^{p^*}(\mathbb{R}^d)}^r\right)^{p/r} \sim N^{p/r}$$

Thus inserting v_N in (8), we find that for N large,

$$N^{\theta} \gtrsim N^{p/r},$$

which requires $\theta \ge p/r$.

Counterexample for the case p = 2 and $\theta > 1/p - 1/r$. Define $u : \mathbb{R}^d \to \mathbb{R}$ by

$$u(x) = \frac{f(|x|)}{|x|^{(d-p)/p}}.$$
(47)

where f(r) is chosen as in (18) with $\varepsilon > 0$ small. Then *u* is radially symmetric decreasing and hence (28) holds. Therefore,

$$\sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x = \int_{\mathbb{R}^d} \frac{|f(|x|)|^p}{|x|^d} \, \mathrm{d}x = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f(r)|^p}{r} \, \mathrm{d}r = |\mathbb{S}^{d-1}| \frac{2}{\varepsilon p}.$$
(48)

We also have by the ground state representation for p = 2,

$$\int_{\mathbb{R}^d} |\nabla u|^p - \left(\frac{d-p}{p}\right)^p \sup_{y \in \mathbb{R}^d} \int_{\mathbb{R}^d} \frac{|u(x)|^p}{|x-y|^p} \, \mathrm{d}x = \int_{\mathbb{R}^d} \frac{|f(|x|)|^p}{|x|^{d-p}} \, \mathrm{d}x = \varepsilon^{p-1} |\mathbb{S}^{d-1}| \frac{2}{p}.$$
(49)

Note that the analogue of (49) is more complicated for $p \neq 2$ (see [7]), which is why our counterexample only works for p = 2.

By (36), we have for $r < \infty$

$$\|u\|_{L^{p^*,r}}^r = \int_{\mathbb{R}^d} \frac{|f(x)|^r}{|x|^d} \, \mathrm{d}x = |\mathbb{S}^{d-1}| \int_0^\infty \frac{|f(s)|^r}{s} \, \mathrm{d}s = |\mathbb{S}^{d-1}| \frac{2}{\varepsilon r}.$$

Moreover, for $r = \infty$, by (21) and (27),

$$||u||_{p^*,\infty} = |B(0,1)|^{1/p^*}$$

Therefore, if (8) holds, then

$$\varepsilon^{(p-1)\theta}\varepsilon^{-(1-\theta)} \ge \varepsilon^{-p/r},\tag{50}$$

for $\varepsilon > 0$ small, which requires that $p\theta - 1 \ge -p/r$ namely

$$\theta \ge \frac{1}{p} - \frac{1}{r}.$$

The proof of Theorem 2 is complete.

Acknowledgements Part of the work has been done when CD was a visiting reseacher at the Institut des Hautes Études and she would like to thank Laure Saint-Raymond for her support and hospitality. We are grateful to Rupert L. Frank for helpful suggestions, leading to our consideration of Lorentz norms in Theorem 2. We also thank Hajer Bahouri, Morris Brooks, Albert Cohen, Patrick Gérard, Kihyun Kim, Mathieu Lewin, Hoai-Minh Nguyen, Julien Sabin, Nikita Simonov, Thomas Sørensen, Jakob Stern, Hanne van den Bosch and Jean van Schaftingen for interesting and inspiring discussions. We acknowledge the support from the Deutsche Forschungsgemeinschaft through the DFG project Nr. 426365943. CD also acknowledges the support from the Jean-Paul Gimon Fund and from the Erasmus+ programme.

Funding Open Access funding enabled and organized by Projekt DEAL.

Data Availibility Data sharing not applicable to this article as no datasets were generated or analysed during the current study.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Benguria, R.D., Loss, M.: Connection between the Lieb-Thirring conjecture for Schrödinger operators and an isoperimetric problem for ovals on the plane. In: Conca, C., Manásevich, R., Uhlmann, G., Vogelius, M.S. (eds.) Partial Differential Equations and Inverse Problems, pp. 53–61. American Mathematical Society, Providence, R.I. (2004)
- 2. Burchard, A.: A short course on rearrangement inequalities. Lecture notes (2009). http://www.math. utoronto.ca/almut/rearrange.pdf
- Brezis, H., Vázquez, J.-L.: Blow-up solutions of some nonlinear elliptic problems. Rev. Mat. Univ. Madrid 10, 443–469 (1997)
- Caffarelli, L., Kohn, R., Nirenberg, L.: First order interpolation inequalities with weights. Compos. Math. 53(3), 259–275 (1984)
- 5. Fefferman, C.: The uncertainty principle. Bull. Amer. Math. Soc. 9, 129–206 (1983)
- Frank, R.L.: A simple proof of Hardy-Lieb-Thirring inequalities. Commun. Math. Phys. 290, 789–800 (2009)
- Frank, R.L., Seiringer, R.: Non-linear ground state representations and sharp Hardy inequalities. J. Funct. Anal. 255, 3407–3430 (2008)
- Frank, R.L., Lieb, E.H., Seiringer, R.: Hardy-Lieb-Thirring inequalities for fractional Schrödinger operators. J. Amer. Math. Soc. 21(4), 925–950 (2008)
- 9. Grafakos, L.: Classical Fourier analysis. In: Graduate Texts in Mathematics. Springer, Berlin (2014)
- Keller, J.B.: Lower bounds and isoperimetric inequalities for eigenvalues of the Schrödinger equation. J. Math. Phys. 2, 262–266 (1961)
- 11. Kwong, M.K.: Uniqueness of positive solutions of $\Delta u u + u^p = 0$ in \mathbb{R}^n . Arch. Rational Mech. Anal. **105**, 243–266 (1989)

- Lieb, E.H., Thirring, W.E.: Inequalities for the moments of the Eigenvalues of the Schrödinger Hamiltonian and their relation to Sobolev inequalities. In: The Stability of Matter: From Atoms to Stars, pp. 135–169. Princeton, Princeton University Press (2001)
- Nguyen, H.-M., Squassina, M.: Fractional Caffarelli-Kohn-Nirenberg inequalities. J. Funct. Anal. 274, 2661–2672 (2018)
- Nguyen, H.-M., Squassina, M.: On Hardy and Caffarelli-Kohn-Nirenberg inequalities. J. d'Anal. Math. 275, 773–797 (2019)
- Mallick, A., Nguyen, H.-M.: The Caffarelli-Kohn-Nirenberg inequalities for radial functions. Preprint (2022). arXiv:2211.04885
- Nagy, S.Z.: Über integralgleichungen Zwischen einer Funktion und Ihrer Ableitung. Acta Sci. Math. 10, 64–74 (1941)
- 17. Palatucci, G., Pisante, A.: Improved Sobolev embeddings, profile decomposition, and concentration-compactness for fractional Sobolev spaces. Cal. Var. PDE **50**, 799–829 (2014)
- van Schaftingen, J.: Interpolation inequalities between Sobolev and Morrey-Campanato spaces: a common gateway to concentration-compactness and Gagliardo-Nirenberg interpolation inequalities. Port. Math. 71, 159–175 (2014)
- 19. Sawyer, E., Wheeden, R.L.: Weighted inequalities for fractional integrals on Euclidean and homogeneous spaces. Amer. J. Math. **114**, 813–874 (1992)
- Solovej, J.P., Østergaard Sørensen, T., Spitzer, W.: The relativistic Scott correction for atoms and molecules. Comm. Pure Appl. Math. 63, 39–118 (2010)
- 21. Terracini, S.: On positive entire solutions to a class of equations with a singular coefficient and critical exponent. Adv. Differ. Equ. 1, 241–264 (1996)
- Tran, M.-P., Nguyen, T.-N.: A remark on the interpolation inequality between Sobolev spaces and Morrey space. Funct. Anal. Appl. 54(3), 200–207 (2020)
- 23. Vazquez, J.L., Zuazua, E.: The Hardy inequality and the asymptotic behaviour of the heat equation with an inverse-square potential. J. Funct. Anal. **173**, 103–153 (2000)
- Weinstein, M.I.: Nonlinear Schrödinger equations and sharp interpolation estimates. Commun. Math. Phys. 87, 567–576 (1983)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.